

Durham E-Theses

A universal framework for hydrodynamics

JAIN, AKASH

How to cite:

JAIN, AKASH (2018) A universal framework for hydrodynamics, Durham theses, Durham University. Available at Durham E-Theses Online: http://etheses.dur.ac.uk/12707/

Use policy

The full-text may be used and/or reproduced, and given to third parties in any format or medium, without prior permission or charge, for personal research or study, educational, or not-for-profit purposes provided that:

- a full bibliographic reference is made to the original source
- a link is made to the metadata record in Durham E-Theses
- the full-text is not changed in any way

The full-text must not be sold in any format or medium without the formal permission of the copyright holders.

Please consult the full Durham E-Theses policy for further details.

Akash Jain

A thesis presented for the degree of

Doctor of Philosophy



Centre for Particle Theory & Department of Mathematical Sciences

Durham University

United Kingdom

June 2018

Akash Jain

Submitted for the degree of Doctor of Philosophy

June 2018

ABSTRACT: In this thesis, we present a universal framework for hydrodynamics starting from the fundamental considerations of symmetries and the second law of thermodynamics, while allowing for additional gapless modes in the low-energy spectrum. Examples of such fluids include superfluids and fluids with surfaces. Typically, additional dynamical modes in hydrodynamics also need to be supplied with their own equations of motion by hand, like the Josephson equation for superfluids and the Young-Laplace equation for fluid surfaces. However, we argue that these equations can be derived within the hydrodynamic framework by a careful off-shell generalisation of the second law. This potentially provides a universal framework for a large class of hydrodynamic theories, based on their underlying symmetries and gapless modes. Motivated by this newly found universality, we present an all-order analysis of the second law of thermodynamics and propose a classification scheme for the allowed hydrodynamic transport, including arbitrary gapless modes, independent spin current, and background torsion.

In the second half of this thesis, we look at the construction of null fluids which are a new viewpoint of Galilean fluids. These are essentially fluids coupled to spacetime backgrounds carrying a covariantly constant null isometry, but with additional constraints imposed on the background gauge field and affine connection to reproduce the correct Galilean degrees of freedom. We discuss the Galilean version of quantum anomalies and their effect on hydrodynamics. Finally, we follow our relativistic discussion to allow for arbitrary gapless modes in Galilean hydrodynamics and present a classification scheme for the second law abiding hydrodynamic transport at all orders in the derivative expansion.

We apply these abstract ideas to review the theory of ordinary relativistic/Galilean hydrodynamics and provide novel constructions for relativistic/Galilean (non-Abelian) superfluid dynamics and surface transport. We also comment on the possible application to the theory of magnetohydrodynamics.

Dedicated to my parents

Smt. Neelam Jain & Sh. Neeraj Jain

for shaping me into who I am

and to my brother

Abhinav

for helping along the way

Declaration

The work in this thesis is based on research carried out in the Department of Mathematical Sciences at Durham University. The results presented here are derived from the following independent and collaborative works:

- N. Banerjee, S. Dutta and A. Jain, *Equilibrium partition function for nonrelativistic fluids*, *Phys. Rev.* **D92** (2015) 081701, [1505.05677].
- N. Banerjee, S. Dutta and A. Jain, Null Fluids A New Viewpoint of Galilean Fluids, Phys. Rev. D93 (2016) 105020, [1509.04718].
- A. Jain, Galilean Anomalies and Their Effect on Hydrodynamics, Phys. Rev. D93 (2016) 065007, [1509.05777].
- A. Jain, Theory of non-Abelian superfluid dynamics, Phys. Rev. D95 (2017) 121701, [1610.05797].
- N. Banerjee, S. Dutta and A. Jain, First Order Galilean Superfluid Dynamics, Phys. Rev. D96 (2017) 065004, [1612.01550].
- J. Armas, J. Bhattacharya, A. Jain and N. Kundu, On the surface of superfluids, JHEP 06 (2017) 090, [1612.08088].

While this thesis was being prepared, the following collaborative works were also published by the author, results from which have not been included in this thesis:

- N. Banerjee, S. Atul Bhatkar and A. Jain, Second order Galilean fluids & Stokes' law, Phys. Rev. D97 (2018) 096018, [1711.09076].
- J. Armas, J. Gath, A. Jain and A. V. Pedersen, *Dissipative hydrodynamics with higher-form symmetry*, *JHEP* 05 (2018) 192, [1803.00991].
- P. Burda, R. Gregory and A. Jain, Holographic Reconstruction of Bubbles, 1804.05202.

No part of this thesis has been submitted for a degree or qualification in this or any other institution.

Copyright © 2018 Akash Jain.

The copyright of this thesis rests with the author. No quotation from it should be published without the author's prior written consent and information derived from it should be acknowledged.

Acknowledgements

First and foremost I would like to thank my supervisor, Ruth Gregory, for her guidance and encouragement throughout my PhD. I have learned as much about the intricacies of the academic life itself as I have about Physics through her. The three years that I have spent with Ruth have been, without doubt, one of the most exciting times of my life, academic or otherwise. I cannot name many colleagues whose supervisors took them to Canada every year, arranged day trips, accompanied them to restaurants, or invited them over for meals! Thank you, Ruth, for making my PhD such a memorable experience.

I would especially like to thank my collaborators Nabamita Banerjee and Suvankar Dutta for their unwavering trust in my potential when my own self-confidence dwindled at times. During every step of my struggle, they have always looked out for me; perhaps worried even more than I did at times. This thesis would not exist without their invaluable contributions. I would like to offer my special thanks to my friends and collaborators Jyotirmoy Bhattacharya and Jay Armas, for their patience and calm during our long discussions and for putting up with my disorganised, and often pointless, doubts and ideas. Many parts of this thesis are a direct product of their hours spent on explaining crucial concepts to me. I would also like to thank Mukund Rangamani, who supervised me during the coursework phase of my PhD. I am particularly grateful to Philipp Burda, Felix Haehl, Nabil Iqbal, Aurora Ireland, Simon Ross, and many other friends and collaborators, who have helped me get through my PhD.

A huge credit for this thesis goes to my family in Durham—Leo, Mike, Omar, Marisa, Lara, Marina, and many others for sharing this journey with me. For helping me step out of my comfort zone and for tirelessly trying to get me into western music. I am leaving here invariably changed compared to my four years younger self, and for better or worse, these guys have been behind every bit of it. They have made my PhD experience orders of magnitude more fun and exciting. A huge shout-out also to my friends from back home—Deepak, Rahul, Pratik, Aranya, Megha, Sukruti, Sunidhi, and Alka.

I owe a huge debt of gratitude to my family, my father Neeraj Jain, mother Neelam Jain, and brother Abhinav Jain, who stood by me every step of the way. I remember my mother who dedicated every bit of her life nourishing her children and who passed away last September; she would be so proud.

I would like to acknowledge the support and hospitality I received from Durham University, Perimeter Institute, IISER Bhopal, IISER Pune, ICTS Bangalore, NBI Copenhagen, GGI Florence, ICTP Trieste, University of Oviedo, and DAMTP Cambridge during various stages of my PhD. This work is financially supported by Durham Doctoral Scholarship offered by Durham University.

Contents

Abstract iii							
Notation and conventions xiii							
1	Intr	Introduction					
	1.1	Fundar 1.1.1 1.1.2 1.1.3	mentals of hydrodynamics	5 6 8 12			
	1.2	Hydro	dynamics with gapless modes	15			
	1.3	Overvi	ew and organisation	18			
2 Relativistic hydrodynamics							
	2.1	Prepar 2.1.1 2.1.2 2.1.3 2.1.4 The hy 2.2.1 2.2.2 2.2.3	ing the background	23 23 25 27 29 31 31 33 35			
	2.3	 2.2.4 Local s 2.3.1 2.3.2 2.3.3 2.3.4 2.3.5 2.3.6 	Hydrostatic principle	 37 39 39 43 45 47 48 50 			
	2.4	Summa	ary and torsionless limit	51			
3	App	olicatio	lications: relativistic hydrodynamics				
	3.1	Relativ 3.1.1 3.1.2	vistic fluids	55 55 56			

Bi	Bibliography 121						
6	Out	clook 11	7				
	5.3	Galilean fluid surfaces	4				
		5.2.3 One-derivative corrections	9				
		5.2.2 Ideal null superfluids	7				
		5.2.1 Goldstone modes and Josephson equation	6				
	5.2	Galilean superfluids	6				
		5.1.2 One-derivative corrections	2				
	0.1	5.1.1 Ideal null fluids	1				
	51	Galilean fluids 10	1				
5	Арг	plications: Galilean hydrodynamics 10	1				
		4.3.5 Classification of hydrodynamic transport	9				
		4.3.4 Transcendental anomalies	6				
		4.3.3 Anomaly induced transport	5				
		4.3.2 Null reduction to Galilean hydrodynamics	3				
		4.3.1 Hydrodynamics on null backgrounds	1				
	4.3	Null fluids	1				
		4.2.4 The non-relativistic limit	0				
		4.2.3 Reference frame transformations	8				
		4.2.2 Noether currents and conservation equations	7				
		4.2.1 Newton-Cartan backgrounds	4				
	4.2	Null reduction to Newton-Cartan backgrounds	3				
		4.1.4 Galilean anomalies	1				
		4.1.3 Ward identities	0				
		4.1.2 Null backgrounds	7				
	- 1 .1	4.1.1 From Poincaré to Galilean algebra	5				
-	<u> </u>	1 Calilean field theories and null backgrounds					
4	Gal	ilean hydrodynamics 7	5				
	3.3	Relativistic fluid surfaces	0				
		3.2.4 Breaking of non-Abelian internal symmetries	8				
		3.2.3 One-derivative corrections	2				
		3.2.2 Ideal superfluids	0				
	0.2	3.2.1 Coldstone modes and Iosephson equation 55	9 0				
	3.2	Relativistic superfluids 55	9				

Notation and conventions

Spacetime and indices

We use mostly positive metric convention throughout this work. Number of physical spacetime dimensions are denoted by d. The Greek letters $(\mu, \nu, ...)$ are used to denote the d-dimensional spacetime indices, while the Greek letters $(\alpha, \beta, ...)$ are used for the indices on the d-dimensional frame bundle. Similarly, the Roman letters (i, j, ...) denote the (d-1)-dimensional spatial indices, with the corresponding frame bundle indices denoted by the Roman letters (a, b, ...). The time coordinate, on the other hand, is denoted by t. We also introduce (d+1)-dimensional null backgrounds as a tool for Galilean hydrodynamics. Indices on it are denoted by the Roman letters (M, N, ...), while those on its frame bundle by (A, B, ...). In the context of anomaly inflow mechanism, the "bulk" indices will be denoted by a hat. Einstein summation convention is implied everywhere.

We use the round and square brackets to denote totally symmetric and totally antisymmetric combinations of a tensor respectively. For a 2-tensor, $M^{\mu\nu} = M^{(\mu\nu)} + M^{[\mu\nu]}$. We often also use the angular brackets to denote traceless symmetric combinations.

Differential forms

We denote differential forms by bold-faced characters. If A and B are m and n-rank differential forms respectively, X^{μ} is a vector field, f(x) is a function, and $g = \det g_{\mu\nu}$ is the metric determinant, then

$$\begin{aligned} \boldsymbol{A} &= \frac{1}{m!} A_{\mu_1 \dots \mu_m} dx^{\mu_1} \wedge \dots, \\ \boldsymbol{\epsilon} &= \sqrt{|g|} dx^0 \wedge \dots \wedge dx^d & \text{[Volume form]} \\ \boldsymbol{A} \wedge \boldsymbol{B} &= \frac{1}{(m+n)!} \left(\frac{(m+n)!}{m!n!} A_{[\mu_1 \dots \mu_m} B_{\nu_{m+1} \dots \nu_{m+n}]} \right) dx^{\mu_1} \wedge \dots, & \text{[Exterior product]} \\ \star \boldsymbol{A} &= \frac{1}{(d-m)!} \left(\frac{1}{m!} A^{\mu_1 \dots \mu_m} \boldsymbol{\epsilon}_{\mu_1 \dots \mu_m \nu_1 \dots \nu_{d-m}} \right) dx^{\nu_1} \wedge \dots, & \text{[Hodge dual]} \\ \star \star \boldsymbol{A} &= \operatorname{sgn}(g)(-)^{m(d-m)}, \\ \iota_X \boldsymbol{A} &= \frac{1}{(m-1)!} \left(X^{\mu} A_{[\mu\nu_1 \dots \nu_{m-1}]} \right) dx^{\nu_1} \wedge \dots, & \text{[Interior product]} \\ d\boldsymbol{A} &= \frac{1}{(m+1)!} \left((m+1)\partial_{[\mu_1} A_{\mu_2 \dots \mu_{m+1}]} \right) dx^{\mu_1} \wedge \dots, & \text{[Exterior derivative]} \\ \pounds_X \boldsymbol{A} &= \iota_X d\boldsymbol{A} + d(\iota_X \boldsymbol{A}), & \text{[Lie derivative]} \\ f(x) \boldsymbol{\epsilon} &= \int d^d x \sqrt{|g|} f(x). & \text{[Integration]} \end{aligned}$$

Hydrodynamics, or fluid dynamics, is a scientific discipline which concerns itself with the study of fluids. The word "hydrodynamics" derives from the Greek prefix "hydro-" meaning "water" and literally translates to "the study of motion of water". The word "fluid", on the other hand, comes from the Latin word "fluere" meaning "to flow", and is used as an umbrella term for liquids and gases together. Today, at least within the high energy physics community, both hydrodynamics and fluid dynamics are used interchangeably to describe the physics of continuum systems ranging from the infinitesimal quark-gluon plasma to the infinite universe itself.

Over seventy per cent of Earth's surface is covered with water, which undoubtedly contributes the most vital ingredient to our biology, chemistry, as well as our sociology. Major civilisations in the history of humankind have flourished around water bodies, from ancient Mesopotamia all the way to present-day London. Understanding the physics of water, or more generally of fluids, has therefore played a key role in shaping the human civilisation. The earliest records of scientific enquiry into the statics and dynamics of fluids date back to Archimedes in his treatise On Floating Bodies, published around 250 BC. However, it is generally accepted that a pragmatic, if not scientific, understanding of fluids existed even in ancient human civilisations. Over the two millennia following Archimedes' principle of buoyancy, the field of hydrodynamics attracted the attention of many inquisitive minds, slowly taking shape into a scientific discipline in its own right. Rapid progress was made in the seventeenth and eighteenth centuries alongside the development of calculus and the laws of thermodynamics, to which we owe much of our current understanding of hydrodynamics. Major contributors include Torricelli, Newton, Pascal, Bernoulli, Euler, d'Alembert, Lagrange, Laplace, Poisson, Poiseuille, Hagen, Navier, Stokes, Prandtl, Reynolds, and Taylor to name a few. In the early twentieth century, principles of hydrodynamics were reconciled with Einstein's theory of relativity paving way to our modern understanding of accelerator physics and cosmology. Exotic phenomena in hydrodynamics like dissipation, vorticity, turbulence, diffusion, superfluidity, and multifluid models, and their relation to thermodynamics and statistical mechanics were also developed in the twentieth century owing to the efforts of Ertel, Lichnerowicz, Kamerlingh-Onnes, Landau, Onsager, Prigogine, Khalatnikov, Eckart, Carter, Israel, and Stewart among many others. The state of the art in the field is the standard book by Landau and Lifshitz on Fluid Mechanics [10] published in 1959. To this day, hydrodynamics continues to be one of the most active topics of research in many areas of science including engineering, condensed matter physics, medicine, biophysics, and even high energy physics.

Conventionally, hydrodynamics has been an empirical field of study under the jurisdiction of material sciences and had little to do with the realm of high energy physics, which concerns itself with the fundamental machinery of our universe. However, this began to change towards the end of the twentieth century, when a series of advancements in string theory intricately wove hydrodynamics with the physics of black holes. String theory is a candidate for the quantum theory of gravity, which aims to describe all the four forces of nature within a grand unified framework. Three of these forces: electromagnetic, strong, and weak, have already been unified into a quantum field theory called the standard model of particle physics, with great experimental success at particle accelerator experiments such as the Large Hadron Collider. On the other hand, our current understanding of the fourth fundamental force, gravity, comes from a classical field theory known as the general theory of relativity. It has also been experimentally tested to great lengths, with its latest confirmation coming from the recent gravitational wave detection by LIGO and VIRGO collaborations in 2015 [11]. Despite these success stories and decades of research efforts, quantising gravity and reconciling it with the standard model of particle physics has proven to be very hard. This is where string theory comes in, with its revolutionary idea that the fundamental building blocks of our universe are not various species of point-like particles, as proposed by the standard model, but just a single species of tiny strings. In string theory, point-like particles are understood as emerging from different oscillatory modes of the same fundamental string. A quantum theory of strings gives us a natural framework where gravity and quantum field theories can, in principle, emerge from the same fundamental principles. Whether or not string theory is the answer to life, the universe and everything, it has already taught us some valuable lessons about nature. Perhaps the most important of these lessons is the concept of *holography*. It is the observation that the physics of quantum gravity in spacetimes with a certain asymptotic structure can be equivalent to that of a one-lower dimensional quantum field theory living at its boundary [12, 13]; see [14] for a review. The most popular example of holography comes from the AdS/CFTcorrespondence [15–17], which relates quantum gravity on asymptotically Anti-de Sitter (AdS) spacetimes in the bulk to conformal field theories at the boundary. See [18] for an introductory review.

Holographic dualities have proven to be invaluable for both quantum gravity as well as field theory research. On the one hand, they relate quantum field theories in a weak coupling regime to highly quantum phenomena on the gravity side and can be used to get valuable insights into the quantum nature of gravity. For example, it is widely felt that spacetime itself could be seen as emerging from the entanglement structure of the boundary field theory [9, 19]. More generally, there seems to be an intricate relationship between quantum gravity in the bulk and quantum information theory at the boundary [20-23], which can be used to probe hard questions in gravity like the black hole information paradox [24]. On the other hand, holography also relates classical general relativity to strongly coupled quantum field theories. The strong coupling regime can be very hard to access directly in quantum field theories due to the breakdown of the conventional perturbative approach using Feynman diagrams. Interestingly, the theory of strong interactions between quarks and gluons, called quantum chromodynamics (QCD), is weakly coupled at high energies but is strongly coupled at our day-to-day energy scales. Due to this feature of QCD, many strong interaction phenomena remain very hard to describe theoretically. Although novel techniques are being developed to directly tackle strongly coupled phenomena in quantum

field theories, like conformal bootstrap [25] and lattice field theories [26], holography is one of our best analytical techniques to gain insights into these problems. This has led to a whole new field of research called *holographic condensed matter physics*, aimed at utilising holographic techniques to study qualitative features of real-world condensed matter systems. See for example [27] for an extensive review.

A particular niche of holography, called the *fluid/qravity correspondence*, dualises classical perturbations around black hole horizons in the bulk to a strongly-coupled fluid living at the boundary. The fact that the linearised Einstein equations governing black hole fluctuations can be remapped to the linearised dynamical equations of hydrodynamics, had been known since the late 1980s under the name of *membrane paradigm* [28–30]. But the full non-linear incarnation of this correspondence was only made explicit for the first time in 2007 [31]. Due to this correspondence, every solution to hydrodynamic equations of motion at the boundary gives rise to a dynamical black hole solution in the bulk. This has been a powerful tool in the literature, for example, to test the stability of black hole configurations using our intuitive notion of stability in hydrodynamics. In this language, the well known Gregory-Laflamme instability in black strings [32] can be holographically understood as the Rayleigh-Plateau instability in hydrodynamics [33, 34], which explains why a thin stream of fluid tends to break into smaller packets. Reversing the duality, the fluid/gravity correspondence has also led to some novel insights into hydrodynamics itself, and its strongly-coupled applications like the quark-gluon plasma [35, 36]. For instance, it was experimentally discovered that the ratio of shear viscosity to entropy density for a quark-gluon plasma takes a universal value of $\hbar/4\pi k_{\rm B}$ [37], which can be theoretically explained by holographic models [38, 39]. In fact, actual heavy-ion collisions at particle accelerator experiments can themselves be modelled holographically using the fluid/gravity correspondence [40, 41]. On other fronts, one of the most interesting aspects of this correspondence has been to provide a test-bed for new models of hydrodynamic phenomena. by providing an analytically manageable tool for otherwise intractable computations. Some examples involve recent holographic studies into exotic phenomena in hydrodynamics like vortices, turbulence and chaos [42–45].

These holographic explorations have also fuelled new research into hydrodynamics in its own right, leading to new insights into its fundamental principles and applications. Perhaps the most important of these insights have been realising the signatures of quantum anomalies in hydrodynamics [46, 47], which in turn have found crucial applications in the physics of neutron stars [48]. We also have a microscopically motivated derivation of a large sector of hydrodynamics using hydrostatic partition functions [2, 49–51], along with a classification of entire hydrodynamic transport to all orders in derivative expansion [1, 52]. Other important advancements include new models for superfluid dynamics [1, 53], magnetohydrodynamics [54], boost non-invariant [55, 56] and translation non-invariant hydrodynamics [57, 58], formation of surfaces and lumps [6, 59], and hydrodynamics with higher-form symmetries [8, 60]. Novel applications of hydrodynamics to condensed matter systems like high-temperature superconductors [61] and graphene [62] have also been realised recently. However, there are still a lot of fundamental questions that remain unanswered. For instance, we still do not satisfactorily understand how inherently reversible microscopic field theories can lead to irreversible phenomena, such as dissipation and the second law of thermodynamics, in the macroscopic limit. Also, our understanding of non-equilibrium processes in finite-temperature field theories is not very robust, with hydrodynamics practically being the only laboratory where we have some degree of control. The hope is that if we can understand how hydrodynamics emerges as an effective field theory from the microscopic degrees of freedom, we will be able to, at least, frame these questions in the right language. Considerable progress has been made towards a Schwinger-Keldysh formalism for hydrodynamics and writing down a Wilsonian effective action [63–71], which sheds light on some of these fundamental issues.

On an independent front, our understanding of Galilean hydrodynamics has also been revolutionised over the past decade. The holographic correspondence for Galilean (Schrödinger) hydrodynamics was first set up in 2008 [72], shortly after its relativistic counterpart. The authors employed the standard prescription of null reduction [73–75] (see [76] for a review), which reduces a (d + 1)-dimensional relativistic fluid to a *d*-dimensional Galilean fluid, and prescribed a holographic map between *d*-dimensional Galilean hydrodynamics and (d + 2)-dimensional black holes. Exploiting null reduction, a new framework for Galilean fluids has been proposed, called null fluids, which reorganises Galilean hydrodynamics into one-higher dimensional anisotropic relativistic hydrodynamics [2–5, 7]. Other frameworks of Galilean hydrodynamics, utilising Newton-Cartan geometries [77–80], have also been proposed recently [81, 82]. Besides Galilean, there have also been other incarnations of non-relativistic hydrodynamics in the literature with different underlying symmetry groups. Most notably, Lifshitz hydrodynamics has attracted a lot of attention; see e.g. [55, 83] for some recent work and [84] for a review of Lifshitz holography.

We should mention that the references provided here are only intended to be indicative of the ongoing activity in the field of hydrodynamics. Given the sheer magnitude of relevant papers being published every year, the references mentioned here are in no way exhaustive, or even representative, of all the interesting research being carried out in hydrodynamics and related areas. The readers interested in learning more are encouraged to consult the mentioned references and follow the bibliographies therein.

In this thesis, we review some of the recent advancements mentioned above to which the author has contributed during the course of his PhD [1–6]. We also discuss some new results and insights following from these works, which were found during the preparation of this thesis. We present a universal framework for hydrodynamics starting from the fundamental considerations such as symmetries and the second law of thermodynamics, while allowing for arbitrary gapless modes in the low-energy spectrum. The most natural examples of such systems are (non-Abelian) superfluids, which we discussed in [1], with gapless Goldstone modes arising due to a spontaneously broken internal symmetry. Such modes can also arise due to a spontaneous or explicit breaking of spacetime symmetries, which can cause the formation of surfaces, as we described in [6], and potentially a myriad of other interesting phenomena like momentum relaxation and boost non-invariance. In fact, the theory of magnetohydrodynamics can also be understood within the same framework, with the components of a dynamical U(1) gauge field serving as gapless modes. Typically, such additional dynamical modes in hydrodynamics also need to be supplied with their own equations of motion by hand, like the Josephson equation for superfluids, Young-Laplace equation for fluid surfaces, and Maxwell's equations for magnetohydrodynamics. However, we realised in [1] that these equations can be derived within the hydrodynamic framework by a careful off-shell generalisation of the second law of thermodynamics. This potentially provides a universal framework for a large class of hydrodynamic theories, based on their underlying symmetries and gapless modes. Motivated by this newly found universality, we extend the classification scheme of hydrodynamic transport, proposed by the authors of [52], to include arbitrary gapless modes, while also allowing for an independent spin-current and background torsion. This classification scheme first appeared for (non-Abelian) superfluids in [1]. We expect this construction to naturally extend to hydrodynamics with higher form symmetries as well, which we set up in [8], but we do not explore this direction in this thesis. In the second half of this thesis, we look at the construction of null fluids, based on our work in [2–5], which are a new viewpoint of Galilean fluids. These are essentially fluids coupled to spacetime backgrounds carrying a covariantly constant null isometry, but with additional constraints imposed on the background gauge field and affine connection to reproduce the correct Galilean degrees of freedom. We discuss the Galilean version of quantum anomalies and their effect on hydrodynamics, taken from our work in [5]. Finally, we follow our relativistic discussion to allow for arbitrary gapless modes in Galilean hydrodynamics and present a classification scheme for the second law abiding hydrodynamic transport at all orders in the derivative expansion. As far as we are aware, an all-order analysis of Galilean transport has not appeared in the literature before.

In the remainder of this introductory chapter, we present a quick recap of the fundamentals of hydrodynamics in section 1.1, followed by an introduction to hydrodynamics with gapless modes in section 1.2. Finally, in section 1.3 we provide a comprehensive overview of the main results of this thesis.

1.1 Fundamentals of hydrodynamics

Hydrodynamics is the low-energy effective description of a generic finite temperature quantum system near its thermodynamic equilibrium. The qualifier "near" essentially means that we are only ever allowed to leave the global thermodynamic equilibrium perturbatively so that locally we still have a notion of thermodynamic equilibrium at every spacetime point. In precise terms, hydrodynamics describes physical systems whose fluctuations are on the length scales much larger compared to the inherent length scales of the system, like mean-free path. Even in the absence of a (quasi-)particle description, temperature itself can set such an inherent length scale, $\beta = c\hbar/k_{\rm B}T$, which for a system at room temperature is roughly 10^{-6} meters. Within this narrow regime of applicability, hydrodynamics is quite universal; the fundamental equations governing it can be applied to water at the scale of a raft, quark-gluon plasma at the scale of particle accelerators, and even cosmology at the scale of the universe itself. In this section, we build hydrodynamics starting from the fundamental laws of thermodynamics. We briefly discuss both the relativistic and Galilean versions of hydrodynamics and work out the first corrections to the respective constitutive relations as we leave the thermodynamic equilibrium. This should set the groundwork for the material presented in this thesis.

1.1.1 Thermodynamics

Let us say that we are given a volume of gas in equilibrium and we want to describe it using our understanding of thermodynamics. The macroscopic variables we should choose to describe this system are based on the *statistical ensemble* we are working in. For example, if the gas is held in a closed container of fixed volume and is not allowed to exchange heat or conserved charges (like particles) with its surroundings, it is said to be in the microcanonical ensemble. The thermodynamic parameters in this ensemble are the volume V of the container and the total entropy S_{tot} and total charge Q_{tot} of the gas. Their canonical conjugates, on the other hand, are the thermodynamic observables: pressure P, temperature T, and chemical potential μ . The macroscopic state of the gas itself is represented by a *thermodynamic potential* specified as a function of thermodynamic parameters, which in the microcanonical ensemble happens to be the internal energy $E_{tot}(V, S_{tot}, Q_{tot})$ of the gas. We can read out various macroscopic observables in terms of the internal energy using the celebrated *first law of thermodynamics*

$$\delta E_{\rm tot} = T \delta S_{\rm tot} + \mu \delta Q_{\rm tot} - P \delta V. \tag{1.1}$$

It states the principle of energy conservation in thermodynamics: the change in the internal energy of a system is equal to the heat $T\delta S_{tot}$ supplied to the system, plus the energy $\mu\delta Q_{tot}$ gained by charge influx, minus the work $P\delta V$ done by the system. Based on the application we have in mind, we can also work with any of the other thermodynamic ensembles; they are all related to each other via a Legendre transform. In this work, we are mainly interested in the so-called grand canonical ensemble. It describes a given volume of gas which is free to exchange heat and charge with its surroundings but is held at a fixed temperature and chemical potential. The thermodynamic potential for this ensemble is called the grand potential $\Omega(V, T, \mu)$. It is related to the internal energy of the microcanonical ensemble via the Legendre transform $\Omega = E - TS_{tot} - \mu Q_{tot}$. Using the first law of thermodynamics, we can work out its variation to be

$$\delta\Omega = -S_{\rm tot}\delta T - Q_{\rm tot}\delta\mu - P\delta V. \tag{1.2}$$

For completeness, we should also mention the third standard statistical ensemble called the canonical ensemble. The associated thermodynamic potential is given by the Helmholtz free energy $F_{\text{tot}}(V, T, Q_{\text{tot}})$ defined as $F_{\text{tot}} = E_{\text{tot}} - TS_{\text{tot}}$. It describes a given volume of gas that is allowed to exchange heat with its surroundings but not charge.

The topic of interest of this work is fluids. From a thermodynamic perspective, fluids are homogeneous systems. That is to say that no particular subvolume of a fluid is more interesting than any generic subvolume of the same fluid. In precise terms, homogeneity can be defined as a property of a thermodynamic system that under a scaling of all the extensive parameters by some arbitrary function λ , the thermodynamic potential also scales by the same factor. For example, in the microcanonical ensemble we have

$$E_{\text{tot}}(\lambda V, \lambda S_{\text{tot}}, \lambda Q_{\text{tot}}) = \lambda E_{\text{tot}}(V, S_{\text{tot}}, Q_{\text{tot}}).$$
(1.3)

In other ensembles, this statement equivalently implies $\Omega(\lambda V, T, \mu) = \lambda \Omega(V, T, \mu)$ and $F_{\text{tot}}(\lambda V, T, \lambda Q_{\text{tot}}) = \lambda F_{\text{tot}}(V, T, Q_{\text{tot}})$. If we differentiate this equation with respect to λ and set $\lambda = 1$, we arrive at the *Euler equation* for homogeneous fluids

$$E_{\rm tot} = TS_{\rm tot} + \mu Q_{\rm tot} - PV, \qquad (1.4)$$

or equivalently $\Omega = -PV$ and $F_{\text{tot}} = \mu Q_{\text{tot}} - PV$. Note that in the grand canonical ensemble, homogeneity completely fixes the volume dependence of Ω . To see this, note that the pressure of a fluid in the grand canonical ensemble is given by $P = -\partial\Omega/\partial V = -\Omega/V$ which implies $P = P(T,\mu)$ and $\Omega(V,T,\mu) = -P(T,\mu)V$. The functional dependence of P on T and μ , i.e. $P = P(T,\mu)$, is known as the equation of state of the fluid, which completely specifies its macroscopic state. It is made manifest by the *Gibbs-Duhem equation*, obtained by substituting the Euler equation into the first law of thermodynamics, leading to

$$\delta P = S\delta T + Q\delta\mu. \tag{1.5}$$

Here we have defined the entropy density $S = S_{\text{tot}}/V$ and charge density $Q = Q_{\text{tot}}/V$ of the fluid, which are more natural quantities for a homogeneous system. In terms of the densities, the Euler relation itself becomes

$$E + P = TS + \mu Q, \tag{1.6}$$

where $E = E_{\text{tot}}/V$ is the internal energy density. Lastly, we can also derive a local version of the first law of thermodynamics

$$\delta E = T\delta S + \mu\delta Q,\tag{1.7}$$

which is more relevant in fluid dynamics. It states that the change in the internal energy distribution of a fluid can be attributed to the change in the heat and charge distributions. Consequently, the equation of state for a fluid in the microcanonical ensemble is E = E(S, Q) and similarly in the canonical ensemble F = F(T, Q), where $F = F_{tot}/V$ is the Helmholtz free energy density.

We should clarify some confusion in the literature regarding the usage of the phrase "equation of state". By our definition, specifying the equation of state completely specifies the macroscopic thermodynamic state of the fluid. This would not be the case if we instead decided to specify, for example, P(T, Q) rather than $P(T, \mu)$. For concreteness, let us look at ideal gases, which are popularly known to have an "equation of state" $PV = Q_{\text{tot}}k_{\text{B}}T$. In this context, Q_{tot} counts the number of gas atoms/molecules in a given volume and k_{B} is the Boltzmann constant. Dividing by the volume, we get a local version of this equation

 $P(T,Q) = Qk_{\rm B}T$. Realising that $Q = \partial P/\partial \mu$, we can integrate this equation to get

$$P(T,\mu) = f(T) \exp\left(\frac{\mu}{k_{\rm B}T}\right).$$
(1.8)

In fact, this would have been the true equation of state of an ideal gas, which, as of yet, is unspecified due to the appearance of an arbitrary function f(T). For instance, we could take this function to be a power law $f(T) = \alpha T^{c_P}$ for some constants α and c_P . Using the Gibbs-Duhem and Euler equations, we can read out the energy, charge, and entropy densities of an ideal gas to be

$$S = \frac{P}{T} \left(c_P - \frac{\mu}{k_{\rm B}T} \right), \qquad Q = \frac{P}{k_{\rm B}T}, \qquad E = (c_P - 1)P. \tag{1.9}$$

These are, obviously, standard results in the thermodynamics of ideal gases. Importantly, note that the supposed equation of state $PV = Q_{\text{tot}}k_{\text{B}}T$ does not contain any information about the specific heat c_P , which however is contained in $P(T, \mu)$.

1.1.2 Relativistic hydrodynamics

Hydrodynamics describes small fluctuations of a homogeneous system around its thermodynamic equilibrium. To this end, we introduce a slowly varying fluid velocity field $u^{\mu}(t, \mathbf{x})$ normalised as $u^{\mu}u^{\nu}\eta_{\mu\nu} = -1$, which in equilibrium would have been just δ^{μ}_{t} . Here $\eta_{\mu\nu} = \text{diag}(-1, 1, 1, ...)$ is the pseudo-Riemannian Minkowski metric which is used to raise/lower the μ, ν, \ldots indices. It should be noted that u^{μ} characterises a course-grained macroscopic velocity of the fluid at any given spacetime point and not the individual velocities of the microscopic constituents that make up the fluid. The assumption that we can treat fluids as a continuous media, ignoring their microscopic mechanics, is known as the *continuum assumption* of hydrodynamics. Within this premise, we also promote the thermodynamic parameters $T(t, \mathbf{x})$ and $\mu(t, \mathbf{x})$ to slowly varying scalar fields over the spacetime manifold. Here, by "slowly varying" we mean that the spacetime derivatives of u^{μ} , T, and μ are much smaller compared to the quantities themselves, allowing us to treat derivatives as a well-defined perturbative parameter around the thermodynamic equilibrium. As a corollary, we can always revert back to the thermodynamic regime by setting all the derivatives to zero.

The dynamical setup we have in mind is as follows: let us say that we are given a fluid configuration $u^{\mu}(0, \mathbf{x})$, $T(0, \mathbf{x})$, and $\mu(0, \mathbf{x})$ at some initial time t = 0, along with their first time derivatives. We would like to be able to solve an initial value problem and determine the fluid configuration at all later times t > 0. For this purpose, we need a set of equations of motion for u^{μ} , T, and μ . For the scalars T and μ , we can obtain a first approximation to such an equation by demanding the entropy and charge densities to be conserved along the fluid flow, i.e.

$$\partial_{\mu} \left(S u^{\mu} \right) = 0, \qquad \partial_{\mu} \left(Q u^{\mu} \right) = 0. \tag{1.10a}$$

Here S and Q are seen as functions of T and μ . The evolution of the fluid velocity, on the

other hand, is governed by the relativistic Navier-Stokes equation for force balance

$$(E+P) u^{\mu} \partial_{\mu} u^{\nu} = -P^{\nu\rho} \partial_{\rho} P, \qquad (1.10b)$$

where $P^{\mu\nu} = \eta^{\mu\nu} + u^{\mu}u^{\nu}$ is the projector against the fluid velocity. The fluid acceleration on the left is balanced on the right by the gradient of pressure. The fluids that can be described by this simple set of equations are known as *ideal fluids*.

As the name suggests, the dynamical equations (1.10) are quite idealised. In a generic fluid flow, the flow of entropy might not necessarily align with the flow of charge. Moreover, there could be frictional (viscous) forces in the Navier-Stokes equation due to nearby layers of the fluid dragging along each other. These effects introduce correction terms in eq. (1.10) appearing at second derivative order or higher, which are not fixed by thermodynamics. To efficiently organise these derivative corrections, let us define an *energy-momentum tensor* and *charge current* for our fluid as

$$T^{\mu\nu} = (E+P)u^{\mu}u^{\nu} + P\eta^{\mu\nu} + T^{\mu\nu}_{der}, \qquad J^{\mu} = Qu^{\mu} + J^{\mu}_{der}.$$
 (1.11)

Here $T_{der}^{\mu\nu}$ and J_{der}^{μ} represent the possible derivative corrections. Using thermodynamics, we can show that the dynamical equations (1.10) can now be represented as the energy-momentum and charge conservation equations

$$\partial_{\mu}T^{\mu\nu} = 0, \qquad \partial_{\mu}J^{\mu} = 0. \tag{1.12}$$

Thanks to the Noether theorem, these conservation equations are guaranteed to hold even in the presence of derivative corrections. Consequently, derivative corrections only enter the dynamical equations via the corrections in $T^{\mu\nu}$ and J^{μ} . The plan henceforth is to write down the most generic derivative corrections in the currents $T^{\mu\nu}$ and J^{μ} allowed by symmetries, truncated to a finite order in the derivative expansion. These expressions are known as the *fluid constitutive relations*.

As such, when we include derivative corrections in $T^{\mu\nu}$ and J^{μ} , there is no guarantee that the entropy remains conserved. This is not surprising, as we do not expect entropy to stay conserved in an arbitrary dynamical process. However, in accordance with the *second law of thermodynamics*, we still expect there to be a local notion of an entropy current

$$J_{\rm S}^{\mu} = S u^{\mu} + J_{\rm S,der}^{\mu}, \tag{1.13}$$

which has a non-negative divergence at every spacetime point, i.e.

$$\partial_{\mu}J_{\rm S}^{\mu} \ge 0. \tag{1.14}$$

This is, in fact, a very non-trivial condition, which needs to be imposed by hand at every derivative order and implies some strict constraints on the tensor structures that can enter the constitutive relations. To glimpse the full potential of this inequality, let us consider the zero derivative order constitutive relations from a combinatorial viewpoint. They are still given by eq. (1.11), except that P, E, and Q are now viewed as arbitrary functions

of T and μ , without the thermodynamics relating them. Similarly, the most generic zero order entropy current is given by eq. (1.13). Let us compute the divergence of $J_{\rm S}^{\mu}$; we find

$$\partial_{\mu}J_{S}^{\mu} = u^{\mu}\partial_{\mu}S + S\partial_{\mu}u^{\mu} = \left(-\frac{\partial S}{\partial E}(E+P) - \frac{\partial S}{\partial Q}Q + S\right)\partial_{\mu}u^{\mu} + \mathcal{O}(\partial^{2}).$$
(1.15)

In the second step we have viewed S as a function of E and Q and used the first order equations of motion

$$\partial_{\mu}T^{\mu\nu} = 0 \implies u^{\mu}\partial_{\mu}E = -(E+P)\partial_{\mu}u^{\mu} + u_{\nu}\partial_{\mu}T^{\mu\nu}_{der},$$
$$u^{\mu}\partial_{\mu}u^{\nu} = \frac{1}{E+P}\left(P^{\nu\rho}\partial_{\rho}P - P^{\nu}_{\ \rho}\partial_{\mu}T^{\mu\rho}_{der}\right),$$
$$\partial_{\mu}J^{\mu} = 0 \implies u^{\mu}\partial_{\mu}Q = -Q\partial_{\mu}u^{\mu} - \partial_{\mu}J^{\mu}_{der},$$
(1.16)

to eliminate $u^{\mu}\partial_{\mu}E$ and $u^{\mu}\partial_{\mu}Q$. We want the right-hand side of eq. (1.15) to be non-negative for any fluid profile. Even if we start with a positive $\partial_{\mu}u^{\mu}$, there is no guarantee that it remains so during the course of the flow, causing a violation of the second law. The only way of ensuring that the second law is identically upheld is for us to demand the coefficient of $\partial_{\mu}u^{\mu}$ in eq. (1.15) to vanish. If at the ideal order, we identify the temperature and chemical potential of the fluid with their thermodynamic values, i.e. $\partial S/\partial E = 1/T + \mathcal{O}(\partial)$ and $\partial S/\partial Q = -\mu/T + \mathcal{O}(\partial)$, we can immediately read out this constraint as the Euler scaling relation. Therefore thermodynamic relations, which are equalities, follow from requiring the inequality of second law.

Let us briefly return to the fluid variables u^{μ} , T, and μ . At the ideal order, they are uniquely defined by their respective values in thermodynamic equilibrium. However, as we leave the equilibrium, they can admit arbitrary derivative redefinitions

$$T \to T + T_{der}, \qquad \mu \to \mu + \mu_{der}, \qquad u^{\mu} \to u^{\mu} + u^{\mu}_{der}.$$
 (1.17)

It essentially implies that not all the derivative corrections in $T_{der}^{\mu\nu}$ and J_{der}^{μ} are physical; some of them can be attributed to the inherent ambiguity in the fluid variables. Often, it is convenient to work in a hydrodynamic frame which uniquely specifies these fluid variables. For instance, in the Landau frame one chooses $T^{\mu\nu}u_{\nu} = -Eu^{\mu}$ and $J^{\mu}u_{\mu} = -Q$, which defines the fluid velocity to be along the flow of energy. There is also the Eckart frame with $T^{\mu\nu}u_{\mu}u_{\nu} = E$ and $J^{\mu} = Qu^{\mu}$, which instead aligns the fluid velocity with the flow of charge. In the bulk of this paper, we work in a third frame called the hydrostatic frame. However, we do not have enough tools to define it yet.

As an example, let us consider one-derivative corrections to the fluid constitutive relations. For simplicity, we only include tensor structures that preserve parity. We return to a more general analysis in section 3.1. To start with, let us write down the most generic one-derivative corrections in Landau frame

$$T_{\rm der}^{\mu\nu} = -\zeta P^{\mu\nu}\Theta - \eta \sigma^{\mu\nu} + \mathcal{O}(\partial^2),$$

$$J_{\rm der}^{\mu} = \kappa P^{\mu\nu}\partial_{\nu}T - \sigma P^{\mu\nu}\partial_{\nu}\mu + \mathcal{O}(\partial^2),$$
(1.18)

where we have defined

$$\Theta = \partial_{\mu} u^{\mu}, \qquad \sigma^{\mu\nu} = P^{\mu\rho} P^{\nu\sigma} \left(\partial_{(\mu} u_{\nu)} - \frac{1}{3} P_{\mu\nu} \Theta \right). \tag{1.19}$$

We have chosen not to include any terms involving $u^{\mu}\partial_{\mu}T$, $u^{\mu}\partial_{\mu}\mu$, or $u^{\mu}\partial_{\mu}u^{\nu}$, as they can be eliminated using the first order equations of motion (1.16). Here the bulk viscosity ζ , shear viscosity η , thermal conductivity κ , and electric conductivity σ are arbitrary functions of T and μ , known as *transport coefficients*.

To see what constraints are imposed by the second law of thermodynamics, let us start with the entropy current (1.13) and compute its divergence, but this time paying attention to the derivative corrections. We find

$$\partial_{\mu}J_{\rm S}^{\mu} = -\frac{1}{T}T_{\rm der}^{\mu\nu}\partial_{\mu}u_{\nu} - J_{\rm der}^{\mu}\partial_{\mu}\frac{\mu}{T} + \partial_{\mu}\left(J_{\rm S,der}^{\mu} + \frac{\mu}{T}J_{\rm der}^{\mu}\right)$$
$$= \frac{\zeta}{T}\Theta^{2} + \frac{\eta}{T}\sigma^{\mu\nu}\sigma_{\mu\nu} + T\sigma P^{\mu\nu}\partial_{\mu}\frac{\mu}{T}\partial_{\nu}\frac{\mu}{T}$$
$$- \left(\kappa - \frac{\mu}{T}\sigma\right)P^{\mu\nu}\partial_{\mu}T\partial_{\nu}\frac{\mu}{T} + \partial_{\mu}\left(J_{\rm S,der}^{\mu} + \frac{\mu}{T}J_{\rm der}^{\mu}\right) + \mathcal{O}(\partial^{3}).$$
(1.20)

Here again, we have used the equations of motion to express the entropy current divergence in the desired form. The last term implies that we can choose the derivative corrections in the entropy current to be

$$J_{\rm S,der}^{\mu} = -\frac{\mu}{T} J_{\rm der}^{\mu} = -\frac{\mu}{T} \kappa P^{\mu\nu} \partial_{\nu} T + \frac{\mu}{T} \sigma P^{\mu\nu} \partial_{\nu} \mu + \mathcal{O}(\partial^2).$$
(1.21)

The remaining term in the last line cannot be made positive definite and hence must vanish. This relates the thermal conductivity to the electric conductivity. On the other hand, the terms in the first line are manifestly positive semi-definite, which tells us that their coefficients must be non-negative, leading to the non-negativity of the viscosities and electric conductivity. Together, we have

$$\kappa = \sigma \mu / T,$$
 $\zeta \ge 0,$ $\eta \ge 0,$ $\sigma \ge 0.$ (1.22)

In summary, the constitutive relations of a parity-preserving relativistic fluid, corrected up to one-derivative order, are given as

$$T^{\mu\nu} = (E+P)u^{\mu}u^{\nu} + P\eta^{\mu\nu} - \zeta P^{\mu\nu}\Theta - \eta \sigma^{\mu\nu} + \mathcal{O}(\partial^2),$$

$$J^{\mu} = Qu^{\mu} - \sigma T\partial_{\mu}\frac{\mu}{T} + \mathcal{O}(\partial^2).$$
(1.23)

They are characterised by an equation of state $P = P(T, \mu)$ at the ideal order and three non-negative transport coefficients η , ζ , and σ at the first order. These are the standard textbook results and can be found, for example, in [10].

1.1.3 Galilean hydrodynamics

In our discussion above, we assumed the fluid to be relativistic and described its dynamics by a set of Lorentz-invariant dynamical equations. However, for most applications in our day-to-day non-relativistic lives, it is more useful to formulate a Galilean version of hydrodynamics. In a Galilean setting, we still have $T(t, \mathbf{x})$ and $\mu(t, \mathbf{x})$ as fundamental variables of our fluid, but the Galilean fluid velocity is taken to be $u^i(t, \mathbf{x})$ instead, without any normalisation condition. Furthermore, thermodynamics of Galilean fluids can also admit an independent mass density R alongside the charge density, which is absent in a relativistic description. We call the associated chemical potential $\mu_m(t, \mathbf{x})$ with

$$dP = SdT + \mu_m dR + \mu dQ, \qquad E + P = TS + \mu_m R + \mu Q. \tag{1.24}$$

To first approximation, the dynamical equations for T, $\mu_{\rm m}$, and μ are provided by the entropy, mass, and charge conservation equations respectively, while that for u^i is provided by the Navier-Stokes equation, i.e.

$$\partial_t S + \partial_i \left(S u^i \right), \qquad \partial_t R + \partial_i \left(R u^i \right), \qquad \partial_t Q + \partial_i \left(Q u^i \right), R \left(\partial_t u^i + u^j \partial_j u^i \right) = -\partial^i P.$$
(1.25)

Similar to the relativistic case, the fluids which are governed by this set of equations are known as ideal Galilean fluids.

To classify the possible derivative corrections that these equations can admit, it is useful to convert them into the Noether conservation equations. With this in mind, let us define a set of macroscopic observables for our Galilean fluid

$$\rho = R + \rho_{der}, \qquad \rho^{i} = Ru^{i} + \rho^{i}_{der},$$

$$q = Q + q_{der}, \qquad j^{i} = Qu^{i} + j^{i}_{der},$$

$$\epsilon = E + \frac{1}{2}Ru^{2} + \epsilon_{der}, \qquad \epsilon^{i} = \left(E + P + \frac{1}{2}Ru^{2}\right)u^{i} + \epsilon^{i}_{der},$$

$$p^{ij} = Ru^{i}u^{j} + P\delta^{ij} + p^{ij}_{der}.$$
(1.26)

They are the mass density, mass flux, charge density, charge flux, energy density, energy flux, and stress tensor of the fluid respectively. Note that our definition of the energy density ϵ contains contributions from both the internal energy density E as well as the kinetic energy density $\frac{1}{2}Ru^2$, as we would expect for a non-relativistic system. In the expressions above, ρ_{der} , ρ_{der}^i , q_{der} , j_{der}^i , ϵ_{der} , ϵ_{der}^i , and p_{der}^{ij} represent the possible derivative corrections that the fluid can admit as we leave the thermodynamic equilibrium. In line with our relativistic discussion, we could exploit the redefinition freedom in the hydrodynamic fields to set some of these corrections to zero. One such choice ubiquitous in Galilean hydrodynamics is the so-called mass frame, which sets $\rho_{der} = \rho_{der}^i = q_{der} = \epsilon_{der} = 0$ and aligns the fluid velocity with the flow of mass. Using the definitions in eq. (1.26), we can convert the ideal order equations of motion (1.25) into

$$\partial_t \rho + \partial_i \rho^i = 0, \qquad \partial_t q + \partial_i j^i = 0,$$

$$\partial_t \epsilon + \partial_i \epsilon^i = 0, \qquad \partial_t \rho^i + \partial_j p^{ij} = 0.$$
 (1.27)

These are the Noether conservation equations for a Galilean system, which we take to be the fundamental equations of motion for Galilean hydrodynamics. Note that we do not require them to admit any explicit derivative corrections; all the corrections enter implicitly via the Galilean currents and densities.

To find these corrections up to any given derivative order, we need to first write down the most generic expressions for various currents and densities allowed by symmetries, called the Galilean fluid constitutive relations. Listing out the tensor structures that respect spatial rotations is quite trivial, as these symmetries are manifest in the index structure of the Galilean observables. However, the Galilean boost symmetry is not manifest; under a boost $x^i \to x^i - \psi^i t$, the fluid velocity shifts as $u^i \to u^i + \psi^i$, and accordingly the Galilean observables mix in a non-trivially manner

$$\rho^{i} \to \rho^{i} - \rho\psi^{i}, \qquad j^{i} \to j^{i} - q\psi^{i}, \qquad p^{ij} \to p^{ij} - 2\rho^{(i}\psi^{j)} + \rho\psi^{i}\psi^{j},$$

$$\epsilon \to \epsilon + \frac{1}{2}\rho\psi^{2} - \rho^{i}\psi_{i}, \qquad \epsilon^{i} \to \epsilon^{i} - \epsilon\psi^{i} + \frac{1}{2}\psi^{2}\left(\rho^{i} - \rho\psi^{i}\right) - \left(p^{ij} - \psi^{i}\rho^{j}\right)\psi_{j}, \qquad (1.28)$$

while leaving the conservation equations invariant. Writing down the tensor structures that are invariant under this transformation can be quite cumbersome. For a detailed account, see the discussion in [10]. However, we can make our life considerably simpler if we introduce an auxiliary coordinate x^- and arrange the Galilean observables into one-higher dimensional Poincaré-invariant structures

$$T^{MN} = \begin{pmatrix} \times & \epsilon & \epsilon^{j} \\ \epsilon & \rho & \rho^{j} \\ \epsilon^{i} & \rho^{i} & p^{ij} \end{pmatrix}, \qquad J^{M} = \begin{pmatrix} \times \\ q \\ j^{i} \end{pmatrix}.$$
 (1.29)

Here we have chosen the coordinate system $(x^M) = (x^-, t, x^i)$. The coordinates x^- and t are taken to be null with respect to the 5-dimensional spacetime, i.e. $ds^2 = -2dx^-dt + \delta_{ij}dx^i dx^j$. The notation "×" above represents some arbitrary unphysical quantities introduced to complete the 5-dimensional tensor structures; they can be safely ignored for our purposes. Barring these, note that the tensors T^{MN} and J^M are manifestly x^- -independent. To ensure that they remain so, we only allow for x^- -independent Poincaré transformations on the 5-dimensional spacetime. With some elementary algebra, we can convince ourselves that the residual symmetries are: t and x^i translations, x^i -rotations, U(1) charge transformations, x^- translations, and x^- - x^i rotations

$$x^{-} \to x^{-} + \frac{1}{2}\psi^{2}t - \psi_{i}x^{i}, \qquad t \to t, \qquad x^{i} \to x^{i} - \psi^{i}t.$$
 (1.30)

The spacetime translations, spatial rotations, and U(1) transformations map respectively to their Galilean counterparts, while the x^- translations are identified with an emergent U(1) symmetry which is responsible for the conservation of mass. The remaining $x^{-}-x^{i}$ rotations actually map to the non-trivial Galilean boosts, as can be readily verified by applying them to eq. (1.29). We can also arrange the equations of motion (1.27) into a natural 5-dimensional form

$$\partial_M T^{MN} = 0, \qquad \partial_M J^M = 0. \tag{1.31}$$

In deriving these, we have noted that the action of ∂_{-} yields zero. These equations are precisely the conservation laws of relativistic hydrodynamics given in eq. (1.12). We have, therefore, mapped our Galilean fluids into one-higher dimensional relativistic fluids. Note however that these relativistic fluids are different from those discussed in section 1.1.2; they are anisotropic due the presence of a preferred coordinate x^{-} . We call them null fluids.

At this point, we can essentially repeat the entire analysis we did for relativistic fluids, step-by-step. For starters, the ideal Galilean fluid constitutive relations (1.26) can be mapped into ideal null fluids as

$$T^{MN} = Ru^{M}u^{N} + 2(E+P)u^{(M}V^{N)} + P\eta^{MN} + T^{MN}_{der}, \qquad J^{M} = Qu^{M} + J^{M}_{der}, \qquad (1.32)$$

where we have defined

$$u^{M} = \begin{pmatrix} \frac{1}{2}u^{2} \\ 1 \\ u^{i} \end{pmatrix}, \qquad V^{M} = \begin{pmatrix} 1 \\ 0 \\ 0 \end{pmatrix}, \qquad \eta^{MN} = \begin{pmatrix} 0 & -1 & 0 \\ -1 & 0 & 0 \\ 0 & 0 & \delta^{ij} \end{pmatrix}.$$
 (1.33)

 η^{MN} is just the higher dimensional Minkowski metric written in null coordinates, while V^{M} is the covariant representation of the null coordinate x^{-} . On the other hand, u^{M} is a doubly normalised $(u^{M}u_{M} = 0, u^{M}V_{M} = -1)$ null fluid velocity, responsible for the name "null fluids". The tensors T_{der}^{MN} and J_{der}^{M} represent the possible derivative corrections these constitutive relations can admit. The mass frame condition in this language also takes a natural form: $V_{M}T_{der}^{MN} = V_{M}J_{der}^{M} = 0$.

Once we have written down the derivative corrections in accordance with the symmetries, we should impose the second law of thermodynamics. It essentially implies the existence of an entropy density and flux

$$s = S + s_{der}, \qquad s^i = Su^i + s^i_{der}, \qquad J^M_S = \begin{pmatrix} \times \\ s \\ s^i \end{pmatrix}, \qquad (1.34)$$

whose divergence is positive semi-definite at every spacetime point, i.e.

$$\partial_t s + \partial_i s^i \ge 0 \quad \Longleftrightarrow \quad \partial_M J^M_{\mathcal{S}} \ge 0.$$
 (1.35)

Similar to the relativistic fluids, this requirement imposes some strict constraints on the Galilean fluid constitutive relations as well. We do not repeat the calculational details here, but following our relativistic steps from eq. (1.15) onward, we can easily infer that the

one-derivative corrections to the null fluid constitutive relations are given as

$$T^{MN} = Ru^{M}u^{N} + 2(E+P)u^{(M}V^{N)} + P\eta^{MN} - \eta \sigma^{MN} - \zeta P^{MN}\Theta - 2V^{(M}P^{N)R} \left(\kappa \frac{1}{T}\partial_{R}T + (\kappa_{Q} + \overline{\kappa}_{Q}) T\partial_{R}\frac{\mu}{T}\right) + \mathcal{O}(\partial^{2}), J^{M} = Qu^{M} - (\kappa_{Q} - \overline{\kappa}_{Q}) P^{MN}\frac{1}{T}\partial_{N}T - \sigma TP^{MN}\partial_{N}\frac{\mu}{T} + \mathcal{O}(\partial^{2}).$$
(1.36)

Here we have chosen to work in the mass frame and focused on the parity-preserving sector. We have also defined a projector $P^{MN} = \eta^{MN} + 2u^{(M}V^{N)}$ transverse to u^{M} and V^{M} , along with the one-derivative structures

$$\Theta = \partial_M u^M, \qquad \sigma^{MN} = P^{MR} P^{NS} \left(\partial_{(R} u_{S)} - \frac{1}{3} P_{RS} \Theta \right). \tag{1.37}$$

5 of the 6 transport coefficients appearing in eq. (1.36) satisfy a set of 4 inequality constraints among them

 $\zeta \ge 0, \qquad \eta \ge 0, \qquad \sigma \ge 0, \qquad \kappa \ge \kappa_Q^2 / \sigma,$ (1.38)

while the remaining transport coefficient $\overline{\kappa}_Q$ is completely arbitrary.

These results can easily be converted to the 4-dimensional language, providing onederivative corrections to the Galilean fluid constitutive relations in mass frame

$$j^{i} = Qu^{i} - (\kappa_{Q} - \overline{\kappa}_{Q}) \frac{1}{T} \partial^{i} T - \sigma T \partial^{i} \frac{\mu}{T} + \mathcal{O}(\partial^{2}),$$

$$\epsilon^{i} = \left(E + P + \frac{1}{2} R u^{2}\right) u^{i} - \kappa \frac{1}{T} \partial^{i} T - (\kappa_{Q} + \overline{\kappa}_{Q}) T \partial^{i} \frac{\mu}{T} - \eta \sigma^{ij} u_{j} - \zeta u^{i} \Theta + \mathcal{O}(\partial^{2}),$$

$$p^{ij} = R u^{i} u^{j} + P \delta^{ij} - \eta \sigma^{ij} - \zeta \delta^{ij} \Theta + \mathcal{O}(\partial^{2}).$$
(1.39)

Written like this, the coefficients can be identified as the bulk viscosity ζ , shear viscosity η , thermal conductivity κ , electric conductivity σ , and two thermo-electric conductivities κ_Q and $\overline{\kappa}_Q$. Note that a Galilean fluid admits many more transport coefficients compared to a relativistic fluid.

This finishes our introductory review of relativistic and Galilean hydrodynamics. We have been quite sketchy in our approach, not spending too much time on the subtleties of the hydrodynamic description. We return to these issues in chapters 2 and 4, where we approach hydrodynamics from a more formal standpoint.

1.2 Hydrodynamics with gapless modes

A major part of this thesis is concerned with how the presence of arbitrary gapless modes in the low-energy spectrum modifies the hydrodynamic framework. So that these results do not get lost in the deluge of technicalities, we outline the basic principles here. For concreteness, we choose the gapless mode in question to be a Goldstone mode φ , arising due to the spontaneous breaking of a U(1) symmetry. Hydrodynamic systems that admit such a Goldstone mode are known as superfluids. Under a global U(1) transformation, the Goldstone field φ transforms as $\varphi \to \varphi - \Lambda$, so it is more helpful to deal with its derivative $\xi_{\mu} = \partial_{\mu}\varphi$ directly, which is U(1)-invariant. The vector field ξ^{μ} is called the superfluid velocity and is taken to be $\mathcal{O}(\partial^0)$ in the hydrodynamic derivative expansion. Note that the superfluid velocity is exact, i.e. $\partial_{[\mu}\xi_{\nu]} = 0$. The dynamics of φ is governed by the *Josephson equation*

$$u^{\mu}\xi_{\mu} = u^{\mu}\partial_{\mu}\varphi = \mu + \mu_{\rm der}, \qquad (1.40)$$

where μ_{der} represents the possible derivative corrections that this equation can admit. Although well motivated from a physical standpoint, the Josephson equation is still imposed by hand in the conventional treatment of superfluid dynamics. From a hydrodynamic perspective, where all the dynamics is supposed to emerge on the general grounds of symmetries and thermodynamics, this is quite unnatural. It would be much more natural if we can somehow derive this equation from within the framework of hydrodynamics. More generally, such a derivation will be helpful when we might not have a physical intuition about the low-energy dynamics of a gapless mode. To illustrate how our construction works, let us forget the Josephson equation for now. We will see it emerge as a corollary of the second law of thermodynamics.

The equations of motion for u^{μ} , T, and μ , on the other hand, are the Noether conservation equations (1.12) as before. However, for our purposes, it is helpful to record a version of these equations which is valid even when φ is taken off-shell¹

$$\partial_{\mu}T^{\mu\nu} = K\xi^{\nu}, \qquad \partial_{\mu}J^{\mu} = -K. \tag{1.41}$$

Here we have formally denoted $K = \delta S/\delta \varphi$ with S being an effective action for hydrodynamics. The equation of motion for φ can correspondingly be represented as K = 0. If we had started with some other gapless modes than φ , these equations could be considerably different based on the symmetry properties of the gapless modes.

Similar to our previous discussion, we require the fluid to satisfy the second law of thermodynamics. We require the existence of an entropy current $J_{\rm S}^{\mu}$ whose divergence is positive semi-definite everywhere. However, this time we impose this even on arbitrary off-shell configurations of φ . This is quite a non-trivial step, and is responsible for why the second law is able to fix the Josephson equation. To see this, let us work out the ideal superfluid constitutive relations. Using symmetries, at the ideal order the most generic expressions for $T^{\mu\nu}$ and J^{μ} are given as

$$T^{\mu\nu} = (E+P)u^{\mu}u^{\nu} + P\eta^{\mu\nu} + R_{s}\xi^{\mu}\xi^{\nu} + \mathcal{O}(\partial), J^{\mu} = Qu^{\mu} + Q_{s}\xi^{\mu} + \mathcal{O}(\partial),$$
(1.42)

where the coefficients are arbitrary functions of T, μ , and $\mu_s = -\frac{1}{2}\xi^{\mu}\xi_{\mu}$. On the other

¹The easiest way to derive these equations is to consider a low energy effective action for φ in thermodynamic equilibrium, i.e. $S[g_{\mu\nu}, A_{\mu}, \varphi]$, in the presence of arbitrary background sources $g_{\mu\nu}$ and A_{μ} to couple to the currents $T^{\mu\nu}$ and J^{μ} respectively. Eq. (1.41) follows from here if we require our action to be invariant under an infinitesimal diffeomorphism: $g_{\mu\nu} \to g_{\mu\nu} + \pounds_{\chi}g_{\mu\nu}, A_{\mu} \to A_{\mu} + \pounds_{\chi}A_{\mu}, \varphi \to \varphi + \pounds_{\chi}\varphi$ and a U(1) gauge transformation $A_{\mu} \to A_{\mu} + \partial_{\mu}\varphi, \varphi \to \varphi - \Lambda$.

hand, K is just an arbitrary scalar which is unfixed for now. We have not included any dependence on $u^{\mu}\xi_{\mu} = u^{\mu}\partial_{\mu}\varphi$, as this corresponds to the time derivative of φ which can in principle be eliminated using the φ equation of motion. In principle, we could also include a term $2\lambda u^{(\mu}\xi^{\nu)}$ in the energy-momentum tensor, but this gets switched off by the second law of thermodynamics in the end, so we omit it here for the clarity of notation. Using the constitutive relations (1.42), the scalar components of the conservation equations can be evaluated to be

$$\partial_{\mu}T^{\mu\nu} = K\xi^{\nu} \implies u^{\mu}\partial_{\mu}E + R_{s}u^{\mu}\partial_{\mu}\mu_{s} + (E+P)\Theta = -u^{\lambda}\xi_{\lambda}\left(K - \partial_{\mu}\left(R_{s}\xi^{\mu}\right)\right),$$

$$\zeta^{\nu}\left((E+P)u^{\mu}\partial_{\mu}u_{\nu} + \partial_{\nu}P - R_{s}\partial_{\nu}\mu_{s}\right) = \zeta^{2}\left(K - \partial_{\mu}\left(R_{s}\xi^{\mu}\right)\right),$$

$$\partial_{\mu}J^{\mu} = -K \implies u^{\mu}\partial_{\mu}Q + Q\Theta + \partial_{\mu}\left(Q_{s}\xi^{\mu}\right) = -K,$$
(1.43)

where $\zeta^{\mu} = P^{\mu\nu}\xi_{\nu}$. We use these to eliminate the one-derivative scalars $u^{\mu}\partial_{\mu}E$, $\zeta^{\nu}u^{\mu}\partial_{\mu}u_{\nu}$, and $u^{\mu}\partial_{\mu}Q$ respectively. Let us start with an arbitrary zero-derivative entropy current $J_{\rm S}^{\mu} = Su^{\mu} + S_s\xi^{\mu}$ and compute its divergence

$$\partial_{\mu}J_{S}^{\mu} = \frac{\partial S}{\partial E}u^{\mu}\partial_{\mu}E + \frac{\partial S}{\partial Q}u^{\mu}\partial_{\mu}Q + \frac{\partial S}{\partial\mu_{s}}u^{\mu}\partial_{\mu}\mu_{s} + S\Theta$$

$$= -\left(\frac{\partial S}{\partial E}u^{\lambda}\xi_{\lambda} + \frac{\partial S}{\partial Q}\right)(K - \partial_{\mu}(R_{s}\xi^{\mu}))$$

$$\left(S_{s} - (Q_{s} + R_{s})\frac{\partial S}{\partial Q}\right)\partial_{\mu}\xi^{\mu} - \frac{\partial S}{\partial Q}\xi^{\mu}\partial_{\mu}(Q_{s} + R_{s}) + \xi^{\mu}\partial_{\mu}S_{s}$$

$$+ \left(\frac{\partial S}{\partial\mu_{s}} - R_{s}\frac{\partial S}{\partial E}\right)u^{\mu}\partial_{\mu}\mu_{s} + \left(S - (E + P)\frac{\partial S}{\partial E} - Q\frac{\partial S}{\partial Q}\right)\Theta.$$
(1.44)

We have taken S to be a function of E, Q, and μ_s , and used the first order equations of motion in the second step. Every term in the last two lines of this expression is linearly independent and cannot be made positive semi-definite. Thus all the respective coefficients must vanish, leading to

$$E + P = TS + \mu Q,$$
 $R_s = T \frac{\partial S}{\partial \mu_s},$ $Q_s = -R_s,$ $S_s = 0.$ (1.45)

As before, we have identified the temperature and chemical potential with their thermodynamic values: $\partial S/\partial E = 1/T + \mathcal{O}(\partial)$ and $\partial S/\partial Q = -\mu/T + \mathcal{O}(\partial)$. With this, the first condition above can be read as the thermodynamic Euler equation. The first law of thermodynamics and the Gibbs-Duhem equations for a superfluid also follow from the differential expression for S, leading to

$$dE = TdS + \mu dQ - R_s d\mu_s, \qquad dP = SdT + Qd\mu + R_s d\mu_s.$$
(1.46)

Moving on, we still need to take care of the first term in eq. (1.44). It can be made explicitly positive semi-definite if we choose

$$K = -\frac{\alpha}{T} \left(u^{\mu} \xi_{\mu} - \mu \right) + \partial_{\mu} \left(R_s \xi^{\mu} \right) + \mathcal{O}(\partial^2), \qquad (1.47)$$

for some non-negative transport coefficient α . Using this, the φ equation of motion K = 0

takes the form

$$u^{\mu}\xi_{\mu} = \mu + \frac{T}{\alpha}\partial_{\mu}\left(R_{s}\xi^{\mu}\right) + \mathcal{O}(\partial).$$
(1.48)

This is precisely the Josephson equation. Note that this equation is still only accurate at the zero-derivative order. There can be further one-derivative corrections which we have not analysed here. We return to these in section 3.2.

The framework we have presented here is actually applicable to much more than just superfluids. For a quick example, consider a neutral fluid with an energy momentum tensor $T^{\mu\nu}$. The hydrodynamic fields are u^{μ} and T, and their dynamics is provided by the energy-momentum conservation equation $\partial_{\mu}T^{\mu\nu} = 0$. Let us add to this system a dynamical U(1) gauge field A_{μ} , with field strength $F_{\mu\nu}$, and call the respective equations of motion to be $J^{\mu} = 0$. Due to the global part of this U(1) gauge symmetry, the system can also admit a chemical potential μ , whose dynamics is provided by $\partial_{\mu}J^{\mu} = 0$. In an A_{μ} -off-shell configuration, the energy-momentum conservation equation modifies to $\partial_{\mu}T^{\mu\nu} = F^{\nu\rho}J_{\rho}$. This is essentially the setup for relativistic magnetohydrodynamics [54]. To see this, let us write down the most generic parity-preserving expression for $T^{\mu\nu}$

$$T^{\mu\nu} = (E+P)u^{\mu}u^{\nu} + P\eta^{\mu\nu} + \alpha_{\rm BB}B^{\mu}B^{\nu} + \mathcal{O}(\partial), \qquad (1.49)$$

where $B^{\mu} = \frac{1}{2} \epsilon^{\mu\nu\rho\sigma} u_{\nu} F_{\rho\sigma}$. All the coefficients are taken to be functions of T, μ , and B^2 . Similar to the superfluid case, we have chosen not to include any terms involving $E_{\mu} = F_{\mu\nu}u^{\nu}$, as they can be eliminated using the A_{μ} equations of motion. Performing the same analysis as before, we can read out the thermodynamic relations for magnetohydrodynamics

$$E + P = TS + \mu Q - \alpha_{\rm BB}B^2$$
, $dE = TdS + \mu dQ - \frac{1}{2}\alpha_{\rm BB}dB^2$. (1.50)

On the other hand, the A_{μ} equations of motion take the form

$$J^{\mu} = Qu^{\mu} - \sigma \left(T \partial^{\mu} \frac{\mu}{T} - E^{\mu} \right) - T \partial_{\nu} \left(\alpha_{\rm BB} F^{\nu\mu} \right) + \mathcal{O}(\partial^2) = 0, \qquad (1.51)$$

for some non-negative transport coefficient σ . These are the finite temperature incarnation of Maxwell's equations. They primarily tell us that on-shell Q and E^{μ} are $\mathcal{O}(\partial^1)$, which is a characteristic of magnetohydrodynamics. These results can be directly compared to [54].

This is where we close our discussion of gapless modes for now. Later in chapter 2, we revisit the construction of hydrodynamics with arbitrary gapless modes in a more generalised setting. We also briefly mention another application of gapless modes to hydrodynamics with surfaces.

1.3 Overview and organisation

The material presented in this thesis is quite technical and involved, so we dedicate the remainder of this introductory chapter to chart the essential points. We also take this opportunity to segregate the novel results presented in this thesis from the review material derived from previous works.

The main goal of chapter 2 is to present a universal framework of relativistic hydrodynamics that includes arbitrary gapless modes in its spectrum. Alongside these novel results, the chapter also serves as a comprehensive review of the off-shell formalism of relativistic hydrodynamics, building on the fundamental considerations of symmetries and thermal field theories. We start section 2.1 with a fundamental discussion of symmetries and conserved currents in quantum field theories, motivating their coupling to curved torsional background manifolds. We follow it with a brief review of Einstein-Cartan geometries, giving way to a general discussion on Ward identities and anomalies. Having set the stage, we introduce the formal aspects of relativistic hydrodynamics in section 2.2; we review the essentials of thermal field theory and use it motivate the hydrodynamic fields, their equations of motion, and hydrodynamic constitutive relations. Importantly, we highlight in this section how additional gapless dynamical modes can be dealt with in a hydrodynamic framework, which includes e.g. Goldstone modes of broken internal symmetries for superfluid dynamics and dynamical gauge fields for magnetohydrodynamics. This aspect of chapter 2 is novel to this thesis and has not been published separately. We cap off this section with some comments on the hydrostatic principle and the role of thermal equilibrium in hydrodynamics.

The second law of thermodynamics is at the heart of our modern understanding of hydrodynamics, therefore it gets its own section 2.3. We discuss the hydrodynamic incarnation of the law and explore the classification of hydrodynamic transport it implies. The discussion mainly follows the off-shell formalism of non-Abelian superfluid dynamics we proposed in [1], extended to include spin currents, torsional backgrounds, and arbitrary gapless modes. We find that all the hydrodynamic transport can be classified into one of the following five classes: Class A for anomaly induced transport, Class H_S and H_V for hydrostatic transport that governs the behaviour of the fluid in an equilibrium configuration, Class D for dissipative transport that causes the production of entropy when we leave equilibrium, and finally Class \overline{D} for transport that is neither hydrostatic nor dissipative. There is also a Class S for transport that causes the flow of entropy without any flow of energy-momentum or charge; we mention it for completeness, but it does not play any role in our later discussion. Class H_S and \overline{D} are characterised by certain transport coefficients that are arbitrary functions of the thermodynamic variables: temperature and chemical potentials. Class D, as well, is characterised by arbitrary transport coefficients, but specific linear combinations of these are required to be sign-definite by the second law of thermodynamics. On the other hand, Class A and H_V transport is completely fixed up to some dimensionless constants, called anomaly coefficients and transcendental anomaly coefficients respectively. For Class A, H_V , and H_S our construction draws heavily from the work of [52], but our characterisation of Class D, \overline{D} , and S is novel.

In chapter 3, we apply the abstract concepts from chapter 2 to some concrete examples. In section 3.1, we revisit the 4-dimensional relativistic fluids from section 1.1.2, but this time from the viewpoint of the off-shell formalism, allowing also for the parity-violating effects. Truncated to one-derivative order, we find a total of 4 constants and 4 transport coefficients classified as $(1_A, 1_{H_S}, 3_{H_V}, 3_D, 0_{\overline{D}})$. The Class H_S transport coefficient is the ideal order pressure P, while the Class D coefficients are the non-negative shear viscosity η , bulk viscosity ζ , and electric conductivity σ . These results are standard in the literature and have been taken directly from [52]. Next, in section 3.2, we present a generalisation of these results to one-derivative order relativistic superfluids, taken from our work in [4]. We illustrate how the Josephson equation for superfluids naturally emerges from the off-shell formalism. We find that the spectrum of relativistic superfluids is substantially richer than the ordinary fluids, characterised by a total of 2 constants and 31 transport coefficients with classification $(1_A, 5_{H_S}, 1_{H_V}, 15_D, 11_{\overline{D}})$. The 15 dissipative transport coefficients, which include viscosities, conductivities, and their numerous generalisations, satisfy 7 non-trivial inequalities among them. We also find our first non-trivial examples of first-order Class H_S and \overline{D} transport coefficients. We also illustrate how these results can be extended to non-Abelian superfluids. In section 3.3, we study the modification of the constitutive relations of a relativistic fluid near a surface and outline a derivation of the Young-Laplace equation. We find that at the one-derivative order the surface dynamics of a fluid is characterised by two additional transport coefficients: an arbitrary Class H_S surface tension γ and a non-negative Class D transport coefficient α . These results are derived from our work in [6].

In chapter 4, we adapt the off-shell formalism to Galilean hydrodynamics. In a series of collaborative papers [2-5], we have developed a new language to study Galilean fluids. called null fluids, based on the technique of null reduction. These are essentially "relativistic" fluids living on a suitably engineered spacetime manifold, called a null background, with one higher dimension compared to the Galilean fluid of interest. In section 4.1, we start from the Galilean symmetry algebra and review the construction of null backgrounds in detail. Following our work in [5], we discuss the Ward identities for Galilean-invariant field theories and propose a classification of the plausible 't Hooft anomalies. We follow this with a prescription for null reduction in section 4.2, which dimensionally reduces the (d+1)-dimensional null background formulation of a Galilean field theory to its conventional d-dimensional formulation. In the d-dimensional picture, we choose to work in the socalled Newton-Cartan framework, which is a covariant language for Galilean field theories coupled to curved spacetime backgrounds and manifests almost all the Galilean symmetries. Alongside, we also include a self-contained review of torsional Newton-Cartan geometries. Finally, in section 4.3, we discuss hydrodynamics on null backgrounds along with its null reduction to obtain Galilean hydrodynamics. We provide a classification scheme implied by the second law of thermodynamics and explore the constraints imposed by it up to all orders in the derivative expansion. Except for the technical details in Class A and H_V , we find the qualitative results to be exactly the same as for the relativistic case. In particular, the second law imposes some strict equality constraints at every derivative-order in the hydrostatic sector, while only requires some inequalities on the one-derivative transport coefficients in the non-hydrostatic sector and none thereafter.

We apply the technique of null fluids to study some examples of Galilean hydrodynamics in chapter 5. In section 5.1, we consider the 4-dimensional ordinary Galilean fluids from section 1.1.3, but in the off-shell formalism. The results have been directly taken from our work in [3]. Including the ideal order pressure, we find a total of 5 constants and 7 transport coefficients classified as $(1_{\rm A}, 1_{\rm H_S}, 4_{\rm H_V}, 5_{\rm D}, 1_{\overline{\rm D}})$. There are the shear viscosity η , bulk viscosity ζ , electric conductivity σ , thermal conductivity κ , and thermo-electric conductivity κ_Q in Class D, while there is another thermo-electric conductivity $\overline{\kappa}_Q$ in Class $\overline{\rm D}$. The inequalities in Class D are $\eta \geq 0$, $\zeta \geq 0$, $\sigma \geq 0$ and $\kappa \geq \kappa_Q^2/\sigma$. Generalising to Galilean superfluids in section 5.2, taken from our work in [4], we find that these numbers skyrocket. We find a total of 2 constants and 52 transport coefficients, classified as $(1_{\rm A}, 7_{\rm H_S}, 1_{\rm H_V}, 25_{\rm D}, 20_{\overline{\rm D}})$. The 25 transport coefficients in Class D satisfy 9 inequalities among them. The remaining 7 coefficients in Class H_S and 20 in Class $\overline{\rm D}$ are totally unconstrained. In section 5.3, we briefly comment on the surface dynamics in Galilean fluids taken from our work in [6]. Like the relativistic case, we find that the surface transport includes an arbitrary surface tension γ in Class H_S and a non-negative transport coefficient α in Class D. We also derive the Galilean version of the Young-Laplace equation.

Finally, we close in chapter 6 with some commentary on the results of this thesis and a discussion of possible research directions to be explored in the future.
In chapter 1 we gave a basic introduction to the principles of relativistic hydrodynamics. We focused on a fluid living on a flat spacetime without any external electromagnetic fields and heuristically derived its constitutive relations up to one-derivative order. In this chapter we deal with issues more formally, accounting for some technicalities which we had swept under the rug in our heuristic introduction. In addition to the spacetime Poincaré symmetry, we permit our fluid to have an internal, possibly non-Abelian, symmetry Lie group G. In the most familiar example of electromagnetically charged fluids this group is just U(1), but we can also consider more exotic groups like the SU(3) × SU(2) × U(1) of the standard model. The hydrodynamics thus constructed has a direct application to the quark-gluon plasma encountered in accelerator experiments.

A crucial difference compared to most of the existing literature on relativistic hydrodynamics is that we consider a background spacetime with torsion and work in the vielbein formalism. This allows us to describe fluids with an independent "spin current". A secondary motivation for this is our discussion on Galilean fluids in chapter 4, which appears to be mathematically more natural in the presence of torsion. We provide a self-contained introduction to the vielbein formalism and torsion as we go along, but for more details, an excellent review can be found in [85].

2.1 Preparing the background

2.1.1 Symmetries, currents, and background sources

Symmetries are sacrosanct in physics. Our understanding of the universe is based on, and is very often guided by, the principles of symmetries. In this chapter, we are interested in relativistic quantum field theories which respect spacetime Poincaré symmetries. The corresponding Lie-algebra is generated by

Spacetime translations: P_{α} , Lorentz transformations: $J_{\alpha\beta}$, (2.1a)

with commutation relations

$$[P_{\alpha}, P_{\beta}] = 0, \qquad [J_{\alpha\beta}, P_{\gamma}] = i \left(\eta_{\alpha\gamma} P_{\beta} - \eta_{\beta\gamma} P_{\alpha}\right), [J_{\alpha\beta}, J_{\gamma\delta}] = i \left(\eta_{\alpha\gamma} J_{\beta\delta} - \eta_{\alpha\delta} J_{\beta\gamma} - \eta_{\beta\gamma} J_{\alpha\delta} + \eta_{\beta\delta} J_{\alpha\gamma}\right).$$
(2.1b)

Here $\eta_{\alpha\beta}$ is the pseudo-Riemannian flat Minkowski metric diag(-1, +1, +1, ...). The indices $\alpha, \beta, ...$ run over the *d* spacetime coordinates. It will be helpful to familiarise oneself with this algebra, as we return to it time and again in this work. The generators $J_{\alpha\beta}$ naturally

span a Lorentz algebra SO(d-1,1) which are interpreted as Lorentz-boost and angular momentum operators. The spacetime translation generators P_{α} are mutually commuting SO(d-1,1) vectors which are seen as spacetime momentum operators. Occasionally, we also permit our theories of interest to admit a Lie group G of internal symmetries, like U(1) for fields charged under electromagnetism or $SU(3) \times SU(2) \times U(1)$ for the full standard model. We call the associated Lie algebra \mathfrak{g} , which is endowed with a Lie bracket denoted by $[\circ, \circ]$ and a positive semi-definite inner product denoted by $\circ \cdot \circ$, where " \circ " is a placeholder for the elements of \mathfrak{g} .

Let us say that the theory we are seeking to describe is defined over a *d*-dimensional spacetime manifold \mathcal{M} . We denote the coordinates on \mathcal{M} by the Greek letters μ, ν, \ldots . Given the symmetries we are working with, Noether's theorem postulates that the spectrum of our theory must contain a set of associated conserved currents

Energy-momentum tensor:
$$T^{\mu}_{\ \alpha}$$
, Spin current: $\Sigma^{\mu\alpha}_{\ \beta}$,
Charge current: J^{μ} . (2.2)

The indices α, β, \ldots can now be seen as coordinates on a frame bundle $F\mathcal{M}$. To compute the quantum expectation values of these currents in a path-integral formalism, it is convenient to couple the theory to some non-dynamical background sources, one for each current. To achieve this, we introduce on our spacetime manifold \mathcal{M} a set of fields

Vielbein:
$$e^{\alpha}_{\ \mu}$$
, Spin connection: $C^{\alpha}_{\ \mu\beta}$,
Gauge connection: A_{μ} . (2.3)

If $\mathcal{Z}[e^{\alpha}_{\mu}, C^{\alpha}_{\mu\beta}, A_{\mu}]$ is the field theory generating functional for our theory, which we almost always exchange for $W = -i \ln \mathcal{Z}$, then we can compute the quantum expectation values of various Noether currents via

$$\delta W = \int_{\mathcal{M}} \mathrm{d}^{d} x \sqrt{-g} \Big(\langle T^{\mu}_{\ \alpha} \rangle \delta e^{\alpha}_{\ \mu} + \langle \Sigma^{\mu \alpha}_{\ \beta} \rangle \delta C^{\beta}_{\ \mu \alpha} + \langle J^{\mu} \rangle \cdot \delta A_{\mu} \Big).$$
(2.4)

The statement of symmetries can also be made precise in this language, by requiring that the generating functional $W[e^{\alpha}_{\ \mu}, C^{\alpha}_{\ \mu\beta}, A_{\mu}]$ is invariant under an infinitesimal local diffeomorphism, $\mathfrak{so}(d, 1)$ rotation, and \mathfrak{g} transformation (together represented by diff × $\mathfrak{so}(d-1, 1) \times \mathfrak{g}$) of the background fields, modulo plausible anomalies. In terms of a set of parameters $\mathfrak{X} = (\chi^{\mu}, \Lambda^{\Sigma \alpha}_{\chi}{}_{\beta}, \Lambda_{\chi})$, these variations are defined as

$$\delta_{\chi} e^{\alpha}{}_{\mu} = \pounds_{\chi} e^{\alpha}{}_{\mu} - \Lambda_{\chi}^{\Sigma\alpha}{}_{\beta} e^{\beta}{}_{\mu},$$

$$\delta_{\chi} C^{\alpha}{}_{\mu\beta} = \pounds_{\chi} C^{\alpha}{}_{\mu\beta} + \partial_{\mu} \Lambda_{\chi}^{\Sigma\alpha}{}_{\beta} + [C_{\mu}, \Lambda_{\chi}^{\Sigma}]^{\alpha}{}_{\beta},$$

$$\delta_{\chi} A_{\mu} = \pounds_{\chi} A_{\mu} + \partial_{\mu} \Lambda_{\chi} + [A_{\mu}, \Lambda_{\chi}],$$
(2.5a)

where \pounds_{χ} denotes a Lie derivative along χ^{μ} . These transformations form an algebra: $[\delta_{\chi}, \delta_{\mathcal{P}}] = \delta_{[\chi,\mathcal{P}]}$ where $[\chi, \mathcal{P}] = \delta_{\chi} \mathcal{P} = -\delta_{\mathcal{P}} \chi$, provided that we define the action of χ on another set of symmetry parameters $\mathcal{P} = (\psi^{\mu}, \Lambda_{\psi}^{\Sigma \alpha}{}_{\beta}, \Lambda_{\psi})$ as

$$\delta_{\chi}\psi^{\mu} = \pounds_{\chi}\psi^{\mu},$$

$$\delta_{\chi}\Lambda_{\psi}^{\Sigma\alpha}{}_{\beta} = \pounds_{\chi}\Lambda_{\psi}^{\Sigma\alpha}{}_{\beta} + [\Lambda_{\psi}^{\Sigma},\Lambda_{\chi}^{\Sigma}]^{\alpha}{}_{\beta} - \pounds_{\psi}\Lambda_{\chi}^{\Sigma\alpha}{}_{\beta}$$

$$\delta_{\chi}\Lambda_{\psi} = \pounds_{\chi}\Lambda_{\psi} + [\Lambda_{\psi},\Lambda_{\chi}] - \pounds_{\psi}\Lambda_{\chi}.$$
(2.5b)

One might wonder what these transformations have to do with the Poincaré symmetry algebra we originally started with. The simplest way to see the connection is to split the operator δ_{χ} in terms of the Poincaré generators as

$$\delta_{\chi} = -i \,\chi^{\mu} e^{\alpha}{}_{\mu} P_{\alpha} - \frac{i}{2} \left(\Lambda^{\Sigma \,\alpha}{}_{\beta} + \chi^{\mu} C^{\alpha}{}_{\mu\beta} \right) J^{\beta}{}_{\alpha} - i \left(\Lambda_{\chi} + \chi^{\mu} A_{\mu} \right) \cdot Q. \tag{2.6}$$

Explicitly computing the commutator $[\delta_{\mathfrak{X}}, \delta_{\mathcal{P}}]$ in terms of eq. (2.5) and equating it to $\delta_{[\mathfrak{X},\mathcal{P}]}$, we can verify that the generators almost satisfy the Poincaré algebra (2.1), except that the spacetime translations no longer commute with each other. Instead, their commutator is sourced by the spacetime torsion, curvature, and field strength associated with the background fields (see the next subsection for the respective definitions) leading to

$$[P_{\alpha}, P_{\beta}] = -i e_{\alpha}^{\ \mu} e_{\beta}^{\ \nu} \left(T^{\gamma}_{\ \mu\nu} P_{\gamma} + \frac{1}{2} R^{\ \gamma}_{\mu\nu} J^{\delta}_{\ \gamma} + F_{\mu\nu} \cdot Q \right).$$
(2.7)

This also matches our intuitive understanding of the "curved" backgrounds. More importantly, when the background fields are reverted to being flat, for example by setting $e^{\alpha}_{\ \mu} = \delta^{\alpha}_{\ \mu}$ and $C^{\alpha}_{\ \mu\beta} = A_{\mu} = 0$, we recover the full Poincaré algebra (2.1).

2.1.2 Einstein-Cartan geometries

Although not directly relevant as a model of our universe, we see that spacetime geometries with torsion naturally appear as a tool when trying to couple Poincaré invariant quantum field theories with arbitrary background sources. Commonly known as Einstein-Cartan geometries, these were first introduced by Élie Cartan in 1922. We review here some elementary aspects of these geometries which we require in the course of this work. For a more detailed review, please refer to a standard reference like [85]. Let us consider a *d*-dimensional spacetime manifold \mathcal{M} with a metric $g_{\mu\nu}$ and a generic metric compatible affine connection $\Gamma^{\lambda}{}_{\mu\sigma}$ given by

$$\Gamma^{\lambda}{}_{\mu\sigma} = \frac{1}{2}g^{\lambda\rho} \left(\partial_{\mu}g_{\rho\sigma} + \partial_{\sigma}g_{\rho\mu} - \partial_{\rho}g_{\mu\sigma} + \mathcal{T}_{\rho\mu\sigma} - \mathcal{T}_{\mu\sigma\rho} - \mathcal{T}_{\sigma\mu\rho}\right).$$
(2.8)

The antisymmetric part of the affine connection $T^{\lambda}{}_{\mu\nu} = -T^{\lambda}{}_{\nu\mu}$ is called the Cartan torsion tensor. We also define on \mathcal{M} a \mathfrak{g} -valued gauge field A_{μ} , and take D_{μ} to be the covariant derivative operator associated with $\Gamma^{\lambda}{}_{\mu\sigma}$ and A_{μ} . We can work out the Riemann curvature tensor and gauge field strength associated with these connections as

$$R_{\mu\nu}{}^{\lambda}{}_{\sigma} = 2\partial_{[\mu}\Gamma^{\lambda}{}_{\nu]\sigma} + 2\Gamma^{\lambda}{}_{[\mu\rho}\Gamma^{\rho}{}_{\nu]\sigma}, \qquad F_{\mu\nu} = 2\partial_{[\mu}A_{\nu]} + [A_{\mu}, A_{\nu}]. \tag{2.9}$$

Physical theories living on \mathcal{M} are required to respect spacetime diffeomorphisms and internal *G*-transformations. Let us denote an infinitesimal such transformation by parameters $\mathfrak{X} = (\chi^{\mu}, \Lambda_{\chi})$, where χ^{μ} is a vector field and Λ_{χ} is a g-valued scalar field. The derivative operator D_{μ} acts on these fields as

$$D_{\mu}\chi^{\nu} = \partial_{\mu}\chi^{\nu} + \Gamma^{\nu}{}_{\mu\rho}\chi^{\rho}, \qquad D_{\mu}\Lambda_{\chi} = \partial_{\mu}\Lambda_{\chi} + [A_{\mu},\Lambda_{\chi}].$$
(2.10)

Under the action of \mathfrak{X} , the background fields transform as

$$\delta_{\chi}g_{\mu\nu} = \pounds_{\chi}g_{\mu\nu} = 2D_{(\mu}\chi_{\nu)} - T_{(\mu\nu)\sigma}\chi^{\sigma},$$

$$\delta_{\chi}\Gamma^{\lambda}{}_{\mu\sigma} = \pounds_{\chi}\Gamma^{\lambda}{}_{\mu\sigma} + \partial_{\mu}\partial_{\sigma}\chi^{\lambda} = D_{\mu}\left(D_{\sigma}\chi^{\lambda} + \chi^{\nu}T^{\lambda}{}_{\nu\sigma}\right) + \chi^{\nu}R_{\nu\mu}{}^{\lambda}{}_{\sigma},$$

$$\delta_{\chi}A_{\mu} = \pounds_{\chi}A_{\mu} + D_{\mu}\Lambda_{\chi} = D_{\mu}\left(\Lambda_{\chi} + \chi^{\nu}A_{\nu}\right) + \chi^{\nu}F_{\nu\mu}.$$
(2.11)

To this end, our background geometry is described by a spacetime manifold \mathcal{M} with fields $(g_{\mu\nu}, \Gamma^{\lambda}{}_{\mu\sigma}, A_{\mu})$ modded by local diff $\times \mathfrak{g}$ transformations.

In the case of torsional geometries, however, it is more natural to shift to the vielbein formalism, which we describe in the following. The condition of local flatness of a manifold allows us to define an isomorphism between the tangent bundle $T\mathcal{M}$ and a pseudo-Riemannian frame bundle $F\mathcal{M} = \mathbb{R}^{(d-1,1)}$, realised in terms of a vielbein $e^{\alpha}{}_{\mu}$ and its inverse $e_{\alpha}{}^{\mu}$. The vielbein is essentially a d^2 component matrix, defined by the requirement that it maps the metric $g_{\mu\nu}$ on \mathcal{M} to a flat Minkowski metric $\eta_{\alpha\beta}$ on $F\mathcal{M}$,

$$g_{\mu\nu}e_{\alpha}^{\ \mu}e_{\beta}^{\ \nu} = \eta_{\alpha\beta}, \qquad g^{\mu\nu}e^{\alpha}_{\ \mu}e^{\beta}_{\ \nu} = \eta^{\alpha\beta}. \tag{2.12}$$

We often use the vielbein to freely switch between the spacetime indices μ, ν, \ldots and the flat indices α, β, \ldots . Note that the defining equation (2.12) for a vielbein has $\frac{1}{2}d(d+1)$ components, so it leaves $\frac{1}{2}d(d-1)$ of its components undetermined. This redundancy can be attributed to a SO(d-1, 1) Lorentz symmetry on $F\mathcal{M}$ that acts on the vielbein as $e^{\alpha}{}_{\mu} \rightarrow O^{\alpha}{}_{\beta}e^{\beta}{}_{\mu}$ where $O^{\alpha}{}_{\gamma}O^{\beta}{}_{\delta}\eta_{\alpha\beta} = \eta_{\gamma\delta}$. It is trivial to check that eq. (2.12) is invariant under this transformation. We can define a spin-connection for fields transforming in some non-trivial representation of this symmetry

$$C^{\alpha}{}_{\mu\beta} \equiv e_{\beta}{}^{\sigma} \left(e^{\alpha}{}_{\rho} \Gamma^{\rho}{}_{\mu\sigma} - \partial_{\mu} e^{\alpha}{}_{\sigma} \right), \qquad (2.13)$$

which has the same amount of information as $\Gamma^{\lambda}{}_{\mu\sigma}$. One can check that with this connection, $D_{\mu}e^{a}{}_{\nu} = 0$, which is tantamount to the metric compatibility of $\Gamma^{\lambda}{}_{\mu\sigma}$. Switching momentarily to differential forms notation we define

$$\boldsymbol{e}^{\alpha} = e^{\alpha}{}_{\mu} \mathrm{d}x^{\mu}, \qquad \boldsymbol{C}^{\alpha}{}_{\beta} = C^{\alpha}{}_{\mu\beta} \mathrm{d}x^{\mu}, \qquad \boldsymbol{A} = A_{\mu} \mathrm{d}x^{\mu}, \tag{2.14}$$

in terms of which the torsion, curvature, and field strength can be expressed as

$$\boldsymbol{T}^{\alpha} = \frac{1}{2} \operatorname{T}^{\alpha}{}_{\mu\nu} \mathrm{d}x^{\mu} \wedge \mathrm{d}x^{\nu} = \mathrm{d}\boldsymbol{e}^{\alpha} + \boldsymbol{C}^{\alpha}{}_{\beta} \wedge \boldsymbol{e}^{\beta} = \left(\partial_{[\mu}\boldsymbol{e}^{\alpha}{}_{\nu]} + \boldsymbol{C}^{\alpha}{}_{[\mu\beta}\boldsymbol{e}^{\beta}{}_{\nu]}\right) \mathrm{d}x^{\mu} \wedge \mathrm{d}x^{\nu}$$

$$= \frac{1}{2} \boldsymbol{e}^{\alpha}{}_{\lambda} \operatorname{T}^{\lambda}{}_{\mu\nu} \mathrm{d}x^{\mu} \wedge \mathrm{d}x^{\nu},$$

$$\boldsymbol{R}^{\alpha}{}_{\beta} = \frac{1}{2} R_{\mu\nu}{}^{\alpha}{}_{\beta} \mathrm{d}x^{\mu} \wedge \mathrm{d}x^{\nu} = \mathrm{d}\boldsymbol{C}^{\alpha}{}_{\beta} + \boldsymbol{C}^{\alpha}{}_{\gamma} \wedge \boldsymbol{C}^{\gamma}{}_{\beta} = \left(\partial_{[\mu}\boldsymbol{C}^{\alpha}{}_{\nu]\beta} + \boldsymbol{C}^{\alpha}{}_{[\mu\gamma}\boldsymbol{C}^{\gamma}{}_{\nu]\beta}\right) \mathrm{d}x^{\mu} \wedge \mathrm{d}x^{\nu}$$

$$= \frac{1}{2} \boldsymbol{e}^{\alpha}{}_{\lambda} \boldsymbol{e}_{\beta}{}^{\sigma} R_{\mu\nu}{}^{\lambda}{}_{\sigma} \mathrm{d}x^{\mu} \wedge \mathrm{d}x^{\nu},$$

$$\boldsymbol{F} = \frac{1}{2} F_{\mu\nu} \mathrm{d}x^{\mu} \wedge \mathrm{d}x^{\nu} = \mathrm{d}\boldsymbol{A} + \boldsymbol{A} \wedge \boldsymbol{A}.$$
(2.15)

Under an infinitesimal local diff $\times \mathfrak{so}(d-1,1) \times \mathfrak{g}$ transformation parametrised by $\mathfrak{X} = (\chi^{\mu}, \Lambda^{\Sigma^{\alpha}}_{\chi}_{\beta}, \Lambda_{\chi})$ various background fields vary according to

$$\delta_{\chi}e^{\alpha}{}_{\mu} = \pounds_{\chi}e^{\alpha}{}_{\mu} - \Lambda^{\Sigma\alpha}_{\chi\beta}e^{\beta}{}_{\mu} = e^{\alpha}{}_{\nu}D_{\mu}\chi^{\nu} + \chi^{\nu}T^{\alpha}{}_{\nu\mu} - \left(\Lambda^{\Sigma\alpha}_{\chi\beta} + \chi^{\nu}C^{\alpha}{}_{\nu\beta}\right)e^{\beta}{}_{\mu},$$

$$\delta_{\chi}C^{\alpha}{}_{\mu\beta} = \pounds_{\xi}C^{\alpha}{}_{\mu\beta} + D_{\mu}\Lambda^{\Sigma\alpha}_{\chi} = D_{\mu}\left(\Lambda^{\Sigma\alpha}_{\chi\beta} + \chi^{\nu}C^{\alpha}{}_{\nu\beta}\right) + \chi^{\nu}R_{\nu\mu}{}^{\alpha}{}_{\beta},$$

$$\delta_{\chi}A_{\mu} = \pounds_{\chi}A_{\mu} + D_{\mu}\Lambda_{\chi} = D_{\mu}\left(\Lambda_{\chi} + \chi^{\nu}A_{\nu}\right) + \chi^{\nu}F_{\nu\mu}.$$
 (2.16)

These follow trivially from the respective definitions of various quantities and are exactly the same as advertised in eq. (2.5). In the vielbein formalism, therefore, our background geometry is a *d*-dimensional spacetime manifold \mathcal{M} with fields $(e^{\alpha}{}_{\mu}, C^{\alpha}{}_{\mu\beta}, A_{\mu})$, but this time modded by diff $\times \mathfrak{so}(d-1, 1) \times \mathfrak{g}$ transformations.

This is the extent of the Einstein-Cartan geometries that we require in this chapter. There are, however, some interesting identities and relations which might be useful in an explicit computation. We refer the reader to [85] for a detailed review.

2.1.3 Noether theorem and Ward identities

Now that we have our background and symmetries ready, in this subsection we briefly outline the characteristic features of the physical theories that are coupled to it. Let the theory in question be described by some dynamical fields φ^I with respective equations of motion $\mathcal{E}_I \approx 0.^1$ We assume for now that our theory is described by an action $S[e^{\alpha}{}_{\mu}, C^{\alpha}{}_{\mu\beta}, A_{\mu}; \varphi^I]$. Since we are dealing with low energy effective descriptions in this work, we do not require the theory to be UV complete. This essentially means that φ^I could be some effective degrees of freedom relevant at our energy scale and S could be a Wilsonian effective action, with all the heavier modes integrated out.

Under an infinitesimal variation of the background and dynamical fields, we can parametrise the variation of the action as

$$\delta S = \int \mathrm{d}^d x \sqrt{-g} \Big(T^{\mu}_{\ \alpha} \ \delta e^{\alpha}_{\ \mu} + \Sigma^{\mu\alpha}_{\ \beta} \ \delta C^{\beta}_{\ \mu\alpha} + J^{\mu} \cdot \delta A_{\mu} + \mathcal{E}_I \ \delta \varphi^I \Big). \tag{2.17}$$

¹Throughout this work we denote on-shell equalities by " \approx ", whereas we keep "=" reserved for off-shell statements. Later we also introduce the symbol " \simeq " for "partially on-shell" statements, where only the hydrodynamic fields have been taken on-shell.

Here we have identified the Noether currents as being probed by the background fields. The variation of the action with respect to φ^{I} , on the other hand, gives us the equations of motion \mathcal{E}_{I} . Since the theory we seek to describe is invariant under the action of a set of symmetry parameters \mathcal{X} , we must require $\delta_{\mathcal{X}}S = 0$, modulo plausible anomalies. We already know how \mathcal{X} acts on the background fields, but we cannot say anything about the dynamical fields φ^{I} without knowing their explicit transformation properties. Nevertheless, provided that the symmetries act homogeneously on these fields, using differentiation by parts we can write down a generic statement

$$\mathcal{E}^{I}\delta_{\chi}\varphi_{I} = \frac{1}{\sqrt{-g}}\partial_{\mu}\left(\sqrt{-g}\,\mathcal{N}_{\chi}^{I\mu}\mathcal{E}_{I}\right) \\ + \left[\chi^{\alpha}\mathcal{O}_{\alpha}^{I} + \left(\Lambda_{\chi}^{\Sigma\alpha}{}_{\beta} + \chi^{\nu}C^{\alpha}{}_{\nu\beta}\right)\mathcal{O}^{I\beta}{}_{\alpha} + \left(\Lambda_{\chi} + \chi^{\mu}A_{\mu}\right)\cdot\mathcal{O}^{I}\right]\mathcal{E}_{I}.$$
 (2.18)

Here \mathcal{O}^{I} 's are some \mathfrak{X} -independent differential operators, while the operator $\mathcal{N}_{\chi}^{I\mu}$ is linear in \mathfrak{X} . The explicit form of these operators depends on how \mathfrak{X} acts on φ^{I} and can be left abstract for the purposes of this generic discussion. Plugging the field variations from eqs. (2.16) and (2.18) into eq. (2.17) and requiring $\delta_{\mathfrak{X}}S$ to vanish modulo anomalies, we can read out a set of identities

$$\underline{D}_{\mu}T^{\mu}{}_{\alpha} = e_{\alpha}{}^{\nu} \left(T^{\beta}{}_{\nu\mu}T^{\mu}{}_{\beta} + R_{\nu\mu}{}^{\gamma}{}_{\beta}\Sigma^{\mu\beta}{}_{\gamma} + F_{\nu\mu} \cdot J^{\mu} \right) + \mathcal{O}_{\alpha}^{I}\mathcal{E}_{I},$$

$$\underline{D}_{\mu}\Sigma^{\mu\alpha\beta} = T^{[\beta\alpha]} + \Sigma_{\mathrm{H}}^{\perp\alpha\beta} + \mathcal{O}^{I\alpha\beta}\mathcal{E}_{I},$$

$$\underline{D}_{\mu}J^{\mu} = \mathbf{J}_{\mathrm{H}}^{\perp} + \mathcal{O}^{I}\mathcal{E}_{I}.$$
(2.19)

We have defined the notation $\underline{\mathbf{D}}_{\mu} = \mathbf{D}_{\mu} + e_{\alpha}{}^{\nu}\mathbf{T}^{\alpha}{}_{\mu\nu}$ to avoid clutter. The Hall currents $\Sigma_{\mathrm{H}}^{\perp\alpha}{}_{\beta}$ and $\mathbf{J}_{\mathrm{H}}^{\perp}$ are a manifestation of anomalies, which we discuss in detail in section 2.1.4. For now it suffices to say that they are completely fixed in terms of $R_{\mu\nu}{}^{\alpha}{}_{\beta}$ and $F_{\mu\nu}$ up to some constants. On-shell, when $\mathcal{E}_{I} \approx 0$, the Ward identities (2.19) imply a set of conservation laws for $T^{\mu}{}_{\alpha}$, $\Sigma^{\mu\alpha}{}_{\beta}$, and J^{μ} respectively

$$\underline{D}_{\mu}T^{\mu}{}_{\alpha} \approx e_{\alpha}{}^{\nu} \left(T^{\beta}{}_{\nu\mu}T^{\mu}{}_{\beta} + R_{\nu\mu}{}^{\gamma}{}_{\beta}\Sigma^{\mu\beta}{}_{\gamma} + F_{\nu\mu} \cdot J^{\mu} \right),$$

$$\underline{D}_{\mu}\Sigma^{\mu\alpha\beta} \approx T^{[\beta\alpha]} + \Sigma^{\perp\alpha\beta}_{\mathrm{H}},$$

$$\underline{D}_{\mu}J^{\mu} \approx \mathrm{J}_{\mathrm{H}}^{\perp}.$$
(2.20)

These are the Noether conservation laws corresponding to our symmetries, which along with the identities eq. (2.19), form the backbone of all our analysis in this work.

When the background fields are switched off, i.e. $e^{\alpha}{}_{\mu} = \delta^{\alpha}{}_{\mu}$ and $C^{\alpha}{}_{\mu b} = A_{\mu} = 0$, the conservation laws reduce to their better known form

$$\partial_{\mu}T^{\mu}_{\ \alpha} \approx 0, \qquad \partial_{\mu}\Sigma^{\mu\alpha\beta} \approx T^{[\beta\alpha]}, \qquad \partial_{\mu}J^{\mu} \approx 0.$$
 (2.21)

2.1.4 Anomalous symmetries

In our discussion above, we referred to the so-called Hall currents $\Sigma_{\rm H}^{\perp\alpha\beta}$ and $J_{\rm H}^{\perp}$, which we claimed were a manifestation of our symmetries being anomalous. Let us take a brief detour and discuss what these currents actually are and how they arise using the anomaly inflow mechanism. In relativistic field theories, the anomaly inflow mechanism is an efficient way to classify flavour and gravitational/Lorentz anomalies [86]. A detailed introductory discussion can be found in section 2 of [87]. In essence, we consider that our *d*-dimensional spacetime manifold \mathcal{M} lives on the boundary of a (d + 1)-dimensional *bulk* manifold \mathcal{B} . We denote the bulk coordinates with a hat and near the boundary choose a basis $(x^{\hat{\mu}}) = (x^{\perp}, x^{\mu})$, where x^{\perp} corresponds to the depth into the bulk. All the field content $e^{\hat{\alpha}}_{\hat{\mu}}$, $C^{\hat{\alpha}}_{\hat{\mu}\hat{\beta}}$ and $A_{\hat{\mu}}$ is extended down into the bulk with the requirement that all the x^{\perp} components vanish at the boundary.

We keep our theory of interest on \mathcal{M} , whose field theory generating functional $W_{\mathcal{M}}$ is not necessarily invariant under the symmetries of the theory, i.e. is anomalous. In the bulk we keep some topological theory with generating functional $W_{\mathcal{B}}$, which is invariant under all the symmetries up to some non-trivial boundary terms. The full theory described by $W = W_{\mathcal{M}} + W_{\mathcal{B}}$ is assumed to be invariant under all the symmetries. It is actually this non-trivial bulk term $W_{\mathcal{B}}$ which induces anomaly in the boundary theory, hence the name anomaly inflow. In the notation of differential forms, $W_{\mathcal{B}}$ can be expressed as an integration of a (d + 1)-rank form I,

$$W_{\mathcal{B}} = \int_{\mathcal{B}} \boldsymbol{I}.$$
 (2.22)

The requirement that $W_{\mathcal{B}}$ should be symmetry-invariant up to a boundary term can be recast into the requirement that $\mathcal{P} = d\mathbf{I}$ should be invariant under all the symmetries. \mathcal{P} is called the *anomaly polynomial* of the theory, which encodes all the non-trivial information about anomaly. It is evident that \mathcal{P} needs to be closed, symmetry invariant, and should not be expressible as exterior derivative of a symmetry invariant form. Its explicit form, however, depends on the background field content of the theory. In the current context, it is given by the Chern-Simons anomaly polynomial $\mathcal{P}_{\rm CS}$ for even dimensional boundary theories, while no such terms are possible in odd spacetime dimensions. Here $\mathcal{P}_{\rm CS}$ is a "polynomial" made out of the Chern classes of the field strength \mathbf{F} and the Pontryagin classes of the curvature tensor $\mathbf{R}^{\hat{\alpha}}_{\hat{\beta}}$. In d = 4, for example, the most generic such anomaly polynomial is given as

$$\boldsymbol{\mathcal{P}}_{\rm CS} = C \, \operatorname{tr}[\boldsymbol{F} \wedge \boldsymbol{F} \wedge \boldsymbol{F}] + C_g \, \operatorname{tr} \boldsymbol{F} \wedge \boldsymbol{R}^{\hat{\alpha}}{}_{\hat{\beta}} \wedge \boldsymbol{R}^{\hat{\beta}}{}_{\hat{\alpha}}, \qquad (2.23)$$

where the trace is defined over the adjoint representation of the Lie-algebra \mathfrak{g} . The constants C and C_g are called the anomaly coefficients. Look at e.g. [87] for more details. In generality, for d = 4k - 2 or d = 4k, there are k + 1 possible anomaly coefficients.

Let us parametrise an infinitesimal variation of the bulk generating functional $W_{\mathcal{B}}$ as

$$\delta W_{\mathcal{B}} = \int_{\mathcal{B}} \mathrm{d}^{d+1} x \sqrt{-g_{d+1}} \left(\Sigma_{\mathrm{H}}{}^{\hat{\mu}\hat{\alpha}}{}_{\hat{\beta}} \delta C^{\hat{\beta}}{}_{\hat{\mu}\hat{\alpha}} + \mathrm{J}_{\mathrm{H}}^{\hat{\mu}} \cdot \delta A_{\hat{\mu}} \right) + \int_{\mathcal{M}} \mathrm{d}^{d} x \sqrt{-g} \left(\Sigma_{\mathrm{BZ}}{}^{\mu\alpha}{}_{\beta} \delta C^{\beta}{}_{\mu\alpha} + \mathrm{J}_{\mathrm{BZ}}^{\mu} \cdot \delta A_{\mu} \right). \quad (2.24)$$

The currents in the bulk subscripted by "H" are called *Hall currents*. In terms of the anomaly polynomial, they can be formally written down as a compact formula²

$$\star_{(d+1)} \Sigma_{\mathrm{H}}{}^{\hat{\alpha}}{}_{\hat{\beta}} = \frac{\partial \mathcal{P}}{\partial \mathbf{R}^{\hat{\beta}}{}_{\hat{\alpha}}}, \qquad \star_{(d+1)} \mathbf{J}_{\mathrm{H}} = \frac{\partial \mathcal{P}}{\partial \mathbf{F}}.$$
(2.25)

Here the Hodge duality operator $\star_{(d+1)}$ is defined in the bulk. For our d = 4 example in eq. (2.23), we can explicitly compute the Hall currents to be

$$\Sigma_{\mathrm{H}}{}^{\hat{\mu}\hat{\alpha}}{}_{\hat{\beta}} = \frac{1}{2}C_g \,\epsilon^{\hat{\mu}\hat{\nu}\hat{\rho}\hat{\sigma}\hat{\tau}} \operatorname{tr} F_{\hat{\nu}\hat{\rho}}R_{\hat{\sigma}\hat{\tau}}{}^{\hat{\alpha}}{}_{\hat{\beta}},$$

$$J_{\mathrm{H}}^{\hat{\mu}} = \frac{3}{4}C \,\epsilon^{\hat{\mu}\hat{\nu}\hat{\rho}\hat{\sigma}\hat{\tau}}F_{\hat{\nu}\hat{\rho}}F_{\hat{\sigma}\hat{\tau}} + \frac{1}{4}C_g \,\epsilon^{\hat{\mu}\hat{\nu}\hat{\rho}\hat{\sigma}\hat{\tau}}R_{\hat{\nu}\hat{\rho}}{}^{\hat{\alpha}}{}_{\hat{\beta}}R_{\hat{\sigma}\hat{\tau}}{}^{\hat{\beta}}{}_{\hat{\alpha}}.$$
(2.26)

The term coupling to C is called G (flavour) anomaly, because it only affects the G-symmetry sector, while C_g is called a mixed anomaly as it affects both Lorentz as well as G sectors.

On the other hand, the bulk generating functional induces the so called *Bardeen-Zumino* currents subscripted by "BZ" at the boundary. They can be represented as derivatives of I, i.e.

$$\star \boldsymbol{\Sigma}_{\mathrm{BZ}}{}^{\hat{\alpha}}{}_{\hat{\beta}} = \frac{\partial \boldsymbol{I}}{\partial \boldsymbol{R}^{\hat{\beta}}{}_{\hat{\alpha}}}, \qquad \star \mathbf{J}_{\mathrm{BZ}} = \frac{\partial \boldsymbol{I}}{\partial \boldsymbol{F}}.$$
(2.27)

These are, of course, not symmetry-covariant. But since the full partition function W is symmetry-covariant, if we define the variation of the boundary piece $W_{\mathcal{M}}$ as

$$\delta W_{\mathcal{M}} = \int_{\mathcal{M}} \mathrm{d}^{d} x \sqrt{-g} \left(T^{\mu}{}_{\alpha} \,\delta e^{\alpha}{}_{\mu} + \Sigma_{\mathrm{cons}}{}^{\mu\alpha}{}_{\beta} \,\delta C^{\beta}{}_{\mu\alpha} + J^{\mu}_{\mathrm{cons}} \cdot \delta A_{\mu} \right), \tag{2.28}$$

the full boundary covariant currents

$$T^{\mu}_{\ \alpha}, \qquad \Sigma^{\mu\alpha}_{\ \beta} = \Sigma_{\rm cons}^{\ \mu\alpha}{}_{\beta} + \Sigma_{\rm BZ}^{\ \mu\alpha}{}_{\beta}, \qquad J^{\mu} = J^{\mu}_{\rm cons} + J^{\mu}_{\rm BZ}, \tag{2.29}$$

are symmetry-covariant. Moreover, demanding $\delta_{\chi}W = 0$ under an infinitesimal symmetry variation χ precisely leads to the conservation laws (2.20) for the boundary currents. There are also a similar set of (non-anomalous) conservation laws in the bulk, but given that there are no dynamical fields in the bulk, they are trivially satisfied.

As opposed to the covariant boundary currents, the currents obtained from $W_{\mathcal{M}}$, i.e. $(T^{\mu}_{\alpha}, \Sigma_{\text{cons}}{}^{\mu\alpha}{}_{\beta}, J^{\mu}_{\text{cons}})$, are called the *consistent currents*. Even though they are non-covariant under symmetries, they are the physical Noether currents of the boundary theory. They

²We should clarify what we mean by differentiating or dividing with respect to a 2-form. Since \mathcal{P} is a polynomial in 2-forms, which mutually commute, we can intuitively define this differentiation and division as we would for an ordinary polynomial. Such an operation converts a (d+2)-form into a *d*-form.

also satisfy a set of conservation laws akin to eq. (2.20),

$$\underline{D}_{\mu}T^{\mu}_{\ \alpha} \approx e_{\alpha}^{\ \nu} \left(T^{\beta}_{\ \nu\mu}T^{\mu}_{\ \beta} + R_{\nu\mu}^{\ \gamma}_{\ \beta}\Sigma_{\rm cons}^{\ \mu\beta}_{\ \gamma} + F_{\nu\mu} \cdot J^{\mu}_{\rm cons} \right) + T_{\rm H,cons}^{\ \perp}_{\ \alpha},$$

$$\underline{D}_{\mu}\Sigma^{\mu\alpha\beta}_{\rm cons} \approx T^{[\beta\alpha]} + \Sigma^{\perp\alpha\beta}_{\rm H,cons},$$

$$\underline{D}_{\mu}J^{\mu}_{\rm cons} \approx J^{\perp}_{\rm H,cons},$$
(2.30)

but the "consistent anomaly" is instead given as

$$T_{\rm H,cons}{}^{\perp}{}_{\alpha} = e_{\alpha}{}^{\nu} \left(R_{\nu\mu}{}^{\gamma}{}_{\beta}\Sigma_{\rm BZ}{}^{\mu\beta}{}_{\gamma} + F_{\nu\mu} \cdot J_{\rm BZ}^{\mu} \right),$$

$$\Sigma_{\rm H,cons}{}^{\perp}{}_{\beta} = \Sigma_{\rm H}{}^{\perp}{}_{\beta} - \underline{\rm D}_{\mu}\Sigma_{\rm BZ}{}^{\mu\alpha}{}_{\beta}, \qquad J_{\rm H,cons}{}^{\perp} = J_{\rm H}{}^{\perp} - \underline{\rm D}_{\mu}J_{\rm BZ}^{\mu}.$$
(2.31)

In particular, note that written in terms of the consistent currents, the energy-momentum conservation also gets an anomaly. In most of this work, however, we are working directly with the covariant currents.

2.2 The hydrodynamic setup

We are interested in a universal low energy effective description of finite-temperature near-equilibrium field theories. By itself, it is an absurd thing to seek as it would obviously depend on the field theory under consideration. But in most physical scenarios when such an effective description would actually be useful, like describing your mug of coffee as you add milk to it, we probably would not know anything about the underlying microscopic field theory, let alone an action principle. Within its narrow regime of applicability, hydrodynamics aims at providing a universal framework to study low energy fluctuations of a finite temperature field theory around its thermodynamic equilibrium with unknown microscopics, guided by the fundamental principles like symmetries and other empirical physical expectations like the second law of thermodynamics.

2.2.1 Thermal equilibrium

Before considering departures from equilibrium, it will be helpful to revisit some elementary facts about quantum field theories at thermodynamic equilibrium. The material presented in this section can be found in any standard text on thermal field theories like [88, 89]. Consider a relativistic background manifold \mathcal{M} constructed in the previous section with fields $(e^{\alpha}_{\mu}, C^{\alpha}_{\mu\beta}, A_{\mu})$. \mathcal{M} is said to admit an equilibrium if there exists a set of symmetry parameters $\mathcal{K} = (K^{\mu}, \Lambda_{K}^{\Sigma}, \Lambda_{K})$ with $K_{\mu}K^{\mu} < 0$, whose action on \mathcal{M} is an isometry

$$\delta_{\mathcal{K}} e^{\alpha}{}_{\mu} = \delta_{\mathcal{K}} C^{\alpha}{}_{\mu\beta} = \delta_{\mathcal{K}} A_{\mu} = 0. \tag{2.32}$$

 \mathcal{K} can be interpreted as a set of parameters defining a relativistic observer, with respect to whom the background is time-independent. In \mathcal{K} 's reference frame, the time-evolution of

an observable is generated by a time-independent Hamiltonian operator $\mathcal{H}_{\mathcal{K}}$ given by

$$\mathcal{H}_{\mathcal{K}} = \int_{\Sigma} \mathrm{d}^{d-1} x \sqrt{-g} \left[T^{0}_{\ \alpha} K^{\alpha} + \left(\Lambda^{\Sigma \ \alpha}_{K \ \beta} + K^{\mu} C^{\alpha}_{\ \mu\beta} \right) \Sigma^{0\beta}_{\ \alpha} + \left(\Lambda_{K} + K^{\mu} A_{\mu} \right) \cdot J^{0} \right].$$
(2.33)

Here the integral is performed over some Cauchy slice Σ with transverse timelike coordinate x^0 . Using the conservation laws (2.20) and the isometry condition (2.32), we can check that $\mathcal{H}_{\mathcal{K}}$ is conserved, i.e. it does not depend on the choice of Σ .³

Let us consider a generic quantum field theory coupled to an equilibrium-admitting background manifold \mathcal{M} . Turning on a constant global temperature $T_0 = 1/\beta_0$, the classic result in quantum statistical mechanics states that in thermodynamic equilibrium, a system can be described by a grand-canonical Gibbs density matrix

$$\rho^{\text{eqb}} = \exp\left(-\beta_0 \mathcal{H}_{\mathcal{K}}\right). \tag{2.34}$$

We can also write down the associated thermal partition function

$$\mathcal{Z}^{\text{eqb}} = \operatorname{tr}\exp\left(-\beta_0 \mathcal{H}_{\mathcal{K}}\right). \tag{2.35}$$

Here the trace is taken over a complete set of basis on the Hilbert space of our theory. Once we are provided with \mathcal{Z}^{eqb} as a functional of the time-independent (*K*-invariant) background fields $(e^{\alpha}_{\ \mu}, C^{\alpha}_{\ \mu\beta}, A_{\mu})$, we know everything about the macroscopic behaviour of our equilibrium state. The thermal expectation values and correlators of all the macroscopic thermodynamic observables (i.e. conserved currents) can be obtained by taking functional derivatives of \mathcal{Z}^{eqb} . To wit,

$$\langle T^{\mu}{}_{\alpha}\rangle_{\beta_0} = \frac{1}{\sqrt{-g}} \frac{\delta W^{\text{eqb}}}{\delta e^{\alpha}{}_{\mu}}, \qquad \langle \Sigma^{\mu\alpha}{}_{\beta}\rangle_{\beta_0} = \frac{1}{\sqrt{-g}} \frac{\delta W^{\text{eqb}}}{\delta C^{\beta}{}_{\mu\alpha}}, \qquad \langle J^{\mu}\rangle_{\beta_0} = \frac{1}{\sqrt{-g}} \frac{\delta W^{\text{eqb}}}{\delta A_{\mu}}, \quad (2.36)$$

where $W^{\text{eqb}} = -\ln \mathcal{Z}^{\text{eqb}}$.

To compute \mathcal{Z}^{eqb} for a given quantum field theory, it is most natural to work in the so-called imaginary-time formalism. Let us consider that the field theory we are interested in is described at equilibrium by an effective action (see section 2.1.3) given as

$$S[e^{\alpha}{}_{\mu}, C^{\alpha}{}_{\mu\beta}, A_{\mu}; \varphi^{i}] = \int_{\mathcal{M}} \mathrm{d}^{d}x \sqrt{-g} \,\mathcal{L}(e^{\alpha}{}_{\mu}, C^{\alpha}{}_{\mu\beta}, A_{\mu}; \varphi^{i}).$$
(2.37)

Here φ^i are some effective dynamical degrees of freedom relevant at our energy scales. We are only interested in the time-independent configurations of the dynamical fields, therefore we choose $\delta_{\mathcal{K}}\varphi^i = 0$. Together with eq. (2.32), it implies that $K^{\mu}\partial_{\mu}\mathcal{L} = 0$. If we choose a basis $(x^{\mu}) = (x^0, x^i)$ on \mathcal{M} such that $K^{\mu} = \delta^{\mu}_{0}$, for these configurations we have $\partial_0 \mathcal{L} = 0$ and consequently

$$S[e^{\alpha}{}_{\mu}, C^{\alpha}{}_{\mu\beta}, A_{\mu}; \varphi^{i}] = \int \mathrm{d}x^{0} \int_{\Sigma} \mathrm{d}^{d-1}x \sqrt{-g} \,\mathcal{L}(e^{\alpha}{}_{\mu}, C^{\alpha}{}_{\mu\beta}, A_{\mu}; \varphi^{i}).$$
(2.38)

 $^{^{3}}$ Actually, this statement is only correct for non-anomalous field theories. For anomalous field theories, one has to take into account the bulk piece of the Hamiltonian as well.

Here again, Σ is a Cauchy slice spanned by x^i 's whose choice is irrelevant owing to the time independence of the integrand. To compute the thermal partition function, we Wickrotate the time coordinate $x^0 \to -i\tau$ and compactify τ into a circle with period β_0 . The coordinates (τ, x^i) span a Euclidean manifold $\mathcal{M}_{\rm E}$, which is an analytic continuation of \mathcal{M} . This allows us to analytically continue the effective action

$$S \to iS^{\text{eqb}} = \int_{\mathcal{M}_{\text{E}}} (-i\mathrm{d}\tau) \,\mathrm{d}^{d-1}x \left(\sqrt{-g}\,\mathcal{L}\right)_{x^0 \to i\tau} = i\beta_0 \int_{\Sigma} \mathrm{d}^{d-1}x \sqrt{-g}\,\mathcal{L}^{\text{eqb}}.$$
 (2.39)

The superscript "eqb" in the Euclidean Lagrangian $\mathcal{L}^{\text{eqb}} = -\mathcal{L}$ signifies that the background and dynamical fields making up \mathcal{L}^{eqb} are manifestly \mathcal{K} -invariant. S^{eqb} is generally known as equilibrium effective action in the literature. It can be used to compute the thermal partition function \mathcal{Z}^{eqb} , defined as the analytic continuation of the field theory generating functional

$$\mathcal{Z} = \int \mathcal{D}\varphi^{i} \exp\left(iS\right) \to \mathcal{Z}^{\text{eqb}} = \int \mathcal{D}\varphi^{i} \exp\left(-S^{\text{eqb}}\right).$$
(2.40)

Once we have \mathcal{Z}^{eqb} , we can go ahead and compute all the thermal expectation values and correlators. Note that the thermodynamic equilibrium is effectively described by a (d-1)-dimensional Euclidean quantum field theory with action S^{eqb} .

All of the discussion above was focused on finite temperature states at the thermodynamic equilibrium. That is to say that the density matrix ρ^{eqb} and the thermal partition function \mathcal{Z}^{eqb} are both time-independent, by the virtue of the Hamiltonian operator $\mathcal{H}_{\mathcal{K}}$ being foliation independent. Ideally, we would like to extend this story to arbitrary non-equilibrium states and processes, but it turns out to be quite a non-trivial task to perform. Therefore for simplicity, let us choose to work with the non-equilibrium states that are still sufficiently close to equilibrium. Let us consider a subset of states described by a set of symmetry parameters $\mathcal{B} = (\beta^{\mu}, \Lambda^{\Sigma}_{\beta}, \Lambda_{\beta})$ that are sufficiently close to the equilibrium parameters \mathcal{K} . Effectively, we are looking at a field theory with dynamical fields $\varphi^{I} = \{\mathcal{B}, \varphi^{i}\}$. Similar to eq. (2.33), one can define a Hamiltonian operator $\mathcal{H}_{\mathcal{B}}$, however, given that \mathcal{B} is not an isometry, $\mathcal{H}_{\mathcal{B}}$ is not conserved and depends on the choice of foliation used to define it. Consequently, the associated Gibbs density matrix $\rho = \exp(-\beta_0 \mathcal{H}_{\mathcal{B}})$ and the partition function $\mathcal{Z} = \operatorname{tr} \exp(-\beta_0 \mathcal{H}_{\mathcal{B}})$ are also time-dependent. We can no longer compute the time-dependent \mathcal{Z} using the neat trick of imaginary-time formalism. Nevertheless, we are not really interested in the specifics of the microscopic finite-temperature field theory, but only the universal near-equilibrium behaviour. This can be quite readily captured by hydrodynamics, as we outline below.

2.2.2 Hydrodynamic fields

In the absence of an effective action describing the finite-temperature low-energy regime, we take the Noether theorem as our starting point. We require our theory of interest to respect the spacetime Poincaré transformations and some internal global *G*-transformations. This implies the existence of the associated conserved currents $T^{\mu}_{\ \alpha}$, $\Sigma^{\mu\alpha}_{\ \beta}$, and J^{μ} . After integrating out all the massive modes from the theory as we approach the deep IR, the only relevant degrees of freedom would be some effective massless modes φ^I along with their associated equations of motion $\mathcal{E}_I \approx 0$. For a generic low-energy theory, therefore, the conserved currents would be some expressions in terms of φ^I and background fields $(e^{\alpha}{}_{\mu}, C^{\alpha}{}_{\mu\beta}, A_{\mu})$, with appropriate transformation properties. Noether's theorem implies that these currents satisfy the identities (2.19), where the operators $\mathcal{O}^I_{\alpha}, \mathcal{O}^{I\alpha}{}_{\beta}$, and \mathcal{O}^I encode how the fields φ^I transform under symmetries.

Due to these identities, the conservation laws (2.20) can serve as a placeholder for a vector, a $\mathfrak{so}(d, 1)$ -valued scalar, and a \mathfrak{g} -valued scalar linear combinations of the equations of motion $\mathcal{E}_I \approx 0$. That is to say that out of the dynamic fields φ^I , eq. (2.20) can serve as equations of motion for some dynamical degrees of freedom packaged into

$$\mathcal{B} = \left(\begin{array}{c} \beta^{\mu} , & \Lambda_{\beta}^{\Sigma} , & \Lambda_{\beta} \end{array} \right). \tag{2.41}$$

We denote the remaining dynamical fields which are not captured in \mathcal{B} by φ^i . The choice of this splitting of degrees of freedom near thermodynamic equilibrium is naturally motivated from our discussion in the previous subsection. In the conventional treatment of hydrodynamics, it is more convenient to rewrite \mathcal{B} in terms of the so called *hydrodynamic fields*

Velocity: u^{μ} with $u^{\mu}u_{\mu} = -1$, Temperature: T, Spin chemical potential: $\mu^{\Sigma} \in \mathfrak{so}(d, 1)$, G chemical potential: $\mu \in \mathfrak{g}$,

defined via the relations

$$\beta^{\mu} = \frac{1}{T} u^{\mu}, \qquad \Lambda^{\Sigma \alpha}_{\beta}{}_{\beta} + \beta^{\mu} C^{\alpha}{}_{\mu\beta} = \frac{1}{T} \mu^{\Sigma \alpha}{}_{\beta}, \qquad \Lambda_{\beta} + \beta^{\mu} A_{\mu} = \frac{1}{T} \mu.$$
(2.42)

These hydrodynamic fields, as the name suggests, form the fundamental dynamical variables of hydrodynamics. Note that a priori there is no unique definition of these fields. Therefore, like in any field theory, they are permitted to undergo arbitrary field redefinition involving themselves and the other fields φ^i and $(e^{\alpha}{}_{\mu}, C^{\alpha}{}_{\mu\beta}, A_{\mu})$ without changing any physics. Let us push this to the back our minds for the moment. We return to this issue in section 2.3.

Once we have made this logical split between the degrees of freedom, we can formally decompose $\mathcal{O}^I \mathcal{E}_I$'s appearing in eq. (2.18) into

$$\mathcal{O}^{I}_{\alpha}\mathcal{E}_{I} = P^{\text{ext}}_{\alpha} + \mathcal{O}^{i}_{\alpha}\mathcal{E}_{i}, \qquad \mathcal{O}^{I\alpha}_{\ \beta}\mathcal{E}_{I} = S^{\text{ext}\,\alpha}_{\ \beta} + \mathcal{O}^{i\alpha}_{\ \beta}\mathcal{E}_{i}, \qquad \mathcal{O}^{I}\mathcal{E}_{I} = Q^{\text{ext}} + \mathcal{O}^{i}\mathcal{E}_{i}. \tag{2.43}$$

Here P_{α}^{ext} , $S^{\text{ext }\alpha}{}_{\beta}$, and Q^{ext} are some linear combinations of our original equations of motion which provide dynamics for our hydrodynamic fields. Using the identities (2.19), these can also be expressed as

$$P_{\alpha}^{\text{ext}} = \underline{D}_{\mu} T^{\mu}{}_{\alpha} - e_{\alpha}{}^{\nu} \left(T^{\beta}{}_{\nu\mu} T^{\mu}{}_{\beta} + R_{\nu\mu}{}^{\gamma}{}_{\beta} \Sigma^{\mu\beta}{}_{\gamma} + F_{\nu\mu} \cdot J^{\mu} \right) - \mathcal{O}_{\alpha}^{i} \mathcal{E}_{i},$$

$$S^{\text{ext}}{}^{\alpha\beta} = \underline{D}_{\mu} \Sigma^{\mu\alpha\beta} - T^{[\beta\alpha]} - \Sigma_{\text{H}}^{\perp\alpha\beta} - \mathcal{O}^{i\alpha\beta} \mathcal{E}_{i},$$

$$Q^{\text{ext}} = \underline{D}_{\mu} J^{\mu} - J_{\text{H}}^{\perp} - \mathcal{O}^{i} \mathcal{E}_{i}.$$
(2.44)

From here we can interpret P_{α}^{ext} as "external energy-momentum", $S^{\text{ext}\,\alpha}{}_{\beta}$ as "external spin", and Q^{ext} as "external charge" sources, because when they are non-vanishing, possibly due to some external agents, they would violate the respective conservation equations. When the system is thermodynamically isolated, however, these sources vanish on-shell and can be taken to be the equations of motion for our hydrodynamic fields

$$P_{\alpha}^{\text{ext}} \approx 0, \qquad S^{\text{ext}\,\alpha}{}_{\beta} \approx 0, \qquad Q^{\text{ext}} \approx 0.$$
 (2.45)

Due to eq. (2.44), these equations are the same as the conservation laws (2.20) as we originally intended.

A definition: a fluid configuration is said to be *thermodynamically isolated* if the hydrodynamic fields \mathcal{B} are taken on-shell by setting the external sources P_{α}^{ext} , $S^{\text{ext }\alpha}{}_{\beta}$, and Q^{ext} to zero, while still keeping the remaining fields φ^i off-shell. Equalities in this "partially on-shell" configuration are denoted by the symbol " \simeq ",

$$\underline{D}_{\mu}T^{\mu}{}_{\alpha} \simeq e_{\alpha}{}^{\nu} \left(T^{\beta}{}_{\nu\mu}T^{\mu}{}_{\beta} + R_{\nu\mu}{}^{\gamma}{}_{\beta}\Sigma^{\mu\beta}{}_{\gamma} + F_{\nu\mu} \cdot J^{\mu} \right) + \mathcal{O}^{i}_{\alpha}\mathcal{E}_{i},$$

$$\underline{D}_{\mu}\Sigma^{\mu\alpha\beta} \simeq T^{[\beta\alpha]} + \Sigma^{\perp\alpha\beta}_{\mathrm{H}} + \mathcal{O}^{i\alpha\beta}\mathcal{E}_{i},$$

$$\underline{D}_{\mu}J^{\mu} \simeq \mathbf{J}^{\perp}_{\mathrm{H}} + \mathcal{O}^{i}\mathcal{E}_{i}.$$
(2.46)

To this end, hydrodynamics is characterised by the most generic expressions for a set of conserved currents $(T^{\mu}{}_{\alpha}, \Sigma^{\mu\alpha}{}_{\beta}, J^{\mu})$ and some quantities \mathcal{E}_i written in terms of the hydrodynamic fields $(u^{\mu}, T, \mu^{\Sigma}, \mu)$ (or equivalently \mathcal{B}), other gapless fields φ^i , and background fields $(e^{\alpha}{}_{\mu}, C^{\alpha}{}_{\mu\beta}, A_{\mu})$. These expressions are called the *hydrodynamic constitutive relations*. The fields \mathcal{B} and φ^i are the dynamical field content of hydrodynamics. The equations of motion for \mathcal{B} are given by eq. (2.46), while those for φ^i are given by $\mathcal{E}_i \approx 0$. This makes the system of equations closed.

2.2.3 Constitutive relations

Before we start writing down the hydrodynamic constitutive relations, we need to get some technical aspects in order. We do not know anything about the fields φ^i , and even if we did we could cook up an infinite tower of distinct tensor structures that could potentially enter the constitutive relations. Let us deal with the latter problem first. Recall that we are working under the assumption of "low-energy", which we gladly used to integrate out all the massive degrees of freedom from our theory. We must, therefore, take care that at all times we are only probing energies which are well below the energy-scale set by the first massive excitation. Heuristically, the inverse temperature defines a length scale called the mean-free path of the system. When a particle interpretation is available, the mean-free path can be understood as the average length a particle travels between collisions. We require that the perturbations in hydrodynamics are over the length scales much larger compared to this mean-free path so that the actual constituents of the theory stay irrelevant. Practically, this implies that hydrodynamics is only valid in a regime where the derivatives of background and dynamical fields are small. Within this regime, therefore, we can treat derivatives as a perturbative parameter and expand the constitutive relations order by order in derivatives. This is commonly referred to as the *derivative expansion* of hydrodynamics. At any given order in the derivative expansion, the constitutive relations only contain a finite number of tensor structures, each coming with an arbitrary multiplier function called a *transport coefficient*. Truncated to a finite derivative order, therefore, hydrodynamics is completely characterised by a finite set of transport coefficients entering the constitutive relations. At this level, these transport coefficients are completely arbitrary, and depending on the microscopic field theory at play, can be different for different hydrodynamic systems. For a given system, if the microscopic field theory is known they can be explicitly computed, or otherwise, they can be measured in an experiment. Admittedly, the number of transport coefficients grows factorially as we increase the derivative order, so hydrodynamics is only a useful framework at the first few derivative orders.

When it comes to the fields φ^i , hydrodynamics is generally evasive. It is certainly true that we need their transformation properties to make any real progress with the constitutive relations. Moreover, depending on their number and transformation properties, the spectrum of transport coefficients would be totally different. The easiest case to consider is when these gapless modes are not present at all and the hydrodynamic fields are the only relevant low-energy effective degrees of freedom. This is true of theories with a "mass-gap" where there is a finite energy difference between the ground state and the first excited state. The hydrodynamics following from here is commonly referred to as the ordinary fluid dynamics or simply fluid dynamics. Another interesting case to study is when φ^{i} 's are Goldstone modes of a spontaneously broken symmetry, in which case we explicitly know their transformation properties. The breaking of internal symmetries leads to the curiously termed superfluid dynamics. On the other hand, the breaking of spacetime symmetries could lead to interesting phenomena in hydrodynamics like the formation of surfaces or momentum relaxation. Trying something more exotic, we can take φ^i to be a dynamical U(1) gauge field. This leads to the interesting physics of magnetohydrodynamics. We briefly touch upon some of the examples of these cases in the course of this work.

With some of these technicalities out of the way, we can conclude that hydrodynamics is characterised by a finite set of transport coefficients. Their actual number and contribution to the constitutive relations depend on the derivative order we are working at and what additional gapless modes we have at hand. We also commented that the explicit functional form of these transport coefficients depends on the fluid under consideration and could be obtained either by a carefully designed experiment or a field theoretic computation. If this feels rather disheartening to someone, we can, in fact, impose some empirical physical requirements on the constitutive relations and significantly bring down the number of independent transport coefficients that need to be measured. The most important of these empirical requirements, and as far as we know universally applicable, is the *local second law of thermodynamics*. We describe the statement in the section 2.3.

To avoid any confusion, we should clarify that in our entire discussion above we have only been interested in the constitutive relations and transport coefficients of a fluid. They are essentially a set of parameters characterising a fluid, such as the viscosity and conductivity, as a function of the thermodynamic variables such as the temperature and chemical potential. We also discussed how these properties could influence the velocity, temperature, or chemical potential profiles of the fluid via the conservation laws serving as equations of motion. We have not, however, discussed the actual solutions of these equations of motion, known as *fluid configurations*. There is a lot of interesting physics to be understood in these configurations, but in this work, we only focus on the fundamental principles governing fluids and their constitutive relations.

Choosing derivative order for constituent fields: To be able to use the derivative expansion when writing down the constitutive relations, we need to decide where various fields in the theory stand with respect to each other. The choice we make depends on the physical application we have in mind and affects the explicit form of the constitutive relations. But the framework of hydrodynamics works with any such choice. For reference, we always choose the temperature $T \sim \mathcal{O}(\partial^0)$. During most of this work, we are working with the choice that the fields corresponding to Abelian symmetries are $\mathcal{O}(\partial^0)$, while those corresponding to non-Abelian symmetries are $\mathcal{O}(\partial^1)$:

- Translations sector: $u^{\mu}, T, e^{\alpha}{}_{\mu} \sim \mathcal{O}(\partial^0).$
- SO(d-1,1) sector: $\mu^{\Sigma\alpha}{}_{\beta}, C^{\alpha}{}_{\mu\beta} \sim \mathcal{O}(\partial^1)$. Consequently, $T^{\alpha}{}_{\mu\nu} \sim \mathcal{O}(\partial^1), R_{\mu\nu}{}^{\alpha}{}_{\beta} \sim \mathcal{O}(\partial^2)$.
- G-sector: We split G into $G^{ab} \times G^{nab}$ where G^{ab} is its largest Abelian subgroup. It implies a splitting of various fields $\mu = \mu^{ab} + \mu^{nab}$, $A_{\mu} = A_{\mu}^{ab} + A_{\mu}^{nab}$ and $F_{\mu\nu} = F_{\mu\nu}^{ab} + F_{\mu\nu}^{nab}$. We choose μ^{ab} , $A_{\mu}^{ab} \sim \mathcal{O}(\partial^0)$ and μ^{nab} , $A_{\mu}^{nab} \sim \mathcal{O}(\partial^1)$. Consequently, $F_{\mu\nu}^{ab} \sim \mathcal{O}(\partial^1)$ and $F_{\mu\nu}^{nab} \sim \mathcal{O}(\partial^2)$.
- The choice for φ^i depends on their symmetry properties.

For the connections $C^{\alpha}{}_{\mu\beta}$ and A_{μ} , this choice ensures that the action of the covariant derivative operator D_{μ} on the constituent fields always raises a derivative order and can be treated perturbatively. We have chosen the derivative orders of the chemical potentials to match up with their connection counterparts.

We must stress that although this choice is suitable for the systems studied in this work, the hydrodynamic setup works for any arbitrary choice. For example, when focusing on hydrodynamics with spin, we typically take the spin chemical potential $\mu^{\Sigma} \sim \mathcal{O}(\partial^0)$; see e.g. [90]. In particular, the generalities of hydrodynamics discussed in this chapter go through irrespective of the choice of derivative orders for the constituent fields.

2.2.4 Hydrostatic principle

We started out this section with an introduction to finite temperature field theories at thermodynamic equilibrium in section 2.2.1. We motivated that in addition to the gapless dynamical modes φ^i relevant at equilibrium, near equilibrium states are described by the so-called hydrodynamic modes \mathcal{B} . In the limit that we approach the thermodynamic equilibrium, the fields \mathcal{B} approach a set of background isometry data \mathcal{K} . Later in sections 2.2.2 and 2.2.3 we developed a generic formalism of hydrodynamics with dynamical fields \mathcal{B} and φ^i based on symmetries and the low-energy approximation. In this subsection, we would like to subject our construction to some internal consistency conditions by requiring that the theory of hydrodynamics admits a consistent thermodynamic equilibrium.

Let us say that we have coupled our theory of hydrodynamics to an equilibrium-admitting background manifold \mathcal{M} with timelike isometry \mathcal{K} . We can define the so-called *equilibrium fluid configurations* as solutions of the equations of motion which respect the timelike isometry, i.e. $\delta_{\mathcal{K}}\mathcal{B} = \delta_{\mathcal{K}}\varphi^i = 0$. We require that there must always exist an equilibrium fluid configuration, called the *hydrostatic configuration*, for which $\mathcal{B} = \mathcal{K}$.⁴ We dub this as the *weak hydrostatic principle*. It is a very non-trivial statement, as for $\mathcal{B} = \mathcal{K}$ the hydrodynamic equations of motion (2.46) must be rendered trivial. One can imagine that this would not hold true for arbitrary hydrodynamic constitutive relations. Therefore, the weak hydrostatic principle imposes some non-trivial constraints on the form of the constitutive relations and the independent transport coefficients that can appear in it.

The weak hydrostatic principle stated above is a necessary condition for the existence of equilibrium but is still not sufficient. To guarantee the existence of a well-defined equilibrium configuration, we must require that the conserved currents and φ^i -profiles in a hydrostatic configuration can be generated from a hydrostatic effective action [49, 50].⁵ We call this the strong hydrostatic principle. The effective action $S^{\text{hs}}[e^{\alpha}{}_{\mu}, C^{\alpha}{}_{\mu\beta}, A_{\mu}; \varphi^i]$ is given as a functional of the fields $(e^{\alpha}{}_{\mu}, C^{\alpha}{}_{\mu\beta}, A_{\mu})$ and φ^i respecting the timelike isometry \mathcal{K} , defined over a spatial hypersurface. S^{hs} is a special case of a generic equilibrium effective action S^{eqb} discussed in section 2.2.1, wherein the constituent fields vary over length-scales much larger than the mean free path of the system, and hence can be written down order-by-order in derivatives. The φ^i -profiles are generated by extremising the effective action with respect to the variations in φ^i

$$\mathcal{E}_i = \frac{1}{\sqrt{-g}} \frac{\delta S^{\rm hs}}{\delta \varphi^i} \approx 0. \tag{2.47}$$

On the other hand, the hydrostatic conserved currents are obtained by varying the action with respect to the background fields

$$T^{\mu}_{\ \alpha}\big|_{\mathcal{B}=\mathcal{K}} = \frac{1}{\sqrt{-g}} \frac{\delta S^{\rm hs}}{\delta e^{\alpha}_{\ \mu}}, \qquad \Sigma^{\mu\alpha}_{\ \beta}\big|_{\mathcal{B}=\mathcal{K}} = \frac{1}{\sqrt{-g}} \frac{\delta S^{\rm hs}}{\delta C^{\beta}_{\ \mu\alpha}}, \qquad J^{\mu}\big|_{\mathcal{B}=\mathcal{K}} = \frac{1}{\sqrt{-g}} \frac{\delta S^{\rm hs}}{\delta A_{\mu}}.$$
 (2.48)

We require the hydrostatic effective action to be invariant under \mathcal{K} preserving diff $\times \mathfrak{so}(d-1,1) \times \mathfrak{g}$ transformations, up to anomalies. The hydrodynamic equations of motion (2.46) evaluated at $\mathcal{B} = \mathcal{K}$ turn out to be the Bianchi identities associated with these symmetries for arbitrary φ^i -offshell configurations. Therefore the weak hydrostatic principle

⁴Technically, one only needs $\mathcal{B} = \mathcal{K} + \mathcal{O}(\partial)$ to define a hydrostatic configuration. But we can always fix part of the redefinition freedom in \mathcal{B} to exactly set $\mathcal{B} = \mathcal{K}$ in a hydrostatic configuration. We call the class of hydrodynamic frames that satisfy this condition to be *hydrostatic frames*.

⁵When there are no other dynamical fields φ^i in the hydrodynamic description, the hydrostatic effective action reduces to a hydrostatic partition function(al). They have also been called equilibrium effective action and equilibrium partition function in the literature respectively.

follows as a corollary of the strong hydrostatic principle.

In practice, we can express S^{hs} as an integral of some scalar density \mathcal{L}^{hs} made out of the \mathcal{K} respecting fields $(e^{\alpha}{}_{\mu}, C^{\alpha}{}_{\mu\beta}, A_{\mu})$ and φ^{i} , truncated at our desired derivative order. \mathcal{L}^{hs} can also contain Chern-Simons terms made out of the connections $C^{\alpha}{}_{\mu\beta}$ and A_{μ} , which are covariant under symmetries up to some boundary terms. It is the terms like these that are responsible for anomalies. Having done that, we can use eq. (2.48) to compare the hydrostatic constitutive relations with those generated using S^{hs} and read out the constraints imposed by the hydrostatic effective action. This procedure has been successfully implemented for a large number of cases in the literature [2, 49–51, 91]. In this work, however, we do not pursue this direction any further. As it turns out, all of these constraints follow from an even more restrictive physical requirement: the second law of thermodynamics, which we describe in the next section. See section 2.3.4 for the reasoning how the second law captures the strong hydrostatic principle.

2.3 Local second law of thermodynamics

2.3.1 Second law and classification of transport

Given a set of hydrodynamic constitutive relations $(T^{\mu}_{\ \alpha}, \Sigma^{\mu\alpha}_{\ \beta}, J^{\mu}, \mathcal{E}_i)$, the local second law of thermodynamics requires that there must exist an entropy current $J^{\mu}_{\rm S}$ whose divergence is non-negative for all thermodynamically isolated fluid configurations, i.e.

$$\underline{\mathbf{D}}_{\mu}J_{\mathbf{S}}^{\mu}\simeq\Delta\geq0.$$
(2.49)

To respect this inequality, we require that Δ evaluates to a quadratic form with positive semi-definite eigenvalues. The sacred "global" second law of thermodynamics follows from here if we define the total entropy as $S_{\text{tot}} = \int d^d x \sqrt{-g} J_{\text{S}}^0$. Taking a ∂_0 derivative and dropping the boundary terms,

$$\partial_0 S_{\text{tot}} = \int \mathrm{d}^d x \; \partial_0 \left(\sqrt{-g} \; J_{\mathrm{S}}^0 \right) = \int \mathrm{d}^d x \sqrt{-g} \; \underline{\mathrm{D}}_\mu J_{\mathrm{S}}^\mu \gtrsim 0, \tag{2.50}$$

we can see that the total entropy always increases. To understand why we should expect a stronger "local" version of the second law to hold, note that in hydrodynamics we are only ever leaving the equilibrium perturbatively (in derivatives). This implies that locally the system always stays in thermodynamic equilibrium while the variations away from it happen at much larger scales. So locally at every point in the spacetime, the entropy must be produced, leading to the local second law of thermodynamics. At its face value, the second law (2.49) is an inequality, but it is an inequality that must be satisfied for all thermodynamically isolated configurations. Consequently, if even a rogue implausible fluid configuration could potentially lead a transport coefficient to violate the inequality, the second law forces the said transport coefficient to zero. In this sense, the local second law of thermodynamics is a very strong statement and can significantly bring down the number of independent transport coefficients. Note that in its original form, the second law is only defined over thermodynamically isolated fluid configurations. We can bypass this limitation by noting that eq. (2.49) can be expressed as an off-shell statement by adding to it a linear combination of the external sources

$$\underline{\mathbf{D}}_{\mu}J_{\mathbf{S}}^{\mu} + \beta^{\alpha}P_{\alpha}^{\mathrm{ext}} + \left(\Lambda_{\beta}^{\Sigma\beta}{}_{\alpha} + \beta^{\mu}C_{\ \mu\alpha}^{\beta}\right)S^{\mathrm{ext}\,\alpha}{}_{\beta} + \left(\Lambda_{\beta} + \beta^{\mu}A_{\mu}\right)\cdot Q^{\mathrm{ext}} = \Delta \ge 0.$$
(2.51)

An important comment is in order. Recall that since the hydrodynamic fields \mathcal{B} were arbitrarily defined to be solved for using the conservation laws, they admit an arbitrary redefinition freedom. In writing eq. (2.51), we have fixed this freedom off-shell by choosing the hydrodynamic fields to be equal to the multipliers for the external sources, which a priori could have been arbitrary. Note however that on-shell, this does not completely fix the redefinition freedom. We can redefine the hydrodynamic fields with arbitrary combinations of the external sources and on-shell they are still equivalent. Thanks to the hydrostatic principle, we know that the external sources identically vanish upon setting $\mathcal{B} = \mathcal{K}$ and hence can be represented entirely in terms of $\delta_{\mathcal{B}} e^{\alpha}{}_{\mu}$, $\delta_{\mathcal{B}} C^{\beta}{}_{\mu\alpha}$, $\delta_{\mathcal{B}} A_{\mu}$, $\delta_{\mathcal{B}} \varphi^{i}$, and their derivatives. To fix the residual redefinition freedom, therefore, all we need to do is choose a set of vector, $\mathfrak{so}(d-1, 1)$ -valued scalar, and \mathfrak{g} -valued scalar combinations of these constituent fields to eliminate from the constitutive relations using the equations of motion. For definiteness, in this work we choose our constitutive relations to be independent of $u^{\mu}\delta_{\mathcal{B}}e^{\alpha}{}_{\mu}$, $u^{\mu}\delta_{\mathcal{B}}C^{\alpha}{}_{\mu\beta}$, and $u^{\mu}\delta_{\mathcal{B}}A_{\mu}$.

As it turns out, eq. (2.51) can be expressed in a more useful form if instead of the entropy current, one works with its Legendre transform *free energy current* and the associated *free energy Hall current*

$$N^{\mu} = J^{\mu}_{\rm S} + \beta^{\alpha} T^{\mu}{}_{\alpha} + \left(\Lambda^{\Sigma\beta}_{\beta}{}^{\alpha}{}_{\alpha} + \beta^{\mu} C^{\beta}{}_{\mu\alpha}\right) \Sigma^{\mu\alpha}{}_{\beta} + (\Lambda_{\beta} + \beta^{\mu} A_{\mu}) \cdot J^{\mu} + \mathcal{N}^{i\mu}_{\beta} \mathcal{E}_{i},$$
$$N^{\perp}_{\rm H} = \left(\Lambda^{\Sigma\beta}_{\beta}{}_{\alpha} + \beta^{\mu} C^{\beta}{}_{\mu\alpha}\right) \Sigma^{\perp\alpha}_{\rm H}{}_{\beta} + (\Lambda_{\beta} + \beta^{\mu} A_{\mu}) \cdot J^{\perp}_{\rm H},$$
(2.52)

where $\mathcal{N}_{\beta}^{i\mu}$ has been defined in eq. (2.18). Using the definition of external sources from eq. (2.44) and performing some differentiation by parts, eq. (2.51) can be expressed as the so-called *adiabaticity equation*,

$$\underline{\mathbf{D}}_{\mu}N^{\mu} - \mathbf{N}_{\mathrm{H}}^{\perp} = T^{\mu}{}_{\alpha}\delta_{\mathcal{B}}e^{\alpha}{}_{\mu} + \Sigma^{\mu\alpha}{}_{\beta}\delta_{\mathcal{B}}C^{\beta}{}_{\mu\alpha} + J^{\mu}\cdot\delta_{\mathcal{B}}A_{\mu} + \mathcal{E}_{i}\delta_{\mathcal{B}}\varphi^{i} + \Delta, \qquad \Delta \ge 0.$$
(2.53)

Here we have made use of the symmetry variations defined in eqs. (2.16) and (2.18) to simplify the expressions. This equation is at the heart of entire hydrodynamics and forms the basis for most of the discussion presented in this work.

We would like to study what constraints does imposing the second law of thermodynamics imply on the hydrodynamic constitutive relations. In other words, we would like to find the most generic hydrodynamic constitutive relations $(T^{\mu}{}_{\alpha}, \Sigma^{\mu\alpha}{}_{\beta}, J^{\mu}, \mathcal{E}_i)$ allowed by the adiabaticity equation (2.53) for some choice of the free energy current N^{μ} and quadratic form Δ . Conventionally, one would list the most generic tensor structures that can enter the constitutive relations and free energy current up to a particular derivative order, append them with arbitrary transport coefficients, plug them into eq. (2.53), and read out the constraints arising from demanding $\Delta \geq 0$. This is the approach we took in the example we studied in chapter 1. However, this path becomes incredibly cumbersome as we go to higher derivative orders or include non-trivial φ^i fields. Arguably, it is more fruitful to directly inspect the adiabaticity equation (2.53) and classify all the solutions it can admit. This allows us to outline a generic algorithm to generate the hydrodynamic constitutive relations with arbitrary additional gapless modes and at arbitrarily high derivative orders, without having to rely on the counting of independent tensor structures.

For the clarity of notation in the following discussion, let us define

$$\Phi = \begin{pmatrix} e^{\alpha}{}_{\mu} \\ C^{\alpha}{}_{\mu\beta} \\ A_{\mu} \\ \varphi^{i} \end{pmatrix}, \qquad \mathcal{C} = \begin{pmatrix} T^{\mu}{}_{\alpha} \\ \Sigma^{\mu\alpha}{}_{\beta} \\ J^{\mu} \\ \mathcal{E}_{i} \end{pmatrix}.$$
 (2.54)

 \mathfrak{C} and Φ can be seen as elements of an extended vector space \mathfrak{V} . In this notation, \mathfrak{C} denotes the hydrodynamic constitutive relations which are written in terms of the fields \mathfrak{B} and Φ . This allows us to express the adiabaticity equation (2.53) in a compact form

$$\underline{\mathbf{D}}_{\mu}N^{\mu} = \mathbf{N}_{\mathbf{H}}^{\perp} + \mathcal{C} \cdot \delta_{\mathcal{B}} \Phi + \Delta, \qquad \Delta \ge 0.$$
(2.55)

Looking at the form of eq. (2.55), the tensor structures appearing in the hydrodynamic constitutive relations $(T^{\mu}_{\ \alpha}, \Sigma^{\mu\alpha}_{\ \beta}, J^{\mu}, \mathcal{E}_i)$ can be naturally split into two independent sectors and 5 independent classes

• Hydrostatic sector: In this sector, constitutive relations and free energy-current are constructed out of the independent tensor structures that, or any of their linear combinations, do not vanish in a hydrostatic configuration, i.e. upon setting $\mathcal{B} = \mathcal{K}$ where \mathcal{K} is a background isometry. See section 2.2.4 for details on this language. We commonly use the terminology that they or any of their linear combinations cannot contain an instance of " $\delta_{\mathcal{B}}$ ". Hydrostatic constitutive relations are required to satisfy a non-dissipative version of the adiabaticity equation

$$\underline{\mathbf{D}}_{\mu}N_{\mathrm{hs}}^{\mu} = \mathbf{N}_{\mathrm{H}}^{\perp} + \mathcal{C}_{\mathrm{hs}} \cdot \delta_{\mathcal{B}}\Phi.$$
(2.56)

The respective solutions can be classified in three classes

- 1. Class A (anomaly induced transport): These are the constitutive relations which are induced by anomalies in our symmetries. They are completely fixed in terms of the anomaly polynomial, which in turn is characterised by a set of constant anomaly coefficients. See section 2.3.2.
- 2. Class H_V (hydrostatic vector transport): These are the hydrostatic constitutive relations which are characterised by a free energy flow transverse to the fluid velocity, excluding Class A transport. They, as well, are completely determined up to a set of constants. See section 2.3.3.

- 3. Class H_S (hydrostatic scalar transport): These are the hydrostatic constitutive relations which are characterised by a free energy flow which has a component along the fluid velocity. They can be obtained from a generic hydrostatic scalar \mathcal{N} which is only defined up to total derivatives. See section 2.3.4.
- Non-hydrostatic sector: In this sector, constitutive relations and free energy current are made out of independent tensor structures that vanish in a hydrostatic configuration. In practise, they involve all the tensor structures that have at least one instance of $\delta_{\mathcal{B}}\Phi$ or their derivatives. Being faithful to our terminology, they contain at least one instance of " $\delta_{\mathcal{B}}$ ". Non-hydrostatic constitutive relations are required to satisfy a non-anomalous version of the adiabaticity equation

$$\underline{\mathbf{D}}_{\mu}N^{\mu}_{\mathrm{nhs}} = \mathcal{C}_{\mathrm{nhs}} \cdot \delta_{\mathcal{B}}\Phi + \Delta, \qquad \Delta \ge 0.$$
(2.57)

Keep note that due to our choice while fixing the residual hydrodynamic redefinition freedom, we are not allowed to use the non-hydrostatic tensor structures which depend on $u^{\mu}\delta_{\mathcal{B}}e^{\alpha}{}_{\mu}$, $u^{\mu}\delta_{\mathcal{B}}C^{\beta}{}_{\mu\alpha}$, or $u^{\mu}\delta_{\mathcal{B}}A_{\mu}$. The respective solutions can be classified in two classes

- 4. Class $\overline{\mathbb{D}}$ (non-dissipative transport): These non-hydrostatic constitutive relations are characterised by no entropy production. They are completely determined by $\overline{\mathfrak{D}}_n|_{n\geq 0} \in \mathfrak{V} \times \mathfrak{V}$ with $\overline{\mathfrak{D}}_n^{\mathrm{T}} = -(-)^n \overline{\mathfrak{D}}_n$. See section 2.3.5.
- 5. Class D (dissipative transport): These non-hydrostatic constitutive relations are the only ones that are responsible for the production of entropy. They are completely determined by $\mathfrak{D}_n|_{n\geq 0} \in \mathfrak{V} \times \mathfrak{V}$ with $\mathfrak{D}_n^{\mathrm{T}} = (-)^n \mathfrak{D}_n$. See section 2.3.5.

Together, the hydrostatic and non-hydrostatic sectors make up the most generic hydrodynamic constitutive relations that respect the local second law of thermodynamics.

A comment is due on the form of the adiabaticity equation in the respective sectors. Since a hydrostatic fluid configuration is a state of thermal equilibrium, we know that there should be no entropy production in the hydrostatic sector. Consequently, Δ in the hydrostatic sector is zero. In fact, by the virtue of being a quadratic form, Δ must at least be quadratic in " δ_B ". Coming to the free energy current, N^{μ} must also be made purely out of the hydrostatic data. It can obviously not contain multiple instances of δ_B , as the RHS of eq. (2.55) only contains one. On the other hand, if N^{μ} contained one δ_B , there will be a term in its gradient which has δ_B acted upon by a derivative, which cannot be matched with the RHS either. In essence, therefore, the hydrostatic sector is completely parametrised by a hydrostatic N^{μ}_{hs} that satisfies eq. (2.56). Since anomalies are being taken care of in the hydrostatic sector, the non-hydrostatic sector is required to satisfy the non-anomalous adiabaticity equation (2.57).

There is also a sixth class of "hydrodynamic transport"

6. Class S (entropy transport): This class contains solutions to the adiabaticity equation with vanishing constitutive relations but non-trivial free energy or

entropy transport. These are completely characterised by an antisymmetric tensor $X^{\mu\nu}$ and matrices $\mathfrak{S}_{mn}|_{m,n\geq 1}$ with $\mathfrak{S}_{mn}^{\mathrm{T}} = \mathfrak{S}_{nm}$. See section 2.3.6.

However, Class S solutions are not genuine hydrodynamic transport. They merely characterise the multitude of entropy currents that satisfy the second law for the same set of constitutive relations.

In the following subsections, we cover all of these classes in detail. Most of the results presented here are directly taken from [52], which discusses the classification of the hydrodynamic transport for fluids without an independent spin current, background torsion, or extra gapless modes. The extension thereof to include φ^i fields corresponding to broken non-Abelian internal symmetries was presented in our work [1]. Work in [1] also presented a new understanding of the non-hydrostatic sector which we have adopted in this work.⁶ The discussion presented here is a direct generalisation of [1] to include a spin current, background torsion, and arbitrary φ^i fields.

2.3.2 Anomaly induced transport

Before delving into the full analysis of the adiabaticity equation and the constraints it imposes on the constitutive relations, let us first deal with the affect of anomalies in hydrodynamics. The anomaly induced constitutive relations, called Class A, lie within the hydrostatic sector and show up in the hydrostatic adiabaticity equation (2.56) in the form of the free energy Hall current $N_{\rm H}^{\perp}$ defined in eq. (2.52). We need to find a particular solution to eq. (2.56) for a generic $N_{\rm H}^{\perp}$, which we can append to the most generic solutions of the non-anomalous adiabaticity equation to obtain the most generic hydrodynamic constitutive relations. The end result of a significant amount of literature written on this problem [47, 52, 87, 92–94] is that this particular solution can be most naturally obtained using the so-called transgression machinery. In the following we present a skimmed version of this construction.

Let us begin by defining the "shadow connections"

$$\hat{\boldsymbol{A}} = \boldsymbol{A} + \mu \, \boldsymbol{u} = (A_{\mu} + \mu \, u_{\mu}) \, \mathrm{d}x^{\mu}, \qquad \hat{\boldsymbol{C}}^{\alpha}{}_{\beta} = \boldsymbol{C}^{\alpha}{}_{\beta} + \mu^{\Sigma \, \alpha}{}_{\beta} \, \boldsymbol{u} = \left(C^{\alpha}{}_{\mu\beta} + \mu^{\Sigma \, \alpha}{}_{\beta} \, u_{\mu} \right) \, \mathrm{d}x^{\mu}.$$
(2.58)

These are defined so as to satisfy $\Lambda_{\beta} + \beta^{\mu} \hat{A}_{\mu} = \Lambda_{\beta}^{\Sigma \alpha}{}_{\beta} + \beta^{\mu} \hat{C}^{\alpha}{}_{\mu\beta} = 0$. Let us denote the anomaly polynomial of the theory by \mathcal{P} , which is made out of A and $C^{\alpha}{}_{\beta}$, and can be used to obtain the Hall currents using eq. (2.25). We can also compute the associated free energy Hall current defined in eq. (2.52) using the formula

$$\star_{(d+1)} \mathbf{N}_{\mathrm{H}} = \frac{1}{T} \star_{(d+1)} \left(\mu^{\Sigma} \cdot \boldsymbol{\Sigma}_{\mathrm{H}} + \mu \cdot \mathbf{J}_{\mathrm{H}} \right).$$
(2.59)

⁶In [52], the non-hydrostatic and entropy transport is together classified into 5 classes: B, C, D', \overline{H}_S , and \overline{H}_V , in contrast to our 3 classes: D, \overline{D} , and S. Class C is characterised by terms in the entropy/free energy current which are topologically conserved, and is contained within Class S. Class D' is equal to Class D \cup (S \setminus C), which contains the solutions of the adiabaticity equation corresponding to the most generic quadratic form Δ . Finally, the three non-hydrostatic non-dissipative Classes B, \overline{H}_S , and \overline{H}_V are an alternate (and redundant) parametrisation of Class \overline{D} .

The "shadow anomaly polynomial" $\hat{\mathcal{P}}$ can be obtained starting from \mathcal{P} by replacing all instances of A and $C^{\alpha}{}_{\beta}$ with \hat{A} and $\hat{C}^{\alpha}{}_{\beta}$ respectively. A similar procedure can be used to define the shadow Hall currents as well. Having done that, let us consider

$$\star \boldsymbol{\Sigma}_{\boldsymbol{\mathcal{P}}} = \frac{\boldsymbol{u}}{\mathrm{d}\boldsymbol{u}} \wedge \star_{(d+1)} \left(\boldsymbol{\Sigma}_{\mathrm{H}} - \hat{\boldsymbol{\Sigma}}_{\mathrm{H}} \right), \qquad \star \boldsymbol{J}_{\boldsymbol{\mathcal{P}}} = \frac{\boldsymbol{u}}{\mathrm{d}\boldsymbol{u}} \wedge \star_{(d+1)} \left(\boldsymbol{J}_{\mathrm{H}} - \hat{\boldsymbol{J}}_{\mathrm{H}} \right), \\ \star \boldsymbol{q}_{\boldsymbol{\mathcal{P}}} = -\frac{\boldsymbol{u}}{\mathrm{d}\boldsymbol{u} \wedge \mathrm{d}\boldsymbol{u}} \wedge \left(\boldsymbol{\mathcal{P}} - \hat{\boldsymbol{\mathcal{P}}} + T \mathrm{d}\boldsymbol{u} \wedge \star_{(d+1)} \hat{\boldsymbol{N}}_{\mathrm{H}} \right).$$
(2.60)

In terms of these abstract definitions, the Class A constitutive relations are given as

$$(T^{\mu\alpha})_{\mathcal{A}} = q^{\mu}_{\mathcal{P}} u^{\alpha} + q^{\alpha}_{\mathcal{P}} u^{\mu}, \qquad (\Sigma^{\mu\alpha\beta})_{\mathcal{A}} = \Sigma^{\mu\alpha\beta}_{\mathcal{P}}, \qquad (J^{\mu})_{\mathcal{A}} = J^{\mu}_{\mathcal{P}}.$$
(2.61a)

It can be explicitly checked that they satisfy the hydrostatic adiabaticity equation (2.56) with the free energy current

$$(N^{\mu})_{\mathcal{A}} = \frac{1}{T} \left(-q^{\mu}_{\mathcal{P}} + \mu^{\Sigma} \cdot \Sigma^{\mu}_{\mathcal{P}} + \mu \cdot J^{\mu}_{\mathcal{P}} \right).$$
(2.61b)

Since the anomaly polynomial does not make any reference to the gapless modes φ^i , it is natural that the respective equations of motion $(\mathcal{E}_i)_A = 0$ in Class A. Curiously, in Class A there is no entropy transport either

$$(J_{\rm S}^{\mu})_{\rm A} = 0.$$
 (2.61c)

To verify the validity of these claims, one can define an effective action describing the anomalous transport [93]

$$S_{\rm A} = \int_{\mathcal{B}} \mathcal{V}_{\mathcal{P}}, \qquad \mathcal{V}_{\mathcal{P}} = \frac{u}{\mathrm{d}u} \wedge \left(\mathcal{P} - \hat{\mathcal{P}}\right).$$
 (2.62)

The integrand $\mathcal{V}_{\mathcal{P}}$ is called a transgression form. After some algebraic manipulations, one can show that the infinitesimal variation of $S_{\rm A}$ is given by

$$\delta S_{\mathcal{A}} = \int_{\mathcal{B}} \mathrm{d}^{d+1} x \sqrt{-g_{d+1}} \left(\Sigma_{\mathcal{H}}{}^{\hat{\mu}\hat{\alpha}}{}_{\hat{\beta}} \, \delta C^{\hat{\beta}}{}_{\hat{\mu}\hat{\alpha}} - \hat{\Sigma}_{\mathcal{H}}{}^{\hat{\mu}\hat{\alpha}}{}_{\hat{\beta}} \, \delta \hat{C}^{\hat{\beta}}{}_{\hat{\mu}\hat{\alpha}} + \mathrm{J}^{\hat{\mu}}_{\mathcal{H}} \, \delta A_{\hat{\mu}} - \hat{\mathrm{J}}^{\hat{\mu}}_{\mathcal{H}} \, \delta \hat{A}_{\hat{\mu}} \right) + \int_{\mathcal{M}} \mathrm{d}^{d} x \sqrt{-g} \left(\Sigma_{\mathcal{P}}{}^{\mu\alpha}{}_{\beta} \, \delta C^{\beta}{}_{\mu\alpha} + J_{\mathcal{P}}{}^{\mu} \, \delta A_{\mu} + q_{\mathcal{P}}{}^{\mu} \, \delta u_{\mu} \right). \quad (2.63)$$

The effective action S_A is manifestly invariant under all the symmetries of the theory. Consequently, under the action of an infinitesimal set of symmetry parameters \mathcal{X} , the variation $\delta_{\mathcal{X}}S_A$ is trivially zero. In particular, if we choose $\mathcal{X} = \alpha \mathcal{B}$, requiring $\delta_{\mathcal{X}}S_A = 0$ for an arbitrary scalar field α precisely leads to the identity

$$\underline{\mathbf{D}}_{\mu}(N^{\mu})_{\mathbf{A}} - \mathbf{N}_{\mathbf{H}}^{\perp} = (T^{\mu}_{\ \alpha})_{\mathbf{A}} \delta_{\mathcal{B}} e^{\alpha}_{\ \mu} + (\Sigma^{\mu\alpha}_{\ \beta})_{\mathbf{A}} \delta_{\mathcal{B}} C^{\beta}_{\ \mu\alpha} + (J^{\mu})_{\mathbf{A}} \cdot \delta_{\mathcal{B}} A_{\mu}, \tag{2.64}$$

which is nothing but the hydrostatic adiabaticity equation for Class A. Hence we have verified that Class A constitutive relations do satisfy the adiabaticity equation.

2.3.3 Transcendental anomalies

While we are on the topic of anomalies, let us consider a particular extension of Class A to accommodate more generic solutions of the hydrostatic adiabaticity equation. These are called *transcendental anomalies* or *hydrostatic vectors* in the literature, denoted by Class H_V . We justify this nomenclature in the next subsection. The most natural route to these "anomalies" is using an auxiliary U(1)_T symmetry introduced in [52]. We briefly review this method below.

Let us introduce an auxiliary Abelian global symmetry $U(1)_{\mathsf{T}}$ in our theory and an associated background gauge field A_{μ}^{T} . We take the corresponding chemical potential to be $\mu^{\mathsf{T}} = T$, i.e. $\Lambda_{\beta}^{\mathsf{T}} + \beta^{\mu} A_{\mu}^{\mathsf{T}} = 1$. The anomaly polynomial for this enlarged theory is denoted by \mathcal{P}_{T} , which is known as the *thermal anomaly polynomial*. Generically, it is characterised by more constants compared to the physical anomaly polynomial \mathcal{P} due to the presence of additional terms involving the Chern classes of $F_{\mu\nu}^{\mathsf{T}}$. In the limit that $F_{\mu\nu}^{\mathsf{T}}$ is taken to zero, \mathcal{P}_{T} reduces to \mathcal{P} . With this in mind, let us decompose

$$\boldsymbol{\mathcal{P}}_{\mathsf{T}} = \boldsymbol{\mathcal{P}} + \boldsymbol{\mathcal{P}}_{\mathsf{H}_{\mathsf{V}}} = \boldsymbol{\mathcal{P}} + \sum_{j \ge 1} \left(\boldsymbol{F}^{\mathsf{T}} \right)^{\wedge j} \wedge \boldsymbol{\mathcal{P}}_{\mathsf{H}_{\mathsf{V}}, j}.$$
 (2.65)

 $\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}$ contains terms which are at least linear in $F_{\mu\nu}^{\mathsf{T}}$, wherein $\mathcal{P}_{\mathrm{H}_{\mathrm{V}},j}$ are (d+2-2j)-rank anomaly polynomials made out of \mathbf{F} and \mathbf{R} . For d = 4k - 2 and d = 4k, the extra piece $\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}$ involves k(k+1) and $(k+1)^2$ constants respectively.

We can use the method prescribed in the previous subsection to find a particular anomaly induced solution to the extended hydrostatic adiabaticity equation

$$\underline{\mathbf{D}}_{\mu}N^{\mu} - \mathbf{N}_{\mathrm{H}}^{\perp} - \mathbf{J}_{\mathrm{TH}}^{\perp} = T^{\mu}{}_{\alpha}\delta_{\mathcal{B}}e^{\alpha}{}_{\mu} + \Sigma^{\mu\alpha}{}_{\beta}\delta_{\mathcal{B}}C^{\beta}{}_{\mu\alpha} + J^{\mu}\cdot\delta_{\mathcal{B}}A_{\mu} + \mathcal{E}_{i}\delta_{\mathcal{B}}\varphi^{i} + J^{\mu}_{\mathsf{T}}\delta_{\mathcal{B}}A^{\mathsf{T}}_{\mu},$$
(2.66)

where $\delta_{\mathcal{B}}A_{\mu}^{\mathsf{T}} = \pounds_{\beta}A_{\mu}^{\mathsf{T}} + \partial_{\mu}\Lambda_{\beta}^{\mathsf{T}} = \beta^{\nu}F_{\nu\mu}^{\mathsf{T}}$. The free energy Hall current \mathbf{N}_{H} in this extended theory is again taken to be eq. (2.59), except that Σ_{H} and \mathbf{J}_{H} are now being defined using \mathcal{P}_{T} as opposed to \mathcal{P} . Similarly, \mathbf{J}_{TH} is the Hall current associated with the U(1)_T symmetry, i.e.

$$\star_{(d+1)} \mathbf{J}_{\mathsf{TH}} = \frac{\partial \boldsymbol{\mathcal{P}}_{\mathsf{T}}}{\partial \boldsymbol{F}^{\mathsf{T}}} = \sum_{j \ge 1} j(\boldsymbol{F}^{\mathsf{T}})^{\wedge (j-1)} \wedge \boldsymbol{\mathcal{P}}_{\mathsf{H}_{\mathsf{V}},j}.$$
(2.67)

Consider taking a limit of this extended theory wherein we set $F_{\mu\nu}^{\mathsf{T}} \to 0$. Doing this, the $\delta_{\mathbb{B}} A_{\mu}^{\mathsf{T}}$ term in eq. (2.66) identically vanishes. If we ignore the terms in \mathcal{P}_{T} which are linear in $F_{\mu\nu}^{\mathsf{T}}$ for now (i.e. stick to j > 1), the U(1)_T Hall current \mathbf{J}_{TH} alse vanishes in this limit. Consequently, the modified adiabaticity equation (2.66) precisely reduces to the original hydrostatic adiabaticity equation (2.56). Unfortunately, for the terms in \mathcal{P}_{T} which are linear in $F_{\mu\nu}^{\mathsf{T}}$ (i.e. j = 1), the associated Hall current $\star_{(d+1)}\mathbf{J}_{\mathsf{TH}} = \mathcal{P}_{\mathsf{H}_{\mathsf{V}},1} = d\mathbf{I}_{\mathsf{H}_{\mathsf{V}},1}$ does not vanish. We can cheat our way out of this situation, however, by noting that we can shift $N^{\mu} \to N^{\mu} + (\star \mathbf{I}_{\mathsf{H}_{\mathsf{V}},1})^{\mu}$ to get rid of the extra piece in the adiabaticity equation. This leads to a gauge-non-invariant free energy current, which in principle should not be an issue because the free energy or entropy currents are not physical observables of our theory and are, therefore, not restricted to be gauge invariant. However, if one insists on having a

gauge-invariant entropy current, $\mathcal{P}_{H_{V},1}$ must be set to zero.

Modulo this technicality, we can conclude that if we are given a set of anomaly induced constitutive relations in the extended theory which satisfy eq. (2.66), we can take $F_{\mu\nu}^{\mathsf{T}} \to 0$ to gain a solution to our original adiabaticity equation (2.53). Interestingly, we find that the solution thus obtained is not merely Class A. Even though $\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}$ vanishes in this limit, the hydrodynamic constitutive relations it induces, called Class H_{V} , do survive. They satisfy the non-anomalous version of the hydrostatic adiabaticity equation (2.56).

To obtain the explicit solution, let us define a shadow $U(1)_{\mathsf{T}}$ gauge field $\hat{A}^{\mathsf{T}} = A^{\mathsf{T}} + Tu$. We can follow through the exact same analysis we did in the previous subsection, but this time with the extra Abelian gauge field, and obtain the extended anomaly induced constitutive relations. We skip the details and directly write down the solutions in $F_{\mu\nu}^{\mathsf{T}} \to 0$ limit. Since the entire construction is linear in the anomaly polynomial, we can treat \mathcal{P} and $\mathcal{P}_{\mathsf{H}_{\mathsf{V}}}$ segments of the thermal anomaly polynomial \mathcal{P}_{T} independently. The physical anomaly polynomial \mathcal{P} has no dependence on $F_{\mu\nu}^{\mathsf{T}}$ and leads to the familiar Class A constitutive relations we discussed in the previous subsection. The $\mathcal{P}_{\mathsf{H}_{\mathsf{V}}}$ piece, however, is novel. The key point to note is that although $\mathcal{P}_{\mathsf{H}_{\mathsf{V}}}$ and the associated Hall currents

$$\star_{(d+1)}(\boldsymbol{\Sigma}_{\mathrm{H}})_{\mathrm{H}_{\mathrm{V}}} = \frac{\partial \boldsymbol{\mathcal{P}}_{\mathrm{H}_{\mathrm{V}}}}{\partial \boldsymbol{R}}, \quad \star_{(d+1)}(\mathbf{J}_{\mathrm{H}})_{\mathrm{H}_{\mathrm{V}}} = \frac{\partial \boldsymbol{\mathcal{P}}_{\mathrm{H}_{\mathrm{V}}}}{\partial \boldsymbol{F}}, \quad \star_{(d+1)}(\mathbf{J}_{\mathrm{TH}})_{\mathrm{H}_{\mathrm{V}}} = \frac{\partial \boldsymbol{\mathcal{P}}_{\mathrm{H}_{\mathrm{V}}}}{\partial \boldsymbol{F}^{\mathsf{T}}}, \\ \star_{(d+1)}(\mathbf{N}_{\mathrm{H}})_{\mathrm{H}_{\mathrm{V}}} = \frac{1}{T} \star_{(d+1)} \left(\mu^{\Sigma} \cdot (\boldsymbol{\Sigma}_{\mathrm{H}})_{\mathrm{H}_{\mathrm{V}}} + \mu \cdot (\mathbf{J}_{\mathrm{H}})_{\mathrm{H}_{\mathrm{V}}} \right), \quad (2.68)$$

vanish upon setting $F_{\mu\nu}^{\mathsf{T}} \to 0$ (except j = 1), the respective shadow quantities do not. These are precisely responsible for the non-trivial Class H_V constitutive relations. We find

$$(T^{\mu\alpha})_{\mathrm{H}_{\mathrm{V}}} = q^{\mu}_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} u^{\alpha} + q^{\alpha}_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} u^{\mu}, \qquad (\Sigma^{\mu\alpha\beta})_{\mathrm{H}_{\mathrm{V}}} = \Sigma^{\mu\alpha\beta}_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}}, \qquad (J^{\mu})_{\mathrm{H}_{\mathrm{V}}} = J^{\mu}_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}}, (N^{\mu})_{\mathrm{H}_{\mathrm{V}}} = J^{\mu}_{\mathsf{T}\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} + \frac{1}{T} \left(-q^{\mu}_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} + \mu^{\Sigma} \cdot \Sigma_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} + \mu \cdot J^{\mu}_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} \right) - (\star \mathbf{I}_{\mathrm{H}_{\mathrm{V}},1})^{\mu}, \qquad (2.69)$$

along with $(\mathcal{E}_i)_{\mathrm{H}_{\mathrm{V}}} = 0$, where

$$\star \boldsymbol{\Sigma}_{\boldsymbol{\mathcal{P}}_{H_{V}}} = -\frac{\boldsymbol{u}}{\mathrm{d}\boldsymbol{u}} \wedge \star_{(d+1)} (\hat{\boldsymbol{\Sigma}}_{H})_{H_{V}} \Big|_{\boldsymbol{F}^{\mathsf{T}} \to 0}, \qquad \star \boldsymbol{J}_{\boldsymbol{\mathcal{P}}_{H_{V}}} = -\frac{\boldsymbol{u}}{\mathrm{d}\boldsymbol{u}} \wedge \star_{(d+1)} (\hat{\boldsymbol{J}}_{H})_{H_{V}} \Big|_{\boldsymbol{F}^{\mathsf{T}} \to 0}, \star \boldsymbol{J}_{\mathsf{T}\boldsymbol{\mathcal{P}}_{H_{V}}} = \frac{\boldsymbol{u}}{\mathrm{d}\boldsymbol{u}} \wedge \star_{(d+1)} \left((\boldsymbol{J}_{\mathsf{T}H})_{H_{V}} - (\hat{\boldsymbol{J}}_{\mathsf{T}H})_{H_{V}} \right) \Big|_{\boldsymbol{F}^{\mathsf{T}} \to 0}, \star \boldsymbol{q}_{\boldsymbol{\mathcal{P}}_{H_{V}}} = \frac{\boldsymbol{u}}{\mathrm{d}\boldsymbol{u} \wedge \mathrm{d}\boldsymbol{u}} \wedge \left(\hat{\boldsymbol{\mathcal{P}}}_{H_{V}} - T\mathrm{d}\boldsymbol{u} \wedge \star_{(d+1)} \left((\hat{\boldsymbol{N}}_{H})_{H_{V}} + (\hat{\boldsymbol{J}}_{\mathsf{T}H})_{H_{V}} \right) \right) \Big|_{\boldsymbol{F}^{\mathsf{T}} \to 0}.$$
(2.70)

The Class H_V constitutive relations satisfy the non-anomalous version of the adiabaticity equation (2.56). Unlike Class A however, these constitutive relations do induce a non-trivial entropy transport

$$\left(J_{\rm S}^{\mu}\right)_{\rm H_{\rm V}} = J_{\mathsf{T}\mathcal{P}_{\rm H_{\rm V}}}^{\mu}.$$
(2.71)

Finally, following our discussion around eq. (2.62), we can also write down an effective action to generate Class H_V constitutive relations

$$S_{\mathrm{H}_{\mathrm{V}}} = -\int_{\mathcal{B}} \frac{\boldsymbol{u}}{\mathrm{d}\boldsymbol{u}} \wedge \hat{\boldsymbol{\mathcal{P}}}_{\mathrm{H}_{\mathrm{V}}} \Big|_{\boldsymbol{F}^{\mathsf{T}} \to 0}.$$
 (2.72)

Varying this action and invoking its invariance under symmetries, we can verify that the Class H_V constitutive relations indeed satisfy the non-anomalous adiabaticity equation.

2.3.4 Hydrostatic transport

Over the previous two subsections, we have covered two special classes of solutions of the hydrostatic adiabaticity equation. First, the Class A constitutive relations form a particular solution to eq. (2.56) which takes care of all the anomaly induced hydrodynamic transport. Second, the Class H_V constitutive relations satisfy the non-anomalous version of eq. (2.56) and are characterised by a set of undermined constants. We are now in a position to specify the remaining hydrostatic constitutive relations as well.

We would first like to note that there can be no non-trivial solutions to the hydrostatic adiabaticity equation (2.56) with $N_{\rm hs}^{\mu} = 0$. Consequently, the hydrostatic constitutive relations are characterised by the most generic hydrostatic free energy current $N_{\rm hs}^{\mu}$. We choose a decomposition

$$N_{\rm hs}^{\mu} = \left(\mathcal{N}\beta^{\mu} + \Theta_{\mathcal{N}}^{\mu}\right) + \mathbb{N}^{\mu},\tag{2.73}$$

where $\mathbb{N}^{\mu}\beta_{\mu} = 0$. \mathcal{N} is the most generic scalar made out of the independent hydrostatic data. $\Theta^{\mu}_{\mathcal{N}}$ is a \mathcal{N} -dependent non-hydrostatic vector defined via

$$\underline{\mathbf{D}}_{\mu}(\mathcal{N}\beta^{\mu}) = \frac{1}{\sqrt{-g}} \delta_{\mathcal{B}}\left(\sqrt{-g}\,\mathcal{N}\right) = \Phi \cdot \mathcal{C}_{\mathrm{H}_{\mathrm{S}}} - \underline{\mathbf{D}}_{\mu}\Theta_{\mathcal{N}}^{\mu},\tag{2.74}$$

which ensures that $\underline{D}_{\mu}(\mathcal{N}\beta^{\mu} + \Theta^{\mu}_{\mathcal{N}})$ has a bare " $\delta_{\mathcal{B}}$ " to match the RHS of eq. (2.53). Eq. (2.74) also defines the constitutive relations $\mathcal{C}_{H_{S}}$ associated with \mathcal{N} , called the Class H_{S} constitutive relations. The respective free energy current is simply $(N^{\mu})_{H_{S}} = \mathcal{N}\beta^{\mu} + \Theta^{\mu}_{\mathcal{N}}$.

On the other hand, \mathbb{N}^{μ} is the most generic hydrostatic vector transverse to β^{μ} , such that $\underline{\mathbf{D}}_{\mu}\mathbb{N}^{\mu} - \mathbf{N}_{\mathrm{H}}^{\perp}$ has exactly one bare " $\delta_{\mathcal{B}}$ ", so that it can fit the hydrostatic adiabaticity equation. We have already seen two classes of free energy currents which meet this criteria in sections 2.3.2 and 2.3.3,

$$\mathbb{N}^{\mu} = (N^{\mu})_{\mathcal{A}} + (N^{\mu})_{\mathcal{H}_{\mathcal{V}}}.$$
(2.75)

The corresponding constitutive relations C_A and C_{H_V} are given in eq. (2.61) and eq. (2.69) respectively. This also justifies the name "hydrostatic vectors" for Class H_V in contrast to the "hydrostatic scalars" for Class H_S . As it happens, eq. (2.75) is already the most generic form of \mathbb{N}^{μ} allowed by the requirements laid out. However, we refrain from giving a detailed proof here; interested readers can consult [52, 87, 92]. Therefore, Classes A, H_V , and H_S make up the complete set of hydrostatic constitutive relations.

By definition, the hydrostatic constitutive relations completely determine the physics in a hydrostatic fluid configuration (obtained by setting $\mathcal{B} = \mathcal{K}$ where \mathcal{K} is a timelike background isometry). But as we discussed in section 2.2.4, to maintain consistency with the thermodynamic equilibrium, constitutive relations in a hydrostatic configuration must admit a hydrostatic effective action. Luckily for us, this notion is already hardwired in the second law requirement. Indeed, the Class H_S constitutive relations in a hydrostatic configuration can be generated from an effective action [51]

$$S_{\mathrm{H}_{\mathrm{S}}}^{\mathrm{hs}} = \beta_0 \int_{\Sigma} \mathrm{d}^{d-1} x \sqrt{-g} \left(\mathcal{N} \right)_{\mathcal{B} = \mathcal{K}}.$$
 (2.76)

The hydrostatic effective actions for Classes A and H_V , on the other hand, are trivially obtained by setting $\mathcal{B} = \mathcal{K}$ in eqs. (2.62) and (2.72) respectively. Furthermore, the hydrostatic effective action thus obtained is actually the most generic such effective action we can write down in a hydrostatic configuration. Therefore, within the hydrostatic sector, the constraints arising form requiring the hydrostatic principle or those following from the second law are exactly equivalent.

2.3.5 Non-hydrostatic transport

Having figured out the hydrostatic transport, let us now move on to the non-hydrostatic sector. To remind ourselves, these are the constitutive relations which contain at least one instance of $\delta_{\mathcal{B}} \Phi$ or their derivatives. To this end, let us introduce some new notation. Let us define a totally symmetric covariant derivative operator

$$\mathbf{D}^{n} = \left(\mathbf{D}_{(\mu_{1}}\mathbf{D}_{\mu_{2}}\dots\mathbf{D}_{\mu_{n}})\right).$$

$$(2.77)$$

 D^n forms a basis for all the differential operators, as the antisymmetric derivatives can always be replaced by combinations involving curvatures and field strengths. Given a differential operator \mathcal{O} , we define its conjugate \mathcal{O}^{\dagger} via

$$X_1\left(\mathcal{O}X_2\right) = \left(\mathcal{O}^{\dagger}X_1\right)X_2 + \underline{\mathbf{D}}_{\mu}\left(\cdots\right)^{\mu}, \qquad (2.78)$$

where X_1, X_2 are some arbitrary fields. Expressed in terms of the basis operators, we have $(D^n)^{\dagger} = (-)^n D^n$.

With this new derivative operator in place, the most generic non-hydrostatic constitutive relations can be expressed in a compact form

$$\mathcal{C}_{\rm nhs} = -\sum_{n=0}^{\infty} \frac{1}{2} \Big(\mathfrak{C}_n \cdot (\mathbf{D}^n \delta_{\mathcal{B}} \Phi) + \mathbf{D}^n (\mathfrak{C}_n \cdot \delta_{\mathcal{B}} \Phi) \Big).$$
(2.79)

 $\mathfrak{C}_n \in \mathfrak{V} \times \mathfrak{V}$ are matrices with additional *n* symmetric indices to be contracted with \mathbb{D}^n . The last term in eq. (2.79) is taken purely for convenience and can be absorbed into the first via differentiation by parts. Let us factor \mathfrak{C}_{nhs} into a so-called dissipative (Class D) and a non-dissipative (Class $\overline{\mathrm{D}}$) class parametrised by

$$\mathfrak{D}_n = \frac{1}{2} \left(\mathfrak{C}_n + (-)^n \mathfrak{C}_n^{\mathrm{T}} \right), \qquad \overline{\mathfrak{D}}_n = \frac{1}{2} \left(\mathfrak{C}_n - (-)^n \mathfrak{C}_n^{\mathrm{T}} \right), \qquad (2.80)$$

respectively. Beginning with the non-dissipative piece first, as it is simpler, we find that

$$\mathcal{C}_{\overline{\mathrm{D}}} \cdot \delta_{\mathcal{B}} \Phi = \underline{\mathrm{D}}_{\mu} \Theta_{\overline{\mathfrak{D}}}^{\mu}, \tag{2.81}$$

where $\underline{D}_{\mu}\Theta_{\overline{\mathfrak{D}}}^{\mu}$ is a total derivative piece obtained after successive differentiation by parts. Comparing it to eq. (2.57) we can infer that the Class \overline{D} constitutive relations identically satisfy the adiabaticity equation with $(N^{\mu})_{\overline{D}} = \Theta_{\overline{\mathfrak{D}}}^{\mu}$ and $\Delta_{\overline{D}} = 0$, with no constraints imposed. Hence the name non-dissipative.

Moving on to the more non-trivial dissipative constitutive relations, we instead find

$$\mathcal{C}_{\mathrm{D}} \cdot \delta_{\mathcal{B}} \Phi = -\sum_{n=0}^{\infty} \delta_{\mathcal{B}} \Phi \cdot \mathfrak{D}_{n} \cdot (\mathrm{D}^{n} \delta_{\mathcal{B}} \Phi) + \underline{\mathrm{D}}_{\mu} \Theta_{\mathfrak{D},0}^{\mu}.$$
 (2.82)

To muster it into a form apt for the adiabaticity equation, we need to do a little more work. We start by noting that the above equation can be massaged into

$$C_{D} \cdot \delta_{\mathcal{B}} \Phi = -\delta_{\mathcal{B}} \Phi \cdot \boldsymbol{\eta} \cdot \delta_{\mathcal{B}} \Phi - 2\delta_{\mathcal{B}} \Phi \cdot \boldsymbol{\eta} \cdot (\Upsilon_{1} \cdot \delta_{\mathcal{B}} \Phi) + \underline{D}_{\mu} \Theta_{\mathfrak{D},0}^{\mu}$$

$$= -\underbrace{\left((1+\Upsilon_{1}) \cdot \delta_{\mathcal{B}} \Phi\right) \cdot \boldsymbol{\eta} \cdot \left((1+\Upsilon_{1}) \cdot \delta_{\mathcal{B}} \Phi\right)}_{\text{quadratic form}} + \underbrace{\left(\Upsilon_{1} \cdot \delta_{\mathcal{B}} \Phi\right)^{\mathrm{T}} \cdot \boldsymbol{\eta} \cdot (\Upsilon_{1} \cdot \delta_{\mathcal{B}} \Phi)}_{\text{residue}} + \underbrace{\underline{D}_{\mu} \Theta_{\mathfrak{D},0}^{\mu}}_{\text{total derivative}}, \qquad (2.83)$$

where $\mathfrak{D}_{0(n)}$ denotes the *n*th derivative piece in $\mathfrak{D}_{(0)}$ and $\eta = \mathfrak{D}_{0(0)}$. On the other hand, Υ_1 is a differential operator

$$\Upsilon_1 = \frac{1}{2} \eta^{-1} \sum_{n=1}^{\infty} \left(\mathfrak{D}_{0(n)} + \mathfrak{D}_n \mathbf{D}^n \right).$$
(2.84)

The quadratic form piece in eq. (2.83) is of most interest to us, as it contributes to Δ . The total derivative piece on the other hand is a contribution to the free energy current N^{μ} . However, we would like to get rid of the residue piece, which has at least 4 derivatives. Using differentiation by parts, this piece can be rewritten as

$$(\Upsilon_1 \cdot \delta_{\mathcal{B}} \Phi) \cdot \boldsymbol{\eta} \cdot (\Upsilon_1 \cdot \delta_{\mathcal{B}} \Phi) = \delta_{\mathcal{B}} \Phi \cdot \left(\Upsilon_1^{\dagger} \cdot \boldsymbol{\eta} \cdot \Upsilon_1 \cdot \delta_{\mathcal{B}} \Phi\right) + \underline{D}_{\mu} \Theta_{\mathfrak{D},1}^{\mu}.$$
(2.85)

Putting this back in eq. (2.83) we get

$$\mathcal{C}_{\mathrm{D}} \cdot \delta_{\mathcal{B}} \Phi = -\underbrace{\left((1 + \Upsilon_{1} + \Upsilon_{2}) \cdot \delta_{\mathcal{B}} \Phi \right) \cdot \boldsymbol{\eta} \cdot \left((1 + \Upsilon_{1} + \Upsilon_{2}) \cdot \delta_{\mathcal{B}} \Phi \right)}_{\text{quadratic form}} + \underbrace{\left((2\Upsilon_{1} + \Upsilon_{2}) \cdot \delta_{\mathcal{B}} \Phi \right) \cdot \boldsymbol{\eta} \cdot (\Upsilon_{2} \cdot \delta_{\mathcal{B}} \Phi)}_{\text{residue}} + \underbrace{\underline{D}_{\mu} \left(\Theta_{\mathfrak{D},0}^{\mu} + \Theta_{\mathfrak{D},1}^{\mu} \right)}_{\text{total derivative}}, \quad (2.86)$$

where Υ_2 is another differential operator

$$\Upsilon_2 = -\frac{1}{2}\boldsymbol{\eta}^{-1} \cdot \Upsilon_1^{\dagger} \cdot \boldsymbol{\eta} \cdot \Upsilon_1.$$
(2.87)

Comparing eq. (2.86) to eq. (2.83), hopefully the reader can make out a repeating pattern. The quadratic form piece now has some additional higher derivative terms, whereas we have pushed the residue piece to 5th derivative order. We can repeat this procedure iteratively to push the residue piece to arbitrarily high derivative orders and obtain

$$\mathcal{C}_{\mathrm{D}} \cdot \delta_{\mathcal{B}} \Phi = -\left(\Upsilon \cdot \delta_{\mathcal{B}} \Phi\right) \cdot \boldsymbol{\eta} \cdot \left(\Upsilon \cdot \delta_{\mathcal{B}} \Phi\right) + \underline{\mathrm{D}}_{\mu} \Theta_{\mathfrak{D}}^{\mu}, \tag{2.88}$$

where

$$\Upsilon_{i+1}\big|_{i=1}^{\infty} = -\eta^{-1} \cdot \left(\sum_{k=1}^{d-1} \Upsilon_k^{\dagger} + \frac{1}{2} \Upsilon_i^{\dagger}\right) \left(\eta \cdot \Upsilon_i\right), \qquad \Upsilon = 1 + \sum_{n=1}^{\infty} \Upsilon_n, \qquad (2.89)$$

and $\Theta_{\mathfrak{D}}^{\scriptscriptstyle M} = \sum_{n=0}^{\infty} \Theta_{\mathfrak{D},n}^{\scriptscriptstyle M}$. Finally, comparing eq. (2.88) to eq. (2.57), we can see that Class D constitutive relations satisfy the adiabaticity equation with

$$(N^{\mu})_{\mathrm{D}} = \Theta_{\mathfrak{D}}^{\mu}, \qquad \Delta_{\mathrm{D}} = (\Upsilon \cdot \delta_{\mathfrak{B}} \Phi) \cdot \boldsymbol{\eta} \cdot (\Upsilon \cdot \delta_{\mathfrak{B}} \Phi).$$
 (2.90)

The second law requirement $\Delta \geq 0$ only gives a constraint on the first derivative nonhydrostatic constitutive relations by forcing all the eigenvalues of $\eta = \mathfrak{D}_{0(0)}$ to be nonnegative. Apart from these inequalities, we do not get any constraints from the second law in the non-hydrostatic sector. In particular, unlike the hydrostatic sector, the second law does not switch off any transport coefficients. This argument was first presented by Sayantani Bhattacharyya in [91, 95]. Our presentation above can be seen as a succinct summary of her seminal work.

2.3.6 Entropy transport

In the previous subsections we have taken up the task of classifying the most generic hydrodynamic constitutive relations \mathcal{C} , which admit a free energy current N^{μ} and a quadratic form Δ such that the adiabaticity equation (2.55) is satisfied. However, the N^{μ} and Δ which achieve this goal for a given \mathcal{C} do not have to be unique. In fact, we can shift N^{μ} and Δ with an arbitrary solution of

$$\underline{\mathbf{D}}_{\mu}(N^{\mu})_{\mathrm{S}} = \Delta_{\mathrm{S}},\tag{2.91}$$

and still satisfy the adiabaticity equation for the same set of constitutive relations C. The solutions of eq. (2.91) are called Class S "constitutive relations". They satisfy the non-anomalous adiabaticity equation (2.55) with vanishing constitutive relations $C_{\rm S} = 0$. The associated entropy transport however is non-trivial

$$(J_{\rm S}^{\mu})_{\rm S} = (N^{\mu})_{\rm S},\tag{2.92}$$

hence the name *entropy transport*. It should be noted however that Class S constitutive relations are not genuine hydrodynamic transport, they merely parametrise the multitude of entropy/free energy currents which meet the second law requirement for the same set of constitutive relations.

Firstly, there are some trivial solutions in Class S. Consider an arbitrary antisymmetric tensor $X^{\mu\nu}$. If we take $(N^{\mu})_{\rm S} = \underline{\mathcal{D}}_{\nu} X^{\mu\nu} + \frac{1}{2} T^{\mu}{}_{\nu\rho} X^{\nu\rho}$, the divergence $\underline{\mathcal{D}}_{\mu}(N^{\mu})_{\rm S}$ is identically zero and satisfies eq. (2.91) with $\Delta_{\rm S} = 0$. To obtain the remaining non-trivial solutions, let

us start with the most generic allowed quadratic form

$$\Delta_{\rm S} = \sum_{m,n \ge 0} ({\rm D}^m \delta_{\mathcal{B}} \Phi) \cdot \mathfrak{S}_{mn} \cdot ({\rm D}^n \delta_{\mathcal{B}} \Phi), \qquad (2.93)$$

where $\mathfrak{S}_{mn} \in \mathfrak{V} \times \mathfrak{V}$ satisfy $\mathfrak{S}_{mn}^{\mathrm{T}} = \mathfrak{S}_{nm}$. We know that an arbitrary \mathfrak{S}_{mn} cannot correspond to the Class S constitutive relations, as some part of it gives rise to the Class D transport. As we saw in section 2.3.5, the Class D transport is characterised by a quadratic form given in terms of a matrix \mathfrak{D}_n . So to pass as valid Class S transport, we must eliminate \mathfrak{D}_n worth of components from \mathfrak{S}_{mn} . To see this, let us perform successive differentiation by parts on eq. (2.93) and obtain

$$\Delta_{\rm S} = \delta_{\mathbb{B}} \Phi \cdot \mathfrak{S}_{00} \cdot \delta_{\mathbb{B}} \Phi + 2 \sum_{n \ge 1} \delta_{\mathbb{B}} \Phi \cdot \mathfrak{S}_{0n} \cdot ({\rm D}^n \delta_{\mathbb{B}} \Phi) + \sum_{m,n \ge 1} ({\rm D}^m \delta_{\mathbb{B}} \Phi) \cdot \mathfrak{S}_{mn} \cdot ({\rm D}^n \delta_{\mathbb{B}} \Phi)$$
$$= \delta_{\mathbb{B}} \Phi \cdot \mathfrak{S}_{00} \cdot \delta_{\mathbb{B}} \Phi + 2\delta_{\mathbb{B}} \Phi \cdot \mathfrak{S}_{01} \cdot ({\rm D}^1 \delta_{\mathbb{B}} \Phi) + \sum_{n \ge 2} \delta_{\mathbb{B}} \Phi \cdot \left(2\mathfrak{S}_{0n} + \mathfrak{S}_n\right) \cdot ({\rm D}^n \delta_{\mathbb{B}} \Phi)$$
$$+ \underline{\mathrm{D}}_{\mu} \Theta_{\mathfrak{S}}^{\mu}. \tag{2.94}$$

In the second step, we have isolated a series of differential operators \mathfrak{S}_n via the relation

$$\sum_{m,n\geq 1} (-)^m \mathcal{D}^m \Big(\mathfrak{S}_{mn} \cdot (\mathcal{D}^n \delta_{\mathfrak{B}} \Phi) \Big) \equiv \sum_{n\geq 2} \mathfrak{S}_n \cdot (\mathcal{D}^n \delta_{\mathfrak{B}} \Phi).$$
(2.95)

If we choose $\mathfrak{S}_{00} = \mathfrak{S}_{01} = 0$ and $\mathfrak{S}_{0n} = -\mathfrak{S}_n/2$ for $n \ge 2$, we can satisfy eq. (2.91) with $(N^{\mu})_{\mathrm{S}} = \Theta_{\mathfrak{S}}^{\mu}$. These are the \mathfrak{D}_m worth of conditions on \mathfrak{S}_{mn} we were alluding to. Finally,

$$(N^{\mu})_{\rm S} = \underline{\mathbf{D}}_{\nu} X^{\mu\nu} + \frac{1}{2} \mathbf{T}^{\mu}{}_{\nu\rho} X^{\nu\rho} + \Theta^{\mu}_{\mathfrak{S}}, \qquad (2.96)$$

parametrises the most generic Class S constitutive relations.

2.4 Summary and torsionless limit

This chapter has been quite technical, so let us summarise the important points. Hydrodynamics is characterised by its constitutive relations: the most generic expressions of $(T^{\mu}_{\ \alpha}, \Sigma^{\mu\alpha}_{\ \ \beta}, J^{\mu}, \mathcal{E}_i)$ in terms of the dynamical fields $(u^{\mu}, T, \mu^{\Sigma}, \mu)$ (or \mathcal{B}) and φ^i , and the background fields $(e^{\alpha}_{\ \mu}, C^{\alpha}_{\ \ \mu\beta}, A_{\mu})$, arranged in a derivative expansion. The equations of motion for the dynamical fields are given by eq. (2.46) and $\mathcal{E}_i \approx 0$. Up to a given derivative order, the constitutive relations contain all the possible tensor structures made out of its constituent fields and their derivatives, multiplied with arbitrary transport coefficients. The actual number of such transport coefficients, however, depends on the possible tensor structures we can write down, which in turn depends on the derivative order we are working at and on the number and transformation properties of the gapless modes φ^i . In general, the explicit functional form of these transport coefficients depends on the system in question, but we can bring the number down considerably by imposing the second law of thermodynamics. It requires that there must exist an entropy current J^{μ}_{S} , or equivalently a free energy current N^{μ} , such that the adiabaticity equation (2.53) is satisfied for some quadratic form $\Delta \geq 0$. This requirement constrains the hydrodynamic constitutive relations to be in one of the five classes

$$\mathcal{C} = \mathcal{C}_{\rm hs} + \mathcal{C}_{\rm nhs} = \left(\mathcal{C}_{\rm A} + \mathcal{C}_{\rm H_{\rm V}} + \mathcal{C}_{\rm H_{\rm S}}\right) + \left(\mathcal{C}_{\overline{\rm D}} + \mathcal{C}_{\rm D}\right). \tag{2.97}$$

In addition, there is also a Class S worth of redundancies in the entropy or free energy currents we can choose from, to satisfy the second law for a given set of constitutive relations.

For most applications of hydrodynamics, it is useful to work with the energy-momentum tensor and charge current only and no spin current. The spin current does not necessarily have to be zero. We could just choose not to probe this particular observable by turning off the background torsion. Without torsion, we can also fall back to the good old metric formulation with background fields $(g_{\mu\nu}, A_{\mu})$, as opposed to the vielbein formulation that we have been working with. This is the premise in which most of the work in relativistic field theories, especially hydrodynamics, is done. In this setting, we can use the second equation in the relativistic identities (2.19) to eliminate the antisymmetric part of the energy-momentum tensor from the first equation. Once that is done, we can show that remaining equations can be rewritten as

$$D_{\mu}T_{\rm B}^{\mu\nu} = F^{\nu\rho} \cdot J_{\rho} + D_{\mu}\Sigma_{\rm H}^{\perp\nu\mu} + \mathcal{O}_{\alpha}^{I}\mathcal{E}_{I},$$

$$D_{\mu}J^{\mu} = J_{\rm H}^{\perp} + \mathcal{O}^{I}\mathcal{E}_{I},$$
 (2.98)

where we have defined the so-called symmetric Belinfante energy-momentum tensor

$$T_{\rm B}^{\mu\nu} = T^{(\mu\nu)} + 2 \,\mathrm{D}_{\rho} \Sigma^{(\mu\nu)\rho}. \tag{2.99}$$

This is the quantity that couples to the spacetime metric and appears in the Einstein's equations in general relativity. Later in the text, when focusing on these theories without a spin current, we drop the subscript B in $T_{\rm B}^{\mu\nu}$ for clarity. The corresponding conservation equations on the other hand are given as

$$D_{\mu}T_{\rm B}^{\mu\nu} \approx F^{\nu\rho} \cdot J_{\rho} + D_{\mu}\Sigma_{\rm H}^{\perp\nu\mu},$$

$$D_{\mu}J^{\mu} \approx J_{\rm H}^{\perp},$$
 (2.100)

which are perhaps more familiar to the reader. In the flat space limit, they simply read

$$\partial_{\mu}T_{\rm B}^{\mu\nu} \approx 0, \qquad \partial_{\mu}J^{\mu} \approx 0.$$
 (2.101)

An interesting thing to note is that what used to be a Lorentz anomaly $\Sigma_{\rm H}^{\perp\alpha\beta}$ in the spin conservation, shows up as a gravitational anomaly $D_{\mu}\Sigma_{\rm H}^{\perp\nu\mu}$ in the Belinfante energy-momentum conservation. This is the incarnation in which this anomaly appears in most of the literature.

Moving on to hydrodynamics, the hydrodynamic fields are taken to be merely (u^{μ}, T, μ)

or equivalently $\mathcal{B} = (\beta^{\mu}, \Lambda_{\beta})$. Formally, we can think that the spin chemical potential μ^{Σ} has been "solved for" using the spin conservation equation in terms of the other fields in the theory, and has been substituted back into $T_{\rm B}^{\mu\nu}$, J^{μ} , and \mathcal{E}^{i} . The associated analogue to the adiabaticity equation (2.53) is given by

$$D_{\mu}N^{\mu} - N_{H}^{\perp} = \frac{1}{2}T_{B}^{\mu\nu}\delta_{\mathcal{B}}g_{\mu\nu} + J^{\mu}\cdot\delta_{\mathcal{B}}A_{\mu} + \mathcal{E}_{i}\delta_{\mathcal{B}}\varphi^{i} + \Delta, \qquad \Delta \ge 0.$$
(2.102)

If we define

$$\Phi = \begin{pmatrix} \frac{1}{2}g_{\mu\nu} \\ A_{\mu} \\ \varphi^{i} \end{pmatrix}, \qquad \mathcal{C} = \begin{pmatrix} T^{\mu\nu} \\ J^{\mu} \\ \mathcal{E}_{i} \end{pmatrix}, \qquad (2.103)$$

we can re-express the adiabaticity equation into

$$D_{\mu}N^{\mu} = N_{\rm H}^{\perp} + \mathcal{C} \cdot \delta_{\mathcal{B}} \Phi + \Delta, \qquad \Delta \ge 0, \tag{2.104}$$

which is essentially the same as our previous version in eq. (2.55). Therefore, all our discussion on hydrodynamic classification in section 2.3 still holds true.

This concludes our discussion of the principles of relativistic hydrodynamics. In the next chapter, we consider some explicit applications of these ideas.

Chapter 2. Relativistic hydrodynamics

Applications: relativistic hydrodynamics

In chapter 2 we studied the fundamentals of relativistic hydrodynamics in an abstract language. In this chapter, we take these key ideas and apply them to some explicit examples. We start with the well-known example of ordinary relativistic fluids and make our way to the more involved relativistic superfluids. Later, we also briefly talk about relativistic fluids with surfaces. For concreteness, we stick to d = 4 spacetime dimensions in this chapter.

3.1 Relativistic fluids

Ordinary relativistic fluids are the simplest hydrodynamic systems we can consider. They do not contain any additional gapless modes in their description, i.e. hydrodynamic fields \mathcal{B} are the only relevant dynamical degrees of freedom at low energies. They also do not have an independent spin current and are coupled to spacetime backgrounds without torsion. We already talked about them in section 1.1.2 using the conventional formalism of hydrodynamics. Here we rederive these results using the off-shell formalism of hydrodynamics proposed in chapter 2. The results we obtain here are standard in the literature; see e.g. [52] for a modern review.

3.1.1 Ideal fluids

Hydrodynamic systems whose constitutive relations are truncated to the zeroth order in derivatives are known as *ideal fluids*. With our choice of derivative counting it is easy to see that all except Class H_S constitutive relations start at the one-derivative order, so we can safely ignore them for now. At the ideal order, the hydrostatic scalar density \mathcal{N} characterising Class H_S is given by an arbitrary scalar $P(T, \mu)$ as a function of the temperature T and chemical potential μ of the fluid. Using the definitions of T and μ in terms of \mathcal{B} given in eq. (2.42), we can derive their $\delta_{\mathcal{B}}$ variations as

$$\delta_{\mathcal{B}}T = \frac{T}{2}u^{\mu}u^{\nu}\delta_{\mathcal{B}}g_{\mu\nu}, \qquad \delta_{\mathcal{B}}\frac{\mu}{T} = \frac{1}{T}u^{\mu}\delta_{\mathcal{B}}A_{\mu}.$$
(3.1)

We can use these expressions to compute the divergence of the ideal order free energy current $P\beta^{\mu}$ leading to

$$D_{\mu} (P\beta^{\mu}) = \frac{1}{\sqrt{-g}} \delta_{\mathcal{B}} \left(\sqrt{-g} P \right) = \frac{1}{2} P g^{\mu\nu} \delta_{\mathcal{B}} g_{\mu\nu} + \frac{\partial P}{\partial T} \delta_{\mathcal{B}} T + \frac{\partial P}{\partial \mu} \cdot \delta_{\mathcal{B}} \mu$$
$$= \left((E+P) u^{\mu} u^{\nu} + P g^{\mu\nu} \right) \frac{1}{2} \delta_{\mathcal{B}} g_{\mu\nu} + Q u^{\mu} \cdot \delta_{\mathcal{B}} A_{\mu}, \qquad (3.2)$$

where we have defined

$$S = \frac{\partial P}{\partial T}, \qquad Q = \frac{\partial P}{\partial \mu}, \qquad E = T \frac{\partial P}{\partial T} + \mu \cdot \frac{\partial P}{\partial \mu} - P.$$
 (3.3)

Comparing eq. (3.2) with eq. (2.74), we can easily read out the ideal fluid constitutive relations, free energy, and entropy currents

$$T^{\mu\nu} = (E+P) u^{\mu}u^{\nu} + P g^{\mu\nu} + \mathcal{O}(\partial), \qquad J^{\mu} = Q u^{\mu} + \mathcal{O}(\partial),$$
$$N^{\mu} = \frac{1}{T} P u^{\mu} + \mathcal{O}(\partial), \qquad J^{\mu}_{S} = S u^{\mu} + \mathcal{O}(\partial).$$
(3.4)

From here we can identify E as the energy density, P as the isotropic pressure, Q as the charge density, and S as the entropy density of the fluid. Due to eq. (3.3), these quantities satisfy the standard thermodynamic relations

Gibbs-Duhem equation:
$$dP = S dT + Q \cdot d\mu$$
,
Euler scaling relation: $E + P = ST + Q \cdot \mu$,
First law of thermodynamics: $dE = T dS + \mu \cdot dQ$. (3.5)

We see that the constitutive relations of an ideal fluid are completely characterised by its equation of state $P = P(T, \mu)$.

We can also work out the explicit first order equations of motion

$$\frac{1}{\sqrt{-g}}\delta_{\mathcal{B}}\left(\sqrt{-g}T\left(E+P\right)u_{\mu}\right) + TQ\cdot\delta_{\mathcal{B}}A_{\mu} + \mathcal{O}(\partial^{2}) = 0,$$
$$\frac{1}{\sqrt{-g}}\delta_{\mathcal{B}}\left(\sqrt{-g}TQ\right) + \mathcal{O}(\partial^{2}) = 0,$$
(3.6)

which provide dynamics for β^{μ} and Λ_{β} respectively. In this form, it is clear that the hydrostatic principle is satisfied, i.e. upon promoting \mathcal{B} to a background isometry \mathcal{K} , the equations of motion are trivially satisfied. It is also clear that we can use these equations to eliminate $u^{\mu}\delta g_{\mu\nu}$ and $u^{\mu}\delta A_{\mu}$ respectively from our non-hydrostatic constitutive relations. In a more readable form, these equations imply

$$D_{\mu} (Su^{\mu}) = \mathcal{O}(\partial^{2}), \qquad D_{\mu} (Qu^{\mu}) = \mathcal{O}(\partial^{2}),$$
$$(E+P)P^{\mu\nu} \left(\frac{1}{T}\partial_{\nu}T + u^{\rho}D_{\rho}u_{\nu}\right) + QP^{\mu\nu} \left(TD_{\nu}\frac{\mu}{T} + u^{\rho}F_{\rho\nu}\right) = \mathcal{O}(\partial^{2}), \qquad (3.7)$$

where $P^{\mu\nu} = g^{\mu\nu} + u^{\mu}u^{\nu}$ is the projector against u^{μ} . They can be identified as the ideal order entropy conservation, charge continuity equation, and the relativistic Navier-Stokes equation respectively. We see that the ideal fluid indeed satisfies the second law of thermodynamics.

3.1.2 One-derivative corrections

Having discussed the constitutive relations of an ideal fluid, we would now like to explore the most generic one-derivative corrections they can admit, allowed by the second law of

${\cal P}_{\sf T}$	$T^{\mu u}$	J^{μ}	N^{μ}
$C \boldsymbol{F} \wedge \boldsymbol{F} \wedge \boldsymbol{F}$	$2\mu^2 C u^{(\mu} \left(3B^{\nu)} + 2\mu\omega^{\nu)} \right)$	$3\mu C \left(2B^{\mu}+\mu\omega^{\mu}\right)$	$\frac{\mu^2}{T}C\left(3B^\mu + \mu\omega^\mu\right)$
$C_1 \mathbf{F}_{T} \wedge \mathbf{F}_{T} \wedge \mathbf{F}_{T}$	$4T^3C_1u^{(\mu}\omega^{\nu)}$		$4T^2C_1\omega^\mu$
$C_2 F_{T} \wedge F_{T} \wedge F$	$2T^2 C_2 u^{(\mu} \left(B^{\nu)} + 2\mu \omega^{\nu)} \right)$	$T^2 C_2 \omega^{\mu}$	$TC_2 \left(B^\mu + 3\mu\omega^\mu \right)$
$C_0 oldsymbol{F}_{T} \wedge oldsymbol{F} \wedge oldsymbol{F}$	$4\mu T C_0 u^{(\mu} \left(B^{\nu)} + \mu \omega^{\nu)} \right)$	$2TC_0 \left(B^\mu + \mu \omega^\mu \right)$	$ \mu C_0 \left(2B^{\mu} + \mu \omega^{\mu} \right) + \frac{1}{2} C_0 \epsilon^{\mu\nu\rho\sigma} A_{\nu} F_{\rho\sigma} - \frac{1}{3} C_0 \epsilon^{\mu\nu\rho\sigma} A_{\nu} A_{\rho} A_{\sigma} $

Table 3.1: One-derivative Class A and Class H_V constitutive relations for a (3+1)-dimensional relativistic fluid. Lie algebra traces are understood in columns \mathcal{P}_{T} , $T^{\mu\nu}$, and N^{μ} . Note that the C_0 term is in the thermal anomaly polynomial is linear in \mathbf{F}_{T} , hence the associated free energy current is not gauge-invariant.

\mathfrak{D}_0	$T^{\mu\nu}$	J^{μ}
$-T\zeta \begin{pmatrix} P^{\mu\nu}P^{\rho\sigma} & \vdots \\ \vdots & \vdots \end{pmatrix}$	$-\zeta P^{\mu\nu}\Theta$	
$-T\eta \begin{pmatrix} P^{\mu\langle\rho}P^{\sigma\rangle\nu} & \cdot \\ \cdot & \cdot \end{pmatrix}$	$-\eta \sigma^{\mu \nu}$	
$-T\sigma \begin{pmatrix} \cdot & \cdot \\ \cdot & P^{\mu\rho} \end{pmatrix}$		$-\sigma P^{\mu\nu} \left(T \mathcal{D}_{\nu} \frac{\mu}{T} - E_{\nu}\right)$

Table 3.2: One-derivative Class D constitutive relations for a (3+1)-dimensional relativistic fluid.

thermodynamics. First, some standard definitions:

$$\Theta = \mathcal{D}_{\mu}u^{\mu}, \qquad \sigma^{\mu\nu} = P^{\mu\langle\rho}P^{\sigma\rangle\nu}\mathcal{D}_{\rho}u_{\sigma} = P^{\mu\rho}P^{\nu\sigma}\left(\mathcal{D}_{(\mu}u_{\nu)} - \frac{1}{3}P_{\mu\nu}\Theta\right),$$
$$\mathfrak{a}^{\mu} = u^{\nu}\mathcal{D}_{\nu}u^{\mu}, \qquad \omega^{\mu} = \epsilon^{\mu\nu\rho\sigma}u_{\nu}\partial_{\rho}u_{\sigma}, \qquad E^{\mu} = F^{\mu\nu}u_{\nu}, \qquad B^{\mu} = \frac{1}{2}\epsilon^{\mu\nu\rho\sigma}u_{\nu}F_{\rho\sigma}. \tag{3.8}$$

In words, they are the expansion, shear, acceleration, and vorticity of the fluid, along with the electric and magnetic fields defined in the rest frame of the fluid. Angular brackets denote a traceless symmetric combination.

Let us start with the hydrostatic sector. We can check that there are no one-derivative scalars to make up \mathcal{N} , so Class H_S is empty. There are however non-trivial Class A and Class H_V constitutive relations at one-derivative order. They are collectively characterised by a 6-rank thermal anomaly polynomial

$$\mathcal{P}_{\mathsf{T}} = C \operatorname{tr}[\mathbf{F} \wedge \mathbf{F} \wedge \mathbf{F}] + C_1 \mathbf{F}_{\mathsf{T}} \wedge \mathbf{F}_{\mathsf{T}} \wedge \mathbf{F}_{\mathsf{T}} + C_2 \mathbf{F}_{\mathsf{T}} \wedge \operatorname{tr} \mathbf{F} + C_0 \mathbf{F}_{\mathsf{T}} \wedge \operatorname{tr}[\mathbf{F} \wedge \mathbf{F}].$$
(3.9)

C is the standard U(1)³ anomaly coefficient, while C_1 and C_2 are arbitrary constants. Following our discussion in sections 2.3.2 and 2.3.3, we can easily read out the respective constitutive relations. The results have been summarised in table 3.1.

Next, let us move on to the non-hydrostatic constitutive relations. Since the operator " $\delta_{\mathcal{B}}$ " already contains a derivative in its definition, at one-derivative order the only relevant non-hydrostatic constitutive relations are parametrised by \mathfrak{C}_0 . If we split \mathfrak{C}_0 into \mathfrak{D}_0 and $\overline{\mathfrak{D}}_0$ according to eq. (2.80), we can check that these are its symmetric and anti-symmetric parts respectively. Given the tensor structures at hand, we can check that there are no
available terms that we can use to construct an antisymmetric matrix $\overline{\mathfrak{D}}_0$. Consequently, Class \overline{D} is empty. There are, however, three possible terms that we can write down in \mathfrak{D}_0 , leading to the non-trivial Class D constitutive relations. The results have been summarised in table 3.2. The associated quadratic form Δ is given by

$$T\Delta = \zeta \Theta^2 + \eta \,\sigma^{\mu\nu}\sigma_{\mu\nu} + \sigma P^{\mu\nu} \left(T \mathcal{D}_{\mu}\frac{\mu}{T} - E_{\mu}\right) \left(T \mathcal{D}_{\nu}\frac{\mu}{T} - E_{\nu}\right). \tag{3.10}$$

Since each tensor structure on the right hand side is manifestly positive definite, the second law requirement merely states that the dissipative transport coefficients ζ , η , and σ are all non-negative. They are identified as the bulk viscosity, shear viscosity, and electric conductivity of the fluid respectively.

To summarise, the constitutive relations of a relativistic fluid up to the first order in derivatives are given as

$$T^{\mu\nu} = (E+P) u^{\mu} u^{\nu} + P g^{\mu\nu} - \zeta P^{\mu\nu} \Theta - \eta \sigma^{\mu\nu} + 2 \left(3\mu^2 C + 2\mu T C_0 + C_2 T^2 \right) u^{(\mu} B^{\nu)} + 4 \left(\mu^3 C + \mu^2 T C_0 + T^3 C_1 + C_2 \mu T^2 \right) u^{(\mu} \omega^{\nu)} + \mathcal{O}(\partial^2), J^{\mu} = Q u^{\mu} - \sigma P^{\mu\nu} \left(T D_{\nu} \frac{\mu}{T} - E_{\nu} \right) + (6\mu C + 2T C_0) B^{\mu} + \left(3\mu^2 C + 2\mu T C_0 + C_2 T^2 \right) \omega^{\mu} + \mathcal{O}(\partial^2).$$
(3.11)

They satisfy the second law of thermodynamics with the free energy and entropy currents

$$N^{\mu} = \frac{1}{T} P u^{\mu} + \frac{1}{T} \left(3\mu^{2}C + 2\mu T C_{0} + T^{2}C_{2} \right) B^{\mu} + \frac{1}{T} \left(\mu^{3}C + \mu^{2}T C_{0} + 4T^{3}C_{1} + 3\mu T^{2}C_{2} \right) \omega^{\mu} + C_{0} \epsilon^{\mu\nu\rho\sigma} \left(\frac{1}{2} A_{\nu}F_{\rho\sigma} - \frac{1}{3}A_{\nu}A_{\rho}A_{\sigma} \right) + \mathcal{O}(\partial^{2}),$$

$$J^{\mu}_{S} = S u^{\mu} + \frac{\mu}{T} \sigma P^{\mu\nu} \left(T D_{\nu} \frac{\mu}{T} - E_{\nu} \right) + 2 \left(\mu C_{0} + T C_{2} \right) B^{\mu} + \left(\mu^{2}C_{0} + 3T^{2}C_{1} + 2\mu T C_{2} \right) \omega^{\mu} + C_{0} \epsilon^{\mu\nu\rho\sigma} \left(\frac{1}{2} A_{\nu}F_{\rho\sigma} - \frac{1}{3}A_{\nu}A_{\rho}A_{\sigma} \right) + \mathcal{O}(\partial^{2}),$$
(3.12)

provided that the zeroth order transport coefficients are derived from a single function $P(T,\mu)$ via the thermodynamic relations (3.5) and the first order transport coefficients η , ζ , and σ are non-negative.

It should be appreciated that, a priori, the constitutive relations could admit many more tensor structures: 3 at the ideal order and 7 at the one-derivative order after modding out field redefinition and on-shell equivalence, each coming with their own transport coefficients. However, the second law of thermodynamics fixed these 10 coefficients in terms of just 1 arbitrary function, 3 arbitrary non-negative functions, and 4 constants. For example, there could be independent terms in the charge current: $\kappa P^{\mu\nu}\partial_{\nu}T$ and σE^{μ} , interpreted as thermal and electric conductivities respectively. Instead, the second law fixes the thermal conductivity to be $\kappa = \mu \sigma/T$.

Before moving on to the next example, we should note that we have derived the fluid constitutive relations (3.11) in a very specific hydrodynamic frame, fixed by requiring the off-shell second law to hold. We can always transform to our preferred frame of choice by performing a field redefinition $u^{\mu} \to u^{\mu} + \delta u^{\mu}$, $T \to T + \delta T$, $\mu \to \mu + \delta \mu$ where $u_{\mu} \delta u^{\mu} = 0$. For example, one quite ubiquitous frame in relativistic hydrodynamics is the so called Landau frame defined as $T^{\mu\nu}_{\text{Landau}} u_{\nu} = -E u^{\mu}$ and $J^{\mu}_{\text{Landau}} u_{\mu} = Q u^{\mu}$. This corresponds to the field redefinition $u^{\mu} \to u^{\mu} - \frac{1}{E+P} P^{\mu}_{\ \rho} T^{\rho\sigma} u_{\sigma}$ in eq. (3.11), leading to

$$T_{\text{Landau}}^{\mu\nu} = (E+P) u^{\mu} u^{\nu} + P g^{\mu\nu} - \zeta P^{\mu\nu} \Theta - \eta \sigma^{\mu\nu} + \mathcal{O}(\partial^2),$$

$$J_{\text{Landau}}^{\mu} = Q u^{\mu} - \sigma P^{\mu\nu} \left(T \mathcal{D}_{\nu} \frac{\mu}{T} - E_{\nu} \right) + \xi_B B^{\mu} + \xi_{\omega} \omega^{\mu} + \mathcal{O}(\partial^2), \qquad (3.13)$$

where we have defined the parity-odd conductivities

$$\xi_B = (6\mu C + 2TC_0) - \frac{Q}{E+P} \left(3\mu^2 C + 2\mu TC_0 + C_2 T^2 \right)$$

$$\xi_\omega = \left(3\mu^2 C + 2\mu TC_0 + C_2 T^2 \right) - \frac{2Q}{E+P} \left(\mu^3 C + \mu^2 TC_0 + T^3 C_1 + C_2 \mu T^2 \right). \quad (3.14)$$

This is perhaps the better known form of the one-derivative order relativistic fluid constitutive relations, which were first derived in full generality in [50].

3.2 Relativistic superfluids

For our next example, we study relativistic superfluids. They describe the hydrodynamic regime of a quantum field theory with a spontaneously broken internal symmetry. The corresponding Goldstone modes act as gapless degrees of freedom in the hydrodynamic description, besides u^{μ} , T, and μ , providing us with an example of non-trivial φ^i fields. In general, we could break an arbitrary Lie group G of internal symmetries to any Lie subgroup H, but for now we content ourselves with breaking G = U(1) to $H = \{1\}$. This leads to Abelian superfluids, which we briefly covered in section 1.2. We return to the non-Abelian superfluids in section 3.2.4. Constitutive relations for an Abelian superfluid have already been obtained in the literature [51, 53, 96] using the conventional "on-shell" formalism. Here we employ the off-shell formalism of hydrodynamics discussed in chapter 2 to rederive these results. One important distinction between the two formalisms is that in the conventional formalism, one typically imposes an equation of motion for the Goldstone mode, known as the Josephson equation, by hand. However, in the off-shell formalism, this equation automatically pops out of the second law. The results presented here appeared in our works [1, 4].

3.2.1 Goldstone modes and Josephson equation

Consider a microscopic theory which is charged under a global U(1) symmetry. We are seeking to describe low energy fluctuations in this theory in a phase where the U(1) symmetry is spontaneously broken, perhaps due to condensation of a charged scalar operator. The U(1) phase φ of this operator becomes an additional field in the fluid description, upon which the respective constitutive relations can depend. Under an infinitesimal symmetry transformation parametrised by $\chi = (\chi^{\mu}, \Lambda_{\chi})$, the field φ transforms as $\delta_{\chi} \varphi = \chi^{\mu} \partial_{\mu} \varphi - \Lambda_{\chi}$. We can write down a covariant derivative of φ as

$$\xi_{\mu} = \partial_{\mu}\varphi + A_{\mu}, \qquad (3.15)$$

commonly known as the superfluid velocity. It satisfies $2\partial_{[\mu}\xi_{\nu]} = F_{\mu\nu}$. In superfluid dynamics, it is customary to choose the derivative order of φ to be $\mathcal{O}(\partial^{-1})$, which would imply that ξ_{μ} is $\mathcal{O}(\partial^{0})$. The phase field φ provides us with our first example of an additional gapless mode " φ^{i} " discussed in section 2.2.3. For the corresponding equation of motion " $\mathcal{E}_{i} \approx 0$ ", we take

$$K \approx 0. \tag{3.16}$$

Just like $T^{\mu\nu}$ and J^{μ} , the placeholder K is also provided with constitutive relations. Its form is fixed by requiring the second law of thermodynamics to hold.

Following eq. (2.53), we can directly write down the adiabaticity equation for superfluids, leading to

$$D_{\mu}N^{\mu} - N_{\rm H}^{\perp} = \frac{1}{2}T^{\mu\nu}\delta_{\mathcal{B}}g_{\mu\nu} + J^{\mu}\delta_{\mathcal{B}}A_{\mu} + K\delta_{\mathcal{B}}\varphi + \Delta, \qquad \Delta \ge 0, \qquad (3.17)$$

where

$$\delta_{\mathcal{B}}\varphi = \beta^{\mu}\partial_{\mu}\varphi - \Lambda_{\beta} = \frac{1}{T}\left(u^{\mu}\xi_{\mu} - \mu\right).$$
(3.18)

Let us start by considering the adiabaticity equation at the zeroth order in derivatives. Ignoring all the derivatives, eq. (3.17) simply becomes: $-K\delta_{\mathbb{B}}\varphi + \mathcal{O}(\partial) = \Delta \geq 0$. The placeholder K, at this order, is just a function of all the ideal order scalars T, μ , $\mu_s = -\frac{1}{2}\xi^{\mu}\xi_{\mu}$, and $\delta_{\mathbb{B}}\varphi$, such that $-K\delta_{\mathbb{B}}\varphi$ is a positive semi-definite quadratic form. It follows that K must take the form

$$K = -\alpha \delta_{\mathcal{B}} \varphi + \mathcal{O}(\partial), \qquad \Delta = \alpha (\delta_{\mathcal{B}} \varphi)^2 + \mathcal{O}(\partial), \qquad (3.19)$$

for some transport coefficient $\alpha \geq 0$. Using the φ equation of motion $K \approx 0$, it follows that

$$K = -\alpha \delta_{\mathcal{B}} \varphi = -\frac{\alpha}{T} \left(u^{\mu} \xi_{\mu} - \mu \right) \approx \mathcal{O}(\partial) \implies u^{\mu} \xi_{\mu} \approx \mu + \mathcal{O}(\partial).$$
(3.20)

This is known as the *Josephson equation*. Due to this relation, $u^{\mu}\xi_{\mu}$ is not independent on-shell and hence is not used as an independent scalar while writing down the superfluid constitutive relations.

3.2.2 Ideal superfluids

Similar to the ordinary ideal fluids, Class H_S is the only non-empty class for ideal superfluids as well. However, the hydrostatic scalar density \mathcal{N} characterising Class H_S is given by a three-variable function $P(T, \mu, \mu_s)$ this time, where T is the temperature, μ is the chemical potential, while $\mu_s = -\frac{1}{2}\xi^{\mu}\xi_{\mu}$ is the superfluid potential. We have omitted the only other possible ideal order scalar $u^{\mu}\xi_{\mu}$ in the functional dependence, because it is not independent on-shell. The $\delta_{\mathcal{B}}$ variations of T and μ are given in eq. (3.1), while for μ_s we find

$$\delta_{\mathcal{B}}\mu_{s} = \frac{1}{2}\xi^{\mu}\xi^{\nu}\delta_{\mathcal{B}}g_{\mu\nu} - \xi^{\mu}\delta_{\mathcal{B}}A_{\mu} - \xi^{\mu}\mathcal{D}_{\mu}\delta_{\mathcal{B}}\varphi.$$
(3.21)

We can use it to compute the divergence

$$D_{\mu} \left(\beta^{\mu} P\right) = \frac{1}{\sqrt{-g}} \delta_{\mathcal{B}} \left(\sqrt{-g} P\right) = \frac{1}{2} P g^{\mu\nu} \delta_{\mathcal{B}} g_{\mu\nu} + \frac{\partial P}{\partial T} \delta_{\mathcal{B}} T + \frac{\partial P}{\partial \mu} \delta_{\mathcal{B}} \mu + \frac{\partial P}{\partial \mu_s} \delta_{\mathcal{B}} \mu_s,$$

$$= \left((E+P) u^{\mu} u^{\nu} + P g^{\mu\nu} + R_s \xi^{\mu} \xi^{\nu} \right) \frac{1}{2} \delta_{\mathcal{B}} g_{\mu\nu} + (Q u^{\mu} - R_s \xi^{\mu}) \delta_{\mathcal{B}} A_{\mu} + D_{\mu} \left(R_s \xi^{\mu} \right) \delta_{\mathcal{B}} \varphi - D_{\mu} \left(R_s \xi^{\mu} \delta_{\mathcal{B}} \varphi \right), \qquad (3.22)$$

where we have defined

$$S = \frac{\partial P}{\partial T}, \qquad Q = \frac{\partial P}{\partial \mu}, \qquad R_s = \frac{\partial P}{\partial \mu_s}, \qquad E = T\frac{\partial P}{\partial T} + \mu \frac{\partial P}{\partial \mu} - P.$$
 (3.23)

Comparing eq. (3.22) with eq. (2.74), we can read out the ideal superfluid constitutive relations, free energy, and entropy currents

$$T^{\mu\nu} = (E+P)u^{\mu}u^{\nu} + Pg^{\mu\nu} + R_s\xi^{\mu}\xi^{\nu} + \mathcal{O}(\partial),$$

$$J^{\mu} = Qu^{\mu} - R_s\xi^{\mu} + \mathcal{O}(\partial),$$

$$K = -\alpha\delta_{\mathcal{B}}\varphi + D_{\mu}(R_s\xi^{\mu}) + \mathcal{O}(\partial),$$

$$N^{\mu} = \frac{1}{T}Pu^{\mu} + \delta_{\mathcal{B}}\varphi R_s\xi^{\mu} + \mathcal{O}(\partial),$$

$$J^{\mu}_{S} = N^{\mu} - \frac{1}{T}(T^{\mu\nu}u_{\nu} + \mu J^{\mu}) = Su^{\mu} + \mathcal{O}(\partial).$$
(3.24)

Like before, P can be identified as the isotropic pressure, while E, Q, and S can be identified as the energy, charge, and entropy densities of the superfluid respectively. The quantity R_s on the other hand is known as the superfluid density. They follow a set of thermodynamic relations

Gibbs-Duhem equation:
$$dP = SdT + Qd\mu + R_s d\mu_s$$
,
Euler scaling relation: $E + P = ST + Q\mu$,
First law of thermodynamics: $dE = TdS + \mu dQ - R_s d\mu_s$, (3.25)

which are directly implied by eq. (3.23). We see that an ideal superfluid is also completely characterised by its equation of state $P = P(T, \mu, \mu_s)$. Note that we have included some one-derivative terms in K and N^{μ} which can be ignored when working at ideal order, but are required for the internal consistency with the second law. Let us work out the first order equations of motion. By a direct computation we get

$$\frac{1}{\sqrt{-g}}\delta_{\mathcal{B}}\left(\sqrt{-g}T(E+P)u_{\mu}\right) + QT\delta_{\mathcal{B}}A_{\mu} + \xi_{\nu}D_{\mu}(R_{s}\xi^{\mu}) + \mathcal{O}(\partial^{2}) = 0,$$
$$\frac{1}{\sqrt{-g}}\delta_{\mathcal{B}}\left(\sqrt{-g}QT\right) - D_{\mu}(R_{s}\xi^{\mu}) + \mathcal{O}(\partial^{2}) = 0,$$
$$\alpha\delta_{\mathcal{B}}\varphi - D_{\mu}(R_{s}\xi^{\mu}) + \mathcal{O}(\partial) = 0.$$
(3.26)

They provide dynamics to β^{μ} , Λ_{β} , and φ respectively. In a hydrostatic configuration, all these equations boil down to a scalar equation $D_{\mu}(R_s\xi^{\mu}) = 0$, which determines the equilibrium profile of the phase field φ . Substituting the respective expressions for $\delta_{\mathcal{B}}$ variations, these equations imply

$$D_{\mu} (Su^{\mu}) = \mathcal{O}(\partial^{2}),$$

$$D_{\mu} (Qu^{\mu}) = D_{\mu} (R_{s}\xi^{\mu}) + \mathcal{O}(\partial^{2}),$$

$$(E+P)\zeta^{\mu} \left(\frac{1}{T}\partial_{\mu}T + u^{\rho}D_{\rho}u_{\mu}\right) + Q\zeta^{\mu} \left(T\partial_{\mu}\frac{\mu}{T} + u^{\rho}F_{\rho\mu}\right) + \zeta^{2}D_{\nu}(R_{s}\xi^{\nu}) = \mathcal{O}(\partial^{2}),$$

$$(E+P)P_{\zeta}^{\mu\nu} \left(\frac{1}{T}\partial_{\nu}T + u^{\rho}D_{\rho}u_{\nu}\right) + QP_{\zeta}^{\mu\nu} \left(T\partial_{\nu}\frac{\mu}{T} + u^{\rho}F_{\rho\nu}\right) = \mathcal{O}(\partial^{2}),$$

$$u^{\mu}\xi_{\mu} = \mu + \frac{T}{\alpha}D_{\mu}(R_{s}\xi^{\mu}) + \mathcal{O}(\partial),$$
(3.27)

where we have defined $P^{\mu\nu} = g^{\mu\nu} + u^{\mu}u^{\nu}$, $\zeta^{\mu} = P^{\mu\nu}\xi_{\nu}$, and $P^{\mu\nu}_{\zeta} = g^{\mu\nu} + u^{\mu}u^{\nu} - \frac{1}{\zeta^2}\zeta^{\mu}\zeta^{\nu}$. Note that $\zeta^2 = \mu^2 - 2\mu_s + \mathcal{O}(\partial)$. The first equation is the ideal order entropy conservation. The second equation serves as the charge continuity equation, balancing the flow of charges along u^{μ} and ξ^{μ} . The third and forth equations are components of the relativistic Navier-Stokes equation along and transverse to ζ^{μ} . Finally, the fifth equation is an improvement of the Josephson equation. Note, however, that this equation can admit further one derivative corrections due to the first order constitutive relations discussed in the next subsection; the correction mentioned here is only how the ideal superfluid transport affects the Josephson equation.

3.2.3 One-derivative corrections

We now improve our superfluid constitutive relations with one-derivative corrections. In table 3.3 we have listed various one-derivative tensor structures that we require in our discussion below.

Let us start with the hydrostatic sector. Class H_S , unlike ordinary fluids, is non-empty for superfluids. Ignoring the total derivative terms, the corresponding hydrostatic scalar density \mathcal{N} at the first order in derivatives is now given as

$$\mathcal{N} = P + f_1 S_1 + f_2 S_2 + g_1 \tilde{S}_{e,1} + g_2 \tilde{S}_{e,2}. \tag{3.28}$$

We have chosen not to include the first order hydrostatic scalars: $D_{\mu}\xi^{\mu}$ and $\xi^{\mu}\partial_{\mu}\mu_s$, as the former is a total derivative and the latter can be exchanged for $D_{\mu}(R_s\xi^{\mu})$, which is not

Non-hydrostatic — on-shell independent					
S_1	$\frac{T}{2}P^{\mu\nu}_{\zeta}\delta_{B}g_{\mu\nu}$ $P^{\mu\nu}D_{\mu}u_{\nu}$				
S_2	$\frac{\bar{T}}{2}\zeta^{\mu}\zeta^{\nu}\delta_{\mathcal{B}}g_{\mu\nu}\qquad\qquad \zeta^{\mu}\zeta^{\nu}\mathrm{D}_{\mu}u_{\nu}$				
S_3	$T\zeta^{\mu}\delta_{\mathcal{B}}A_{\mu} \qquad \qquad \zeta^{\mu}\left(T\partial_{\mu}\frac{\mu}{T} - E_{\mu}\right)$				
S_4	$T\delta_{\mathcal{B}}\varphi \qquad \qquad u^{\mu}\xi_{\mu}-\mu=\frac{T}{\alpha}D_{\mu}(R_{s}\xi^{\mu})+\mathcal{O}(\partial)$				
V_1^{μ}	$TP^{\mu\nu}_{\zeta}\zeta^{\rho}\delta_{\mathfrak{B}}g_{\nu\rho} \qquad \qquad 2P^{\mu\nu}_{\zeta}\zeta^{\rho}\mathcal{D}_{(\nu}u_{\rho)}$				
V_2^{μ}	$T P_{\zeta}^{\mu\nu} \delta_{\mathcal{B}} A_{\mu} \qquad P_{\zeta}^{\mu\nu} \left(T \partial_{\nu} \frac{\mu}{T} + E_{\nu} \right)$				
$\sigma^{\mu u}_{\zeta}$	$\frac{T}{2}P_{\zeta}^{\rho\langle\mu}P_{\zeta}^{\nu\rangle\sigma}\delta_{\mathcal{B}}g_{\rho\sigma} \qquad P_{\zeta}^{\mu\rho}P_{\zeta}^{\nu\sigma}\left(\mathcal{D}_{(\rho}u_{\sigma)}-\frac{1}{2}P_{\rho\sigma}^{\zeta}S_{1}\right)$				
\tilde{V}^{μ}_1	$\epsilon^{\mu u ho\sigma}u_ u\zeta_ ho V_{1,\sigma}$				
\tilde{V}_2^{μ}	$\epsilon^{\mu u ho\sigma}u_ u\zeta_ ho V_{2,\sigma}$				
$ ilde{\sigma}^{\mu u}_{\zeta}$	$P_{\zeta}^{\lambda(\mu}\epsilon^{\nu)\rho\sigma\tau}u_{\rho}\zeta_{\sigma}\sigma_{\tau\lambda}^{\zeta}$				
Non-hydrostatic — on-shell dependent					
S_5	$\frac{T}{2}u^{\mu}u^{\nu}\delta_{\mathbb{B}}g_{\mu\nu}$ $\frac{1}{T}u^{\mu}\partial_{\mu}T$				
S_6	$Tu^{\mu}\delta_{\mathcal{B}}A_{\mu}$ $Tu^{\mu}\partial_{\mu}\frac{\mu}{T}$				
S_7	$T\zeta^{\mu}u^{\nu}\delta_{\mathcal{B}}g_{\mu\nu}\qquad \qquad \zeta^{\nu}\left(\frac{1}{T}\partial_{\nu}T+\mathfrak{a}_{\nu}\right)$				
V_3^{μ}	$TP_{\zeta}^{\mu\nu}u^{\rho}\delta_{\mathcal{B}}g_{\nu\rho} \qquad P_{\zeta}^{\mu\nu}\left(\frac{1}{T}\partial_{\nu}T + \mathfrak{a}_{\nu}\right)$				
\tilde{V}_3^{μ}	$\epsilon^{\mu u ho\sigma}u_{ u}\zeta_{ ho}V_{3,\sigma}$				
	Hydrostatic				
$S_{e,1}$	$\frac{1}{T}\zeta^{\mu}\partial_{\mu}T$				
$S_{e,2}$	$T\zeta^{\mu}\partial_{\mu}rac{\mu}{T}$				
$V_{e,1}^{\mu}$	$rac{1}{T}P_{c}^{\mu u}\partial_{ u}T$				
$V_{e,2}^{\mu}$	$TP_{\zeta}^{\mu u}\partial_{ u} u$				
$\tilde{S}_{e,1}$	$T\epsilon^{\mu u ho\sigma}\zeta_{\mu}u_{ u}\partial_{ ho}u_{\sigma}$				
$\tilde{S}_{e,2}$	$\frac{1}{2}T\epsilon^{\mu\nu\rho\sigma}\zeta_{\mu}u_{\nu}F_{\rho\sigma}$				
$\tilde{V}_{e,1}^{\mu}$	$TP_{\zeta}{}^{\mu}{}_{\tau}\epsilon^{\tau u ho\sigma}u_{ u}\partial_{ ho}u_{\sigma}$				
$\tilde{V}_{e,2}^{\mu}$	$\frac{1}{2}TP_{\zeta}{}^{\mu}{}_{\tau}\epsilon^{ au u ho\sigma}u_{ u}F_{ ho\sigma}$				
$ ilde{V}^{\dot{\mu}}_{e.3}$	$\tilde{T}P_{\zeta}{}^{\dot{\mu}}{}_{ au}\epsilon^{ au u ho\sigma}\xi_{ u}\partial_{ ho}u_{\sigma}$				
$\tilde{V}_{e,4}^{\mu}$	$\frac{1}{2}TP_{\zeta}{}^{\mu}{}_{\tau}\epsilon^{\tau\nu\rho\sigma}\xi_{\nu}F_{\rho\sigma}$				
÷	÷				

Table 3.3: Independent first order data for (3 + 1)-dimensional relativistic superfluids. We have not enlisted, neither would we need, all the independent data surviving at equilibrium.

<i>N</i>	$T^{\mu\nu}$	J^{μ}
$f_1 S_{e,1}$	$ \left(\alpha_{E,1} u^{\mu} u^{\nu} + \alpha_{R_s,1} (\zeta^{\mu} \zeta^{\nu} - 2\mu u^{(\mu} \zeta^{\nu)}) + f_1 P_{\zeta}^{\mu\nu} \right) S_{e,1} - 2f_1 \xi^{(\mu} V_{e,1}^{\nu)} + 2f_1 u^{(\mu} \zeta^{\nu)} S_5 - u^{\mu} u^{\nu} \frac{1}{T} \mathcal{D}_{\sigma} \left(T f_1 \zeta^{\sigma} \right) $	$\begin{array}{c} (\alpha_{Q,1}u^{\mu} - \alpha_{R_s,1}\zeta^{\mu}) S_{e,1} \\ + f_1 V^{\mu}_{e,1} \end{array}$
$f_2 S_{e,2}$	$ \left(\alpha_{E,2} u^{\mu} u^{\nu} + \alpha_{R_s,2} (\zeta^{\mu} \zeta^{\nu} - 2\mu u^{(\mu} \zeta^{\nu)}) + f_2 P_{\zeta}^{\mu\nu} \right) S_{e,2} - 2f_2 \xi^{(\mu} V_{e,2}^{\nu)} + 2u^{(\mu} \zeta^{\nu)} f_2 S_6 $	$\begin{array}{l} (\alpha_{Q,2}u^{\mu} - \alpha_{R_{s},2}\zeta^{\mu}) S_{e,2} \\ + f_2 V^{\mu}_{e,2} \\ - u^{\mu} \frac{1}{T} \mathcal{D}_{\nu} (T f_2 \zeta^{\nu}) \end{array}$
$g_1\tilde{S}_{e,1}$	$ \begin{split} \left(\tilde{\alpha}_{E,1} u^{\mu} u^{\nu} + \tilde{\alpha}_{R_s,1} (\zeta^{\mu} \zeta^{\nu} - 2\mu u^{(\mu} \zeta^{\nu)}) \right) \tilde{S}_{e,1} \\ + \left(g_1 u^{\mu} u^{\nu} + \frac{g_1}{\zeta^2} \zeta^{\mu} \zeta^{\nu} \right) \tilde{S}_{e,1} \\ - 2g_1 u^{(\mu} \tilde{V}_{e,3}^{\nu)} - 2u^{(\mu} \epsilon^{\nu)\rho\sigma\tau} \mathcal{D}_{\sigma} \left(Tg_1 u_{\tau} \zeta_{\rho} \right) \end{split} $	$\begin{array}{l} \left(\tilde{\alpha}_{Q,1}u^{\mu}-\tilde{\alpha}_{R_{s},1}\zeta^{\mu}\right)\tilde{S}_{e,1} \\ +g_{1}\tilde{V}_{e,1}^{\mu} \end{array}$
$g_2 \tilde{S}_{e,2}$	$ \begin{pmatrix} \tilde{\alpha}_{E,2} u^{\mu} u^{\nu} + \tilde{\alpha}_{R_{s},2} (\zeta^{\mu} \zeta^{\nu} - 2\mu u^{(\mu} \zeta^{\nu)}) \end{pmatrix} \tilde{S}_{e,2} \\ + \frac{g_{1}}{\zeta^{2}} \zeta^{\mu} \zeta^{\nu} \tilde{S}_{e,1} - 2u^{(\mu} g_{2} \tilde{V}_{e,4}^{\nu)} $	$ \begin{array}{l} \left(\tilde{\alpha}_{Q,2} u^{\mu} - \tilde{\alpha}_{R_s,2} \zeta^{\mu} \right) \tilde{S}_{e,2} \\ + g_2 \tilde{V}^{\mu}_{e,2} \\ + \epsilon^{\mu\nu\rho\sigma} \mathcal{D}_{\nu} (Tg_2 \zeta_{\rho} u_{\sigma}) \end{array} $
\mathcal{N}	K	N^{μ}
$f_1 S_{e,1}$	$\mathcal{D}_{\mu}\left(\zeta^{\mu}\alpha_{R_{s},1}S_{e,1}-f_{1}V_{e,1}^{\mu}\right)$	$\frac{1}{T}f_1\left(u^{\mu}S_{e,1}-\zeta^{\mu}S_5\right)$
$f_2 S_{e,2}$	$\mathrm{D}_{\mu}\left(\zeta^{\mu}\alpha_{R_{s},2}S_{e,1}-f_{1}V_{e,2}^{\mu}\right)$	$\frac{1}{T}f_2\left(u^{\mu}S_{e,2}-\zeta^{\mu}S_6\right)$
$g_1\tilde{S}_{e,1}$	$\mathrm{D}_{\mu}\left(\zeta^{\mu} ilde{lpha}_{R_{s},1} ilde{S}_{e,1}-g_{1} ilde{V}_{e,1}^{\mu} ight)$	$g_1\left(\frac{1}{T}u^{\mu}\tilde{S}_{e,1}+\tilde{V}_3^{\mu}\right)$
$g_2 \tilde{S}_{e,2}$	$\mathrm{D}_{\mu}\left(\zeta^{\mu} ilde{lpha}_{R_{s},2} ilde{S}_{e,2}-g_{2} ilde{V}_{e,2}^{\mu} ight)$	$g_2\left(\frac{1}{T}u^{\mu}\tilde{S}_{e,2}+\tilde{V}_2^{\mu}\right)$

Table 3.4: One-derivative Class H_S constitutive relations for a (3 + 1)-dimensional relativistic superfluid.

hydrostatic due to the Josephson equation. For clarity, let us define the derivatives of the transport coefficients appearing above as

$$df_{i} = \frac{\alpha_{E,i}}{T} dT + T\alpha_{Q,i} d\frac{\mu}{T} - \frac{1}{2} \left(\alpha_{R_{s},i} + \frac{f_{i}}{\zeta^{2}} \right) d\zeta^{2},$$

$$dg_{i} = \frac{\tilde{\alpha}_{E,i}}{T} dT + T\tilde{\alpha}_{Q,i} d\frac{\mu}{T} - \frac{1}{2} \left(\tilde{\alpha}_{R_{s},i} + \frac{g_{i}}{\zeta^{2}} \right) d\zeta^{2},$$
(3.29)

along with

 $\alpha_{E,i} + f_i = \alpha_{S,i}T + \alpha_{Q,i}\mu, \qquad \tilde{\alpha}_{E,i} + g_i = \tilde{\alpha}_{S,i}T + \tilde{\alpha}_{Q,i}\mu.$ (3.30)

Comparing the variation of \mathcal{N} to eq. (2.74) and using these definitions, we can read out the Class H_S constitutive relations. The algebra is quite involved; see the appendix of [4] for detailed steps. For the benefit of the reader, we have summarised the results in table 3.4. Moving on, introduction of the phase field φ does not alter the (thermal) anomaly polynomial in any way, therefore Class A and Class H_V results can be directly imported from table 3.1. The only worthwhile comment is that we do not need to consider C_0 and C_2 terms in Class H_V independently. Due to the presence of a "gauge fixed" version of the gauge field $\xi_{\mu} = A_{\mu} + \partial_{\mu}\varphi$, these terms can be absorbed into Class H_S by shifting $g_1 \to g_1 - \mu C_0 - TC_2$ and $g_2 \to g_2 - C_0$. However, the C_1 term in Class H_V and C term in Class A remain independent.

\mathfrak{D}_0	$T^{\mu u}$	J^{μ}	K
$-T\beta_{11} \begin{pmatrix} P_{\zeta}^{\mu\nu}P_{\zeta}^{\rho\sigma} & \cdot & \cdot \\ \cdot & \cdot & \cdot \end{pmatrix}$	$-eta_{11}P^{\mu u}_{\zeta}S_1$		
$-T\beta_{(12)}\begin{pmatrix} P^{\mu\nu}_{\zeta}\hat{\zeta}^{\rho}\zeta^{\sigma}+\zeta^{\mu}\zeta^{\nu}P^{\rho\sigma}_{\zeta}&\vdots\\\vdots&\vdots&\vdots\end{pmatrix}$	$-\beta_{(12)}\left(P^{\mu\nu}_{\zeta}S_2+\zeta^{\mu}\zeta^{\nu}S_1\right)$		
$-T\beta_{(13)} \begin{pmatrix} \zeta^{\mu} P_{\zeta}^{\rho\sigma} & \vdots \\ \zeta^{\mu} P_{\zeta}^{\rho\sigma} & \vdots \end{pmatrix}$	$-\beta_{(13)}P_{\zeta}^{\mu\nu}S_3$	$-\beta_{(13)}\zeta^{\mu}S_1$	
$-T\beta_{(14)}\begin{pmatrix} \cdot & \cdot & P_{\zeta}^{\mu\nu} \\ \cdot & \cdot & \cdot \\ P_{\zeta}^{\rho\sigma} & \cdot & \cdot \end{pmatrix}$	$-\beta_{(14)}P_{\zeta}^{\mu\nu}S_4$		$-\beta_{(14)}S_1$
$-T\beta_{22}\left(\begin{array}{ccc} \zeta^{\dot{\mu}}\zeta^{\dot{\nu}}\zeta^{\rho}\zeta^{\sigma} & \vdots & \vdots \\ \vdots & \vdots & \vdots \end{array}\right)$	$-eta_{22}\zeta^\mu\zeta^ u S_2$		
$-T\beta_{(23)}\left(\dot{\zeta^{\mu}}\dot{\zeta^{\rho}}\zeta^{\sigma} \vdots \vdots \right)$	$-eta_{(23)}\zeta^{\mu}\zeta^{ u}S_3$	$-\beta_{(23)}\zeta^{\mu}S_2$	
$-T\beta_{(24)} \begin{pmatrix} \vdots & \zeta^{\mu}\zeta^{\nu} \\ \zeta^{\rho}\zeta^{\sigma} & \vdots & \ddots \end{pmatrix}$	$-\beta_{(24)}\zeta^{\mu}\zeta^{\nu}S_4$		$-\beta_{(24)}S_2$
$-T\beta_{33}\left(\begin{array}{c} \zeta^{\mu}\zeta^{ ho} \end{array}\right)$		$-\beta_{33}\zeta^{\mu}S_3$	
$-T\beta_{(34)}\left(\begin{array}{ccc} \cdot & \cdot & \zeta^{\mu} \\ \cdot & \zeta^{ ho} \end{array}\right)$		$-\beta_{(34)}\zeta^{\mu}S_4$	$-\beta_{(34)}S_3$
$-T\beta_{44} \begin{pmatrix} \vdots & \vdots & \vdots \\ \vdots & \vdots & 1 \end{pmatrix}$			$-\beta_{44}S_4$
$-4T\kappa_{11}\begin{pmatrix} \zeta^{(\mu}P_{\zeta}^{\nu)(\rho}\zeta^{\sigma)} & \cdot & \cdot \\ \vdots & \vdots & \vdots \end{pmatrix}$	$-2\kappa_{11}\zeta^{(\mu}V_1^{\nu)}$		
$-2T\kappa_{(12)}\begin{pmatrix} \cdot & \zeta^{(\mu}P_{\zeta}^{\nu)\rho} \\ \zeta^{(\rho}P_{\zeta}^{\sigma)\mu} & \cdot \\ \cdot & \cdot \end{pmatrix}$	$-2\kappa_{(12)}\zeta^{(\mu}V_2^{\nu)}$	$-\kappa_{(12)}V_1^{\mu}$	
$-T\kappa_{22}\left(\begin{array}{cc} & P_{\zeta}^{\mu\rho} & \vdots \\ \vdots & \vdots \end{array}\right)$		$-\kappa_{22}V_2^{\mu}$	
$2T\tilde{\kappa}_{[12]}\begin{pmatrix} & \zeta^{(\mu}\tilde{\epsilon}^{\nu)\rho} & \\ -\tilde{\epsilon}^{\mu(\rho}\zeta^{\sigma)} & & \\ & & \ddots \end{pmatrix}$	$-2\tilde{\kappa}_{[12]}\zeta^{(\mu}\tilde{V}_2^{\nu)}$	$\tilde{\kappa}_{[12]}\tilde{V}^{\mu}_{1}$	
$-T\eta \begin{pmatrix} P_{\zeta}^{\mu\langle\rho}P_{\zeta}^{\sigma\rangle\nu} & \cdot & \cdot \\ \cdot & \cdot & \cdot & \cdot \\ \cdot & \cdot & \cdot & \cdot$	$-\eta\sigma^{\mu u}_{\zeta}$		

Table 3.5: One-derivative Class D constitutive relations for a (3 + 1)-dimensional superfluid. We have defined $\tilde{\epsilon}^{\mu\nu} = \epsilon^{\mu\nu\rho\sigma} u_{\rho}\zeta_{\sigma}$. Note that we have included the dissipative transport coefficient $\beta_{44} = \alpha/T$ in K for completeness.

In the non-hydrostatic sector, we need to write the most generic parametrisation of \mathfrak{C}_0 which decomposes into a symmetric part \mathfrak{D}_0 and an antisymmetric part $\overline{\mathfrak{D}}_0$. They correspond to Class D and Class $\overline{\mathrm{D}}$ constitutive relations respectively. There are a total of 11 transport coefficients in Class $\overline{\mathrm{D}}$: $[\beta_{[ij]}]_{4\times 4}$, $[\kappa_{[ij]}]_{2\times 2}$, $[\tilde{\kappa}_{(ij)}]_{2\times 2}$, and $\tilde{\eta}$. On the other hand, there are 15 transport coefficients in Class D: $[\beta_{(ij)}]_{4\times 4}$, $[\kappa_{(ij)}]_{2\times 2}$, $[\tilde{\kappa}_{[ij]}]_{2\times 2}$, and η , including $\beta_{44} = \alpha/T$ discussed in section 3.2.1. These results have been presented in tables 3.5 and 3.6. The quadratic form Δ is given by

$$T\Delta = \sum_{i,j=1}^{4} S_i \beta_{(ij)} S_j + \left(\sum_{i,j=1}^{2} V_i^{\mu} \kappa_{(ij)} V_{i,\mu} + \sum_{i=1}^{2} V_i^{\mu} \tilde{\kappa}_{[ij]} \tilde{V}_{j,\mu} \right) + \eta \sigma^{\mu\nu} \sigma_{\mu\nu}.$$
(3.31)

Note that

$$(\epsilon^{\mu\nu\rho\sigma}u_{\rho}\zeta_{\sigma})(\epsilon_{\tau\nu\alpha\beta}u^{\alpha}\zeta^{\beta}) = \zeta^2 P_{\zeta}{}^{\mu}_{\tau}.$$
(3.32)

Chapter 3. Applications: relativistic hydrodynamics

$\overline{\mathfrak{D}}_0$	$T^{\mu u}$	J^{μ}	K
$-T\beta_{[12]} \begin{pmatrix} \tilde{P}^{\mu\nu}\zeta^{\rho}\zeta^{\sigma} - \zeta^{\mu}\zeta^{\nu}\tilde{P}^{\rho\sigma} & \vdots \\ \vdots & \vdots & \vdots \end{pmatrix}$	$-\beta_{[12]}\left(\tilde{P}^{\mu\nu}S_2-\zeta^{\mu}\zeta^{\nu}S_1\right)$		
$-T\beta_{[13]} \begin{pmatrix} \zeta^{\rho} P_{\zeta}^{\mu\nu} & \cdot \\ -\zeta^{\mu} P_{\zeta}^{\rho\sigma} & \cdot \end{pmatrix}$	$-\beta_{[13]}P^{\mu\nu}_{\zeta}S_3$	$\beta_{[13]}\zeta^{\mu}S_1$	
$-T\beta_{[14]} \begin{pmatrix} \cdot & \cdot & P_{\zeta}^{\mu\nu} \\ \cdot & \cdot & \cdot \\ -P_{\zeta}^{\rho\sigma} & \cdot & \cdot \end{pmatrix}$	$-\beta_{[14]}P^{\mu\nu}_{\zeta}S_4$		$\beta_{[14]}S_1$
$-T\beta_{[23]}\left(\begin{array}{ccc} & \zeta^{\rho}\zeta^{\mu}\zeta^{\nu} \\ -\zeta^{\mu}\zeta^{\rho}\zeta^{\sigma} & \vdots \end{array}\right)$	$-eta_{[23]}\zeta^\mu\zeta^ u S_3$	$\beta_{[23]}\zeta^{\mu}S_2$	
$-T\beta_{[24]} \begin{pmatrix} \vdots & \vdots & \zeta^{\mu}\zeta^{\nu} \\ -\zeta^{\rho}\zeta^{\sigma} & \vdots & \vdots \end{pmatrix}$	$-eta_{[24]}\zeta^\mu\zeta^ u S_4$		$\beta_{[24]}S_2$
$-Teta_{[34]}\left(\begin{array}{ccc} & \zeta^{\mu} \\ & -\zeta^{ ho} \end{array}\right)$		$-\beta_{[34]}\zeta^{\mu}S_4$	$\beta_{[34]}S_3$
$-2T\kappa_{[12]}\begin{pmatrix} & \zeta^{(\mu}P_{\zeta}^{\nu)\rho} & \\ -\zeta^{(\rho}P_{\zeta}^{\sigma)\mu} & & \\ & & \\ \end{pmatrix}$	$-2\kappa_{[12]}\zeta^{(\mu}V_2^{\nu)}$	$\kappa_{[12]}V_1^{\mu}$	
$4T\tilde{\kappa}_{11}\begin{pmatrix} \zeta^{(\mu}\tilde{\epsilon}^{\nu)(\rho}\zeta^{\sigma)} & \vdots & \vdots \\ \vdots & \vdots & \vdots \end{pmatrix}$	$-2\tilde{\kappa}_{11}\zeta^{(\mu}\tilde{V}_1^{\nu)}$		
$2T\tilde{\kappa}_{(12)} \begin{pmatrix} & \zeta^{(\mu}\tilde{\epsilon}^{\nu)\rho} \\ \tilde{\epsilon}^{\mu(\rho}\zeta^{\sigma)} & \vdots \end{pmatrix}$	$-2\tilde{\kappa}_{(12)}\zeta^{(\mu}\tilde{V}_2^{\nu)}$	$-\tilde{\kappa}_{(12)}\tilde{V}_1^{\mu}$	
$T\tilde{\kappa}_{22} \begin{pmatrix} \vdots & \tilde{\epsilon}^{\mu\rho} & \vdots \\ \vdots & \tilde{\epsilon}^{\mu\rho} & \vdots \end{pmatrix}$		$- ilde{\kappa}_{22} ilde{V}_2^\mu$	
$T\tilde{\eta} \begin{pmatrix} P_{\zeta}^{\lambda(\mu} \tilde{\epsilon}^{\nu)(\rho} P_{\zeta}^{\sigma)}{}_{\lambda} & \vdots \\ \vdots & \vdots & \vdots \end{pmatrix}$	$ ilde\eta ilde\sigma^{\mu u}_\zeta$		

Table 3.6: One-derivative Class \overline{D} constitutive relations for a (3+1)-dimensional superfluid. We have defined $\tilde{\epsilon}^{\mu\nu} = \epsilon^{\mu\nu\rho\sigma} u_{\rho} \zeta_{\sigma}$.

We can use this to define a new basis of vector structures

$$\begin{pmatrix} V_1^{\prime \mu} \\ V_2^{\prime \mu} \end{pmatrix} = \begin{pmatrix} V_1^{\mu} \\ V_2^{\mu} \end{pmatrix} + \begin{pmatrix} 0 & \tilde{\kappa}_{12}^{(a)} / \kappa_{11} \\ 0 & 0 \end{pmatrix} \begin{pmatrix} \tilde{V}_1^{\mu} \\ \tilde{V}_2^{\mu} \end{pmatrix},$$
$$\begin{pmatrix} \kappa_{11}^{\prime} & \kappa_{12}^{\prime} \\ \kappa_{12}^{\prime} & \kappa_{22}^{\prime} \end{pmatrix} = \begin{pmatrix} \kappa_{11} & \kappa_{12} \\ \kappa_{(12)} & \kappa_{22} - \zeta^2 \frac{\tilde{\kappa}_{[12]}}{\kappa_{11}} \end{pmatrix},$$
(3.33)

such that

$$\sum_{i,j=1}^{2} V_{i}^{\prime \mu} \kappa_{(ij)}^{\prime} V_{i,\mu}^{\prime} = \sum_{i,j=1}^{2} V_{i}^{\mu} \kappa_{(ij)} V_{i,\mu} + \sum_{i=1}^{2} V_{i}^{\mu} \tilde{\kappa}_{[ij]} \tilde{V}_{j,\mu}.$$
(3.34)

In this basis, Δ takes the form

$$T\Delta = \sum_{i,j=1}^{4} S_i \beta_{(ij)} S_j + \sum_{i,j=1}^{2} V_i^{\prime \mu} \kappa_{(ij)}^{\prime} V_{i,\mu}^{\prime} + \eta \sigma^{\mu\nu} \sigma_{\mu\nu}.$$
 (3.35)

Provided that T > 0, the condition $\Delta \ge 0$ implies that $\eta \ge 0$ and the matrices $[\beta_{(ij)}]_{4\times 4}$, $[\kappa'_{(ij)}]_{2\times 2}$ have all non-negative eigenvalues. This gives 7 inequalities among the 15 Class D transport coefficients, while the remaining 8 are completely arbitrary.

In summary, the most generic constitutive relations of a (3 + 1)-dimensional relativistic

superfluid up to one-derivative order are given as: the energy-momentum tensor

$$T^{\mu\nu} = (E+P)u^{\mu}u^{\nu} + Pg^{\mu\nu} + R_{s}\xi^{\mu}\xi^{\nu} + u^{\mu}u^{\nu} \left[\sum_{i=1}^{2} \alpha_{E,i}S_{e,i} + \sum_{i=1}^{2} \tilde{\alpha}_{E,i}\tilde{S}_{e,i} - \frac{1}{T}D_{\sigma}(Tf_{1}\zeta^{\sigma}) + \epsilon^{\alpha\rho\sigma\tau}u_{\alpha}D_{\rho}(Tg_{1}u_{\sigma}\zeta_{\tau}) \right] + 2u^{(\mu}\zeta^{\nu)} \left[\sum_{i=1}^{2} f_{i}S_{4+i} - \mu \left(\sum_{i=1}^{2} \alpha_{R_{s,i}}S_{e,i} + \sum_{i=1}^{2} \tilde{\alpha}_{R_{s,i}}\tilde{S}_{e,i} \right) + \frac{1}{2\hat{\mu}_{s}}\epsilon^{\alpha\rho\sigma\tau}\zeta_{\alpha}D_{\rho}(Tg_{1}u_{\sigma}\zeta_{\tau}) \right] + \zeta^{\mu}\zeta^{\nu} \left[\sum_{i=1}^{2} \alpha_{R_{s,i}}S_{e,i} + \sum_{i=1}^{2} \left(\tilde{\alpha}_{R_{s,i}} - \frac{g_{i}}{2\hat{\mu}_{s}} \right) \tilde{S}_{e,i} - \sum_{i=1}^{4} \beta_{2i}S_{i} \right] + 2u^{(\mu} \left[\mu \sum_{i=1}^{2} f_{i}V_{e,i}^{\nu)} - \sum_{i=1}^{2} g_{i}\tilde{V}_{e,2+i}^{\nu)} - P\zeta^{\nu}_{\alpha}\epsilon^{\alpha\rho\sigma\tau}D_{\rho}(Tg_{1}u_{\sigma}\zeta_{\tau}) + 2C_{1}T^{3}\omega^{\nu}) + \mu^{2}C \left(3B^{\nu} + 2\mu\omega^{\nu} \right) \right] - 2\zeta^{(\mu} \left[\sum_{i=1}^{2} f_{i}V_{e,i}^{\nu)} + \sum_{i=1}^{2} \kappa_{1i}V_{i}^{\nu)} + \sum_{i=1}^{2} \tilde{\kappa}_{1i}\tilde{V}_{i}^{\nu} \right] + P_{\zeta}^{\mu\nu} \left[\sum_{i=1}^{2} f_{i}S_{e,i} - \sum_{i=1}^{4} \beta_{1i}S_{i} \right] - \eta\sigma_{\zeta}^{\mu\nu} - \tilde{\eta}\tilde{\sigma}_{\zeta}^{\mu\nu} + \mathcal{O}(\partial^{2}),$$
 (3.36)

and the charge current

$$J^{\mu} = Qu^{\mu} - R_{s}\xi^{\mu} + u^{\mu} \left[\sum_{i=1}^{2} \alpha_{Q,i} S_{e,i} + \sum_{i=1}^{2} \tilde{\alpha}_{Q,i} \tilde{S}_{e,i} - \frac{1}{T} D_{\nu} (Tf_{2}\zeta^{\nu}) + \epsilon^{\alpha\nu\rho\sigma} u_{\alpha} D_{\nu} (Tg_{2}u_{\rho}\zeta_{\sigma}) \right] - \zeta^{\mu} \left[\sum_{i=1}^{2} \alpha_{R_{s},i} S_{e,i} + \sum_{i=1}^{2} \tilde{\alpha}_{R_{s},i} \tilde{S}_{e,i} + \sum_{i=1}^{4} \beta_{3i} S_{i} - \frac{1}{2\hat{\mu}_{s}} \epsilon^{\alpha\nu\rho\sigma} \zeta_{\alpha} D_{\nu} (Tg_{2}u_{\rho}\zeta_{\sigma}) \right] + \sum_{i=1}^{2} f_{i} V^{\mu}_{e,i} + \sum_{i=1}^{2} g_{i} \tilde{V}^{\mu}_{e,i} - \sum_{i=1}^{2} \kappa_{2i} V^{\mu}_{i} - \sum_{i=1}^{2} \tilde{\kappa}_{2i} \tilde{V}^{\mu}_{i} - P_{\zeta}{}^{\mu}_{\alpha} \epsilon^{\alpha\nu\rho\sigma} D_{\nu} (Tg_{2}u_{\rho}\zeta_{\sigma}) + 3\mu C^{(4)} (2B^{\mu} + \mu\omega^{\mu}) + \mathcal{O}(\partial^{2}).$$
(3.37)

The corrected Josephson equation is given by

$$u^{\mu}\xi_{\mu} = \mu + \frac{1}{\beta_{44}} D_{\mu} \left(R_{s}\xi^{\mu} \right) - \sum_{i=1}^{3} \frac{\beta_{4i}}{\beta_{44}} S_{i} + \frac{1}{\beta_{44}} D_{\mu} \left(\zeta^{\mu} \sum_{i=1}^{2} \alpha_{R_{s},i} S_{e,i} + \zeta^{\mu} \sum_{i=1}^{2} \tilde{\alpha}_{R_{s},i} \tilde{S}_{e,i} - \sum_{i=1}^{2} f_{i} V_{e,i}^{\mu} - \sum_{i=1}^{2} g_{i} \tilde{V}_{e,i}^{\mu} \right) + \mathcal{O}(\partial^{2}), \quad (3.38)$$

which can be used to substitute for $S_4 = u^{\mu}\xi_{\mu} - \mu$ in the constitutive relations above. These constitutive relations satisfy the second law of thermodynamics with the free energy current

$$N^{\mu} = \frac{1}{T} P u^{\mu} + \frac{1}{T} \sum_{i=1}^{2} f_{i} \left(u^{\mu} S_{e,i} - \zeta^{\mu} S_{4+i} \right) + \frac{1}{T} \sum_{i=1}^{2} g_{i} \left(u^{\mu} \tilde{S}_{e,i} + \tilde{V}_{2+i}^{\mu} \right) \\ + \frac{1}{T} 3 \mu^{2} C B^{\mu} + \frac{1}{T} \left(\mu^{3} C + 4T^{3} C_{1} \right) \omega^{\mu}.$$
(3.39)

We can also work out the associated entropy current explicitly

$$J_{\rm S}^{\mu} = Su^{\mu} + g_1 \frac{1}{T} \epsilon^{\mu\nu\rho\sigma} u_{\nu} \zeta_{\rho} \partial_{\sigma} T + g_2 T \epsilon^{\mu\nu\rho\sigma} u_{\nu} \xi_{\rho} \partial_{\sigma} \frac{\mu}{T} + 3C_1 T^2 \omega^{\mu} + u^{\mu} \left[\sum_{i=1}^{2} \alpha_{S,i} S_{e,i} + \sum_{i=1}^{2} \tilde{\alpha}_{S,i} \tilde{S}_{e,i} - \frac{1}{T^2} D_{\sigma} (Tf_1 \zeta^{\sigma}) + \frac{\mu}{T^2} D_{\nu} (Tf_2 \zeta^{\nu}) + \frac{1}{T} \epsilon^{\alpha\nu\rho\sigma} u_{\alpha} D_{\nu} (Tg_1 u_{\rho} \zeta_{\sigma}) - \frac{\mu}{T} \epsilon^{\alpha\rho\sigma\tau} u_{\alpha} D_{\nu} (Tg_2 u_{\rho} \zeta_{\sigma}) \right] + \frac{1}{T} \zeta^{\mu} \left[\sum_{i=1}^{4} \mu \beta_{3i} S_i + \frac{1}{2\hat{\mu}_s} \epsilon^{\alpha\rho\sigma\tau} \zeta_{\alpha} D_{\rho} (Tg_1 u_{\sigma} \zeta_{\tau}) - \frac{\mu}{2\hat{\mu}_s} \zeta_{\alpha} \epsilon^{\alpha\nu\rho\sigma} D_{\nu} (Tg_2 u_{\rho} \zeta_{\sigma}) \right] + \frac{\mu}{T} \sum_{i=1}^{2} \kappa_{2i} V_i^{\mu} + \frac{\mu}{T} \sum_{i=1}^{2} \tilde{\kappa}_{2i} \tilde{V}_i^{\mu} - \frac{1}{T} P_{\zeta^{\mu}} \epsilon^{\alpha\nu\rho\sigma} D_{\nu} (Tg_1 u_{\rho} \zeta_{\sigma}) + \frac{\mu}{T} P_{\zeta^{\mu}} \epsilon^{\alpha\nu\rho\sigma} D_{\nu} (Tg_2 u_{\rho} \zeta_{\sigma}).$$
(3.40)

Including the ideal order pressure P, the superfluid constitutive relations up to the first derivative order are parametrised by 2 constants and 31 independent transport coefficients, 7 of which are restricted to be non-negative.

Similar to the ordinary fluids, we can also convert the superfluid constitutive relations to any desired hydrodynamic frame, for example the Landau frame. However, given the complexity of the constitutive relations as they are, we do not perform this exercise here.

3.2.4 Breaking of non-Abelian internal symmetries

In our discussion above, we focused on Abelian superfluids. However, in recent years (see e.g. [97]), non-Abelian superfluids have also started to attract some attention in the literature, in relation to the *p*-wave superfluidity observed in liquid ³He [98, 99]. In this section, we explore how our Abelian results might be generalised to accommodate non-Abelian superfluids. The discussion presented here is directly from our work in [1].

Let us start with a quick recap of the non-Abelian spontaneous symmetry breaking. More details can be found in section 19 of [100]. Consider a microscopic theory which is invariant under spacetime Poincaré transformations and the global action of a semisimple Lie group G (with Lie algebra \mathfrak{g}). Let ψ be a field in the theory transforming under some unitary representation $\mathcal{D}(G)$ of G, i.e. under a $g \in G$ transformation $\psi \to \mathcal{D}(g)\psi$. The field ψ is said to spontaneously break the symmetry from G to its Lie subgroup $H \subset G$ (with Lie subalgebra $\mathfrak{h} \subset \mathfrak{g}$), if its ground state expectation value $\langle \psi \rangle$ is only invariant under H, i.e.

$$\mathcal{D}(h)\langle\psi\rangle = \langle\psi\rangle \quad \text{if and only if} \quad h \in H.$$
(3.41)

 $\mathcal{D}(g) \langle \psi \rangle$ with $g \notin H$ represents the "other" ground states the system could have spontaneously chosen from. Around $\langle \psi \rangle$, the field ψ can be expressed as a group transformation of a reference field $\tilde{\psi}$, i.e. $\psi = \mathcal{D}(\gamma)\tilde{\psi}$, defined by

$$\tilde{\psi}^{\dagger} \mathcal{D}(X) \langle \psi \rangle = \tilde{\psi}^{\dagger} \langle \psi \rangle, \quad \forall X \in \mathfrak{g}.$$
 (3.42)

Roughly speaking, γ corresponds to the fluctuations of ψ which take us to the nearby ground states with no energy cost, while $\tilde{\psi}$ contains the genuine excitations of ψ .

Note that eq. (3.42) is invariant under $\tilde{\psi} \to \mathcal{D}(h)\tilde{\psi}$ with $h \in H$, and hence determines γ only up to a coset equivalence $\gamma \sim \gamma h$. Let us pick a representative from each coset $\gamma = \gamma(\varphi)$ parametrised by a field φ living in the Lie algebra quotient $\mathfrak{g}/\mathfrak{h}$. The field φ can be identified as the *Goldstone modes* of the broken symmetry. Under a $g \in G$ transformation, these modes transform according to

$$\gamma(\varphi) \to g\gamma(\varphi)h(\varphi,g)^{-1}, \quad \tilde{\psi} \to \mathcal{D}(h(\varphi,g))\tilde{\psi},$$
(3.43)

for some $h(\varphi, g) \in H$, such that $\psi \to \mathcal{D}(g)\psi$ and eq. (3.42) remains invariant. From these transformation properties, it is clear that the theory cannot contain a mass term for φ , rendering it gapless.

Let us make contact with the Abelian case, where G = U(1) is broken down to $H = \{1\}$. The Goldstone mode φ is a scalar with $\gamma(\varphi) = e^{-i\varphi}$. Under a $g = e^{i\Lambda} \in U(1)$ transformation, it transforms according to $e^{-i\varphi} \to e^{i\Lambda}e^{-i\varphi}$ implying $\varphi \to \varphi - \Lambda$. This is what we used in our construction of Abelian superfluids.

Once we have our Goldstone modes ready, we need to define their covariant derivative, so as to define the associated superfluid velocity. However, there is a hitch; φ lives in the quotient $\mathfrak{g}/\mathfrak{h}$ and it is not very straightforward to define a covariant derivative on a quotient space. One such notion comes from the Maurer–Cartan form in differential geometry, however, we can use a much simpler setup for our purposes. Let us introduce a set of generators $\{\mathbf{t}_A\} = \{\mathbf{t}_i, \mathbf{t}_a\}$ of G such that the subset $\{\mathbf{t}_i\}$ generates H. We normalise these generators by choosing $\mathbf{t}_A \cdot \mathbf{t}_B = 2\text{Tr} [\mathbf{t}_A \mathbf{t}_B] = \eta_{AB}$, where η_{AB} is a diagonal matrix with entries ± 1 . We define the action of an element $g \in G$ on these generators as $\mathbf{t}_A \to \text{Ad}_g(\mathbf{t}_A) = (\text{Ad}_g)_A^B \mathbf{t}_B = g\mathbf{t}_A g^{-1}$. We can now use these to define a set of projectors

$$P(t_A) = P^B_A t_B = \left((Ad_{\gamma^{-1}})^B_i (Ad_{\gamma})^i{}_A \right) t_B, \qquad \overline{P}(t_A) = \overline{P}^B_A t_B = \left(\delta^B_A - P^B_A \right) t_B. \quad (3.44)$$

It can be checked that they covariantly project out the components of $X \in \mathfrak{g}$ along and against the residual symmetry respectively. Using these, we can package the non-covariant information in the derivatives of the Goldstone modes φ into a covariant object

$$\xi_{\mu} = \overline{\mathbb{P}}\left(i\partial_{\mu}\gamma(\varphi)\gamma(\varphi)^{-1} + A_{\mu}\right) \in \overline{\mathbb{P}}(\mathfrak{g}), \qquad (3.45)$$

identified as the superfluid velocity. As an added benefit of the projectors we defined, we can revert back to the ordinary fluids at any point by setting $\overline{\mathbf{P}} = 0$, $\mathbf{P} = \mathrm{id}_{\mathfrak{g}}$ (identity in \mathfrak{g}).

From this point forward, the analysis is exactly parallel to our Abelian superfluid discussion. We introduce a set of equations of motion for φ

$$K \approx 0 \in \overline{\mathbf{P}}(\mathfrak{g}). \tag{3.46}$$

The adiabaticity equation for this system is given simply by

$$D_{\mu}N^{\mu} - N_{\rm H}^{\perp} = \frac{1}{2}T^{\mu\nu}\delta_{\mathcal{B}}g_{\mu\nu} + J^{\mu}\cdot\delta_{\mathcal{B}}A_{\mu} + K\cdot\delta_{\mathcal{B}}\varphi + \Delta, \qquad \Delta \ge 0.$$
(3.47)

Here we have defined $\delta_{\mathcal{B}}\varphi = \beta^{\mu}\xi_{\mu} - \overline{P}(\mu)/T$. Closely following our Abelian superfluid calculation, at the ideal order we find the set of constitutive relations, free energy, and entropy currents for a non-Abelian superfluid to be

$$T^{\mu\nu} = (E+P)u^{\mu}u^{\nu} + Pg^{\mu\nu} + \xi^{\mu} \cdot R_s \cdot \xi^{\nu} + \mathcal{O}(\partial),$$

$$J^{\mu} = Qu^{\mu} - R_s \cdot \xi^{\mu} + \mathcal{O}(\partial),$$

$$K = -\alpha \cdot \delta_{\mathcal{B}}\varphi + D_{\mu}(R_s\xi^{\mu}) + [\xi_{\mu}, R_s \cdot \xi^{\mu}] + \mathcal{O}(\partial),$$

$$N^{\mu} = \frac{1}{T}Pu^{\mu} + \delta_{\mathcal{B}}\varphi \cdot R_s \cdot \xi^{\mu} + \mathcal{O}(\partial),$$

$$J^{\mu}_{S} = N^{\mu} - \frac{1}{T}(T^{\mu\nu}u_{\nu} + \mu \cdot J^{\mu}) = Su^{\mu} + \mathcal{O}(\partial).$$
(3.48)

Here $\alpha \in \overline{P}(\mathfrak{g}) \times \overline{P}(\mathfrak{g})$ is a positive semi-definite matrix. On the other hand, $R_s \in \overline{P}(\mathfrak{g}) \times_{\text{sym}} \overline{P}(\mathfrak{g})$ is a symmetric matrix, which enters the thermodynamic relations

Gibbs-Duhem equation:
$$dP = SdT + Q \cdot d\mu + R_s^{AB} d\mu_{AB}^s$$
,
Euler scaling relation: $E + P = ST + Q \cdot \mu$,
First law of thermodynamics: $dE = TdS + \mu \cdot dQ - R_s^{AB} d\mu_{AB}^s$, (3.49)

where $\mu_{AB}^s = -\frac{1}{2} \xi_A^{\mu} \xi_B^{\nu} g_{\mu\nu}$. By setting $K \approx 0$, we can compute the non-Abelian Josephson equation

$$u^{\mu}\xi_{\mu} = \overline{\mathbf{P}}(\mu) + T\alpha' \cdot \left(\mathbf{D}_{\mu}(R_{s}\xi^{\mu}) + [\xi_{\mu}, R_{s} \cdot \xi^{\mu}] \right) + \mathcal{O}(\partial), \qquad (3.50)$$

where $\alpha' \cdot \alpha = \overline{P}$. Note, again, that this equation is only accurate at the zero derivative order. At the first derivative order, it can still admit further corrections.

We wrap up our discussion of non-Abelian superfluids here. We could explicitly write down the one-derivative corrections, but nothing out of the ordinary happens compared to the Abelian case.

3.3 Relativistic fluid surfaces

For the last example of this chapter, let us consider the spontaneous breaking of a spacetime symmetry rather than an internal one to generate a gapless mode. In the two hydrodynamic systems that we discussed above, we assumed the fluid to be space-filling, i.e. we assumed that it extends infinitely in every direction. This is a reasonable assumption when describing physics deep in the bulk of a fluid, but starts to break down as we approach an interface or a boundary. The very existence of such a "surface" breaks the translational symmetry of the theory and considerably modifies the hydrodynamic spectrum. Let us call the Goldstone mode of this broken translational generator to be f(x). It is just a scalar field with the transformation property $\delta_{\chi} f = \chi^{\mu} \partial_{\mu} f$. Just like the superfluids, we take f to be $\mathcal{O}(\partial^{-1})$ in the derivative expansion.

A priori, including f in our description breaks the translational symmetry at every spacetime point. This has been utilised elsewhere in the literature to model lattices and study phenomena like momentum relaxation; see e.g. [101]. The corresponding hydrodynamic constitutive relations can be derived straightforwardly following our superfluid analysis, by substituting $\partial_{\mu} f$ for ξ_{μ} . Here, however, we are interested in somehow localising the symmetry breaking into a thin spatial region, so as to form a surface. With this in mind, let us perform a coordinate transformation to bring the surface of interest to f = 0, and introduce a distribution functional $\theta(f)$ centred around it. We assume that $\theta(f)$ completely characterises the f dependence in the constitutive relations. We further assume that $\theta'(f)$ is positive and is only supported within a narrow region around f = 0. The width of this region characterises the thickness of the surface. In the limit that the surface is infinitesimally thin, $\theta(f)$ approaches the Heaviside step function, while $\theta'(f)$ approaches the Dirac delta function. We call f > 0 to be the "inside" of the fluid while f < 0 to be its "outside".

We take $\theta(f)$ to be $\mathcal{O}(\partial^0)$ in the hydrodynamic derivative expansion. Taking its spacetime derivative we find that $\partial_{\mu}\theta(f) = \theta'(f)\partial_{\mu}f$, which we can formally decompose into¹

$$z_{\mu} = -\frac{\partial_{\mu}\theta(f)}{\sqrt{g^{\mu\nu}\partial_{\mu}\theta(f)\partial_{\nu}\theta(f)}} = -\frac{\partial_{\mu}f}{\sqrt{g^{\mu\nu}\partial_{\mu}f\partial_{\nu}f}},$$

$$\tilde{\delta}(f) = -z^{\mu}\partial_{\mu}\theta(f) = \sqrt{g^{\mu\nu}\partial_{\mu}f\partial_{\nu}f}\,\theta'(f).$$
(3.51)

Due to f being $\mathcal{O}(\partial^{-1})$, the inward pointing normal vector to the surface z_{μ} is clearly $\mathcal{O}(\partial^{0})$, while the distribution $\tilde{\delta}(f)$ is $\mathcal{O}(\partial^{1})$. Similarly, one can check that all the higher derivative tensor structures following from $\theta(f)$ can be represented in terms of the derivatives of z_{μ} along with a series of distributions

$$\tilde{\delta}^{(n)}(f) = (-)^{n+1} z^{\mu_1} \dots z^{\mu_{n+1}} \mathcal{D}_{\mu_1} \dots \mathcal{D}_{\mu_{n+1}} \theta(f), \qquad (3.52)$$

with $\tilde{\delta}^{(0)}(f) = \tilde{\delta}(f)$. We should note that $\partial_{\mu}\theta(f) = \theta'(f)\partial_{\mu}f$ is only supported near the surface, so even though the value of z_{μ} does not depend on $\theta(f)$, it is only well defined when used in conjunction with $\tilde{\delta}^{(n)}(f)$. Another thing to note is that, as long as the surface remains thin enough, we can expand the products like $\tilde{\delta}^{(n)}(f)\tilde{\delta}^{(m)}(f)$ into an infinite series $\sum_{r} c_r \tilde{\delta}^{(r)}(f)$ for some coefficients c_r , so we do not need to consider such products independently.

Having set the stage, let us write down the adiabaticity equation for this setup

$$D_{\mu}N^{\mu} - N_{\rm H}^{\perp} = \frac{1}{2}T^{\mu\nu}\delta_{\mathcal{B}}g_{\mu\nu} + J^{\mu}\delta_{\mathcal{B}}A_{\mu} + \frac{Y\delta_{\mathcal{B}}f}{\sqrt{g^{\nu\rho}\partial_{\nu}f\partial_{\rho}f}} + \Delta, \qquad \Delta \ge 0.$$
(3.53)

¹In the original references [6, 59], the symbol n_{μ} has been used for the normal vector to the surface. We replace it with z_{μ} to avoid confusion with the Newton-Cartan clock form.

Here we have chosen the equation of motion for f to be $Y/\sqrt{g^{\mu\nu}\partial_{\mu}f\partial_{\nu}f} \approx 0$, and defined

$$\frac{\delta_{\mathcal{B}}f}{\sqrt{g^{\nu\rho}\partial_{\nu}f\partial_{\rho}f}} = \frac{\beta^{\mu}\partial_{\mu}f}{\sqrt{g^{\nu\rho}\partial_{\nu}f\partial_{\rho}f}} = -\frac{1}{T}u^{\mu}z_{\mu}.$$
(3.54)

Unlike for superfluids, the adiabaticity equation (3.53) does not have any solution at zero derivatives. Sure, we could write down a term $\delta_{\mathcal{B}} f$ in Y, but we agreed that all the dependence on z_{μ} in the constitutive relations must be accompanied by $\tilde{\delta}^{(n)}(f)$. Moving on to the ideal order, the Class H_S constitutive relations are characterised by a free-energy density

$$\mathcal{N} = \theta(f) P_{\rm in}(T,\mu) + \theta(f) P_{\rm out}(T,\mu), \qquad (3.55)$$

where $\bar{\theta}(f) = 1 - \theta(f)$. Here P_{in} and P_{out} are the thermodynamic pressures inside and outside the surface respectively. The only other non-trivial class at ideal order is Class D with $Y \sim -\alpha \theta'(f) \delta_{\mathcal{B}} f$ for some non-negative coefficient α . Together, they imply a set of constitutive relations

$$T^{\mu\nu} = \theta(f)T^{\mu\nu}_{\rm in} + \bar{\theta}(f)T^{\mu\nu}_{\rm out} + \mathcal{O}(\partial),$$

$$J^{\mu} = \theta(f)J^{\mu}_{\rm in} + \bar{\theta}(f)J^{\mu}_{\rm out} + \mathcal{O}(\partial),$$

$$Y = \tilde{\delta}(f)\left(\frac{\alpha}{T}u^{\mu}z_{\mu} + (P_{\rm in} - P_{\rm out})\right) + \mathcal{O}(\partial).$$
(3.56)

Here $T^{\mu\nu}_{\text{in/out}}$ and $J^{\mu}_{\text{in/out}}$ are the energy-momentum tensors and charge currents of the fluid inside/outside the surface respectively, worked out previously in eq. (3.4). We see that there is no dedicated contribution from the surface at the ideal order. On the other hand, we get our first version of the f equation of motion

$$u^{\mu}z_{\mu} \approx -\frac{T}{\alpha}\Delta P + \mathcal{O}(\partial), \quad \text{where} \quad \Delta P = P_{\text{in}} - P_{\text{out}}.$$
 (3.57)

It materialises the natural expectation that in time the surface moves towards the lower pressure. Since ΔP characterises the "change" in pressure across the surface, we take it to be $\mathcal{O}(\partial^1)$ in the derivative expansion. As a corollary of the f equation of motion therefore, $u^{\mu}z_{\mu}$ is not independent on-shell, but is determined in terms of the other one-derivative order scalars in the theory.

Things become more interesting when we include the one-derivative corrections, i.e. we find the constitutive relations that solve eq. (3.53) at two derivative order. In Class H_S, the free-energy density gets improved to admit a surface tension $\gamma(T, \mu)$,

$$\mathcal{N} = \theta(f) P_{\rm in}(T,\mu) + \bar{\theta}(f) P_{\rm out}(T,\mu) - \tilde{\delta}(f) \gamma(T,\mu).$$
(3.58)

In addition, following our discussion in section 3.1.2, we also have two copies of Class D constitutive relations η , ζ , and σ , one on either sides of the surface, and just a single copy of Class A and H_V constants.² All the other classes are empty. This leads to a set of

 $^{^{2}}$ The (thermal) anomaly polynomial is characterised by a set of constants, which are not allowed to vary over the spacetime manifold. Therefore it is not clear if we can take these constants to be different inside and outside the surface.

constitutive relations

$$T^{\mu\nu} = \theta(f)T^{\mu\nu}_{\rm in} + \bar{\theta}(f)T^{\mu\nu}_{\rm out} + \tilde{\delta}(f)\Big((E_{\rm sur} - \gamma)u^{\mu}u^{\nu} - \gamma(g^{\mu\nu} - z^{\mu}z^{\nu})\Big) + \mathcal{O}(\partial^2),$$

$$J^{\mu} = \theta(f)J^{\mu}_{\rm in} + \bar{\theta}(f)J^{\mu}_{\rm out} + \tilde{\delta}(f)Q_{\rm sur}u^{\mu} + \mathcal{O}(\partial^2),$$

$$Y = \tilde{\delta}(f)\left(\frac{\alpha}{T}u^{\mu}z_{\mu} + (P_{\rm in} - P_{\rm out}) - D_{\mu}(\gamma z^{\mu})\right) + \mathcal{O}(\partial^2),$$
(3.59)

where we have defined

$$-d\gamma = S_{sur}dT + Q_{sur}d\mu, \qquad E_{sur} - \gamma = S_{sur}T + Q_{sur}\mu, \qquad (3.60)$$

while $T^{\mu\nu}_{\rm in/out}$ and $J^{\mu}_{\rm in/out}$ are defined according to eq. (3.11). We see that this time we do get some dedicated surface contributions to the energy-momentum tensor and charge current. More interestingly, the f equation of motion also admits a correction to take the form

$$u^{\mu}z_{\mu} \approx -\frac{T}{\alpha} \left(\Delta P - \mathcal{D}_{\mu} \left(\gamma z^{\mu}\right)\right) + \mathcal{O}(\partial).$$
(3.61)

With this, we see that the gradient of pressure still contributes to the time-evolution of the surface, but it can partially be balanced by a surface tension term. To express this equation in a more familiar form, consider that the fluid in question has a constant surface tension and focus on a static surface configuration with $u^{\mu}z_{\mu} \propto u^{\mu}\partial_{\mu}f = 0$. In this regime, the above equation reduces to

$$\Delta P \approx \gamma \mathcal{D}_{\mu} z^{\mu} + \mathcal{O}(\partial). \tag{3.62}$$

This can immediately be recognised as the Young-Laplace equation. So we see that eq. (3.61) is just the Young-Laplace equation generalised to the out-of-equilibrium fluid configurations. Higher derivative corrections to the hydrodynamic constitutive relations and the Young-Laplace equation can also be worked out in a similar manner.

This point of view to describe the surface dynamics of a fluid in terms of a scalar "shape" field f was introduced in [6, 59]. The results presented in [6] even combine the internal and spacetime symmetry breaking and present surface dynamics in a superfluid. Apart from the obvious generalisations, some new hybrid features arise on the surface of a superfluid owing to the emergence of the new ideal order structures $z_{\mu}\xi^{\mu}$ and $\epsilon^{\mu\nu\rho\sigma}u_{\nu}z_{\rho}\xi_{\sigma}$. We do not concern ourselves with these details here, instead, we just note that they can still be understood within the off-shell framework of hydrodynamics we have presented in this work. For more details, we refer the reader to the discussion in [6].

These are all the examples of relativistic hydrodynamics that we consider in this thesis. We comment on some other possible applications in chapter 6. In the next chapter, we revisit the rules of hydrodynamics when the underlying spacetime symmetry group is taken to be Galilean rather than Poincaré, which provides a framework for non-relativistic hydrodynamics. 4

For most practical purposes, the world around us can be regarded as non-relativistic. So it is natural to formulate a non-relativistic version of hydrodynamics, which is more suited to our day-to-day applications ranging from hydraulics to biophysics. Non-relativistic physics emerges as an effective description of the more fundamental relativistic description of our universe, in a limit where the speed of light c is very large compared to the other characteristic speeds under consideration. For a typical relativistic system, however, taking such a limit (sending $c \to \infty$) turns out to be a notoriously non-trivial task to perform. Except in a few special cases, the non-relativistic limit is either not well defined or is not unique,¹ which forces us to resort to other methods. One such method is to study the non-relativistic theories independent of their relativistic parents, using as our guiding principle the Galilean symmetry emergent in the non-relativistic limit. This has, of course, been the conventional view of "non-relativistic" physics for centuries, long before Einstein came up with his theory of relativity in 1905 [103]. In this chapter, we take this approach and set up Galilean hydrodynamics starting from the Galilean symmetry algebra.

To approach Galilean symmetries in a tractable manner, we use the framework of null fluids that we formulated in a series of papers [2–5]. Null fluids are a one-higher dimensional embedding of Galilean fluids, which can be seen as anisotropic relativistic fluids in their own right. Given our handle on relativistic hydrodynamics from chapter 2, they provide an alternate and more natural framework to study Galilean fluids. In section 1.1.3 we motivated null fluids as an emergent structure starting from Galilean fluids. In this chapter, however, we introduce null hydrodynamics from a relativistic standpoint, drawing a comparison with relativistic hydrodynamics discussed in chapter 2 from time to time. Towards the end of the chapter, we return to how these results can be converted to the conventional Galilean hydrodynamics language. For a more detailed discussion on the top-down approach to covariantly organise Galilean symmetries into a higher dimensional structure, see [82].

4.1 Galilean field theories and null backgrounds

4.1.1 From Poincaré to Galilean algebra

Let us start at the very beginning—symmetries. Galilean field theories are known (or defined) to transform covariantly under the action of the so-called *Galilean algebra*.² The

¹For example, Maxwell's electromagnetism is known to have more than one non-relativistic limits [102].

²To be precise, this is actually a central extension of the Galilean algebra, sometimes known as the *Bargmann algebra*. Galilean algebra sits inside the Bargmann algebra as a special case with M = 0.

generators of this algebra are given as

[

Continuity:
$$M$$
, Time-translation: H Translations: P_a ,
Galilean boosts: B_a , Rotations: J_{ab} . (4.1a)

The associated conserved quantities are the mass, energy, momenta, centre of mass velocity,³ and angular momenta respectively. Here the indices a, b, \ldots run over the d-1 flat spatial coordinates. For the familiar universe we live in, d-1 = 3, but it is useful to keep d arbitrary to allow for the study of non-relativistic systems which are effectively confined to a thin film (d-1=2) or a chain (d-1=1). On flat space, these generators have the following non-vanishing commutation relations

$$[B_a, H] = iP_a, \qquad [B_a, P_b] = i\delta_{ab} M,$$

$$[J_{ab}, P_c] = i \left(\delta_{ac} P_b - \delta_{bc} P_a\right), \qquad [J_{ab}, B_c] = i \left(\delta_{ac} B_b - \delta_{bc} B_a\right),$$

$$[J_{ab}, J_{cd}] = i \left(\delta_{ac} J_{bd} - \delta_{ad} J_{bc} - \delta_{bc} J_{ad} + \delta_{bd} J_{ac}\right), \qquad (4.1b)$$

where δ_{ab} is the Kronecker delta. If this algebra is not familiar, it would help to consider it for a minute. Starting at the end, the rotation generators J_{ab} in the last line satisfy the standard SO(d-1) algebra. Furthermore, the commutators in the second line merely state the fact that the translation and boost generators P_a and B_a transform as SO(d-1) vectors. Since J_{ab} commutes with the remaining generators M and H, they transform as SO(d-1) scalars. We also see that when we commute a boost B_a with a time translation H, we end up generating a spatial translation. The characteristic feature of the Galilean algebra is the boost-translation commutator, which measures the mass of a state. This is the only place where M appears in the algebra and is therefore central.

Let us leave the Galilean algebra here for now and consider a (d + 1)-dimensional Poincaré algebra with generators

Spacetime translations:
$$P_A$$
, Lorentz transformations: J_{AB} . (4.2a)

Here the indices A, B, \ldots run over d + 1 higher dimensional coordinates. They satisfy the usual commutation relations

$$[P_A, P_B] = 0, \qquad [J_{AB}, P_C] = i (\eta_{AC} P_B - \eta_{BC} P_A),$$

$$J_{AB}, J_{CD}] = i (\eta_{AC} J_{BD} - \eta_{AD} J_{BC} - \eta_{BC} J_{AD} + \eta_{BD} J_{AC}), \qquad (4.2b)$$

where η_{AB} is the pseudo-Riemannian flat Minkowski metric. The generators J_{AB} naturally span a Lorentz algebra SO(d, 1). The spacetime translation generators P_A are SO(d, 1) vectors and mutually commute. Let us choose a set of null coordinates $(x^A) = (x^-, x^+, x^a)$ such that $\eta_{++} = \eta_{--} = \eta_{+a} = \eta_{-a} = 0$, $\eta_{+-} = -1$, and $\eta_{ab} = \delta_{ab}$. In this coordinate system, consider a subset of the Poincaré generators, all of which commute with the null

³This conserved quantity is surprisingly much less talked about. The point particle version of this "conservation law" is that the centre of mass of a collection of particles moves at a constant velocity in any inertial reference frame.

momenta P_{-} . They are given by

$$M \equiv -P_{-}, \qquad H \equiv -P_{+}, \qquad P_{a}, \qquad B_{a} \equiv -M_{a-}, \qquad M_{ab}. \tag{4.3}$$

In fact, the only generators that do not make the cut are M_{A+} . We can check that these generators are closed under the commutation relations (4.2b), and exactly span the *d*dimensional Galilean algebra (4.1). We, therefore, see that a *d*-dimensional Galilean algebra sits as a subalgebra in a (d+1)-dimensional Poincaré algebra, with one of the null momenta acting as the central mass generator. See [76] for an extensive review of this construction, and an extension to the Schrödinger algebra—Galilean analogue of a conformal algebra appearing naturally in non-relativistic holography [104]—arising from the null reduction of a (d+1)-dimensional relativistic conformal algebra.

This is rather convenient, as instead of starting from a d-dimensional relativistic theory and taking a $c \to \infty$ limit, or trying to write down a d-dimensional Galilean theory directly based on symmetries, we can start with a (d+1)-dimensional relativistic theory and reduce it over a light cone (introduce a null Killing vector) to get a Galilean theory. This is given the name null reduction [73–75] in the literature, also known as light-cone reduction or *discrete light-cone quantisation.* It has some obvious benefits over the other two approaches. We do not need to take a limit, so the prescription is perfectly well defined and unique. Also, we know how to deal with relativistic field theories reasonably well, which we can directly use to our advantage without having to deal with the technicalities of the much less understood Galilean field theories. The idea of null reduction has been used readily in the literature to reproduce known results and to get new insights into non-relativistic physics. Perhaps the most important of these results, at least in the current context, has been to reproduce Newton-Cartan geometries [77, 78] starting from a Bargmann structure (relativistic manifold carrying a covariantly constant null Killing vector) in one higher dimension; see e.g. [105–109]. Newton-Cartan geometries are a covariant representation of spacetime backgrounds which respect Galilean isometries. They are quite useful when describing Galilean physics as they treat space and time coordinates at the same footing, however, Galilean boost symmetry is not manifest in this formalism. As we go ahead, we provide a self-contained review of the aspects of it that we need.

4.1.2 Null backgrounds

As we did for relativistic hydrodynamics in section 2.1, let us start by setting up (d + 1)dimensional *null backgrounds* to which our null fluids are coupled. Roughly speaking, these are extensions of the Bargmann structures we mentioned above, to allow for non-trivial torsion and background gauge fields, in a way that they precisely capture the respective *d*-dimensional Galilean structure. Let the (d + 1)-dimensional theory we are interested in be living on a spacetime manifold \mathcal{M} with coordinates denoted by the indices M, N, \ldots In this context, the indices A, B, \ldots can be seen as coordinates on the frame bundle $F\mathcal{M}$ of \mathcal{M} . Given our Poincaré symmetries, Noether's theorem postulates the existence of a set of associated conserved currents

Energy-momentum tensor:
$$T^{M}_{A}$$
, Spin current: Σ^{MA}_{B} . (4.4a)

For completeness, let us also introduce a Lie group G of internal symmetry which our theory of interest might enjoy. We call the associated Lie algebra \mathfrak{g} , which is endowed with a Lie bracket denoted by $[\circ, \circ]$ and a positive semi-definite inner product denoted by $\circ \circ \circ$. The associated \mathfrak{g} valued conserved current is taken to be

Charge current:
$$J^M$$
. (4.4b)

To probe these Noether currents, it is convenient to couple the theory to a set of relativistic background sources on \mathcal{M} , one corresponding to each current,

Vielbein:
$$e^{A}_{\ M}$$
, Spin connection: $C^{A}_{\ MB}$,
Gauge connection: A_{M} . (4.5)

This is the exact same structure that we introduced in section 2.1 to study relativistic hydrodynamics. $e^{A}_{\ M}$ is a local frame field which furnishes an invertible map between the tangent bundle and the frame bundle. The spin connection $C^{A}_{\ MB}$ and the gauge connection A_{M} are one-form gauge fields valued in $\mathfrak{so}(d, 1)$ and \mathfrak{g} respectively. Along with an affine connection

$$\Gamma^{R}_{MS} \equiv e^{R}_{A} \left(\partial_{M} e^{A}_{S} + e^{B}_{S} C^{A}_{MB} \right), \qquad (4.6)$$

they define a covariant derivative operator D_M on \mathcal{M} . Under an infinitesimal local Poincaré and G transformation parametrised by $\mathfrak{X} = (\chi^M, \Lambda_{\chi}^{\Sigma}, \Lambda_{\chi})$, the variation of the background fields (4.5) is given by eq. (2.16) of our relativistic discussion.⁴

Recall that the Poincaré generator M_{A+} did not commute with the null momenta P_{-} and hence got left out when performing a null reduction to obtain the Galilean algebra. It follows that the respective Noether current Σ^{MA}_{+} does not have any Galilean interpretation either. It holds to reason, therefore, that the associated background source C^{A}_{-M-} must also be switched off for consistency so that these "unphysical" relativistic currents are not probed. With a little bit of algebra, we can show that this implies

$$C^{A}_{\ M} = e^{A}_{\ N} \left(\partial_{M} V^{N} + \Gamma^{N}_{\ MR} V^{N} \right) = e^{A}_{\ N} \mathcal{D}_{M} V^{N} = 0, \tag{4.7}$$

where $V^{M} = e_{-}^{M}$ is a null vector field. Furthermore, we should require that the background sources which do have a Galilean interpretation are left invariant under a local P_{-} transformation. Since P_{-} acts on the background by a Lie shift along V^{M} , it equivalently implies that V^{M} acts as an isometry on all the background fields. These are the defining features of the so-called *Bargmann structures*—spacetimes with a covariantly constant null isometry—which have been shown to be equivalent to the lower dimensional Newton-Cartan backgrounds, at least in the absence of an independent spin connection and gauge field.

⁴All the results from section 2.1 can be directly imported here by changing $d \rightarrow d + 1$ and switching to the uppercase indices.

Before generalising this statement to backgrounds with arbitrary connections, let us do a quick counting exercise. After imposing $e^{A}_{\ M}V^{M} = \delta^{A}_{\ -}$, the vielbein $e^{A}_{\ M}$ has d(d+1)independent components. d^2 of these make up a d-dimensional Galilean version of the vielbein, while the remaining d are identified with the components of a mass gauge field on a Newton-Cartan background. So this counting checks out. On the other hand, the spin connection C^{A}_{MB} , which is antisymmetric in $A \leftrightarrow B$ and satisfies $C^{A}_{M-} = 0$, has d(d-1)/2independent components for each of the d+1 values that the spacetime index M can take. In contrast, the d-dimensional Galilean spin connections also have d(d-1)/2 components, but only for d distinct spacetime indices. The same goes for the gauge connection A_M as well, which has d + 1 g-valued components as opposed to d expected in a d-dimensional Galilean theory. To have an exact mapping, therefore, we must somehow get rid of these extra components. Due to the presence of a preferred vector field V^{M} in the background, the only such components we can eliminate in a Poincaré invariant manner are $V^{\scriptscriptstyle M}C^{\scriptscriptstyle A}_{\scriptscriptstyle \ MB}$ and $V^{M}A_{M}$. They are not gauge-invariant, however, so we cannot merely set them to zero. Therefore, we introduce some arbitrary symmetry parameters $\Lambda_V^{\Sigma} \in \mathfrak{so}(d, 1)$ and $\Lambda_V \in \mathfrak{g}$ on the background and require

$$V^{M}C^{A}_{\ MB} + \Lambda^{\Sigma A}_{V\ B} = V^{M}A_{M} + \Lambda_{V} = 0, \qquad (4.8)$$

This eliminates the extra d(d-1)/2 components from the spin connection and a g-valued component from the gauge connection. We dub this requirement as the "compatibility of null isometry", and call the Bargmann structures that respect it to be *null backgrounds*. It is clear that these null backgrounds are in exact correspondence with the Newton-Cartan backgrounds, both in terms of the field content and symmetries, and as an added benefit have all the Galilean symmetries manifest in disguise of a Poincaré invariant structure.

For later convenience, let us recollect the definition of null backgrounds. We call a manifold \mathcal{M} with relativistic background sources $(e^{A}_{M}, C^{A}_{MB}, A_{M})$ to be a null background if it admits a preferred set of symmetry data $\mathcal{V} = (V^{M}, \Lambda^{\Sigma}_{V}, \Lambda_{V})$ such that

- 1. Action of \mathcal{V} is an isometry: $\delta_{\mathcal{V}} e^{A}{}_{M} = \delta_{\mathcal{V}} C^{A}{}_{MB} = \delta_{\mathcal{V}} A_{M} = 0$,
- 2. $V^{\scriptscriptstyle M}$ is null: $V^{\scriptscriptstyle M}V_{\scriptscriptstyle M}=0,$
- 3. $V^{\scriptscriptstyle M}$ is covariantly constant: $D_{\scriptscriptstyle M}V^{\scriptscriptstyle N}=0$, and
- 4. \mathcal{V} is compatible: $V^{\scriptscriptstyle M}A_{\scriptscriptstyle M} + \Lambda_{\scriptscriptstyle V} = 0, \ V^{\scriptscriptstyle M}C^{\scriptscriptstyle A}{}_{\scriptscriptstyle MB} + \Lambda_{\scriptscriptstyle V}^{\scriptscriptstyle \Sigma A}{}_{\scriptscriptstyle B} = 0.$

Existence of the null isometry \mathcal{V} leads to some nice features on null backgrounds. Firstly, contraction of the spacetime indices of the Riemann curvature tensor $R_{_{MN}}{}^{_{A}}{}_{_{B}}$, field strength $F_{_{MN}}$, and Cartan torsion tensor $T^{_{A}}{}_{_{MN}}$ (defined in eq. (2.15)) with $V^{_{M}}$ vanishes

$$V^{M} T^{A}_{\ MN} = V^{M} F_{MN} = V^{M} R^{A}_{MN \ B} = 0.$$
(4.9a)

These follow trivially from the symmetry variations in eq. (2.16) upon substituting \mathcal{V} for \mathfrak{X} . Furthermore, contraction of the SO(d, 1) indices of $R_{_{MN}}{}^{_{A}}{}_{_{B}}$ and $T^{_{A}}{}_{_{MN}}$ with $V^{_{A}}$ is also

determined to be

$$R_{_{MN}}{}^{_{A}}{}_{_{B}}V_{_{A}} = 0, \qquad T_{_{MN}}{}^{_{A}}V_{_{A}} = 2\partial_{[_{M}}V_{_{N}]} \equiv H_{_{MN}}.$$
 (4.9b)

These can also be easily derived using $D_{[M}D_{N]}V^{R} = D_{[M}V_{N]} = 0$. The presence of these relations is quite natural; they ensure that the respective quantities have the correct number of independent components as we would expect in a *d* dimensional Galilean background. Finally, we note that if φ is an arbitrary tensor transforming in some well behaved representation of diff $\times \mathfrak{so}(d, 1) \times \mathfrak{g}$, then

$$\delta_{\mathcal{V}}\varphi = V^{\scriptscriptstyle M} \mathcal{D}_{\scriptscriptstyle M}\varphi. \tag{4.10}$$

The easiest way to convince oneself of its validity is to see it work for some specific examples of φ . It essentially works because of the compatibility requirement (condition 4) imposed on the connections on a null background, and would not be generically true on a Bargmann structure. For most of this work, we are dealing with tensor structures which respect the null isometry \mathcal{V} , i.e. $\delta_{\mathcal{V}}\varphi = 0$. Through eq. (4.10), this is equivalent to the requirement that φ be covariantly constant along V^M .

4.1.3 Ward identities

Parallel to our relativistic discussion in section 2.1.3, we would like to draw some generic conclusions about the physical theories coupled to null backgrounds. Let the theory in question be described by some dynamical fields φ^I with associated equations of motion $\mathcal{E}_I \approx 0$, and some (effective) action $S[e^A_{\ M}, C^A_{\ MB}, A_M, \mathcal{V}; \varphi^I]$. To be consistent with our null backgrounds setup, we require the dynamical fields to respect the null isometry \mathcal{V} , i.e. $\delta_{\mathcal{V}}\varphi^I = V^M D_M \varphi^I = 0$. This ensures that the fields φ^I only have a dependence on the d spacetime coordinates, which is what we expect for a Galilean theory. Consequently, all the observables in the theory, especially the conserved currents, are covariantly constant along V^M .

Under an infinitesimal symmetry transformation \mathcal{X} , demanding the variation of the action $\delta_{\mathcal{X}}S$ to be zero, modulo anomalies, leads to a set of identities similar to eq. (2.19),

$$\underline{D}_{M}T^{M}{}_{A} = e_{A}{}^{N} \left(T^{B}{}_{NM}T^{M}{}_{B} + R_{NM}{}^{C}{}_{B}\Sigma^{MB}{}_{C} + F_{NM} \cdot J^{M} \right) + \mathcal{O}_{A}^{I}\mathcal{E}_{I},$$

$$\underline{D}_{M}\Sigma^{MAB} = T^{[BA]} + \Sigma^{\perp AB}_{H} + \mathcal{O}^{IAB}\mathcal{E}_{I},$$

$$\underline{D}_{M}J^{M} = J^{\perp}_{H} + \mathcal{O}^{I}\mathcal{E}_{I},$$
(4.11)

where $\underline{\mathbf{D}}_{M} = \mathbf{D}_{M} + e_{A}{}^{N}\mathbf{T}^{A}{}_{MN}$. The Hall currents $\Sigma_{\mathbf{H}}^{\perp A}{}_{B}$ and $\mathbf{J}_{\mathbf{H}}^{\perp}$ appearing here characterise the possible anomalies; they are considered in detail in the next subsection. On-shell, when $\mathcal{E}_{I} \approx 0$, these identities reduce to the Noether conservation laws for various currents

$$\underline{D}_{M}T^{M}{}_{A} \approx e_{A}{}^{N} \left(T^{B}{}_{NM}T^{M}{}_{B} + R_{NM}{}^{C}{}_{B}\Sigma^{MB}{}_{C} + F_{NM} \cdot J^{M} \right),$$

$$\underline{D}_{M}\Sigma^{MAB} \approx T^{[BA]} + \Sigma_{\mathrm{H}}^{\perp AB},$$

$$\underline{D}_{M}J^{M} \approx \mathrm{J}_{\mathrm{H}}^{\perp}.$$
(4.12)

These will be the starting point for our hydrodynamic discussion later.

Consider the following deformation of the Noether currents:

$$T^{\scriptscriptstyle M}_{\scriptscriptstyle A} \to T^{\scriptscriptstyle M}_{\scriptscriptstyle A} + V^{\scriptscriptstyle M}\theta_{1A}, \quad \Sigma^{\scriptscriptstyle MAB} \to \Sigma^{\scriptscriptstyle MAB} + V^{\scriptscriptstyle M}\theta_2^{\scriptscriptstyle [AB]} - \theta_3^{\scriptscriptstyle M[A}V^{\scriptscriptstyle B]}, \quad J^{\scriptscriptstyle M} \to J^{\scriptscriptstyle M} + V^{\scriptscriptstyle M}\theta_4, \quad (4.13)$$

where θ 's are some arbitrary tensor structures. The energy-momentum and charge components in eqs. (4.11) and (4.12) remain invariant, while the spin equation changes by a term

$$\left(\theta_1^{[A} + \underline{\mathbf{D}}_M \theta_3^{M[A}\right) V^{B]}.$$
(4.14)

This term only affects the conservation laws for the spin current components Σ^{MA}_+ which, if we recall, do not show up in the lower dimensional Galilean spectrum. Therefore, as far as the Galilean currents are concerned, shifts given in eq. (4.13) are mere redundancies in the higher dimensional Noether currents. They are not really surprising either; they are merely illustrating the fact that a (d + 1)-dimensional relativistic theory has many more observables compared to a d dimensional Galilean theory. There could be a family of relativistic theories that would give rise to the same Galilean theory upon null reduction. Ideally, we would like to be able to fix these redundancies once and for all by making some suitable choice, however there is no "(d + 1)-dimensional covariant" choice we can make at this point. The procedure for projecting out the unphysical degrees of freedom is essentially the null reduction prescription which leads to Newton-Cartan geometries, as we discuss in section 4.2. When working in the null background formalism while discussing Galilean hydrodynamics, we let these redundancies be, to enjoy working in a manifestly covariant formalism of Galilean physics.

4.1.4 Galilean anomalies

In this subsection, we outline the anomaly inflow mechanism for Galilean field theories and use it to explicitly compute the Hall currents appearing in the conservation laws eq. (4.12). These results were originally discussed in our work in [5], which we directly import here. Similar to our relativistic treatment in section 2.1.4, we introduce a (d + 2)-dimensional bulk manifold \mathcal{B} , on whose boundary our physical manifold \mathcal{M} lives, except that \mathcal{B} is also required to be a null background with compatible null isometry \mathcal{V} . The indices on \mathcal{B} are denoted by a hat. At the boundary, we keep our physical theory with generating functional $W_{\mathcal{M}}$ while in the bulk we introduce a generating functional $W_{\mathcal{B}}$ given by

$$W_{\mathcal{B}} = \int_{\mathcal{B}} \boldsymbol{I}.$$
 (4.15)

The full generating functional $W = W_{\mathcal{M}} + W_{\mathcal{B}}$ is required to be symmetry-invariant. For this to happen, the anomaly polynomial of the theory $\mathcal{P} = d\mathbf{I}$ should be closed, invariant under all the symmetries, and should not be expressible as the exterior derivative of a symmetry-invariant form.

Now comes the interesting part. On even dimensional (d + 1 = 2n) null backgrounds, the allowed anomaly polynomial would take the usual Chern-Simons structure of relativistic theories $\mathcal{P} = \mathcal{P}_{\text{CS}}$, which is made out of the Chern classes of \mathbf{F} and the Pontryagin classes of \mathbf{R} . Note however that neither of \mathbf{F} or \mathbf{R} has a component along $V^{\scriptscriptstyle M}$, hence \mathcal{P}_{CS} is identically zero. This suggests that we cannot get anomalies in an even dimensional null theory, and hence odd dimensional Galilean field theories are anomaly-free. Since odd dimensional relativistic theories are anomaly free as well, this is in fact physically sensible. If we had not required the compatibility of our background connections with the null isometry, $V^{\scriptscriptstyle M}F_{\scriptscriptstyle MN}$ and $V^{\scriptscriptstyle M}R_{\scriptscriptstyle MN}{}^{\scriptscriptstyle A}{}_{\scriptscriptstyle B}$ would not be zero, and we would end up with anomalous odd-dimensional Galilean field theories. This was the case, for example, for the Galilean anomalies found in [110].

The problem is that if we now shift our attention to odd dimensional (d+1 = 2n+1) null backgrounds, there is no Chern-Simons anomaly polynomial to start with. Consequently, our even dimensional Galilean theories are rendered anomaly-free as well. We presented a resolution to this problem in [3, 5], which we now outline. Let us introduce a vector field v^{M} such that $v^{M}v_{M} = 0$ and $v^{M}V_{M} = -1$, and define $\boldsymbol{v} = v_{M}dx^{M}$. With the help of this, we can write down an odd-rank anomaly polynomial

$$\boldsymbol{\mathcal{P}} = \boldsymbol{v} \wedge \boldsymbol{\mathcal{P}}_{\mathrm{CS}}.\tag{4.16}$$

Although this expression makes explicit reference to v_M , one can show that it is invariant under its arbitrary redefinition $v_M \to v_M + \delta v_M$. This follows from the fact that the change δv_M does not have any component along V^M , due to the normalisation property $\delta(v_M V^M) = V^M \delta v_M = 0$. We can convince ourselves that after the introduction of v^M , there are no more terms which can be written in the anomaly polynomial. Returning to the generating functional integrand \mathbf{I} , it can only be defined in the so-called *transverse gauge* where $\Lambda_V^{\Sigma} = \Lambda_V = 0$ leading to $V^M C^A_{\ MB} = V^M A_M = 0$. In this gauge, $\mathbf{I} = -\mathbf{v} \wedge \mathbf{I}_{\text{CS}}$. Taking a differential we obtain

$$d\boldsymbol{I} = \boldsymbol{\mathcal{P}} - d\boldsymbol{v} \wedge \boldsymbol{I}_{\rm CS} = \boldsymbol{\mathcal{P}}.$$
(4.17)

The second term in the middle expression is zero in the transverse gauge, as it does not have any component along V^{M} . Hence, we recover the proposed anomaly polynomial.

Having our anomaly polynomial in place, the rest of the story is an exact parallel of section 2.1.4. In particular, we can use eq. (2.25) to compute the Galilean *Hall currents* as

$$\star_{(d+2)} \boldsymbol{\Sigma}_{\mathrm{H}} = \boldsymbol{v} \wedge \frac{\partial \boldsymbol{\mathcal{P}}_{\mathrm{CS}}}{\partial \boldsymbol{R}}, \qquad \star_{(d+2)} \mathbf{J}_{\mathrm{H}} = \boldsymbol{v} \wedge \frac{\partial \boldsymbol{\mathcal{P}}_{\mathrm{CS}}}{\partial \boldsymbol{F}}.$$
(4.18)

We can also verify that $\Sigma_{\text{H}}{}^{A}{}_{B}V^{B} = 0$, as the SO(d, 1) indices of $\Sigma_{\text{H}}{}^{A}{}_{B}$ come from $\mathbf{R}^{A}{}_{B}$, which have a zero contraction with V^{B} .

This finishes our discussion of (d + 1)-dimensional null backgrounds. We have discussed their structure, symmetries, conservation laws, and equivalence to the lower dimensional Galilean backgrounds. In the next section, we illustrate their equivalence to Newton-Cartan backgrounds in *d*-dimensions.

4.2 Null reduction to Newton-Cartan backgrounds

Over the past decade, we have learned that the correct way to probe the conserved currents of a Galilean field theory is to couple it to a version of (torsional) Newton-Cartan backgrounds, in a way that keeps all the symmetries manifest. These backgrounds were first introduced in 1923 in a series of seminal papers by Élie Cartan [77, 78], to describe a covariant framework for Newtonian gravity. Since then, they have been continually developed and extended upon, particularly in the context of covariant Newtonian gravity and coupling to Galilean field theories; see e.g. [79–82, 111–121] and also [122–126] for some related topics. Alongside, it has also been known that Newton-Cartan backgrounds can be obtained via reducing a relativistic background along a null isometry [105–109]. In the following, we start from our null backgrounds, discussed in the previous section, and establish their equivalence to torsional Newton-Cartan backgrounds. This also provides us with a natural null reduction prescription for (d+1)-dimensional field theories coupled to null backgrounds down to d-dimensional Galilean field theories. We provide a self-contained review of the aspects of Newton-Cartan backgrounds that we require for our subsequent discussion of Galilean hydrodynamics. For more details, we recommend checking out the PhD thesis of Michael Geracie [127], where he discusses the most generic torsional Newton-Cartan geometries and their coupling to Galilean field theories.

Let us consider a foliation of the spacetime manifold as $\mathcal{M} = S^1 \times \mathcal{N}$, where we have compactified the $V^{\scriptscriptstyle M}$ direction into an infinitesimal circle S^1 . The resultant *d*-dimensional null hypersurface \mathcal{N} represents a Galilean (Newton-Cartan) spacetime. As \mathcal{N} is a null hypersurface, to perform such a foliation uniquely we also require an arbitrary null field $v^{\scriptscriptstyle M}$ doubly normalised as $v^{\scriptscriptstyle M}v_{\scriptscriptstyle M} = 0$ and $v^{\scriptscriptstyle M}V_{\scriptscriptstyle M} = -1$. Since we are introducing this vector field by hand just to perform the reduction, the actual physical results are invariant under an arbitrary shift of $v^{\scriptscriptstyle M}$. This choice can be understood as providing a "Galilean frame of reference". When working with hydrodynamics, the null fluid velocity $u^{\scriptscriptstyle M}$ preferentially provides such a frame of reference which we refer to as the "fluid frame". For now however, we stick to an abstract $v^{\scriptscriptstyle M}$ for the sake of generality. Formally, we define null reduction as this choice of $v^{\scriptscriptstyle M}$ and subsequent decomposition of the (d + 1)-dimensional theory into an effective *d*-dimensional one.

Let us first choose a basis on \mathcal{M} as $(x^M) = (x^{\sim}, x^{\mu})$ and partially fix the SO(d, 1) and G-transformation symmetries to set $\mathcal{V} = (\partial_{\sim}, 0, 0)$, i.e. $V^M = \delta^M_{\sim}, \Lambda^{\Sigma A}_V = 0$. The condition $V^M D_M = 0$ simply becomes $D_{\sim} = 0$. At this point, we can already decompose the conservation laws as

$$\underline{D}_{\mu}T^{\mu}{}_{A} \approx e_{A}^{\nu} \left(T^{B}{}_{\nu\mu}T^{\mu}{}_{B} + R_{\nu\mu}{}^{C}{}_{B}\Sigma^{\mu}{}_{C}^{B} + F_{\nu\mu} \cdot J^{\mu} \right),$$

$$\underline{D}_{\mu}\Sigma^{\mu}{}_{B} \approx \frac{1}{2} \left(e_{B\mu}T^{\mu}{}_{A} - e^{A}{}_{\mu}T^{\mu}{}_{B} \right) + \Sigma^{\perp}{}_{\mathrm{H}}{}^{A}{}_{B},$$

$$\underline{D}_{\mu}J^{\mu} \approx J^{\perp}_{\mathrm{H}},$$
(4.19)

up to the ambiguities given in eq. (4.13). Here the background structure can be understood as a *d*-dimensional manifold \mathcal{N} (with indices μ, ν, \ldots) with an "extended" (*d*+1)-dimensional frame bundle (with indices A, B, \ldots). Note that we have not introduced the field v^M yet, so the description is still fully covariant. As we discussed in an appendix of [5], the fact that the Galilean physics can be packaged into this "extended space" structure was first realised, for theories without anomalies or spin currents, by [82], where the authors arrived at this bottom-up starting from the generic *d*-dimensional Newton-Cartan backgrounds. See [127] for a review. The novelty in the null-background formalism is that the symmetries are nicely arranged in terms of a Poincaré invariant structure.

4.2.1 Newton-Cartan backgrounds

Let us now introduce the Galilean frame velocity field v^{M} . We pick a basis on the frame bundle $(x^{A}) = (x^{-}, x^{+}, x^{a})$ such that $V^{A} = \delta^{A}_{-}$ and $v^{A} = \delta^{A}_{+}$. To this end, we can choose a specific representation of the Minkowskian metric η_{AB} and decompose the background fields as

$$\eta_{AB} = \begin{pmatrix} 0 & -1 & 0 \\ -1 & 0 & 0 \\ 0 & 0 & \delta_{ab} \end{pmatrix}, \quad e^{A}_{\ M} = \begin{pmatrix} 1 & -b_{\mu} \\ 0 & n_{\mu} \\ 0 & e^{a}_{\mu} \end{pmatrix},$$
$$C^{A}_{\ \sim B} = 0, \qquad C^{A}_{\ \mu B} = \begin{pmatrix} 0 & 0 & c_{\mu b} \\ 0 & 0 & 0 \\ 0 & c_{\mu}^{\ a} & C^{a}_{\ \mu b} \end{pmatrix}, \qquad A_{M} = \begin{pmatrix} 0 \\ A_{\mu} \end{pmatrix}, \quad (4.20)$$

such that $n_{\mu}v^{\mu} = 1$ and $e^{a}_{\ \mu}v^{\mu} = 0$. The tensor structures appearing in this decomposition define a Newton-Cartan background

Mass gauge field:
$$b_{\mu}$$
, Clock form: n_{μ} Spatial vielbein: $e^{a}_{\ \mu}$,
Temporal spin connection: $c_{\mu}{}^{a}$, Spatial spin connection: $C^{a}_{\ \mu b}$,
Gauge connection: A_{μ} . (4.21)

They serve as the background sources associated with the Galilean symmetry generators given in eq. (4.1). n_{μ} serves as the "time-vielbein" on Galilean backgrounds, while $e^{a}{}_{\mu}$ is the spatial vielbein. b_{μ} is an external U(1) gauge field associated with the conservation of mass, while the connections $A_{\mu} \in \mathfrak{g}$ and $C^{a}{}_{\mu b} \in \mathfrak{so}(d-1)$ are associated with *G*-transformations and spatial rotations respectively. Finally, $c_{\mu}{}^{a}$ couples to the Galilean boosts and can be interpreted as the acceleration of the possibly non-inertial lab frame. Using the vielbein e^{a}_{μ} , we can define an inverse vielbein $e^{a}_{\mu}{}^{\mu}$ via $e^{a}_{\mu}e^{\mu}_{b} = \delta^{a}_{b}$ and $e_{a}{}^{\mu}n_{\mu} = 0$. Together they furnish a resolution of identity

$$e_a{}^{\mu}e^a{}_{\nu} + v^{\mu}n_{\nu} = \delta^{\mu}{}_{\nu}, \qquad (4.22)$$

and can be used to decompose the higher dimensional inverse vielbein

$$e_{A}^{\ M} = \begin{pmatrix} 1 & 0 \\ b_{\nu}v^{\nu} & v^{\mu} \\ b_{\nu}e_{a}^{\ \nu} & e_{a}^{\ \mu} \end{pmatrix}.$$
 (4.23)

We can also define a degenerate spatial projection operator (also called spatial metric) via $h_{\mu\nu} = \delta_{ab}e^a{}_{\mu}e^b{}_{\nu}, h^{\mu\nu} = \delta^{ab}e_a{}^{\mu}e_b{}^{\nu}$, and $h^{\mu}{}_{\nu} = h^{\mu\sigma}h_{\sigma\nu} = e_a{}^{\mu}e^a{}_{\nu}$. They satisfy $h^{\mu\nu}n_{\mu} = h_{\mu\nu}v^{\nu} = 0$. Since there is no non-degenerate metric on \mathcal{N} , raising/lowering of μ, ν, \ldots indices is not permitted. However a, b, \ldots indices can be raised/lowered using the Kronecker delta δ_{ab} . On the other hand, the vielbeins $e^a{}_{\mu}$ and $e_a{}^{\mu}$ can be used to project the tensors on \mathcal{N} to tensors on $\mathbb{R}^{(d-1)}$, and in turn lift the tensors on $\mathbb{R}^{(d-1)}$ to "spatial tensors" on \mathcal{N} (whose n_{μ} or v^{μ} contraction is zero).

Decomposing the null background affine connection, we find that the only non-trivial components are

$$\Gamma^{R}_{\ \mu\nu} = \begin{pmatrix} 0 & c_{\mu}{}^{a}e_{a\nu} - \tilde{D}_{\mu}b_{\nu} \\ 0 & \Gamma^{\lambda}_{\ \mu\nu} = v^{\lambda}\partial_{\mu}n_{\nu} + e_{a}{}^{\lambda}\left(e^{b}{}_{\nu}C^{a}{}_{\mu b} + n_{\nu}c_{\mu}{}^{a} + \partial_{\mu}e^{a}{}_{\nu}\right) \end{pmatrix}.$$
 (4.24)

Here we have identified $\Gamma^{\lambda}_{\mu\nu}$ as the Newton-Cartan affine connection, which along with $C^{a}_{\mu b}$ and A_{μ} , defines a covariant derivative operator \tilde{D}_{μ} . We can check that $\Gamma^{\lambda}_{\mu\nu}$ is the unique affine connection that satisfies

$$\tilde{D}_{\mu}n_{\nu} = 0, \qquad \tilde{D}_{\mu}e^{a}_{\ \nu} + n_{\nu}c_{\mu}^{\ a} = 0 \iff e^{a}_{\ \nu}\tilde{D}_{\mu}v^{\nu} = c_{\mu}^{\ a}, \qquad \tilde{D}_{\mu}e^{\nu}_{a} = 0, \qquad (4.25)$$

which are the Galilean equivalent of the metric compatibility condition. From here, we see that $c_{\mu}{}^{a}$ is indeed the background frame acceleration. For later use, let us define the *globally inertial Galilean frames* as those for which $c_{\mu}{}^{a}$ identically vanishes. However, for a generically curved Newton-Cartan background, there is no guarantee that such a frame would exist.

Under the null background symmetry transformations derived from eq. (2.16), the Newton-Cartan background fields (4.21) transform as

$$\delta_{\chi}b_{\mu} = \pounds_{\chi}b_{\mu} + \partial_{\mu}\Lambda_{\chi}^{m} + \Lambda_{\chi a}^{\tau} e^{a}{}_{\mu}$$

$$\delta_{\chi}n_{\mu} = \pounds_{\chi}n_{\mu}$$

$$\delta_{\chi}e^{a}{}_{\mu} = \pounds_{\chi}e^{a}{}_{\mu} - \Lambda_{\chi}^{\sigma}{}_{b}e^{b}{}_{\mu} - \Lambda_{\chi}^{\tau}{}^{a}n_{\mu},$$

$$\delta_{\chi}c_{\mu}{}^{a} = \pounds_{\chi}c_{\mu}{}^{a} - \Lambda_{\chi}^{\sigma}{}_{b}c_{\mu}{}^{b} + \partial_{\mu}\Lambda_{\chi}^{\tau}{}^{a} + C^{a}{}_{\mu b}\Lambda_{\chi}^{\tau}{}^{b},$$

$$\delta_{\chi}C^{a}{}_{\mu b} = \pounds_{\chi}C^{a}{}_{\mu b} + \partial_{\mu}\Lambda_{\chi}^{\sigma}{}^{a}{}_{b} + [C_{\mu},\Lambda_{\chi}^{\sigma}]^{a}{}_{b},$$

$$\delta_{\chi}A_{\mu} = \pounds_{\chi}A_{\mu} + \partial_{\mu}\Lambda_{\chi} + [A_{\mu},\Lambda_{\chi}].$$
(4.26)

provided that we identify the Galilean symmetry parameters as

$$\mathfrak{X} = \left(\begin{array}{cc} \chi^{\mu} , & \Lambda_{\chi}^{\mathrm{m}} = -\chi^{\sim} , & \Lambda_{\chi}^{\tau \, a} = \Lambda_{\chi}^{\Sigma \, a} + , & \Lambda_{\chi}^{\sigma \, a} = \Lambda_{\chi}^{\Sigma \, a} + , & \Lambda_{\chi} \end{array} \right).$$
(4.27)

We can also work out the transformation properties of v^{μ} and $e_a^{\ \mu}$ as

$$\delta_{\mathfrak{X}}v^{\mu} = \pounds_{\mathfrak{X}}v^{\mu} + e_{a}{}^{\mu}\Lambda_{\mathfrak{X}}^{\tau a}, \qquad \delta_{\mathfrak{X}}e_{a}{}^{\mu} = \pounds_{\mathfrak{X}}e_{a}{}^{\mu} + \Lambda_{\mathfrak{X}}^{\sigma b}{}_{a}e_{b}{}^{\mu}.$$
(4.28)

Note that the diffeomorphisms along $V^{\scriptscriptstyle M}$ take the form of mass gauge transformations on the Newton-Cartan backgrounds, justifying the name "mass gauge field" for b_{μ} . On the other hand, $\Lambda_{\chi} \in \mathfrak{g}$ acts as a *G*-gauge transformation on the respective gauge field A_{μ} . The action of χ^{μ} is a spacetime diffeomorphism, i.e. it acts on all the background fields by a Lie drag, as we would expect. The action of $\Lambda_{\chi}^{\sigma} \in \mathfrak{so}(d-1)$ is also simple; it transforms $C^{a}_{\ \mu b}$ as a gauge field and rotates the SO(d-1) vectors $e^{a}_{\ \mu}$, $c_{\mu}^{\ a}$, and $e_{a}^{\ \mu}$ appropriately. Finally, we have the Galilean boost transformations parametrised by $\Lambda_{\chi}^{\tau a}$, which mix up the background fields non-trivially. They are also known as Milne boosts in the literature.

We can null reduce the null background torsion, curvature, and field strength into the Newton-Cartan language. The only surviving components are

$$F_{\mu\nu}, \qquad \mathbf{T}^{A}_{\ \mu\nu} = \begin{pmatrix} -B_{\mu\nu} \\ -H_{\mu\nu} \\ \mathbf{T}^{a}_{\ \mu\nu} \end{pmatrix}, \qquad R^{A}_{\mu\nu\ B} = \begin{pmatrix} 0 & 0 & -R_{\mu\nu b} \\ 0 & 0 & 0 \\ 0 & -R_{\mu\nu}^{\ a} & R^{\ a}_{\mu\nu\ b} \end{pmatrix}.$$
(4.29)

Here we have defined

Mass torsion:
$$B_{\mu\nu} = 2\tilde{D}_{[\mu}b_{\nu]} - \Omega_{\mu\nu}$$
, Spacetime torsion: $T^{\lambda}{}_{\mu\nu} = 2\Gamma^{\lambda}{}_{[\mu\nu]}$,
Spacetime curvature: $R^{\lambda}_{\mu\nu}{}_{\rho} = 2\left(\partial_{[\mu}\Gamma^{\lambda}{}_{\nu]\rho} + \Gamma^{\lambda}{}_{[\mu|\sigma}\Gamma^{\sigma}{}_{\nu]\rho}\right)$,
Field strength: $F_{\mu\nu} = 2\partial_{[\mu}A_{\nu]} + [A_{\mu}, A_{\nu}]$, (4.30)

where $\Omega_{\mu\nu} = 2e_{a[\nu}c_{\mu]}^{\ a}$ is the (boost-non-invariant) Galilean frame vorticity, and further

$$H_{\mu\nu} = -n_{\lambda} T^{\lambda}{}_{\mu\nu} = -2\partial_{[\mu}n_{\nu]}, \qquad T^{a}{}_{\mu\nu} = e^{a}{}_{\lambda} T^{\lambda}{}_{\mu\nu} = 2\partial_{[\mu}e^{a}{}_{\nu]} + 2c_{[\mu}{}^{a}n_{\nu]} + 2C^{a}{}_{[\mu b}e^{b}{}_{\nu]},$$
$$R_{\mu\nu}{}^{a} = -R_{\mu\nu}{}^{\lambda}{}_{\rho}e^{a}{}_{\lambda}v^{\rho} = -2\left(\partial_{[\mu}c_{\nu]}{}^{a} + C^{a}{}_{[\mu b}c_{\nu]}{}^{b}\right),$$
$$R_{\mu\nu}{}^{a}{}_{b} = R_{\mu\nu}{}^{\lambda}{}_{\rho}e^{a}{}_{\lambda}e^{\rho}{}_{b} = 2\left(\partial_{[\mu}C^{a}{}_{\nu]b} + C^{a}{}_{[\mu c}C^{c}{}_{\nu]b}\right).$$
(4.31)

Using these definitions, we can rewrite the Newton-Cartan affine connection given in eq. (4.24), into a more standard form found in the literature

$$\Gamma^{\lambda}_{\ \mu\nu} = v^{\lambda}\partial_{\mu}n_{\nu} + \frac{1}{2}h^{\lambda\sigma}\left(\partial_{\mu}h_{\sigma\nu} + \partial_{\nu}h_{\sigma\mu} - \partial_{\sigma}h_{\mu\nu}\right) + n_{(\mu}\Omega_{\nu)\sigma}h^{\lambda\sigma} + \frac{1}{2}\left(e^{\lambda}_{a}T^{a}_{\ \mu\nu} - 2e_{a(\nu}T^{a}_{\ \mu)\sigma}h^{\lambda\sigma}\right).$$
(4.32)

As far as we are aware, this was first written down in [82].⁵

Finally, using the (d+1)-dimensional volume form $\epsilon^{MN...}$, we can define the raised and

$$\Gamma^{\lambda}_{\ \mu\nu} = \left(v^{\lambda} - h^{\lambda\sigma}M_{\sigma}\right)\partial_{\mu}n_{\nu} + \frac{1}{2}h^{\lambda\sigma}\left(\partial_{\mu}\overline{h}_{\nu\sigma} + \partial_{\nu}\overline{h}_{\mu\sigma} - \partial_{\sigma}\overline{h}_{\mu\nu}\right),\tag{4.33}$$

where $\overline{h}_{\mu\nu} = h_{\mu\nu} + 2n_{(\mu}M_{\nu)}$. These backgrounds, termed as torsional Newton-Cartan geometries (TNC) in the literature [108, 128], appear naturally in the holography of Lifshitz spacetimes. A further restriction on $H_{\mu\nu}$ setting $h^{\mu\rho}h^{\nu\sigma}H_{\rho\sigma} = 0$ leads to the so called twistless torsional Newton-Cartan geometries (TTNC). If we were to switch off $H_{\mu\nu}$ entirely, we would recover the torsionless Newton-Cartan geometries.

⁵Let us say that our background contains a scalar field χ which transforms as $\delta_{\chi\chi} = -\Lambda_{\chi}^{\rm m}$. We can use it to define boost-invariant torsion tensors $B_{\mu\nu} = v^{\lambda}M_{\lambda}H_{\mu\nu}$ and $T^{\lambda}{}_{\mu\nu} = (h^{\lambda\sigma}M_{\sigma} - v^{\lambda})H_{\mu\nu}$ where $M_{\mu} = b_{\mu} + \partial_{\mu}\chi$. For these special class of Newton-Cartan backgrounds, the affine connection can be expressed as

lowered Newton-Cartan volume elements

$$\varepsilon^{\mu\nu\dots} = v_M \epsilon^{M\mu\nu\dots} = -\epsilon^{\sim\mu\nu\dots}, \qquad \varepsilon_{\mu\nu\dots} = V^M \epsilon_{M\mu\nu\dots} = \epsilon_{\sim\mu\nu\dots}. \tag{4.34}$$

Even though one of them involves an explicit mention of v^M , they are actually independent under an arbitrary frame redefinition. Note that since there is no raising/lowering operator on Newton-Cartan backgrounds, these two volume forms are independent. We can also define an associated Hodge duality operator \star , that uses $\varepsilon^{\mu\nu\dots}$ to map *m*-rank differential forms to (d-m)-rank contravariant forms, while conversely $\varepsilon_{\mu\nu\dots}$ to map *m*-rank contravariant forms to (d-m)-rank differential forms. It can be checked that $\star^2 = (-)^{m(d-m)}$ when acting on an *m*-rank differential or contravariant form.

4.2.2 Noether currents and conservation equations

Let us now move on to the Noether currents T^{M}_{A} , Σ^{MA}_{B} , and J^{M} . We start by recalling that there is a redundancy in the definition of these currents given by eq. (4.13), which allows us to eliminate some components. In the following decomposition, we have denoted these unphysical components by "×", leading to

$$T^{M}_{\ A} = \begin{pmatrix} \times & \times & \times \\ -\rho^{\mu} & -\epsilon^{\mu} & p^{\mu}_{\ a} \end{pmatrix}, \qquad J^{M} = \begin{pmatrix} \times \\ j^{\mu} \end{pmatrix},$$
$$\Sigma^{\sim A}_{\ B} = \times, \qquad \Sigma^{\mu A}_{\ B} = \begin{pmatrix} \times & 0 & \times \\ 0 & \times & -\tau^{\mu}_{\ b} \\ -\tau^{\mu a} & \times & \sigma^{\mu a}_{\ b} \end{pmatrix}.$$
(4.35)

From here we can identify the Noether currents associated with the Galilean symmetry generators given in (4.1)

Mass current:
$$\rho^{\mu}$$
,Energy current: ϵ^{μ} Momentum current: $p^{\mu}{}_{a}$,Temporal spin current: $\tau^{\mu}{}_{a}$,Spatial spin current: $\sigma^{\mu a}{}_{b}$,Charge current: j^{μ} .(4.36)

With these identifications, the physical components of the Noether conservation laws (4.12) can be converted into the respective conservation laws of a Galilean theory with spin current

$$\begin{split} & \text{Mass conservation (continuity):} \quad \underline{D}_{\mu}\rho^{\mu} \approx 0, \\ & \text{Energy conservation (time translation):} \quad \underline{\tilde{D}}_{\mu}\epsilon^{\mu} \approx -v^{\mu}f_{\mu} - p^{\mu}{}_{a}c_{\mu}{}^{a}, \\ & \text{Momentum conservation (spatial translations):} \quad \underline{\tilde{D}}_{\mu}p^{\mu}{}_{a} \approx e_{a}{}^{\mu}f_{\mu} - \rho^{\mu}c_{\mu a}, \\ & \text{Temporal spin conservation (Galilean boosts):} \quad \underline{\tilde{D}}_{\mu}\tau^{\mu}{}_{a} \approx \frac{1}{2}\left(n_{\mu}p^{\mu}{}_{a} - e_{a\mu}\rho^{\mu}\right), \\ & \text{Spatial spin conservation (rotations):} \quad \underline{\tilde{D}}_{\mu}\sigma^{\mu ab} \approx p^{\mu[a}e^{b]}{}_{\mu} - 2\tau^{\mu[a}c_{\mu}{}^{b]} + \sigma_{\mathrm{H}}^{\perp ab}, \\ & \text{Charge conservation (G-transformations):} \quad \underline{\tilde{D}}_{\mu}j^{\mu} \approx \mathrm{j}_{\mathrm{H}}^{\perp}. \end{split}$$

Here $\underline{\tilde{D}}_{\mu} = \tilde{D}_{\mu} + v^{\nu} H_{\nu\mu} - e_a^{\nu} T^a_{\nu\mu}$, while f_{μ} denotes the external Lorentz force due to the presence of background fields

$$f_{\mu} = H_{\mu\nu}\epsilon^{\nu} + B_{\mu\nu}\rho^{\nu} + T^{a}{}_{\mu\nu}p^{\nu}{}_{a} + R_{\nu\mu}{}^{a}\tau^{\nu}{}_{a} + R_{\mu\nu}{}^{a}{}_{b}\sigma^{\nu b}{}_{a} + F_{\mu\nu} \cdot j^{\mu}, \qquad (4.38)$$

which act as a Galilean energy and momentum source. The terms coupling to $c_{\mu a} = e_{a\nu} \underline{\tilde{D}}_{\mu} v^{\nu}$ in eq. (4.37) are due to the chosen Galilean frame v^{μ} not being globally inertial. Finally, we have the Galilean Hall currents $\sigma_{\rm H}^{\perp ab} = \Sigma_{\rm H}^{\perp ab}$ and $j_{\rm H}^{\perp} = J_{\rm H}^{\perp}$, which can be explicitly computed via the reduction of eq. (4.18) leading to

$$\mathbf{j}_{\mathrm{H}}^{\perp} = -\star \left(\frac{\partial \boldsymbol{\mathcal{P}}_{\mathrm{CS}}}{\partial \boldsymbol{F}}\right), \qquad \boldsymbol{\sigma}_{\mathrm{H}}^{\perp a}{}_{b} = -\star \left(\frac{\partial \boldsymbol{\mathcal{P}}_{\mathrm{CS}}}{\partial \boldsymbol{R}^{b}{}_{a}}\right). \tag{4.39}$$

Eq. (4.37) are the complete set of conservation equations for a Galilean invariant field theory coupled to a curved spacetime background. In particular, note the temporal spin conservation equation. It implies that in the absence of a temporal spin current, the mass flux $\rho_a = e_{a\mu}\rho^{\mu}$ must equal the momentum density $p_a = n_{\mu}p^{\mu}{}_a$ in a Galilean theory, which is commonplace in non-relativistic physics.

To get some intuition, it is helpful to see these conservation laws on a flat Galilean background, where all the technicalities of Newton-Cartan backgrounds drop out. To do this, let us choose a basis $(x^{\mu}) = (t, x^i)$ such that $n_{\mu} = \partial_{\mu} t$. We choose to work in a frame given by $v^{\mu} = \delta^{\mu}_{\ t}$, and set $C^{a}_{\ \mu b} = c_{\mu}^{\ a} = b_{\mu} = 0$, and $e^{a}_{\ \mu} = \delta^{a}_{\ \mu}$. Let us also choose the background gauge field A_{μ} to be Abelian and restrict to d = 4 for familiarity. In this simplified case, the conservation laws are given as

$$\partial_t \rho^t + \partial_i \rho^i = 0, \qquad \partial_t \epsilon^t + \partial_i \epsilon^i = E_i j^i, \qquad \partial_t p^t{}_k + \partial_i p^i{}_k = q E_k + \epsilon_{kij} j^i B^j,$$

$$\partial_t \tau^t{}_k + \partial_i \tau^i{}_k = \frac{1}{2} \left(\rho_k - p^t{}_k \right), \qquad \partial_t \sigma^{tjk} + \partial_i \sigma^{ijk} = p^{[kj]},$$

$$\partial_t q^t + \partial_i j^i = 0. \qquad (4.40)$$

Here we have defined the electric field $E_i = F_{it}$ and the magnetic field $B_i = \frac{1}{2} \epsilon_{ijk} F^{jk}$. The first and last equations are obviously the continuity equations for mass and charge respectively. The second equation is the continuity equation for energy, telling us that the rate of change of energy is proportional to the work done by the electric fields. The third equation is the Navier-Stokes equation for balancing force; note that the term on the right is merely the Lorentz force due to the electromagnetic fields. Finally, the two equations in the middle line are the conservation equations for the spin currents. For theories with no spin current, they merely tell us that the mass flux is same as the momentum density, and the stress tensor is symmetric.

4.2.3 Reference frame transformations

Recall that we had introduced an arbitrary vector field v^{M} to facilitate the null reduction. Since the relativistic parent theory does not depend on this choice of v^{M} , after all is said and done, we will be left with an additional v^{M} redefinition freedom in our Galilean theory. This is equivalent to the choice of Galilean reference frames. Let us see how various Galilean quantities defined above transform under this frame transformation: $v^{M} \rightarrow v^{M} + \psi^{M}$. Firstly, the normalisation conditions of v^{M} imply that

$$V^{M}\psi_{M} = 0, \qquad v^{M}\psi_{M} = -\frac{1}{2}\psi^{M}\psi_{M}.$$
 (4.41)

While we do this transformation, we would still like the tangent space vector v^{A} to be equal to δ^{A}_{+} . To achieve this, we need to follow this transformation by a tangent space rotation with matrix

$$O^{A}{}_{B} = \delta^{A}{}_{B} + \psi^{A}V_{B} - V^{A}\left(\psi_{B} + V^{A}V_{B}\frac{1}{2}\psi^{M}\psi_{M}\right) \in \mathrm{SO}(d,1).$$
(4.42)

One can check that $v^A \to O^A_{\ B}(v^B + \psi^B) = v^A$, while $\eta^{AB} \to O^A_{\ C}O^B_{\ D}\eta^{CD} = \eta^{AB}$ and $V^A \to O^A_{\ B}V^B = V^A$. On the other hand

$$e^{A}_{\ M} \to O^{A}_{\ B} e^{B}_{\ M} = e^{A}_{\ M} + \psi^{A} V_{M} - V^{A} \left(\psi_{M} - V_{M} \frac{1}{2} \psi^{R} \psi_{R}\right).$$
(4.43)

Choosing a basis and performing the null reduction, the conditions (4.41) reduce to

$$\psi_{\sim} = 0, \quad v^{\mu}\psi_{\mu} = -\frac{1}{2}h^{\mu\nu}\psi_{\mu}\psi_{\nu}, \qquad n_{\mu}\psi^{\mu} = 0, \quad -\psi^{\sim} + b_{\mu}\psi^{\mu} = -\frac{1}{2}h_{\mu\nu}\psi^{\mu}\psi^{\nu}.$$
(4.44)

Note that using the lowering operator and these identities we have

$$\psi_{\mu} = g_{\mu\nu}\psi^{\nu} = -n_{\mu}\left(\psi^{\sim} - b_{\nu}\psi^{\nu}\right) + h_{\mu\nu}\psi^{\nu} = -n_{\mu}\frac{1}{2}(h_{\rho\sigma}\psi^{\rho}\psi^{\sigma}) + h_{\mu\nu}\psi^{\nu}, \qquad (4.45)$$

therefore all the information in $\psi^{\scriptscriptstyle M}$ can be encoded into ψ^{μ} , which satisfies $n_{\mu}\psi^{\mu} = 0$. We find that n_{μ} , $h^{\mu\nu}$, and $e_a{}^{\mu}$ are invariant under this transformation, while

$$v^{\mu} \to v^{\mu} + \psi^{\mu}, \qquad b_{\mu} \to b_{\mu} + \psi_{\mu}, \qquad h_{\mu\nu} \to h_{\mu\nu} - 2n_{(\mu}\psi_{\nu)}, \qquad e^{a}{}_{\mu} \to e^{a}{}_{\mu} - \psi^{a}n_{\mu}.$$
 (4.46a)

Comparing these to eq. (4.26), we can convince ourselves that we are merely performing a finite Galilean (Milne) boost parametrised by $\Lambda_{\chi}^{\tau a} = \psi^{a}$. Among the connections, A_{μ} , $\Gamma^{\lambda}{}_{\mu\nu}$, and $C^{a}{}_{\mu b}$ are invariant, while $c_{\mu a} \rightarrow c_{\mu a} + \tilde{D}_{\mu}\psi_{a}$. The field strength $F_{\mu\nu}$, temporal torsion $H_{\mu\nu}$, and spatial curvature $R_{\mu\nu}{}^{a}{}_{b}$ are again invariant, while the remaining components transform as

$$T^{a}{}_{\mu\nu} \to T^{a}{}_{\mu\nu} + \psi^{a}H_{\mu\nu}, \qquad B_{\mu\nu} \to B_{\mu\nu} + \psi_{a}T^{a}{}_{\mu\nu} + H_{\mu\nu}\frac{1}{2}\psi^{a}\psi_{a}, R_{\mu\nu}{}^{a} \to R_{\mu\nu}{}^{a} - C^{a}{}_{\mu b}\psi^{b}.$$
(4.46b)

We can repeat the same procedure for Noether currents as well. We get that the mass current ρ^{μ} , charge current j^{μ} , and temporal spin current $\tau^{\mu a}$ are invariant, while the other

currents mix up according to

$$\epsilon^{\mu} \to \epsilon^{\mu} + \frac{1}{2} \rho^{\mu} \psi^{a} \psi_{a} - p^{\mu}{}_{a} \psi^{a}, \qquad p^{\mu}{}_{a} \to p^{\mu}{}_{a} - \rho^{\mu} \psi_{a},$$

$$\sigma^{\mu a}{}_{b} \to \sigma^{\mu a}{}_{b} + \tau^{\mu a} \psi_{b} - \psi^{a} \tau^{\mu}{}_{b}. \qquad (4.46c)$$

One can check that with these transformations, the Lorentz force f_{μ} defined in eq. (4.38) is rendered frame invariant.

Due to the non-trivial nature of these transformation rules, they are quite hard to implement in a purely *d*-dimensional description and have led much of the recent developments in the field of Newton-Cartan geometries. When working in the null background description however, this invariance is automatic because the field v^{M} was only introduced at the end for the purposes of reduction and is not present in the theory by itself.

4.2.4 The non-relativistic limit

Over these last two subsections, we have taken an axiomatic approach to set up Galilean invariant theories, their symmetries, background fields, and conservation equations. It is perhaps instructive to reflect back and consider how this structure could arise in a $c \to \infty$ limit of a relativistic theory. The discussion here is motivated from the work of [121]. Let us start with the d-dimensional Poincaré algebra (2.1). The first thing we should note is that we have one less generator than the Galilean algebra (4.1) that we are hoping to achieve via a non-relativistic limit. There is no analogue of the Galilean mass generator M on the relativistic side. It is quite natural because the relativistic theories do not distinguish between mass and energy; in a relativistic framework mass and energy are not independently conserved, but only their combination. To define a non-relativistic limit therefore, we must be provided with an emergent U(1) symmetry in the relativistic theory when probed at low energies. Typically, this would arise due to the emergence of certain particle number currents at low energies, which are conserved in the absence of enough energy to combine or split particles. Let us denote the generator of this additional U(1) symmetry by M, and the associated conserved current by R^{μ} . This symmetry does not talk to the Poincaré or G sectors of the theory, therefore M commutes with all the other generators.

To take a non-relativistic limit, we need to introduce a *c*-scaling for various relativistic quantities and then take $c \to \infty$. Let us start by picking a basis $(x^A) = (ct, x^a)$ and choose a decomposition for various Poincaré generators as

$$P_{\alpha} = \begin{pmatrix} -c M - \frac{1}{c} H + \dots \\ P_{a} + \dots \end{pmatrix}, \qquad J^{\alpha}_{\ \beta} = \begin{pmatrix} 0 & -c B_{b} + \dots \\ -c B^{a} + \dots & M^{a}_{b} + \dots \end{pmatrix}, \tag{4.47}$$

where ellipses denote the subleading terms in c. The Poincaré commutation relations in eq. (2.1), when truncated to the highest order in c, exactly reproduce the Galilean commutators in eq. (4.1). This suggests that the Noether currents undergo a corresponding decomposition

$$T^{\mu}_{\ \alpha} = \begin{pmatrix} -c \, \rho^{\mu} - \frac{1}{c} \epsilon^{\mu} + \dots \\ p^{\mu}_{\ a} + \dots \end{pmatrix} \qquad \Sigma^{\mu \alpha}{}_{\beta} = \begin{pmatrix} 0 & -c \, \tau^{\mu}_{\ b} + \dots \\ -c \, \tau^{\mu a} + \dots & \sigma^{\mu a}_{\ b} + \dots \end{pmatrix},$$
$$R^{\mu} = \rho^{\mu} + \dots, \qquad J^{\mu} = j^{\mu} + \dots \qquad (4.48)$$

Now let us move on to the non-relativistic limit of the relativistic background fields. In addition to the fields defined in section 2.1, we now also have a gauge field B_{μ} associated with the generator M. We choose the following *c*-scaling for these fields

$$e^{\alpha}{}_{\mu} = \begin{pmatrix} c n_{\mu} \\ e^{a}{}_{\mu} \end{pmatrix}, \quad e_{\alpha}{}^{\mu} = \begin{pmatrix} \frac{1}{c}v^{\mu} \\ e_{a}{}^{\mu} \end{pmatrix}, \quad B_{\mu} = c^{2}n_{\mu} + b_{\mu}, \quad T^{\alpha}{}_{\mu\nu} = \begin{pmatrix} -c H_{\mu\nu} + \frac{1}{c}\Omega_{\mu\nu} \\ T^{a}{}_{\mu\nu} \end{pmatrix},$$
(4.49)

while the gauge field A_{μ} stays as such. The limits of affine and spin connections can be worked out from here, which gives us exactly their Newton-Cartan expressions

$$\Gamma^{\lambda}{}_{\mu\nu} \rightarrow \left(\Gamma^{\lambda}{}_{\mu\nu}\right)_{\rm NC} + \dots, \quad C^{a}{}_{\mu b} \rightarrow \left(C^{a}{}_{\mu b}\right)_{\rm NC} + \dots, \quad C^{a}{}_{\mu t} = c_{\mu}{}^{a} + \dots$$
(4.50)

These scalings are defined based on the requirement that

$$\langle T^{\mu}_{\ \alpha} \rangle \delta e^{\alpha}_{\ \mu} + \langle \Sigma^{\mu\alpha}_{\ \beta} \rangle \delta C^{\beta}_{\ \mu\alpha} + \langle J^{\mu} \rangle \cdot \delta A_{\mu}$$

$$= \rho^{\mu} \delta b_{\mu} - \epsilon^{\mu} \delta n_{\mu} + p^{\mu}_{\ a} \delta e^{a}_{\ \mu} - 2\tau^{\mu}_{\ a} \delta c_{\mu}^{\ a} + \sigma^{\mu a}_{\ b} \delta C^{b}_{\ \mu a} + j^{\mu} \cdot \delta A_{\mu} + \dots$$

$$(4.51)$$

This ensures that in $c \to \infty$ limit, the Galilean sources couple correctly to their respective Galilean currents. As a non-trivial check of these scaling rules, we can take a $c \to \infty$ limit of the δ_{χ} variations defined in eq. (2.16) and find that they indeed reduce to the Galilean ones defined in eq. (4.26), provided that we identify the spin parameters as

$$\Lambda_{\chi}^{\Sigma a}{}_{t} \to \Lambda_{\chi}^{\tau a}, \qquad \Lambda_{\chi}^{\Sigma a}{}_{b} \to \Lambda_{\chi}^{\sigma a}{}_{b}.$$

$$(4.52)$$

Finally, at the highest order in c, the relativistic conservation laws (2.20) also exactly map to the Galilean ones given in eq. (4.37).

With all this machinery in place, we are finally ready to discuss Galilean hydrodynamics. In the next section, we introduce hydrodynamics on null backgrounds and use it work out the principles of Galilean hydrodynamics coupled to Newton-Cartan backgrounds.

4.3 Null fluids

4.3.1 Hydrodynamics on null backgrounds

Following our relativistic discussion in section 2.2, we now formulate a theory of hydrodynamics on null backgrounds. We start with the assumption that the fundamental theory we are seeking to describe respects the spacetime Poincaré transformations and some internal global *G*-transformations, and is left invariant by the action of a null isometry \mathcal{V} . This postulates the existence of the associated \mathcal{V} respecting conserved currents $(T_{A}^{M}, \Sigma_{B}^{M}, J^{M})$. In the low-energy regime, the theory would be described by some effective massless degrees of freedom φ^{I} . Therefore the conserved currents can generically be expressed in terms of φ^{I} , the background fields $(e_{M}^{A}, C_{MB}^{A}, A_{M}, \mathcal{V})$, and their derivatives. Due to Noether's theorem, these currents satisfy the identities (4.11).

Thanks to these identities, the conservation laws (4.12) can serve as a placeholder for a vector, a $\mathfrak{so}(d, 1)$ -valued scalar, and a \mathfrak{g} -valued scalar linear combinations of the equations of motion. That is to say that out of the dynamic fields φ^I , eq. (4.12) can serve as equations of motion for some degrees of freedom packaged into a set of symmetry parameters

$$\mathcal{B} = \left(\begin{array}{cc} \beta^{\scriptscriptstyle M} , & \Lambda^{\scriptscriptstyle \Sigma}_{\beta} , & \Lambda_{\beta} \end{array} \right). \tag{4.53}$$

The remaining dynamical fields and their equations of motion are represented by φ^i and $\mathcal{E}_i \approx 0$ respectively. Recall that the spin conservation equation in eq. (4.12) is only relevant up to an arbitrary vector redefinition (4.14). Therefore we should not treat all the components in Λ_{β}^{Σ} to be independent. We can fix this by choosing $\delta_{\mathcal{B}}V^A = \beta^M \partial_M V^A - \Lambda_{\beta}^{\Sigma A}{}_B V^B = 0$, which can be checked to be a covariant condition. The fields \mathcal{B} can also be represented in terms of a set of hydrodynamic fields

Null velocity:
$$u^{M}$$
 with $u^{M}u_{M} = 0$, $u^{M}V_{M} = -1$,
Temperature: T , Mass chemical potential: μ_{m} ,
Spin chemical potential: $\mu^{\Sigma} \in \mathfrak{so}(d, 1)$ with $\mu^{\Sigma A}{}_{B}V^{B} = 0$,
 G chemical potential: $\mu \in \mathfrak{g}$, (4.54)

defined via

$$\beta^{M} = \frac{1}{T} \left(u^{M} - \mu_{m} V^{M} \right), \qquad \Lambda_{\beta}^{\Sigma_{A}}{}_{B} + \beta^{M} C^{A}{}_{MB} = \frac{1}{T} \mu^{\Sigma_{A}}{}_{B}, \qquad \Lambda_{\beta} + \beta^{M} A_{M} = \frac{1}{T} \mu.$$
(4.55)

We can now go ahead and write down the most generic null fluid constitutive relations: expressions for $(T^{M}_{A}, \Sigma^{MA}_{B}, J^{M}, \mathcal{E}_{i})$ in terms of the dynamical fields φ^{i} and \mathcal{B} , the background fields $(e^{A}_{M}, C^{A}_{MB}, A_{M}, \mathcal{V})$, and their derivatives, arranged in a derivative expansion.

The remainder of the story exactly parallels that for relativistic fluids given in sections 2.2 and 2.3. In the following we recapitulate the key points of the discussion. Upon taking the hydrodynamic fields \mathcal{B} on-shell, the partially on-shell conservation laws are given by

$$\underline{\mathbf{D}}_{M}T^{M}{}_{A} \simeq e_{A}{}^{N} \Big(\mathbf{T}^{B}{}_{NM}T^{M}{}_{B} + R_{NM}{}^{C}{}_{B}\boldsymbol{\Sigma}^{MB}{}_{C} + F_{NM} \cdot J^{M} \Big) + \mathcal{O}^{i}_{A}\mathcal{E}_{i},$$

$$\underline{\mathbf{D}}_{M}\boldsymbol{\Sigma}^{MAB} \simeq T^{[BA]} + \boldsymbol{\Sigma}_{\mathbf{H}}^{\perp AB} + \mathcal{O}^{iAB}\mathcal{E}_{i},$$

$$\underline{\mathbf{D}}_{M}J^{M} \simeq \mathbf{J}_{\mathbf{H}}^{\perp} + \mathcal{O}^{i}\mathcal{E}_{i}.$$
(4.56)

The second law of thermodynamics requires that the null fluid constitutive relations should be accompanied by an entropy current $J_{\rm S}^{\scriptscriptstyle M}$ whose divergence is positive semi-definite, i.e. $\underline{\mathbf{D}}_{\scriptscriptstyle M} J_{\rm S}^{\scriptscriptstyle M} \geq 0$, for all partially on-shell (thermodynamically isolated) fluid configurations. This statement can be re-expressed in a more useful off-shell language by defining a free energy current and a free energy Hall current

$$N^{M} = J^{M}_{S} + \beta^{A} T^{M}_{\ A} + \left(\Lambda^{\Sigma B}_{\beta\ A} + \beta^{M} C^{B}_{\ MA}\right) \Sigma^{MA}_{\ B} + \left(\Lambda_{\beta} + \beta^{M} A_{M}\right) \cdot J^{M} + \mathcal{N}^{iM}_{\beta} \mathcal{E}_{i},$$

$$N^{\perp}_{H} = \left(\Lambda^{\Sigma B}_{\beta\ A} + \beta^{M} C^{B}_{\ MA}\right) \Sigma^{\perp a}_{H\ b} + \left(\Lambda_{\beta} + \beta^{M} A_{M}\right) \cdot J^{\perp}_{H},$$
(4.57)

where $\mathcal{N}_{\beta}^{i_{M}}$ is defined in eq. (2.18). Using these definitions and the partially on-shell conservation laws (4.56), we can convert the second law statement into an adiabaticity equation

$$\underline{\mathbf{D}}_{_{M}}N^{_{M}}-\mathbf{N}_{\mathrm{H}}^{\perp}=T^{_{M}}{}_{_{A}}\delta_{\mathcal{B}}e^{^{A}}{}_{_{M}}+\Sigma^{^{MA}}{}_{_{B}}\delta_{\mathcal{B}}C^{^{B}}{}_{_{MA}}+J^{^{M}}\cdot\delta_{\mathcal{B}}A_{_{M}}+\mathcal{E}_{i}\delta_{\mathcal{B}}\varphi^{i}+\Delta,\quad\Delta\geq0,\quad(4.58)$$

where Δ is a positive semi-definite quadratic form. The $\delta_{\mathcal{B}}$ variation of various fields is given in eq. (2.16). In writing this equation, we have already fixed a large amount of redefinition freedom in the arbitrarily defined hydrodynamic fields. The remaining freedom can be fixed by rendering our constitutive relations to be independent of $u^M \delta_{\mathcal{B}} g_{MN}$, $u^M \delta_{\mathcal{B}} C^A_{MB}$, and $u^M \delta_{\mathcal{B}} A_M$. In this form, the second law requires that for a set of constitutive relations $(T^M_{\ A}, \Sigma^{MA}_{\ B}, J^M, \mathcal{E}_i)$ to be physically allowed, they must admit a free energy current N^M , which satisfies the adiabaticity equation for some positive semi-definite quadratic form Δ . We classify the most generic constitutive relations allowed by this restriction in sections 4.3.3 to 4.3.5. First, let us see how the null fluid constitutive relations can be translated to the *d*-dimensional Galilean fluid language.

4.3.2 Null reduction to Galilean hydrodynamics

Once we are given a set of null fluid constitutive relations $(T^{M}_{A}, \Sigma^{MA}_{B}, J^{M}, \mathcal{E}_{i})$, we can use the null reduction prescription outlined in section 4.2 to obtain the respective constitutive relations for a Galilean fluid. To perform the explicit reduction, however, we need a map between the dynamical fields of a null fluid and those of a Galilean fluid. Upon introduction of a frame velocity v^{M} , the hydrodynamic fields in eq. (4.54) can be decomposed into

$$u^{\mu}$$
 with $u^{\mu}n_{\mu} = 1$, T , $\mu_{\rm m}$, $\mu^{\sigma a}{}_{b}$, $\mu^{\tau a}$, μ , (4.59)

where

$$u^{M} = \begin{pmatrix} b_{\mu}u^{\mu} + \frac{1}{2}u^{a}u_{a} \\ u^{\mu} \end{pmatrix}, \qquad \mu^{\Sigma_{A}}{}_{B} = \begin{pmatrix} 0 & 0 & \mu^{\tau}{}_{b} \\ 0 & 0 & 0 \\ 0 & \mu^{\tau}{}^{a} & \mu^{\sigma}{}^{a}_{b} \end{pmatrix}.$$
(4.60)

In terms of these, we can also find out

$$u_{M} = \begin{pmatrix} -1 \\ u_{\mu} = \bar{u}_{\mu} + b_{\mu} - n_{\mu} \frac{1}{2} u^{a} u_{a} \end{pmatrix}.$$
 (4.61)

We have defined the "spatial" components of the Galilean fluid velocity as $\bar{u}^{\mu} = u^{\mu} - v^{\mu}$ which satisfy $\bar{u}^{\mu}n_{\mu} = 0$, or equivalently on the frame bundle $u^{a} = e^{a}{}_{\mu}u^{\mu} = e^{a}{}_{\mu}\bar{u}^{\mu}$. We can
also define the projection operators against the fluid velocity

$$p^{\mu\nu} = h^{\mu\nu}, \qquad p_{\mu\nu} = h_{\mu\nu} - 2n_{(\mu}\bar{u}_{\nu)} + n_{\mu}n_{\nu}\bar{u}^{\rho}\bar{u}_{\rho}, \qquad p^{\mu}{}_{\nu} = p^{\mu\rho}p_{\rho\nu}.$$
(4.62)

The other dynamical fields φ^i in the fluid description might also admit such a decomposition, but in the absence of their explicit transformation properties, we leave them abstract in this discussion. We later look at some explicit examples in chapter 5.

Under a Galilean frame transformation $v^{M} \rightarrow v^{M} + \psi^{M}$ defined in section 4.2.3, most of these dynamical fields remain unchanged, except the spatial fluid velocity and temporal spin chemical potential, which transform according to

$$u^a \to u^a - \psi^a, \qquad \mu^{\tau a} \to \mu^{\tau a} + \mu^{\sigma a}{}_b \psi^b.$$
 (4.63)

If instead of working in an arbitrary Galilean frame, we decided to work in the naturally defined fluid frame of reference given by $v^{\mu} = u^{\mu}$, we can choose $\psi^{\mu} = \bar{u}^{\mu}$ setting $\bar{u}^{\mu} \to 0$. At a calculational level therefore, it is always best to work in the fluid frame of reference. If required, we can later perform a frame transformation outlined in eq. (4.46) with $\psi^{\mu} = -\bar{u}^{\mu}$ and get back the generic frame results.

Due to the second law of thermodynamics, the null fluid constitutive relations are accompanied by an entropy current $J_{\rm S}^{\rm M}$ and a quadratic form Δ , such that for thermodynamically isolated fluid configurations, we have $\underline{\mathcal{D}}_{\rm M}J_{\rm S}^{\rm M}\simeq\Delta\geq 0$. Upon null reduction, this statement transforms into a local second law for Galilean fluids,

$$\underline{\tilde{D}}_{\mu}s^{\mu} \simeq \Delta \ge 0, \quad \text{where} \quad s^{\mu} = J_{\mathrm{S}}^{\mu}.$$
 (4.64)

Following our null fluid discussion, we know that to satisfy the second law requirement, the Galilean fluid must admit a free energy current $n^{\mu} = N^{\mu}$ such that the following adiabaticity equation is satisfied for any (on-shell or off-shell) fluid configuration

$$\underline{\tilde{D}}_{\mu}n^{\mu} - \mathbf{n}_{\mathrm{H}}^{\perp} = \rho^{\mu}\delta_{\mathcal{B}}b_{\mu} - \epsilon^{\mu}\delta_{\mathcal{B}}n_{\mu} + p^{\mu}{}_{a}\delta_{\mathcal{B}}e^{a}{}_{\mu}
- 2\tau^{\mu a}\delta_{\mathcal{B}}c_{\mu a} + \sigma^{\mu a}{}_{b}\delta_{\mathcal{B}}C^{b}{}_{\mu a} + j^{\mu}\cdot\delta_{\mathcal{B}}A_{\mu} + \mathcal{E}_{i}\delta_{\mathcal{B}}\varphi^{i} + \Delta.$$
(4.65)

Here $n_{\rm H}^{\perp} = N_{\rm H}^{\perp}$ is the anomalous free energy Hall current. The $\delta_{\mathcal{B}}$ variation of various fields used here can be obtained by a substituting \mathcal{X} with \mathcal{B} in eq. (4.26).

To summarise, a Galilean fluid is characterised by its conserved currents, which in the minimal setting are the mass current ρ^{μ} , energy current ϵ^{μ} , and momentum current $p^{\mu}{}_{a}$. Depending on the physical application we have in mind, we can also throw in some possibly non-Abelian charge currents j^{μ} , spin currents $\tau^{\mu a}$ and $\sigma^{\mu ab}$, and equations of motion \mathcal{E}_{i} corresponding to additional gapless modes in the fluid description (like Goldstone modes of broken symmetries). We start with the most generic form of these currents allowed by the Galilean symmetries, called the constitutive relations, in terms of some dynamical fields: fluid velocity u^{μ} , temperature T, chemical potentials $\mu_{\rm m}$, $\mu^{\tau a}$, $\mu^{\sigma ab}$, and μ , and additional gapless fields φ^{i} , arranged in a derivative expansion. Dynamical equations for these fields are given by the conservation equations (4.37). For convenience, we can also introduce some

background fields: frame fields $e^{a}{}_{\mu}$, $e^{\mu}{}_{a}$, and n_{μ} , connections $\Gamma^{\lambda}{}_{\mu\nu}$, $C^{a}{}_{\mu b}$, A_{μ} , and b_{μ} , and a Galilean frame velocity v^{μ} . To be physically relevant, the fluid constitutive relations are required to satisfy the second law of thermodynamics, which essentially requires them to be accompanied by an entropy current s^{μ} whose divergence is locally positive semi-definite. This can equivalently be posed as the requirement that there must exist a free energy current n^{μ} so that the adiabaticity equation (4.65) is satisfied. To work out these constitutive relations in a manner that manifestly preserves the Galilean invariance, it is convenient to work in a one-higher dimensional null fluid picture, where the symmetry structure is exactly that of a relativistic fluid. The Galilean adiabaticity equation also happens to be equivalent to the adiabaticity equation in the null fluid formalism, which can be used to derive the constraints on the Galilean fluid constitutive relations in a more convenient language.

4.3.3 Anomaly induced transport

Having derived the Galilean version of the adiabaticity equation, we would now like to inspect and classify all of its solutions. Since the anomaly inflow mechanism on null backgrounds, as discussed in section 4.1.4, is subtly different from its relativistic counterpart, in this subsection we first focus on a particular anomaly induced solution to eq. (4.58). Later in section 4.3.4, we also discuss its extension to include transcendental anomalies.

Following [5], we note that the shadow connections for null fluids should be defined with respect to the null isometry V^{M} , as opposed to the fluid velocity which was used for relativistic fluids in eq. (2.58), i.e.

$$\hat{\boldsymbol{A}} = \boldsymbol{A} + \mu \boldsymbol{V}, \qquad \hat{\boldsymbol{C}}^{\scriptscriptstyle A}{}_{\scriptscriptstyle B} = \boldsymbol{C}^{\scriptscriptstyle A}{}_{\scriptscriptstyle B} + \mu^{\Sigma \,\scriptscriptstyle A}{}_{\scriptscriptstyle B} \boldsymbol{V}, \qquad (4.66)$$

where $\mathbf{V} = V_M \mathrm{d} x^M$. They satisfy

$$\Lambda_{\beta} + \beta^{M} \hat{A}_{M} = \Lambda_{\beta}^{\Sigma A}{}_{B} + \beta^{M} \hat{C}^{A}{}_{MB} = 0, \qquad \Lambda_{V} + V^{M} \hat{A}_{M} = \Lambda_{V}^{\Sigma A}{}_{B} + V^{M} \hat{C}^{A}{}_{MB} = 0.$$
(4.67)

We can use these to work out the respective shadow field strength and curvature, followed by the shadow anomaly polynomial and Hall currents. In particular, the shadow free energy Hall current is given by

$$\star_{(d+2)} \hat{\mathbf{N}}_{\mathrm{H}} = \frac{1}{T} \star_{(d+2)} \left(\mu \cdot \hat{\mathbf{J}}_{\mathrm{H}} + \mu^{\Sigma} \cdot \hat{\boldsymbol{\Sigma}}_{\mathrm{H}} \right).$$
(4.68)

Let us further define

$$\star \boldsymbol{\Sigma}_{\boldsymbol{\mathcal{P}}} = \frac{\boldsymbol{V}}{\mathrm{d}\boldsymbol{V}} \wedge \star_{(d+2)} \left(\boldsymbol{\Sigma}_{\mathrm{H}} - \hat{\boldsymbol{\Sigma}}_{\mathrm{H}} \right), \qquad \star \boldsymbol{J}_{\boldsymbol{\mathcal{P}}} = \frac{\boldsymbol{V}}{\mathrm{d}\boldsymbol{V}} \wedge \star_{(d+2)} \left(\boldsymbol{J}_{\mathrm{H}} - \hat{\boldsymbol{J}}_{\mathrm{H}} \right), \\ \star \boldsymbol{q}_{\boldsymbol{\mathcal{P}}} = -\frac{\boldsymbol{V}}{\mathrm{d}\boldsymbol{V} \wedge \mathrm{d}\boldsymbol{V}} \wedge \left(\boldsymbol{\mathcal{P}} - \hat{\boldsymbol{\mathcal{P}}} + T \,\mathrm{d}\boldsymbol{V} \wedge \star_{(d+2)} \hat{\boldsymbol{N}}_{\mathrm{H}} \right).$$
(4.69)

In terms of these, we can read out the anomaly induced constitutive relations, also known as the Class A transport, as simply

$$(T^{\scriptscriptstyle M}_{\scriptscriptstyle A})_{\rm A} = q^{\scriptscriptstyle M}_{\mathcal{P}} V_{\scriptscriptstyle A}, \qquad (\Sigma^{\scriptscriptstyle MA}_{\scriptscriptstyle B})_{\rm A} = \Sigma_{\mathcal{P}}^{\scriptscriptstyle A}{}_{\scriptscriptstyle B}, \qquad (J^{\scriptscriptstyle M})_{\rm A} = J^{\scriptscriptstyle M}_{\mathcal{P}}.$$
 (4.70a)

along with $(\mathcal{E}_i)_{\rm A} = 0$. They satisfy the null fluid adiabaticity equation (4.58) with the free energy current

$$(N^{\scriptscriptstyle M})_{\rm A} = \frac{1}{T} \left(-q_{\mathcal{P}}^{\scriptscriptstyle M} + \mu^{\Sigma_{\scriptscriptstyle A}}{}_{\scriptscriptstyle B} \Sigma_{\mathcal{P}}^{\scriptscriptstyle MB}{}_{\scriptscriptstyle A} + \mu \cdot J_{\mathcal{P}}^{\scriptscriptstyle M} \right).$$
(4.70b)

Like in the relativistic case, we note that there is no Class A contribution to the entropy current $J_{\rm S}^{M}$. Upon performing the null reduction, we can see that, in fact, only the energy current ϵ^{μ} , spatial spin current $\sigma^{\mu ab}$, charge current j^{μ} , and free energy current n^{μ} get an anomaly-induced contribution. All the other currents stay untouched by anomalies.

That the solutions (4.70) satisfy the Galilean adiabaticity equation (4.58), can either be established via an explicit computation, or more easily via a bulk effective action

$$S_{\rm A} = \int_{\mathcal{B}} \frac{\boldsymbol{V}}{\mathrm{d}\boldsymbol{V}} \wedge \left(\boldsymbol{\mathcal{P}} - \hat{\boldsymbol{\mathcal{P}}}\right), \qquad (4.71)$$

which generates the anomaly induced constitutive relations at the boundary. The adiabaticity equation follows from here by invoking the invariance of this action under an infinitesimal symmetry variation along \mathcal{B} .

4.3.4 Transcendental anomalies

We can extend the Class A constitutive relations (4.70) to include slightly more generic solutions of the adiabaticity equation (4.58). To this end, like we did in section 2.3.3, we extend our theory by introducing an auxiliary U(1)_T global symmetry, with background gauge field A_M^{T} and chemical potential $\mu^{\mathsf{T}} = T(\Lambda_\beta^{\mathsf{T}} + \beta^M A_M^{\mathsf{T}}) = T$. The associated shadow gauge field is given by $\hat{A}^{\mathsf{T}} = A^{\mathsf{T}} + TV$. Having done that, we can use the respective field strength F_{MN}^{T} to write down new terms in the anomaly polynomial. Unlike the relativistic case, however, we can further involve terms in the anomaly polynomial made out of the Chern classes of $\Omega = du$. The shadow field corresponding to u is given by $\hat{u} = u + \mu_m V$, with the property that $\beta^M \hat{u}_M = 0$. There is no analogue of this in relativistic hydrodynamics. Finally, the full thermal anomaly polynomial for this enlarged theory is given as

$$\mathcal{P}_{\mathsf{T}} = \mathcal{P} + \mathcal{P}_{\mathrm{H}_{\mathrm{V}}}, \qquad \mathcal{P}_{\mathrm{H}_{\mathrm{V}}} = \boldsymbol{u} \wedge \sum_{j \ge 1} \left(\boldsymbol{F}^{\mathsf{T}} \right)^{\wedge j} \wedge \mathcal{P}_{\mathrm{H}_{\mathrm{V}}, j}, \qquad (4.72)$$

where \mathcal{P} is the original anomaly polynomial made out of F and R, while $\mathcal{P}_{H_{V},j}$ are (d+2-2j)-rank anomaly polynomials which can in addition involve Ω . Note that since we now have a preferred vector field u^{M} , we have used it to define the anomaly polynomials instead of an arbitrary vector field v^{M} used in section 4.1.4.

Now, we can use the mechanism outlined in the previous subsection to generate a

particular solution for the adiabaticity equation of the enlarged theory

$$\underline{\mathbf{D}}_{M}N^{M} - \mathbf{N}_{\mathrm{H}}^{\perp} - \mathbf{J}_{\mathrm{TH}}^{\perp}$$

$$= T^{M}_{\ A}\delta_{\mathcal{B}}e^{A}_{\ M} + \Sigma^{MA}_{\ B}\delta_{\mathcal{B}}C^{B}_{\ MA} + J^{M}_{\mathbf{N}}\cdot\delta_{\mathcal{B}}A_{M} + J^{M}_{\mathsf{T}}\delta_{\mathcal{B}}A^{\mathsf{T}}_{M} + J^{M}_{\Omega}\delta_{\mathcal{B}}u_{M} + \mathcal{E}_{i}\delta_{\mathcal{B}}\varphi^{i}$$

$$= \left(T^{M}_{\ A} + J^{N}_{\Omega}P_{NA}u^{M} + J^{M}_{\Omega}u_{A}\right)\delta_{\mathcal{B}}e^{A}_{\ M}$$

$$+ \Sigma^{MA}_{\ B}\delta_{\mathcal{B}}C^{B}_{\ MA} + J^{M}_{\mathbf{N}}\cdot\delta_{\mathcal{B}}A_{M} + J^{M}_{\mathsf{T}}\delta_{\mathcal{B}}A^{\mathsf{T}}_{M} + \mathcal{E}_{i}\delta_{\mathcal{B}}\varphi^{i}.$$
(4.73)

In the second step, we have expanded $\delta_{\mathcal{B}} u_M = (P_{MA} u^N + \delta^N_M u_A) \delta_{\mathcal{B}} e^A_N$. Various Hall currents are now being defined with respect to the thermal anomaly polynomial \mathcal{P}_{T} , in particular

$$\star_{(d+2)} \mathbf{J}_{\mathrm{TH}} = \frac{\partial \boldsymbol{\mathcal{P}}_{\mathsf{T}}}{\partial \boldsymbol{F}^{\mathsf{T}}} = \boldsymbol{u} \wedge \sum_{j \ge 1} j(\boldsymbol{F}^{\mathsf{T}})^{\wedge (j-1)} \wedge \boldsymbol{\mathcal{P}}_{\mathrm{H}_{\mathrm{V}},j},$$
$$\star_{(d+2)} \mathbf{J}_{\Omega\mathrm{H}} = \frac{\partial \boldsymbol{\mathcal{P}}_{\mathsf{T}}}{\partial \boldsymbol{\Omega}} = \boldsymbol{u} \wedge \sum_{j \ge 1} (\boldsymbol{F}^{\mathsf{T}})^{\wedge j} \wedge \frac{\partial \boldsymbol{\mathcal{P}}_{\mathrm{H}_{\mathrm{V}},j}}{\partial \boldsymbol{\Omega}}.$$
(4.74)

On the other hand, we have defined the free energy Hall current to be

$$\star_{(d+2)} \mathbf{N}_{\mathrm{H}} = \frac{1}{T} \star_{(d+2)} \left(\mu^{\Sigma_{A}}{}_{B} \boldsymbol{\Sigma}_{\mathrm{H}}{}^{B}{}_{A} + \mu \cdot \mathbf{J}_{\mathrm{H}} + \mu_{\mathrm{m}} \mathbf{J}_{\Omega \mathrm{H}} \right).$$
(4.75)

In the limit that we take $F_{MN}^{\mathsf{T}} \to 0$, $\mathbf{J}_{\Omega \mathrm{H}}$ vanishes and \mathbf{N}_{H} takes its original unextended form. The quantity $\delta_{\mathcal{B}} A_{M}^{\mathsf{T}} = \beta^{N} F_{NM}^{\mathsf{T}}$ also vanishes. With these simplifications, we can show that the extended adiabaticity equation reduces to

$$\underline{\mathbf{D}}_{M}N^{M} - \mathbf{J}_{\mathsf{TH}}^{\perp} - \mathbf{N}_{\mathsf{H}}^{\perp} = \left(T_{A}^{M} + J_{\Omega}^{N}P_{NA}u^{M} + J_{\Omega}^{M}u_{A}\right)\delta_{\mathcal{B}}e_{M}^{A} + \Sigma_{B}^{MA}\delta_{\mathcal{B}}C_{MA}^{B} + J^{M}\cdot\delta_{\mathcal{B}}A_{M} + \mathcal{E}_{i}\delta_{\mathcal{B}}\varphi^{i}, \quad (4.76)$$

This is almost the form of the adiabaticity equation that we require for our original theory, except for the residual $\star_{(d+2)} \mathbf{J}_{\mathsf{TH}} = \mathbf{u} \wedge \mathcal{P}_{\mathrm{H}_{\mathrm{V}},1}$ piece. Note that $\star_{(d+2)} \mathbf{J}_{\mathrm{TH}}$ is closed, as its derivative does not have any component along $V^{\mathcal{M}}$. If it also happens to be exact, we can absorb it into the free energy current. To this end, we can use the fact that there exists a gauge-non-invariant $\mathbf{I}_{\mathrm{H}_{\mathrm{V}},1}$ such that $\mathrm{d}\mathbf{I}_{\mathrm{H}_{\mathrm{V}},1} = \mathcal{P}_{\mathrm{H}_{\mathrm{V}},1}$, and therefore

$$\star_{(d+2)} \mathbf{J}_{\mathrm{TH}} = \boldsymbol{u} \wedge \mathrm{d} \boldsymbol{I}_{\mathrm{H}_{\mathrm{V}},1} = -\mathrm{d} \left(\boldsymbol{u} \wedge \boldsymbol{I}_{\mathrm{H}_{\mathrm{V}},1} \right) - \boldsymbol{u} \wedge \boldsymbol{\Omega} \wedge \iota_{V} \boldsymbol{I}_{\mathrm{H}_{\mathrm{V}},1}.$$
(4.77)

Here ι_V denotes the interior product along V^M . If we work in the transverse gauge for the null isometry, i.e. $\Lambda_V = \Lambda_V^{\Sigma} = 0$, where $\iota_V \boldsymbol{A} = \iota_V \boldsymbol{C} = 0$, we can always ensure the last piece to vanish as long as $\boldsymbol{\mathcal{P}}_{\mathrm{H}_{\mathrm{V}},1}$ has at least one instance of \boldsymbol{F} or \boldsymbol{R} . In that case, $\star_{(d+2)}\mathbf{J}_{\mathrm{TH}}$ is, in fact, exact. However, if it is purely made out of $\boldsymbol{\Omega}$, i.e. $\boldsymbol{\mathcal{P}}_{\mathrm{H}_{\mathrm{V}},1} = \boldsymbol{\Omega}^{\wedge d/2}$, the respective term decisively does not vanish

$$\boldsymbol{u} \wedge \boldsymbol{\Omega} \wedge \iota_V \boldsymbol{I}_{\mathrm{H}_{\mathrm{V}},1} = -\boldsymbol{u} \wedge \boldsymbol{\Omega}^{\wedge d/2} \neq 0.$$
(4.78)

Consequently, to generate constitutive relations for our Galilean system of interest, we are not allowed to have the term $u \wedge F_{\mathsf{T}} \wedge \Omega^{\wedge d/2}$ in the thermal anomaly polynomial \mathcal{P}_{T} . With

this taken care of, we can rewrite eq. (4.76) into

$$\underline{\mathbf{D}}_{M}\left(N^{M}-\star_{(d+1)}\left(\boldsymbol{u}\wedge\boldsymbol{I}_{\mathrm{H}_{\mathrm{V}},1}\right)^{M}\right)-\mathrm{N}_{\mathrm{H}}^{\perp}$$
$$=\left(T^{M}_{A}+J^{N}_{\Omega}P_{NA}\boldsymbol{u}^{M}+J^{M}_{\Omega}\boldsymbol{u}_{A}\right)\delta_{\mathcal{B}}\boldsymbol{e}^{A}_{\ M}+\boldsymbol{\Sigma}^{MA}_{\ B}\delta_{\mathcal{B}}C^{B}_{\ MA}+J^{M}\cdot\delta_{\mathcal{B}}A_{M}+\mathcal{E}_{i}\delta_{\mathcal{B}}\varphi^{i},\quad(4.79)$$

which is precisely the form that we expect for our original theory. Therefore, we can use the particular solutions of the extended adiabaticity equation to generate new solutions of our original adiabaticity equation.

Let us now return to the actual solutions. Following the procedure of the previous subsection, we note that the \mathcal{P} part of the thermal anomaly polynomial \mathcal{P}_{T} simply generates the Class A constitutive relations as before. On the other hand, the $\mathcal{P}_{\mathsf{H}_{\mathsf{V}}}$ part generates the new Class H_{V} constitutive relations. Let us denote the Hall currents defined using $\mathcal{P}_{\mathsf{H}_{\mathsf{V}}}$ with the wrapper ()_{H_V}. This allows us to define the analogues of eq. (4.69) for Class H_{V} in $F_{_{MN}}^{\mathsf{T}} \to 0$ limit

$$\star \boldsymbol{\Sigma}_{\boldsymbol{\mathcal{P}}_{H_{V}}} = -\frac{\boldsymbol{V}}{d\boldsymbol{V}} \wedge \star_{(d+2)}(\hat{\boldsymbol{\Sigma}}_{H})_{H_{V}} \Big|_{\boldsymbol{F}^{\mathsf{T}} \to 0}, \qquad \star \boldsymbol{J}_{\boldsymbol{\mathcal{P}}_{H_{V}}} = -\frac{\boldsymbol{V}}{d\boldsymbol{V}} \wedge \star_{(d+2)}(\hat{\mathbf{J}}_{H})_{H_{V}} \Big|_{\boldsymbol{F}^{\mathsf{T}} \to 0}, \\ \star \boldsymbol{J}_{\mathsf{T}}\boldsymbol{\mathcal{P}}_{H_{V}} = \frac{\boldsymbol{V}}{d\boldsymbol{V}} \wedge \star_{(d+2)} \left((\mathbf{J}_{\mathsf{T}H})_{H_{V}} - (\hat{\mathbf{J}}_{\mathsf{T}H})_{H_{V}} \right) \Big|_{\boldsymbol{F}^{\mathsf{T}} \to 0}, \\ \star \boldsymbol{J}_{\Omega \boldsymbol{\mathcal{P}}_{H_{V}}} = -\frac{\boldsymbol{V}}{d\boldsymbol{V}} \wedge \star_{(d+2)}(\hat{\mathbf{J}}_{\Omega H})_{H_{V}} \Big|_{\boldsymbol{F}^{\mathsf{T}} \to 0}, \\ \star \boldsymbol{q}_{\boldsymbol{\mathcal{P}}_{H_{V}}} = \frac{\boldsymbol{V}}{(d\boldsymbol{V})^{\wedge 2}} \wedge \left(\hat{\boldsymbol{\mathcal{P}}}_{H_{V}} - T d\boldsymbol{V} \wedge \star_{(d+2)} \left((\hat{\mathbf{N}}_{H})_{H_{V}} + (\hat{\mathbf{J}}_{\mathsf{T}H})_{H_{V}} \right) \right) \Big|_{\boldsymbol{F}^{\mathsf{T}} \to 0}.$$
(4.80)

In terms of these, the Class H_V constitutive relations are given as

$$(T^{M}_{A})_{\mathrm{H}_{\mathrm{V}}} = q^{M}_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} V_{A} + J^{M}_{\Omega \mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} u_{A} + J^{N}_{\Omega \mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} P_{NA} u^{M},$$

$$(\Sigma^{MAB})_{\mathrm{H}_{\mathrm{V}}} = \Sigma^{MAB}_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}}, \qquad (J^{M})_{\mathrm{H}_{\mathrm{V}}} = J^{M}_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}},$$

$$(N^{M})_{\mathrm{H}_{\mathrm{V}}} = J^{M}_{\mathsf{T}\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} + \frac{1}{T} \left(-q^{M}_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} + \mu^{\Sigma} \cdot \Sigma^{M}_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} + \mu \cdot J^{M}_{\mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} + \mu_{\mathrm{m}} J^{M}_{\Omega \mathcal{P}_{\mathrm{H}_{\mathrm{V}}}} \right) - \star_{(d+1)} (\boldsymbol{u} \wedge \boldsymbol{I}_{\mathrm{H}_{\mathrm{V}},1})^{M},$$

$$(4.81)$$

along with $(\mathcal{E}_i)_{\mathrm{H}_{\mathrm{V}}} = 0$. The associated entropy current, like the relativistic case, is non-trivial in Class H_V

$$(J_{\rm S}^{\scriptscriptstyle M})_{\rm H_{\rm V}} = J_{\sf T\boldsymbol{\mathcal{P}}_{\rm H_{\rm V}}}^{\scriptscriptstyle M}.$$
(4.82)

In fact, in contrast to Class A, all the currents, except the temporal spin current $\tau^{\mu a}$, get a contribution form Class H_V.

The Class H_V constitutive relations satisfy the non-anomalous version of the Galilean adiabaticity equation (4.58). It can be verified by noting that these constitutive relations can be generated from the boundary variation of a bulk effective action

$$S_{\mathrm{H}_{\mathrm{V}}} = -\int_{\mathcal{B}} \frac{\boldsymbol{V}}{\mathrm{d}\boldsymbol{V}} \wedge \hat{\boldsymbol{\mathcal{P}}}_{\mathrm{H}_{\mathrm{V}}} \Big|_{\boldsymbol{F}^{\mathsf{T}} \to 0}.$$
(4.83)

The non-anomalous adiabaticity equation follows by making use of the invariance of this effective action under an infinitesimal variation along \mathcal{B} .

Before closing the subsection, let us make an interesting observation about the torsionless case. Because, $2\partial_{[M}V_{N]} = T^{A}{}_{MN}V_{A}$, the 2-form $d\mathbf{V} = 0$ on torsion-less backgrounds. Also, $\mathbf{V} \wedge \hat{\mathbf{F}}^{\mathsf{T}} = T\mathbf{V} \wedge d\mathbf{V} = 0$. It follows that all the terms in the extended anomaly polynomial that have more than two instances of \mathbf{F}^{T} (i.e. j > 2) do not contribute to the Class H_{V} constitutive relations at all. So, for torsionless null hydrodynamics, the Class H_{V} anomaly polynomial simplifies to

$$\boldsymbol{\mathcal{P}}_{\mathrm{H}_{\mathrm{V}}} = \boldsymbol{u} \wedge \boldsymbol{F}^{\mathsf{T}} \wedge \boldsymbol{\mathcal{P}}_{\mathrm{H}_{\mathrm{V}},1} + \boldsymbol{u} \wedge \boldsymbol{F}^{\mathsf{T}} \wedge \boldsymbol{F}^{\mathsf{T}} \wedge \boldsymbol{\mathcal{P}}_{\mathrm{H}_{\mathrm{V}},2}^{(d-2)}, \qquad (4.84)$$

which is characterised by far less number of constants than the full anomaly polynomial. This should be contrasted with the relativistic case, where no such simplification happens in the torsion-less limit.

4.3.5 Classification of hydrodynamic transport

Having discussed two particular classes of solutions, we are now ready to exhaust the most generic solutions of the Galilean adiabaticity equation. As the analysis is an exact analogue of our relativistic discussion in section 2.3, we will be brief. Denoting the hydrodynamic constitutive relations as C and the non-hydrodynamic fields as Φ given in eq. (2.54), which are seen as vectors in a hybrid vector space \mathfrak{V} , the Galilean adiabaticity equation (4.58) can be expressed in a compact form

$$\underline{\mathbf{D}}_{M}N^{M} = \mathbf{N}_{\mathrm{H}}^{\perp} + \mathcal{C} \cdot \delta_{\mathcal{B}} \Phi + \Delta, \qquad \Delta \ge 0.$$
(4.85)

As this equation has the exact same structure as its relativistic counterpart in eq. (2.55), we can follow through the same discussion for classifying all of its solutions. To summarise, the hydrodynamic constitutive relations satisfying eq. (4.85) can be classified into six distinct classes:

- Hydrostatic sector: In this sector, constitutive relations and free energy current are constructed out of independent tensor structures that, or any of their linear combinations, do not vanish in a hydrostatic configuration. That is to say that they cannot contain an instance of " $\delta_{\mathcal{B}}$ ". They are classified into three classes:
 - 1. Class A (anomaly induced transport): These constitutive relations are induced by the anomalies in our symmetries. They are completely fixed in terms of the anomaly polynomial, which in turn is characterised by a set of constant anomaly coefficients. See section 4.3.3.
 - 2. Class $\mathbf{H}_{\mathbf{V}}$ (hydrostatic vector transport): These are the non-anomalous hydrostatic constitutive relations whose corresponding free-energy current flows transverse to β^{M} and V^{M} . They are also completely determined up to some arbitrary constants. See section 4.3.4.
 - 3. Class H_S (hydrostatic scalar transport): These are the hydrostatic constitutive relations whose free energy current has a component that flows along β^M .

They are characterised by a hydrostatic scalar density \mathcal{N} via the formula given in eq. (2.74). See section 2.3.4.

- Non-hydrostatic sector: In this sector, constitutive relations and free energy current are made out of independent tensor structures that vanish in a hydrostatic configuration. That is to say that they involve at least one instance of " $\delta_{\mathcal{B}}$ ". They are classified into two classes:
 - 4. Class $\overline{\mathbf{D}}$ (non-dissipative transport): These are the non-hydrostatic constitutive relations that do not cause any production of entropy. They are completely determined by $\overline{\mathfrak{D}}_n|_{n\geq 0} \in \mathfrak{V} \times \mathfrak{V}$ with $\overline{\mathfrak{D}}_n^{\mathrm{T}} = -(-)^n \overline{\mathfrak{D}}_n$. See section 2.3.5.
 - 5. Class D (dissipative transport): These are the constitutive relations which are solely responsible for the production of entropy. They are completely determined by $\mathfrak{D}_n|_{n\geq 0} \in \mathfrak{V} \times \mathfrak{V}$ with $\overline{\mathfrak{D}}_n^{\mathrm{T}} = (-)^n \overline{\mathfrak{D}}_n$. See section 2.3.5.
- Finally, we also have a set of trivial solutions to the adiabaticity equation
 - 6. Class S (entropy transport): This class contains solutions of the adiabaticity equation with vanishing constitutive relations but non-trivial free energy and entropy transport. These include $N^{M} \sim \mathcal{N}_{V}V^{M} + \underline{\mathbf{D}}_{N}X^{MN} + \frac{1}{2}\mathbf{T}^{M}{}_{NR}X^{R}$ for a scalar \mathcal{N}_{V} and an antisymmetric tensor X^{MN} , for which $\underline{\mathbf{D}}_{M}N^{M}$ is trivially zero. The remainder of the class is completely characterised by a set of matrices $\mathfrak{S}_{mn}|_{m,n\geq 1}$ with $\mathfrak{S}_{mn}^{\mathrm{T}} = \mathfrak{S}_{nm}$. See section 2.3.6.

The Class S constitutive relations are not genuine hydrodynamic transport. They merely characterise the multitude of entropy currents which satisfy the second law for the same set of constitutive relations.

This finishes our discussion of the null/Galilean fluid constitutive relations and their classification following from the second law of thermodynamics. In the next chapter, we study some specific examples of Galilean hydrodynamics to illustrate how this construction works.

Applications: Galilean hydrodynamics

In chapter 4 we formulated a new framework to study Galilean hydrodynamics. In this chapter, we explore how it works in practice. We first study ordinary Galilean fluids at one-derivative order and later extend to Galilean superfluids and Galilean fluids with surfaces. For concreteness, we focus on d = 4, which means that our null fluids are 5-dimensional, but Galilean fluids are 4-dimensional.

5.1 Galilean fluids

Our first example are the ordinary Galilean fluids which we derived heuristically in section 1.1.3. They do not possess any additional gapless modes, neither do that have any spin-current or torsion. The absence of background torsion requires us to set $\partial_{[M}V_{N]} = 0$ in our final results. These Galilean fluids are the most standard examples of Galilean hydrodynamics, or hydrodynamics at large. The results obtained here can, therefore, be verified with a standard text like [10].

5.1.1 Ideal null fluids

Let us start with the zero derivative order ideal null fluids. Class H_S is the only non-empty class at this order, which is characterised by a hydrostatic scalar density \mathcal{N} , given by an arbitrary scalar $P(T, \mu, \mu_m)$ as a function of the temperature T, chemical potential μ , and mass chemical potential μ_m of the fluid. The $\delta_{\mathcal{B}}$ variations of these ideal order scalars are given as

$$\delta_{\mathcal{B}}T = TV^{(M}u^{N)}\delta_{\mathcal{B}}g_{MN}, \quad \delta_{\mathcal{B}}\frac{\mu_{\mathrm{m}}}{T} = \frac{1}{2T}u^{M}u^{N}\delta_{\mathcal{B}}g_{MN}, \quad \delta_{\mathcal{B}}\frac{\mu}{T} = \frac{1}{T}u^{M}\delta_{\mathcal{B}}A_{M}. \tag{5.1}$$

Using these, we can compute the divergence of the ideal order free energy current to be

$$D_{\mu}(P\beta^{M}) = \frac{1}{\sqrt{-g}} \delta_{\mathcal{B}} \left(\sqrt{-g} P \right) = \frac{1}{2} P g^{MN} \delta_{\mathcal{B}} g_{MN} + \frac{\partial P}{\partial T} \delta_{\mathcal{B}} T + \frac{\partial P}{\partial \mu_{m}} \delta_{\mathcal{B}} \mu_{m} + \frac{\partial P}{\partial \mu} \cdot \delta_{\mathcal{B}} \mu,$$

$$= \left(R u^{M} u^{N} + 2E u^{(M} V^{N)} + P P^{MN} \right) \frac{1}{2} \delta_{\mathcal{B}} g_{MN} + Q u^{M} \cdot \delta_{\mathcal{B}} A_{M},$$
(5.2)

where $P^{MN} = g^{MN} + 2u^{(M}V^{N)}$ and we have defined

$$S = \frac{\partial P}{\partial T}, \qquad R = \frac{\partial P}{\partial \mu_{\rm m}}, \qquad Q = \frac{\partial P}{\partial \mu}, \qquad E = T\frac{\partial P}{\partial T} + \mu_{\rm m}\frac{\partial P}{\partial \mu_{\rm m}} + \mu \cdot \frac{\partial P}{\partial \mu} - P. \tag{5.3}$$

Comparing eq. (5.2) with eq. (2.74), we can easily read out the ideal null fluid constitutive relations, free energy, and entropy currents

$$T^{MN} = Ru^{M}u^{N} + 2Eu^{(M}V^{N)} + PP^{MN} + \mathcal{O}(\partial), \qquad J^{M} = Qu^{M} + \mathcal{O}(\partial),$$
$$N^{M} = \frac{1}{T}Pu^{M} + \mathcal{O}(\partial), \qquad J^{M}_{S} = Su^{M} + \mathcal{O}(\partial).$$
(5.4)

Using eq. (5.3), we can also show that the coefficients appearing here follow a set of thermodynamic relations

Gibbs-Duhem equation:
$$dP = S dT + R d\mu_m + Q \cdot d\mu,$$

Euler scaling relation: $E + P = ST + R \mu_m + Q \cdot \mu,$
First law of thermodynamics: $dE = T dS + \mu_m dR + \mu \cdot dQ.$ (5.5)

Similar to the ideal relativistic fluids, we see that the ideal null fluids also are completely characterised by their equation of state $P = P(T, \mu, \mu_m)$. We have arrived at the null fluid constitutive relations that we derived in section 1.1.3.

Having worked out the null fluid constitutive relations, we can perform the null reduction prescribed in section 4.3.2 and read out the respective ideal Galilean fluid constitutive relations in an arbitrary frame of reference. We find

$$\epsilon^{\mu} = \left(E + \frac{1}{2}Ru^{a}u_{a}\right)u^{\mu} + P\bar{u}^{\mu} + \mathcal{O}(\partial), \qquad p^{\mu}{}_{a} = Ru^{\mu}u_{a} + Pe_{a}{}^{\mu} + \mathcal{O}(\partial),$$
$$\rho^{\mu} = Ru^{\mu} + \mathcal{O}(\partial), \qquad j^{\mu} = Qu^{\mu} + \mathcal{O}(\partial),$$
$$n^{\mu} = \frac{1}{T}Pu^{\mu} + \mathcal{O}(\partial), \qquad s^{\mu} = Su^{\mu} + \mathcal{O}(\partial). \tag{5.6}$$

From here we can identify E as the energy density, R as the mass density, P as the isotropic pressure, Q as the charge density, and S as the entropy density of the Galilean fluid. Note that $n_{\mu}p^{\mu}{}_{a} = \rho^{\mu}e_{a\mu} = Ru_{a}$, therefore the boost Ward identity in eq. (4.37) is trivially satisfied with $\tau^{\mu a} = 0$. The same is true for the rotation Ward identity as well, because the stress tensor is symmetric, i.e. $p^{\mu[a}e^{b]}{}_{\mu} = 0$. This is characteristic of torsionless Galilean hydrodynamics.

5.1.2 One-derivative corrections

We are interested in the possible one-derivative corrections that these constitutive relations can admit. With this in mind, let us make some definitions which are useful in the following discussion

$$\Theta = \mathcal{D}_{M}u^{M}, \qquad \sigma^{MN} = P^{M\langle R}P^{S\rangle N}\mathcal{D}_{M}u_{N} = P^{MR}P^{NS}\left(\mathcal{D}_{(M}u_{N)} - \frac{1}{3}P_{RS}\Theta\right),$$
$$\mathfrak{a}^{M} = u^{N}\mathcal{D}_{N}u^{M}, \qquad \omega^{M} = \epsilon^{MNRST}V_{N}u_{R}\partial_{S}u_{T},$$
$$E^{M} = F^{MN}u_{N}, \qquad B^{M} = \frac{1}{2}\epsilon^{MNRST}V_{N}u_{R}F_{ST}.$$
(5.7)

${\cal P}_{\sf T}$	$T^{\scriptscriptstyle MN}$	$J^{\scriptscriptstyle M}$	$N^{\scriptscriptstyle M}$
$C \boldsymbol{u} \wedge \boldsymbol{F} \wedge \boldsymbol{F} \wedge \boldsymbol{F}$	$6\mu^2 C V^{(M}B^{N)}$	$6\mu CB^{\scriptscriptstyle M}$	$3 \frac{\mu^2}{T} C B^{\scriptscriptstyle M}$
$C_2 oldsymbol{u} \wedge oldsymbol{F}_{T} \wedge oldsymbol{F}_{T} \wedge oldsymbol{F}$	$2C_2T^2V^{({}_M}B^{{}_N)}$		$C_2 T B^{\scriptscriptstyle M}$
$C_0 \boldsymbol{u} \wedge \boldsymbol{F}_T \wedge \boldsymbol{F} \wedge \boldsymbol{F}$	$4T\mu C_0 V^{(M}B^{N)}$	$2TC_0B^M$	$\begin{array}{c} 2\mu C_0 B^{\scriptscriptstyle M} \\ -\frac{1}{2}C_0 \epsilon^{\scriptscriptstyle MNRST} u_{\scriptscriptstyle N} A_{\scriptscriptstyle R} F_{\scriptscriptstyle ST} \\ +\frac{1}{3}C_0 \epsilon^{\scriptscriptstyle MNRST} u_{\scriptscriptstyle N} A_{\scriptscriptstyle R} A_{\scriptscriptstyle S} A_{\scriptscriptstyle T} \end{array}$
$C_2' oldsymbol{u} \wedge oldsymbol{F}_{T} \wedge oldsymbol{F}_{T} \wedge oldsymbol{\Omega}$	$2C_2'T^2V^{({}_M}\omega^{{}_N)}$		$C_2'T\omega^{\scriptscriptstyle M}$
$C_0' oldsymbol{u} \wedge oldsymbol{F}_{T} \wedge oldsymbol{\Omega} \wedge oldsymbol{F}$	$2T\mu_{\rm m}C_0'V^{(M}B^{N)} +2T\mu C_0'V^{(M}\omega^{N)} +2TC_0'u^{(M}B^{M)}$	$TC_0'\omega^{\scriptscriptstyle M}$	$ \mu C_0' \omega^{\scriptscriptstyle M} + \mu_{\rm m} C_0' B^{\scriptscriptstyle M} \\ - \frac{1}{2} C_0' \epsilon^{\scriptscriptstyle MNRST} u_{\scriptscriptstyle N} A_{\scriptscriptstyle R} \Omega_{\scriptscriptstyle ST} $

Table 5.1: One-derivative Class A and H_V constitutive relations for a (4+1)-dimensional null fluid. Lie algebra traces are understood in columns \mathcal{P}_{T} , T^{MN} , and J^M . Note that the C_0 and C'_0 terms is in the anomaly polynomial are linear in \mathbf{F}_{T} , hence the associated free energy current is not gauge-invariant.

Angular brackets denote a traceless symmetric combination. Note that E^{M} and B^{M} characterise the components of electric and magnetic fields in the fluid frame of reference. In lab frame, these definitions would be slightly different. We can also null reduce these definitions to obtain their respective Newton-Cartan versions

$$\Theta = \tilde{D}_{\mu}u^{\mu}, \qquad \sigma^{\mu\nu} = p^{\mu\rho}p^{\nu\sigma} \left(p_{\tau(\rho}\tilde{D}_{\sigma)}u^{\tau} - \frac{1}{3}p_{\rho\sigma}\Theta \right),$$

$$\mathfrak{a}_{\mu} = u^{\nu}\tilde{D}_{\nu}u_{\mu}, \qquad \omega^{\mu} = -\epsilon^{\mu\nu\rho\sigma}n_{\nu}\partial_{\rho}u_{\sigma},$$

$$E_{\mu} = F_{\mu\nu}u^{\nu}, \qquad B^{\mu} = -\frac{1}{2}\epsilon^{\mu\nu\rho\sigma}n_{\nu}F_{\rho\sigma}.$$
 (5.8)

To find the derivative corrections, let us start with the hydrostatic sector. Just like the relativistic case, Class H_S is empty, while there are non-trivial Class A and Class H_V constitutive relations. They are collectively characterised by a 7-rank thermal anomaly polynomial

$$\mathcal{P}_{\mathsf{T}} = C \, \boldsymbol{u} \wedge \operatorname{tr}[\mathbf{F} \wedge \mathbf{F} \wedge \mathbf{F}] + C_2 \, \boldsymbol{u} \wedge \boldsymbol{F}_{\mathsf{T}} \wedge \operatorname{tr} \boldsymbol{F} + C_0 \, \boldsymbol{u} \wedge \boldsymbol{F}_{\mathsf{T}} \wedge \operatorname{tr}[\boldsymbol{F} \wedge \boldsymbol{F}] + C_2' \, \boldsymbol{u} \wedge \boldsymbol{F}_{\mathsf{T}} \wedge \boldsymbol{\Omega} + C_0' \, \boldsymbol{u} \wedge \boldsymbol{F}_{\mathsf{T}} \wedge \boldsymbol{\Omega} \wedge \operatorname{tr} \boldsymbol{F}.$$
(5.9)

Note that we have not included a term proportional to $\boldsymbol{u} \wedge \boldsymbol{F}_{\mathsf{T}} \wedge \boldsymbol{\Omega} \wedge \boldsymbol{\Omega}$, as it is not allowed by the adiabaticity equation; see the discussion around eq. (4.76). We have also not included the term $\boldsymbol{u} \wedge \boldsymbol{F}_{\mathsf{T}} \wedge \boldsymbol{F}_{\mathsf{T}} \wedge \boldsymbol{F}_{\mathsf{T}}$, which does not contribute on torsion-less null backgrounds. Following sections 4.3.3 and 4.3.4, we can now easily read out the respective constitutive relations. The results have been summarised in table 5.1.

Next, let us consider the non-hydrostatic constitutive relations. Since the definition of " $\delta_{\mathcal{B}}$ " already contains a derivative, at one-derivative order the only relevant non-hydrostatic constitutive relations are parametrised by \mathfrak{C}_0 . It can be split into a symmetric part \mathfrak{D}_0 and an anti-symmetric part $\overline{\mathfrak{D}}_0$ according to eq. (2.80). Using the tensor structures we have available, we can write down 5 possible terms in \mathfrak{D}_0 making up Class D; they are the

Table 5.2: One-derivative Class D constitutive relations for a (4 + 1)-dimensional null fluid.

$\overline{\mathfrak{D}}_0$		$T^{\mu u}$	J^{μ}
$-2T\overline{\kappa}_Q \left(\begin{array}{c} \cdot \\ -V^{(\mathbf{R}}P^{\mathbf{S})_M} \end{array} \right)$	$\begin{pmatrix} V^{(M}P^{N)R} \\ . \end{pmatrix}$	$-2\overline{\kappa}_Q V^{(M}P^{N)R} \left(T \mathcal{D}_R \frac{\mu}{T} - E_R\right)$	$\overline{\kappa}_Q P^{\scriptscriptstyle MN} \frac{1}{\overline{T}} \partial_{\scriptscriptstyle N} T$

Table 5.3: One-derivative Class \overline{D} constitutive relations for a (4+1)-dimensional null fluid.

bulk viscosity ζ , shear viscosity η , thermal conductivity κ , thermo-electric conductivity κ_Q , and electric conductivity σ . The results have been summarised in table 5.2. On the other hand, there is just one transport coefficient in Class \overline{D} , denoted by $\overline{\kappa}_Q$, which is another kind of thermo-electric conductivity. The corresponding constitutive relations have been summarised in table 5.3. Finally, the quadratic form Δ associated with the Class D constitutive relations is given by

$$T\Delta = \zeta \Theta^2 + \eta \sigma^{MN} \sigma_{MN} + \sigma P^{MN} \left(T \mathcal{D}_M \frac{\mu}{T} - E_M \right) \left(T \mathcal{D}_N \frac{\mu}{T} - E_N \right) + 2\kappa_Q P^{MN} \frac{1}{T} \partial_M T \left(T \mathcal{D}_N \frac{\mu}{T} - E_N \right) + \kappa P^{MN} \frac{1}{T^2} \partial_M T \partial_N T.$$
(5.10)

Demanding it to be positive semi-definite yields a set of 4 inequality constraints among the 5 dissipative transport coefficients

$$\zeta \ge 0, \qquad \eta \ge 0, \qquad \sigma \ge 0, \qquad \kappa \ge \kappa_Q^2 / \sigma.$$
 (5.11)

These are the same as we found in section 1.1.3.

In summary, the constitutive relations of a null fluid up to the first order in derivatives are given as

$$T^{MN} = Ru^{M}u^{N} + 2(E+P)u^{(M}V^{N)} + Pg^{MN} - \eta \sigma^{MN} - \zeta P^{MN}\Theta - 2V^{(M}P^{N)R} \left(\kappa \frac{1}{T}\partial_{R}T + (\kappa_{Q} + \overline{\kappa}_{Q}) \left(TD_{R}\frac{\mu}{T} - E_{R}\right)\right) + 2 \left(3\mu^{2}C + C_{2}T^{2} + 2T\mu C_{0} + T\mu_{m}C_{0}'\right) V^{(M}B^{N)} + 2 \left(C_{2}'T^{2} + T\mu C_{0}'\right) V^{(M}\omega^{N)} + 2TC_{0}'u^{(M}B^{N)} + \mathcal{O}(\partial^{2}), J^{M} = Qu^{M} - (\kappa_{Q} - \overline{\kappa}_{Q}) P^{MN}\frac{1}{T}\partial_{N}T - \sigma P^{MN} \left(TD_{N}\frac{\mu}{T} - E_{N}\right) + (6\mu C + 2TC_{0}) B^{M} + TC_{0}'\omega^{M} + \mathcal{O}(\partial^{2}).$$
(5.12)

The associated free energy and entropy currents are

$$N^{M} = \frac{1}{T} P u^{M} + \left(3 \frac{\mu^{2}}{T} C + C_{2}T + 2\mu C_{0} + \mu_{m}C_{0}^{\prime} \right) B^{M} + \left(TC_{2}^{\prime} + \mu C_{0}^{\prime} \right) \omega^{M} - C_{0} \epsilon^{MNRST} u_{N} \left(\frac{1}{2} A_{R} F_{ST} - \frac{1}{3} A_{R} A_{S} A_{T} \right) - \frac{1}{2} C_{0}^{\prime} \epsilon^{MNRST} u_{N} A_{R} \Omega_{ST} + \mathcal{O}(\partial^{2}), J_{S}^{M} = S u^{M} + \left(\frac{\mu_{m}}{T} \kappa + \frac{\mu}{T} \left(\kappa_{Q} - \overline{\kappa}_{Q} \right) \right) P^{MN} \frac{1}{T} \partial_{N} T + \left(\frac{\mu}{T} \sigma + \frac{\mu_{m}}{T} \left(\kappa_{Q} + \overline{\kappa}_{Q} \right) \right) P^{MN} \left(TD_{N} \frac{\mu}{T} - E_{N} \right) + \left(2C_{2}T + 2\mu C_{0} + \mu_{m}C_{0}^{\prime} \right) B^{M} + \left(2TC_{2}^{\prime} + \mu C_{0}^{\prime} \right) \omega^{M} - C_{0} \epsilon^{MNRST} u_{N} \left(\frac{1}{2} A_{R} F_{ST} - \frac{1}{3} A_{R} A_{S} A_{T} \right) - \frac{1}{2} C_{0}^{\prime} \epsilon^{MNRST} u_{N} A_{R} \Omega_{ST} + \mathcal{O}(\partial^{2}).$$
(5.13)

They satisfy the second law of thermodynamics, provided that the ideal order transport coefficients satisfy the thermodynamic relations and the first order dissipative transport coefficients satisfy their respective inequalities. Finally, performing the null reduction yields us the Galilean fluid constitutive relations

$$\rho^{\mu} = Ru^{\mu} + TC_{0}'B^{\mu} + \mathcal{O}(\partial^{2}),$$

$$\epsilon^{\mu} = \left(E + \frac{1}{2}Ru^{a}u_{a} + TC_{0}'B^{\nu}\bar{u}_{\nu}\right)u^{\mu} + P\bar{u}^{\mu} - \eta\,\sigma^{\mu\nu}\bar{u}_{\nu} - \zeta\,\Theta\bar{u}^{\mu}$$

$$- h^{\mu\nu}\left(\kappa\,\frac{1}{T}\partial_{\nu}T + (\kappa_{Q} + \bar{\kappa}_{Q})\left(TD_{\nu}\frac{\mu}{T} - E_{\nu}\right)\right) + \left(C_{2}'T^{2} + T\mu C_{0}'\right)\omega^{\mu}$$

$$+ \left(3\mu^{2}C + C_{2}T^{2} + 2T\mu C_{0} + T\mu_{m}C_{0}' + \frac{1}{2}TC_{0}'u^{a}u_{a}\right)B^{\mu} + \mathcal{O}(\partial^{2})$$

$$p^{\mu}{}_{a} = Ru^{\mu}u_{a} + Pe^{\mu}{}_{a} - \eta\,\sigma^{\mu\nu}e_{\nu a} - \zeta\,\Theta e^{\mu}{}_{a} + TC_{0}'\left(u^{\mu}B_{a} + B^{\mu}u_{a}\right) + \mathcal{O}(\partial^{2}),$$

$$j^{\mu} = Qu^{\mu} - (\kappa_{Q} - \bar{\kappa}_{Q})h^{\mu\nu}\frac{1}{T}\partial_{\nu}T - \sigma\,h^{\mu\nu}\left(TD_{\nu}\frac{\mu}{T} - E_{\nu}\right)$$

$$+ (6\mu C + 2TC_{0})B^{\mu} + TC_{0}'\omega^{\mu} + \mathcal{O}(\partial^{2}).$$
(5.14)

If we instead wanted these results in the mass frame, we could perform a frame transformation $u^{\mu} \rightarrow u^{\mu} - TC'_0 B^{\mu}/R$. In this frame, the mass current is just $\rho^{\mu}_{\text{mass}} = Ru^{\mu}$, while the other currents look like

$$\epsilon_{\rm mass}^{\mu} = \left(E + \frac{1}{2}Ru^{a}u_{a}\right)u^{\mu} + P\bar{u}^{\mu} - h^{\mu\nu}\left(\kappa\frac{1}{T}\partial_{\nu}T + (\kappa_{Q} + \overline{\kappa}_{Q})\left(TD_{\nu}\frac{\mu}{T} - E_{\nu}\right)\right) + \xi_{B}^{\prime}B^{\mu} + \xi_{\omega}^{\prime}\omega^{\mu} - \eta\,\sigma^{\mu\nu}\bar{u}_{\nu} - \zeta\,\Theta\bar{u}^{\mu} + \mathcal{O}(\partial^{2}),$$
$$p_{\rm mass}^{\mu}{}_{a} = Ru^{\mu}u_{a} + Pe^{\mu}{}_{a} - \eta\,\sigma^{\mu\nu}e_{\nu a} - \zeta\,\Theta e^{\mu}{}_{a} + \mathcal{O}(\partial^{2}),$$
$$j_{\rm mass}^{\mu} = Qu^{\mu} - (\kappa_{Q} - \overline{\kappa}_{Q})\,h^{\mu\nu}\frac{1}{T}\partial_{\nu}T - \sigma\,h^{\mu\nu}\left(TD_{\nu}\frac{\mu}{T} - E_{\nu}\right) + \xi_{B}B^{\mu} + \xi_{\omega}\omega^{\mu} + \mathcal{O}(\partial^{2}).$$
(5.15)

Here we have defined the parity-odd transport coefficients

$$\xi_B = 6\mu C + 2TC_0 - \frac{Q}{R}TC'_0, \qquad \xi_\omega = TC'_0,$$

$$\xi'_B = 3\mu^2 C + C_2T^2 + 2T\mu C_0 - \frac{E + P - \mu_{\rm m}R}{R}TC'_0, \qquad \xi'_\omega = C'_2T^2 + T\mu C'_0. \tag{5.16}$$

This is the standard form of the first order Galilean hydrodynamics, which was first derived in full generality in [3]. Note that these constitutive relations still satisfy $n_{\mu}p^{\mu}{}_{a} = \rho^{\mu}e_{a\mu} = Ru_{a}$ and $p^{\mu[a}e^{b]}{}_{\mu} = 0$, so the boost and rotation Ward identities are identically satisfied.

5.2 Galilean superfluids

In this example, we study Abelian Galilean superfluids. These are essentially Galilean fluids with a spontaneously broken U(1) symmetry. The associated Goldstone phase field φ serves as a gapless mode in the hydrodynamic description. The first theory of Galilean superfluid dynamics was written down by London [129] and was later elaborated upon by Landau and Tisza [130, 131], to describe the phenomenology of liquid ²He. The ideal order results can be found in [10], which we have extended to include one-derivative corrections in our work [4]. The results presented here have been taken directly from [4].

5.2.1 Goldstone modes and Josephson equation

Under an infinitesimal symmetry transformation parametrised by $\mathfrak{X} = (\chi^{M}, \Lambda_{\chi})$, the Goldstone mode φ transforms as $\delta_{\mathfrak{X}} \varphi = \chi^{M} \partial_{M} \varphi - \Lambda_{\chi}$. We can define a covariant superfluid velocity by a taking a derivative of φ , leading to

$$\xi_{\rm M} = \partial_{\rm M} \varphi + A_{\rm M},\tag{5.17}$$

which satisfies $2\partial_{[M}\xi_{N]} = F_{MN}$. We demand $\delta_{\mathcal{V}}\varphi = V^{M}\partial_{M}\varphi - \Lambda_{V} = -1$. The value -1 is purely a choice; different choices yield the same results up to a hydrodynamic frame transformation. With the current choice, we can make a direct comparison of our results with Landau [10]. Due to the compatibility condition of the connection A_{M} , it follows that

$$V^{\scriptscriptstyle M}A_{\scriptscriptstyle M} + \Lambda_V = 0 \implies V^{\scriptscriptstyle M}\xi_{\scriptscriptstyle M} = -1.$$
(5.18)

Let us call the equation of motion for φ to be

$$K \approx 0. \tag{5.19}$$

Following eq. (4.58), we can write down an adiabaticity equation for null superfluids

$$D_{M}N^{M} = \frac{1}{2}T^{MN}\delta_{\mathcal{B}}g_{MN} + J^{M}\delta_{\mathcal{B}}A_{M} + K\delta_{\mathcal{B}}\varphi + \Delta, \qquad \Delta \ge 0.$$
(5.20)

where

$$\delta_{\mathcal{B}}\varphi = \beta^{M}\partial_{M}\varphi - \Lambda_{\beta} = \frac{1}{T}\left(u^{M}\xi_{M} + \mu_{m} - \mu\right).$$
(5.21)

Recycling our relativistic discussion from section 3.2.1, we can solve the adiabaticity equation (5.20) at the zeroth order in derivatives and get the Josephson equation for null superfluids

$$K = -\alpha \delta_{\mathcal{B}} \varphi \approx 0 \quad \Longrightarrow \quad u^{M} \xi_{M} \approx \mu - \mu_{m} + \mathcal{O}(\partial).$$
(5.22)

Note that this equation is slightly different compared to relativistic superfluids, due to the presence of the mass chemical potential $\mu_{\rm m}$ on the right hand side. This is due to the fact that in a Galilean superfluid, a linear combination of U(1)-mass and U(1)-internal symmetry is broken, while the other linear combination remains unbroken. In the null reduced form, it equivalently implies

$$-\mu_s - \frac{1}{2} h_{\mu\nu} \zeta^{\mu} \zeta^{\nu} \approx \mu - \mu_{\rm m} + \mathcal{O}(\partial), \qquad (5.23)$$

where $\zeta^{\mu} = \xi^{\mu} - u^{\mu}$ is the relative velocity of the superfluid component with respect to the ordinary fluid component. The scalar $\mu_s = -\frac{1}{2}\xi^M\xi_M$, on the other hand, is called the superfluid potential. In practise, the Josephson equation can be used to eliminate one of the chemical potentials. In [10] e.g., Landau is working in a setup with no U(1) chemical potential, i.e. $\mu = 0$, and has implicitly eliminated the mass chemical potential via $\mu_m \approx \mu_s + \frac{1}{2}h_{\mu\nu}\zeta^{\mu}\zeta^{\nu}$. Care should be taken when comparing the results, because what we are calling μ_s is denoted by μ in [10].

5.2.2 Ideal null superfluids

Let us now work out the ideal null superfluids. As always, Class H_S is the only nonempty class for ideal null superfluids as well. However, the hydrostatic scalar density \mathcal{N} characterising Class H_S is given by a function of four variables $P(T, \mu, \mu_{\rm m}, \mu_s)$. We have omitted the only other possible ideal order scalar $u^M \xi_M$ in the functional dependence, because it is not independent on-shell due to the Josephson equation. We have already provided the $\delta_{\mathcal{B}}$ variations of T, $\mu_{\rm m}$, and μ in eq. (5.1); for μ_s we find

$$\delta_{\mathcal{B}}\mu_{s} = \frac{1}{2}\xi^{M}\xi^{N}\delta_{\mathcal{B}}g_{\mu\nu} - \xi^{M}\delta_{\mathcal{B}}A_{M} - \xi^{M}D_{M}\delta_{\mathcal{B}}\varphi.$$
(5.24)

We can now compute the divergence of the free energy current

$$D_{M}(\beta^{M}P) = \frac{1}{\sqrt{-g}}\delta_{\mathcal{B}}\left(\sqrt{-g}P\right)$$

$$= \frac{1}{2}Pg^{\mu\nu}\delta_{\mathcal{B}}g_{\mu\nu} + \frac{\partial P}{\partial T}\delta_{\mathcal{B}}T + \frac{\partial P}{\partial\mu}\delta_{\mathcal{B}}\mu + \frac{\partial P}{\partial\mu_{m}}\delta_{\mathcal{B}}\mu_{m} + \frac{\partial P}{\partial\mu_{s}}\delta_{\mathcal{B}}\mu_{s},$$

$$= \left(Ru^{M}u^{N} + 2Eu^{(M}V^{N)} + PP^{MN} + R_{s}\xi^{M}\xi^{N}\right)\frac{1}{2}\delta_{\mathcal{B}}g_{MN}$$

$$+ \left(Qu^{M} - R_{s}\xi^{N}\right)\delta_{\mathcal{B}}A_{M} + D_{M}\left(R_{s}\xi^{M}\right)\delta_{\mathcal{B}}\varphi - D_{M}\left(R_{s}\xi^{M}\delta_{\mathcal{B}}\varphi\right), \qquad (5.25)$$

where we have defined

$$S = \frac{\partial P}{\partial T}, \qquad R = \frac{\partial P}{\partial \mu_{\rm m}}, \qquad Q = \frac{\partial P}{\partial \mu}, \qquad R_s = \frac{\partial P}{\partial \mu_s}, E = T \frac{\partial P}{\partial T} + \mu_{\rm m} \frac{\partial P}{\partial \mu_{\rm m}} + \mu \frac{\partial P}{\partial \mu} - P.$$
(5.26)

Comparing eq. (5.25) with eq. (2.74), we can read out the ideal null superfluid constitutive relations, free energy, and entropy currents

$$T^{MN} = Ru^{M}u^{N} + 2(E+P)u^{(M}V^{N)} + Pg^{MN} + R_{s}\xi^{M}\xi^{N} + \mathcal{O}(\partial),$$

$$J^{M} = Qu^{M} - R_{s}\xi^{M} + \mathcal{O}(\partial),$$

$$K = -\alpha\delta_{B}\varphi + D_{M}(R_{s}\xi^{M}) + \mathcal{O}(\partial),$$

$$N^{M} = \frac{1}{T}Pu^{M} + \delta_{B}\varphi R_{s}\xi^{M} + \mathcal{O}(\partial),$$

$$J^{M}_{S} = N^{M} - \left(T^{MN}\beta_{N} + \frac{\mu}{T}J^{M}\right) = Su^{M} + \mathcal{O}(\partial).$$
(5.27)

In the entropy current, we have ignored a term proportional to V^{M} , as it does not contribute to the second law. Due to the definitions (5.26), the coefficients appearing above satisfy the thermodynamic relations

Gibbs-Duhem equation:
$$dP = SdT + Qd\mu + Rd\mu_m + R_sd\mu_s$$
,
Euler scaling relation: $E + P = ST + Q\mu$,
First law of thermodynamics: $dE = TdS + \mu dQ + \mu_m dR - R_s d\mu_s$. (5.28)

Using eq. (5.27), we can work out the first correction to the Josephson equation, leading to

$$u^{M}\xi_{M} = \mu - \mu_{m} + \frac{T}{\alpha} D_{M} \left(R_{s}\xi^{M} \right) + \mathcal{O}(\partial).$$
(5.29)

We can perform the null reduction on the ideal null superfluid constitutive relations to obtain their respective Galilean form

$$\rho^{\mu} = Ru^{\mu} + R_s \xi^{\mu} + \mathcal{O}(\partial),$$

$$\epsilon^{\mu} = \left(E + \frac{1}{2}Ru^a u_a\right)u^{\mu} + R_s \xi^{\mu} \left(\mu_s + \frac{1}{2}\xi^a \xi_a\right) + P\bar{u}^{\mu} + \mathcal{O}(\partial)$$

$$p^{\mu}{}_a = Ru^{\mu}u_a + Pe_a{}^{\mu} + R_s \xi^{\mu} \xi_a + \mathcal{O}(\partial), \qquad j^{\mu} = Qu^{\mu} - R_s \xi^{\mu} + \mathcal{O}(\partial),$$

$$n^{\mu} = \frac{1}{T}Pu^{\mu} + \mathcal{O}(\partial), \qquad s^{\mu} = Su^{\mu} + \mathcal{O}(\partial). \tag{5.30}$$

Like ordinary Galilean fluids, P is the isotropic pressure of the superfluid, while R, E, Q, and S are its mass, energy, charge, and entropy densities respectively. The coefficient R_s can be identified with the superfluid density. Note that the linear combination $\rho^{\mu} + j^{\mu}$ does not have any R_s dependence, because the associated linear combination of U(1)-mass and U(1)-internal symmetry is unbroken. The Josephson equation, on the other hand, becomes

$$-\mu_s - \frac{1}{2}h_{\mu\nu}\zeta^{\mu}\zeta^{\nu} = \mu - \mu_m + \frac{T}{\alpha}\tilde{D}_{\mu}\left(R_s\xi^{\mu}\right) + \mathcal{O}(\partial).$$
(5.31)

In the limit $\mu = 0$, these results can be compared with their textbook version in [10].

5.2.3 One-derivative corrections

Let us talk about one-derivative corrections to these constitutive relations. In table 5.4, we have provided a list of one-derivative tensor structures that we require in the following discussion. Note that this list is not complete; it only contains terms that show up in our calculation. We have defined $\zeta^{M} = P^{MN} \xi_{N}$ and $P_{\zeta}^{MN} = P^{MN} - \zeta^{M} \zeta^{N} / \zeta^{2}$, along with their null reduced form $\zeta^{\mu} = \xi^{\mu} - u^{\mu}$ and $p_{\zeta}^{\mu\nu} = p^{\mu\nu} - \zeta^{\mu} \zeta^{\nu} / \zeta^{2}$.

Beginning with the hydrostatic sector, we note that the Class H_S constitutive relations are now characterised by a hydrostatic scalar density \mathcal{N} given as

$$\mathcal{N} = P + f_1 S_1 + f_2 S_2 + f_3 S_3 + g_1 \tilde{S}_{e,1} + g_2 \tilde{S}_{e,2} + g_3 \tilde{S}_{e,3}.$$
 (5.32)

We have chosen not to include the first order hydrostatic scalars: $D_M \xi^M$ and $\xi^M \partial_M \mu_s$, as the former is a total derivative and the latter can be exchanged for $D_M(R_s\xi^M)$, which is not hydrostatic due to the Josephson equation. We define the derivatives of the transport coefficients appearing above as

$$df_{i} = \frac{1}{T} \alpha_{E,i} dT + T \alpha_{R_{n},i} d\frac{\mu_{\mathrm{m}}}{T} + T \alpha_{Q,i} d\frac{\mu}{T} - \frac{1}{2} \left(\alpha_{R_{s},i} - \frac{f_{i}}{2\hat{\mu}_{s}} \right) d\zeta^{2},$$

$$dg_{i} = \frac{1}{T} \tilde{\alpha}_{E,i} dT + T \tilde{\alpha}_{R_{n},i} d\frac{\mu_{\mathrm{m}}}{T} + T \tilde{\alpha}_{Q,i} d\frac{\mu}{T} - \frac{1}{2} \left(\tilde{\alpha}_{R_{s},i} - \frac{g_{i}}{2\hat{\mu}_{s}} \right) d\zeta^{2}, \qquad (5.33)$$

along with

$$\alpha_{E,i} + f_i = \alpha_{S,i}T + \alpha_{Q,i}\mu + \alpha_{R_n,i}\mu_{\rm m}, \qquad \tilde{\alpha}_{E,i} + g_i = \tilde{\alpha}_{S,i}T + \tilde{\alpha}_{Q,i}\mu + \tilde{\alpha}_{R_n,i}\mu_{\rm m}.$$
 (5.34)

By performing a $\delta_{\mathfrak{B}}$ variation of \mathcal{N} and comparing it to eq. (2.74), we can read out the respective Class \mathcal{H}_{S} constitutive relations. The results have been summarised in table 5.5. Now for Class A and Class \mathcal{H}_{V} , we can directly import the results from the respective ordinary fluid calculation in table 5.1. We should note, however, that due to the presence of a "gauge fixed" version of the gauge field $\xi_{M} = A_{M} + \partial_{M}\varphi$, three constants in Class \mathcal{H}_{V} , i.e. C_{2} , C_{0} , and C'_{0} , can be removed by shifting g_{1} , g_{2} , and g_{3} . Therefore, only the C'_{2} term in Class \mathcal{H}_{V} and the C term in Class A remain independent for a Galilean superfluid.

Finally, let us consider the non-hydrostatic sector. We need to write the most generic expression for \mathfrak{C}_0 , which decomposes into a symmetric part \mathfrak{D}_0 and an antisymmetric part $\overline{\mathfrak{D}}_0$. They correspond to Class D and Class D constitutive relations respectively. There are a total of 20 transport coefficients in Class D: $[\beta_{[ij]}]_{5\times 5}$, $[\kappa_{[ij]}]_{3\times 3}$, $[\tilde{\kappa}_{(ij)}]_{3\times 3}$, and $\tilde{\eta}$. On the other hand, there are 25 transport coefficients in Class D: $[\beta_{(ij)}]_{5\times 5}$, $[\kappa_{(ij)}]_{3\times 5}$, $[\kappa_{(ij)}]_{3\times 3}$, $[\tilde{\kappa}_{[ij]}]_{3\times 3}$, and η , including $\beta_{55} = \alpha/T$ discussed in section 5.2.1. The results have been summarised in tables 5.6 and 5.7. The associated quadratic form Δ takes the form

$$T\Delta = \sum_{i,j=1}^{5} S_{i}\beta_{(ij)}S_{j} + \left(\sum_{i,j=1}^{3} V_{i}^{M}\kappa_{(ij)}V_{j,M} + \sum_{i,j=1}^{3} V_{i}^{M}\tilde{\kappa}_{[ij]}\tilde{V}_{j,M}\right) + \eta\sigma^{MN}\sigma_{MN}.$$
 (5.35)

	Null background		Newton-Cartan background	
Non-hydrostatic — on-shell independent				
S_1	$rac{T}{2}P_{\zeta}^{_{MN}}\delta_{\mathbb{B}}g_{_{MN}}$	$P^{\scriptscriptstyle MN}_{\zeta} \mathrm{D}_{\scriptscriptstyle M} u_{\scriptscriptstyle N}$	$p_{\zeta}{}^{\mu}{}_{\nu} ilde{\mathrm{D}}_{\mu}u^{ u}$	
S_2	$TV^{\vec{M}}\zeta^{\scriptscriptstyle N}\delta_{\mathfrak{B}}g_{\scriptscriptstyle MN}$	$\zeta^{\scriptscriptstyle M}\left(rac{1}{T}\partial^{'}_{\scriptscriptstyle M}T+u^{\scriptscriptstyle N}H_{\scriptscriptstyle NM} ight)$	$\zeta^{\mu} \left(\frac{1}{T} \partial_{\mu} T + u^{\nu} H_{\nu\mu} \right)$	
S_3	$\frac{T}{2}\zeta^{M}\zeta^{N}\delta_{\mathcal{B}}g_{MN}$	$\zeta^M \zeta^N \mathcal{D}_M u_N$	$\zeta^{\mu}\zeta_{\nu}\mathrm{D}_{\mu}u^{\nu}$	
S_4	$T\zeta^{M}\delta_{\mathcal{B}}A_{M}$	$\zeta^{M}(T\partial_{M}\frac{\mu}{T}-E_{M})$	$\zeta^{\mu} \left(T \partial_{\mu} \frac{\mu}{T} - E_{\mu} \right)$	
\mathcal{S}_5	$I o_{\mathbb{B}} \varphi$	$\frac{u^{\rm m}\xi_{\rm M}+\mu_{\rm m}-\mu}{(1-1)^{1/2}}$	$\frac{-\frac{1}{2}\zeta^{a}\zeta_{a}-\mu_{s}+\mu_{m}-\mu}{2}$	
V_1^M	$TP_{\zeta}^{MR}V^{N}\delta_{\mathcal{B}}g_{RN}$	$P_{\zeta}^{MN}\left(\frac{1}{T}\partial_{N}T+u^{R}H_{RN}\right)$	$p_{\zeta}^{\mu\nu} \left(\frac{1}{T}\partial_{\nu}T + u^{\rho}H_{\rho\nu}\right)$	
V_2^M	$TP_{\zeta}^{MR}\zeta^{N}\delta_{\mathcal{B}}g_{RN}$	$2P_{\zeta}^{MR}\zeta^{N}D_{(R}u_{N)}$	$2p^{\mu\nu}_{\zeta}\zeta^{\sigma}p_{\rho(\sigma}D_{\nu)}u^{\rho}$	
V ₃ ^m	$IP_{\zeta}^{MN} o_{\mathcal{B}} A_N$	$P_{\zeta}^{\text{min}}\left(I \mathcal{O}_{N} \frac{E}{T} - E_{N}\right)$	$\frac{p_{\zeta} \left(I \partial_{\nu} \frac{E}{T} - E_{\nu}\right)}{\left(I \partial_{\nu} \frac{E}{T} - E_{\nu}\right)}$	
$\sigma^{\scriptscriptstyle MN}_\zeta$	$\frac{T}{2}P_{\zeta}^{R\langle M}P_{\zeta}^{N\rangle S}\delta_{\mathfrak{B}}g_{RS}$	$P_{\zeta}^{MR}P_{\zeta}^{NS}\left(\mathbf{D}_{(R}u_{S)}-\frac{1}{2}P_{RS}^{\zeta}S_{1}\right)$	$p_{\zeta}^{\mu\rho}p_{\zeta}^{\nu\sigma}\left(p_{\tau(\rho}\tilde{\mathrm{D}}_{\sigma)}u^{\tau}-\frac{p_{\rho\sigma}^{\zeta}}{2}S_{1}\right)$	
$\tilde{V}_1^{\scriptscriptstyle M}$	$\epsilon^{\scriptscriptstyle MI}$	$V^{RST}V_N u_R \zeta_S V_{1,T}$	$-\epsilon^{\mu u ho\sigma}n_ u\zeta_ ho V_{1,\sigma}$	
$\tilde{V}_2^{\scriptscriptstyle M}$	$\epsilon^{_{MI}}$	$^{NRST}V_{N}u_{R}\zeta_{S}V_{2,T}$	$-\epsilon^{\mu u ho\sigma}n_{\nu}\zeta_{ ho}V_{2,\sigma}$	
V_3^M	$\epsilon^{\scriptscriptstyle MI}$	$^{NRST}V_{N}u_{R}\zeta_{S}V_{3,T}$	$-\epsilon^{\mu u ho\sigma}n_{ u}\zeta_{ ho}V_{3,\sigma}$	
$ ilde{\sigma}^{\scriptscriptstyle MN}_{\zeta}$	$P_{\zeta}^{\scriptscriptstyle L(M)}$	$\epsilon^{N)RSTP}V_{R}u_{S}\zeta_{T}\sigma^{\zeta}_{PL}$	$-p_{\zeta}^{\lambda(\mu}\epsilon^{\mu)\rho\sigma\tau}n_{\rho}\zeta_{\sigma}\sigma_{\tau\lambda}$	
	Ν	Ion-hydrostatic — on-shell dep	pendent	
S_6	$Tu^{\scriptscriptstyle M}V^{\scriptscriptstyle N}\delta_{\mathfrak{B}}g_{\scriptscriptstyle MN}$	$rac{1}{T}u^{\scriptscriptstyle M}\partial_{\scriptscriptstyle M}T$	$\frac{1}{T}u^{\mu}\partial_{\mu}T$	
S_7	$Tu^{\scriptscriptstyle M} \delta_{\mathcal{B}} A_{\scriptscriptstyle M}$	$T u^{\scriptscriptstyle M} \partial_{\scriptscriptstyle M} rac{\mu}{T}$	$T u^\mu \partial_\mu rac{\mu}{T}$	
S_8	$\frac{T}{2}u^{\scriptscriptstyle M}u^{\scriptscriptstyle N}\delta_{\scriptscriptstyle \mathcal{B}}g_{\scriptscriptstyle MN}$	$T u^M \partial_M \frac{\mu_{\rm m}}{T}$	$Tu^{\mu}\partial_{\mu}rac{\mu_{\mathrm{m}}}{T}$	
S_9	$Tu^{\scriptscriptstyle M}\zeta^{\scriptscriptstyle N}\delta_{\scriptscriptstyle {\rm B}}g_{\scriptscriptstyle MN}$	$\zeta^{\scriptscriptstyle M}\left(T\partial_{\scriptscriptstyle M}rac{\mu_{ m m}}{T}+\mathfrak{a}_{\scriptscriptstyle M} ight)$	$\zeta^{\mu}\left(T\partial_{\mu}rac{\mu_{\mathrm{m}}}{T}+\mathfrak{a}_{\mu} ight)$	
$V_4^{\scriptscriptstyle M}$	$TP_{\zeta}^{\scriptscriptstyle MR}u^{\scriptscriptstyle N}\delta_{\mathfrak{B}}g_{\scriptscriptstyle RN}$	$P_{\zeta}^{MN}\left(T\partial_{N}\frac{\mu_{\mathrm{m}}}{T}+\mathfrak{a}_{N}\right)$	$p_{\zeta}^{\mu\nu}\left(T\partial_{\nu}\frac{\mu_{\mathrm{m}}}{T}+\mathfrak{a}_{\nu}\right)$	
$\tilde{V}_4^{\scriptscriptstyle M}$	$\epsilon^{_{MI}}$	$V^{RST}V_N u_R \zeta_S V_{4,T}$	$-\epsilon^{\mu u ho\sigma}n_{\nu}\zeta_{ ho}V_{4,\sigma}$	
		Hydrostatic		
$S_{e,1}$		$\frac{1}{T}\zeta^M\partial_M T$	$\frac{1}{T}\zeta^{\mu}\partial_{\mu}T$	
$S_{e,2}$	$T \zeta^M \partial_M \frac{\mu}{T}$		${ ilde T}\zeta^\mu\partial_\mu {\mu\over T}$	
$S_{e,3}$		$T\zeta^{\scriptscriptstyle M}\partial_{\scriptscriptstyle M}rac{\mu_{ m m}}{T}$	$T\zeta^{\mu}\partial_{\mu}rac{\mu_{\mathrm{m}}}{T}$	
$V_{e,1}^{\scriptscriptstyle M}$		$\frac{1}{T}P_{\zeta}^{MN}\partial_{N}T$	$rac{1}{T} p^{\mu u}_{\zeta} \partial_{ u} T$	
$V_{e,2}^{\scriptscriptstyle M}$	$TP_{\mathcal{L}}^{\widetilde{M}N}\partial_{N}\frac{\mu}{T}$		$Tp_{\zeta}^{\mu u}\partial_{ u}rac{\mu}{T}$	
$V_{e,3}^{\scriptscriptstyle M}$		$TP_{\zeta}^{MN}\partial_{N}rac{\mu_{m}}{T}$	$T p_{\zeta}^{\mu u} \partial_{\nu} rac{\mu_{ m m}}{T}$	
$\tilde{S}_{e,1}$	$T\epsilon^{M}$	$\overline{\zeta_M V_N u_R} \overline{\partial_S u_T}$	$T\epsilon^{\mu\nu\rho\sigma}n_{\mu}\zeta_{\nu}\partial_{\rho}u_{\sigma}$	
$\tilde{S}_{e,2}$	$\frac{1}{2}T\epsilon^{2}$	$^{MNRST}\zeta_{M}V_{N}u_{R}F_{ST}$	$\frac{T}{2}\epsilon^{\mu u ho\sigma}\dot{n}_{\mu}\zeta_{ u}F_{ ho\sigma}$	
$\tilde{S}_{e,3}$	$\frac{1}{2}T\epsilon^{i}$	$^{MNRST}\zeta_{M}V_{N}u_{R}H_{ST}$	$\frac{T}{2}\epsilon^{\mu\nu\rho\sigma}n_{\mu}\zeta_{\nu}H_{\rho\sigma}$	
$\tilde{V}_{e,1}^{\scriptscriptstyle M}$	TP_{ζ}^{M}	$K_{K} \epsilon^{KNRST} V_{N} u_{R} \partial_{S} u_{T}$	$-Tp_{\zeta}{}^{\mu}{}_{\tau}\varepsilon^{\tau\nu\rho\sigma}n_{\nu}\partial_{\rho}u_{\sigma}$	
$\tilde{V}_{e,2}^{\scriptscriptstyle M}$	$\frac{1}{2}TP\zeta$	$\int_{-K}^{M} \epsilon^{KNRST} V_N u_R F_{ST}$	$-\frac{T}{2}p_{\zeta}{}^{\mu}{}_{\tau}\varepsilon^{\tau\nu\rho\sigma}n_{\nu}F_{\rho\sigma}$	
$\tilde{V}_{e,3}^{\scriptscriptstyle M}$	TP_{ζ^N}	$\int_{K}^{A} \epsilon^{KNRST} \xi_{N} u_{R} \partial_{S} u_{T}$	$Tp_{\zeta}^{\mu}{}_{\tau}\varepsilon^{\tau\nu\rho\sigma}\zeta_{\nu}\partial_{\rho}u_{\sigma} + (\mu_s + \frac{1}{2}\zeta^{\mu}\zeta_{\mu})\tilde{V}^{\mu}_{e,1}$	
$\tilde{V}_{e,4}^{\scriptscriptstyle M}$	$\frac{1}{2}TP_{0}$	$\int_{-K}^{M} \epsilon^{\kappa NRST} \xi_N u_R F_{ST}$	$\frac{\frac{T}{2}p_{\zeta}^{\mu}\tau^{\epsilon\tau\nu\rho\sigma}\zeta_{\nu}F_{\rho\sigma}}{+(\mu_{s}+\frac{1}{2}\zeta^{\mu}\zeta_{\mu})\tilde{V}_{e,2}^{\mu}}$	
÷		:	:	

Table 5.4: First order data for 5-dimensional null and 4-dimensional Galilean superfluids. Note that for null fluids coupled to torsionless backgrounds, we must switch off H_{MN} , setting $S_2 = S_{e,1}$, $V_1^M = V_{e,1}^M$, and $\tilde{S}_{e,3} = 0$. However, in the intermediate steps, they are independent.

\mathcal{N}	T^{MN}	$J^{\scriptscriptstyle M}$
$f_1 S_{e,1}$	$ \left(\alpha_{R_n,1} u^{\scriptscriptstyle M} u^{\scriptscriptstyle N} + 2\alpha_{E,1} V^{(\scriptscriptstyle M} u^{\scriptscriptstyle N)} + \alpha_{R_s,1} \lambda^{\scriptscriptstyle MN} + f_1 P_{\zeta}^{\scriptscriptstyle MN} \right) S_{e,1} - 2f_1 \xi^{(\scriptscriptstyle M} V_{e,1}^{\scriptscriptstyle N)} + 2f_1 \zeta^{(\scriptscriptstyle M} V^{\scriptscriptstyle N)} S_6 - 2V^{(\scriptscriptstyle M} u^{\scriptscriptstyle N)} \frac{1}{T} \mathcal{D}_{\scriptscriptstyle R}(Tf_1 \zeta^{\scriptscriptstyle R}) $	$\begin{array}{l} \left(\alpha_{Q,1}u^{\scriptscriptstyle M}-\alpha_{R_s,1}\zeta^{\scriptscriptstyle M}\right)S_{e,1} \\ +f_1V^{\scriptscriptstyle M}_{e,1}\end{array}$
$f_2 S_{e,2}$	$ \begin{pmatrix} \alpha_{R_n,2} u^{\scriptscriptstyle M} u^{\scriptscriptstyle N} + 2\alpha_{E,2} V^{(\scriptscriptstyle M} u^{\scriptscriptstyle N)} + \alpha_{R_s,2} \lambda^{\scriptscriptstyle MN} + f_2 P_{\zeta}^{\scriptscriptstyle MN} \end{pmatrix} S_{e,2} \\ -2f_2 \xi^{(\scriptscriptstyle M} V_{e,2}^{\scriptscriptstyle N)} + 2f_2 \zeta^{(\scriptscriptstyle M} V^{\scriptscriptstyle N)} S_7 $	$\begin{array}{c} (\alpha_{Q,2}u^{\scriptscriptstyle M} - \alpha_{R_s,2}\zeta^{\scriptscriptstyle M}) S_{e,2} \\ + f_2 V^{\scriptscriptstyle M}_{e,2} \\ - u^{\scriptscriptstyle M} \frac{1}{T} \mathcal{D}_{\scriptscriptstyle R}(Tf_2\zeta^{\scriptscriptstyle R}) \end{array}$
$f_3 S_{e,3}$	$ \left(\alpha_{R_n,3} u^{\scriptscriptstyle M} u^{\scriptscriptstyle N} + 2\alpha_{E,3} V^{(\scriptscriptstyle M} u^{\scriptscriptstyle N)} + \alpha_{R_s,3} \lambda^{\scriptscriptstyle MN} + f_3 P^{\scriptscriptstyle MN}_{\zeta} \right) S_{e,3} - 2f_3 \xi^{(\scriptscriptstyle M} V^{\scriptscriptstyle N)}_{e,3} + 2f_3 \zeta^{(\scriptscriptstyle M} V^{\scriptscriptstyle N)} S_8 - u^{\scriptscriptstyle M} u^{\scriptscriptstyle N} \frac{1}{T} \mathcal{D}_{\scriptscriptstyle R}(Tf_3 \zeta^{\scriptscriptstyle R}) $	$\begin{array}{l} (\alpha_{Q,3}u^{\scriptscriptstyle M} - \alpha_{R_s,3}\zeta^{\scriptscriptstyle M})S_{e,3} \\ + f_3V^{\scriptscriptstyle M}_{e,3} \end{array}$
$g_1 ilde{S}_{e,1}$	$ \begin{pmatrix} u^{\scriptscriptstyle M} u^{\scriptscriptstyle N} \tilde{\alpha}_{R_n,1} + 2V^{(\scriptscriptstyle M} u^{\scriptscriptstyle N}) \tilde{\alpha}_{E,1} + \tilde{\alpha}_{R_s,1} \lambda^{\scriptscriptstyle MN} + g_1 \frac{\zeta^{\scriptscriptstyle M} \zeta^{\scriptscriptstyle N}}{\zeta^2} \end{pmatrix} \tilde{S}_{e,1} \\ + 2u^{(\scriptscriptstyle M} V^{\scriptscriptstyle N}) g_1 S_{e,1} + 2u^{(\scriptscriptstyle M} \epsilon^{\scriptscriptstyle N) {\scriptscriptstyle KRST}} \mathcal{D}_{\scriptscriptstyle K} \left(Tg_1 V_R u_S \xi_{\scriptscriptstyle T} \right) \\ - 2g_1 V^{(\scriptscriptstyle M} \tilde{V}_{e,3}^{\scriptscriptstyle N} - 2g_1 u^{(\scriptscriptstyle M} \tilde{V}_{e,1}^{\scriptscriptstyle N}) \end{cases} $	$\begin{array}{l} \left(\tilde{\alpha}_{Q,1}u^{\scriptscriptstyle M}-\tilde{\alpha}_{R_s,1}\zeta^{\scriptscriptstyle M}\right)\tilde{S}_{e,1}\\ +g_1\tilde{V}^{\scriptscriptstyle M}_{e,1}\end{array}$
$g_2 \tilde{S}_{e,2}$	$ \begin{pmatrix} u^{\scriptscriptstyle M} u^{\scriptscriptstyle N} \tilde{\alpha}_{R_n,2} + 2V^{(\scriptscriptstyle M} u^{\scriptscriptstyle N)} \tilde{\alpha}_{E,2} + \tilde{\alpha}_{R_s,2} \lambda^{\scriptscriptstyle MN} + g_2 \frac{\zeta^{\scriptscriptstyle M} \zeta^{\scriptscriptstyle N}}{\zeta^2} \end{pmatrix} \tilde{S}_{e,2} \\ - 2g_4 V^{(\scriptscriptstyle M} \tilde{V}_{e,4}^{\scriptscriptstyle N)} - 2g_2 u^{(\scriptscriptstyle M} \tilde{V}_{e,2}^{\scriptscriptstyle N)} $	$ \begin{array}{l} \left(\tilde{\alpha}_{Q,2} u^{\scriptscriptstyle M} - \tilde{\alpha}_{R_s,2} \zeta^{\scriptscriptstyle M} \right) \tilde{S}_{e,2} \\ + g_2 \tilde{V}_{e,2}^{\scriptscriptstyle M} + \\ \epsilon^{\scriptscriptstyle MNRST} \mathcal{D}_{\scriptscriptstyle N} \left(T g_2 V_{\scriptscriptstyle R} u_{\scriptscriptstyle S} \xi_{\scriptscriptstyle T} \right) \end{array} $
$g_3 \tilde{S}_{e,3}$	$2V^{(M}\epsilon^{N)KRST}D_{K}(g_{3}TV_{R}u_{S}\zeta_{T})$	
\mathcal{N}	K	$N^{\scriptscriptstyle M}$
$f_1 S_{e,1}$	$\mathrm{D}_{\scriptscriptstyle M}\left(\zeta^{\scriptscriptstyle M}lpha_{R_s,1}S_{e,1}-f_1V^{\scriptscriptstyle M}_{e,1} ight)$	$\frac{1}{T}f_1\left(u^{\mu}S_{e,1}-\zeta^{\mu}S_6\right)$
$f_2 S_{e,2}$	$\mathbf{D}_{M}\left(\zeta^{M}\alpha_{Rs,2}S_{e,2}-f_{1}V_{e,2}^{M}\right)$	$\frac{1}{T}f_2 \left(u^{\mu} S_{e,2} - \zeta^{\mu} S_7 \right)$
$f_3 S_{e,3}$	$\mathrm{D}_{\scriptscriptstyle M}\left(\zeta^{\scriptscriptstyle M}lpha_{R_s,3}S_{e,3}-f_3V_{e,3}^{\scriptscriptstyle M} ight)$	$\frac{1}{T}f_3\left(u^{\mu}S_{e,3}-\zeta^{\mu}S_8\right)$
$g_1 S_{e,1}$	$\mathrm{D}_{\scriptscriptstyle M}\left(\zeta^{\scriptscriptstyle M} ilde{lpha}_{R_s,1} S_{e,1} - g_1 V^{\scriptscriptstyle M}_{e,1} ight)$	$g_1\left(\frac{1}{T}u^{\scriptscriptstyle M}S_{e,1}+V_4^{\scriptscriptstyle M}\right)$
$g_2 \tilde{S}_{e,2}$	$\mathrm{D}_{\scriptscriptstyle M}\left(\zeta^{\scriptscriptstyle M} ilde{lpha}_{R_s,2} ilde{S}_{e,2}-g_2 ilde{V}^{\scriptscriptstyle M}_{e,2} ight)$	$g_2\left(rac{1}{T}u^{\scriptscriptstyle M} ilde{S}_{e,2}+ ilde{V}_3^{\scriptscriptstyle M} ight)$
$g_3 \tilde{S}_{e,3}$	×	$g_3 ilde V_1^{\scriptscriptstyle M}$

Table 5.5: One-derivative Class H_S constitutive relations for a (3 + 1)-dimensional relativistic superfluid. We have defined $\lambda^{MN} = \zeta^M \zeta^N + 2\zeta^{(M} u^N) - 2\zeta^{(M} V^N) (u^R \xi_R)$. Note that even thought $\tilde{S}_{e,3}$ is zero on a torsionless background, its variation does lead to non-trivial constitutive relations.

Noting the identity

$$\left(\epsilon^{^{MNRST}}V_{R}u_{S}\zeta_{T}\right)\left(\epsilon_{^{MKLOP}}V^{^{L}}u^{^{O}}\zeta^{^{P}}\right) = \zeta^{^{2}}P_{\zeta_{K}}^{^{N}},\tag{5.36}$$

we can define a new basis for one-derivative vectors

$$\begin{pmatrix} V_1'^M \\ V_2'^M \\ V_3'^M \end{pmatrix} = \begin{pmatrix} V_1^M \\ V_2^M \\ V_3^M \end{pmatrix} + \begin{pmatrix} 0 & a_{12} & a_{13} \\ 0 & 0 & a_{23} \\ 0 & 0 & 0 \end{pmatrix} \begin{pmatrix} \tilde{V}_1'^M \\ \tilde{V}_2'^M \\ \tilde{V}_3'^M \end{pmatrix}, \qquad \kappa_{ij}' = \kappa_{ij} + k_{ij}, \qquad (5.37)$$

\mathfrak{D}_0	$T^{\mu u}$	J^{μ}	K
$-T\beta_{11}\begin{pmatrix}P_{\zeta}^{\scriptscriptstyle MN}P_{\zeta}^{\scriptscriptstyle RS}&\cdot&\cdot\\\cdot&\cdot&\cdot\end{pmatrix}$	$-\beta_{11}P^{\rm\scriptscriptstyle MN}_\zeta S_1$		
$-2T\beta_{(12)}\left(P_{\zeta}^{MN}\zeta^{(R}V^{S)}+P_{\zeta}^{RS}\zeta^{(M}V^{N)}::\right)$	$-\beta_{(12)}P_{\zeta}^{\scriptscriptstyle MN}S_2$		
$-T\beta_{(12)}\left(P_{\zeta}^{MN}\zeta^{R}\zeta^{S}+P_{\zeta}^{RS}\zeta^{M}\zeta^{N}\cdot\cdot\right)$	$\frac{-2\beta_{(12)}\zeta^{MV}V^{N}S_{1}}{-\beta_{(13)}P_{\zeta}^{MN}S_{3}}$		
$\frac{1}{2} P_{(13)} \left(\begin{array}{c} \vdots \\ \vdots \\ P_{\zeta}^{MN} \zeta^{R} \end{array} \right)$	$-\beta_{(13)}\zeta^{\scriptscriptstyle M}\zeta^{\scriptscriptstyle N}S_1$	$\rho \leftarrow MC$	
$-I \beta_{(14)} \begin{pmatrix} P_{\zeta}^{RS} \zeta^{M} & \ddots & \ddots \\ \ddots & \ddots & \ddots & \ddots \\ (\ddots & \ddots & P_{c}^{MN}) \end{pmatrix}$	$-\beta_{(14)}P_{\zeta}^{\mathrm{min}}S_4$	$-\beta_{(14)}\zeta^{m}S_{1}$	
$-T\beta_{(15)}\begin{pmatrix} \cdot & \cdot & \cdot \\ P_{\zeta}^{RS} & \cdot & \cdot \\ & & \cdot \end{pmatrix}$	$-\beta_{(15)} P_{\zeta}^{\scriptscriptstyle MN} S_6$		$-\beta_{(15)}S_1$
$-4T\beta_{22}\begin{pmatrix} V^{(M}\zeta^{N)}\zeta^{(R}V^{S)} & \vdots & \vdots \\ \vdots & \vdots & \vdots & \vdots \end{pmatrix}$	$-2\beta_{22}V^{(M}\zeta^{N)}S_2$		
$-2T\beta_{(23)} \begin{pmatrix} V^{(M}\zeta^{N)}\zeta^{R}\zeta^{S} + V^{(R}\zeta^{S)}\zeta^{M}\zeta^{N} \\ \vdots \\ \ddots \end{pmatrix}$	$-2eta_{(23)}V^{({}^{M}\zeta^{{}^{N})}S_{3}}\ -eta_{(23)}\zeta^{{}^{M}}\zeta^{{}^{N}}S_{2}$		
$-2T\beta_{(24)}\begin{pmatrix} V^{(M}\zeta^{N)}\zeta^{R} \\ V^{(R}\zeta^{S)}\zeta^{M} \\ \vdots \end{pmatrix}$	$-2\beta_{(24)}V^{(M}\zeta^{N)}S_{4}$	$-\beta_{(24)}\zeta^{\scriptscriptstyle M}S_2$	
$-2T\beta_{(25)}\begin{pmatrix} \vdots & \vdots & V^{(\scriptscriptstyle M}\zeta^{\scriptscriptstyle N)} \\ V^{(\scriptscriptstyle R}\zeta^{\scriptscriptstyle S)} & \vdots & \vdots \end{pmatrix}'$	$-2\beta_{(25)}V^{(M}\zeta^{N)}S_{6}$		$-\beta_{(25)}S_2$
$-T\beta_{33}\begin{pmatrix} \zeta^{\scriptscriptstyle M}\zeta^{\scriptscriptstyle N}\zeta^{\scriptscriptstyle N}\zeta^{\scriptscriptstyle N}\zeta^{\scriptscriptstyle S} & \vdots \\ \vdots & \vdots \end{pmatrix}$	$-eta_{33}\zeta^{\scriptscriptstyle M}\zeta^{\scriptscriptstyle N}S_3$		
$-T\beta_{(34)}\begin{pmatrix} \zeta^{\scriptscriptstyle R}\dot{\zeta}^{\scriptscriptstyle S}\zeta^{\scriptscriptstyle M} & \vdots \\ \vdots & \vdots \end{pmatrix}$	$-\beta_{(34)}\zeta^{(\scriptscriptstyle M}\zeta^{\scriptscriptstyle N)}S_4$	$-eta_{(34)}\zeta^{\scriptscriptstyle M}S_3$	
$-T\beta_{(35)}\begin{pmatrix} \vdots & \zeta^{\scriptscriptstyle M}\zeta^{\scriptscriptstyle N}\\ \zeta^{\scriptscriptstyle R}\zeta^{\scriptscriptstyle S} & \cdot & \cdot \end{pmatrix}$	$-eta_{(35)}\zeta^{\scriptscriptstyle M}\zeta^{\scriptscriptstyle N}S_6$		$-\beta_{(35)}S_3$
$-T\beta_{44} \begin{pmatrix} \vdots & \zeta^{\scriptscriptstyle M} \zeta^{\scriptscriptstyle R} & \vdots \end{pmatrix}$		$-\beta_{44}\zeta^{\scriptscriptstyle M}S_4$	
$-T\beta_{(45)}\left(\begin{array}{ccc} \cdot & \cdot & \cdot \\ \cdot & \cdot & \cdot \\ \cdot & \cdot & \cdot \end{array}\right)$		$-\beta_{(45)}\zeta^{\scriptscriptstyle M}S_5$	$-\beta_{(45)}S_4$
$-T\beta_{55}\left(\begin{array}{ccc} \cdot & \cdot & \cdot \\ \cdot & \cdot & 1\end{array} ight)$			$-\beta_{55}S_5$
$-4T\kappa_{11}\begin{pmatrix}V^{({}_{M}}P^{{}_{N})({}_{R}}V^{{}_{S})}&\cdot&\cdot\\\cdot&\cdot&\cdot\\\cdot&\cdot&\cdot\end{pmatrix}$	$2\kappa_{11} V^{(M}V_1^{N)}$		
$-4T\kappa_{(12)}\begin{pmatrix} V^{({}_{M}}P^{{}_{\lambda})({}_{R}}\zeta^{{}_{S})}+\zeta^{({}_{M}}P^{{}_{\lambda})({}_{R}}V^{{}_{S})}& \vdots\\ \vdots & \vdots \end{pmatrix}$	$-2\kappa_{(12)}V^{({}_MV_2^{N)}}\ -2\kappa_{(12)}\zeta^{{}_MV_1^{N}}$		
$-2T\kappa_{(13)}\begin{pmatrix} \cdot & V^{(M}P_{\zeta}^{N)R} & \cdot \\ V^{(R}P_{\zeta}^{S)M} & \cdot & \cdot \\ \cdot & \cdot & \cdot & \cdot \end{pmatrix}$	$-2\kappa_{(13)}V^{(M}V_3^{N)}$	$-\kappa_{(13)}V_1^{\scriptscriptstyle M}$	
$-4T\kappa_{22}\left(\begin{array}{ccc}V^{({}_{M}P_{\zeta}^{N})^{(R}}V^{S})&\cdot&\cdot\\\cdot&\cdot&\cdot\end{array}\right)\right)$	$2\kappa_{22}\zeta^{(\scriptscriptstyle M}V_2^{\scriptscriptstyle N)}$		
$-2T\kappa_{(23)}\begin{pmatrix} \cdot & \zeta^{(M}P_{\zeta}^{N)R} & \cdot \\ \zeta^{(R}P_{\zeta}^{S)M} & \cdot & \cdot \\ \cdot & \cdot & \cdot \end{pmatrix}$	$-2\kappa_{(23)}\zeta^{(\scriptscriptstyle M}V_3^{\scriptscriptstyle N)}$	$-\kappa_{(23)}V_2^{\scriptscriptstyle M}$	
$-T\kappa_{33}\begin{pmatrix} \cdot & \cdot & \cdot \\ \cdot & P^{MR}_{\zeta} & \cdot \end{pmatrix}$		$-\kappa_{33}V_3^{\scriptscriptstyle M}$	
$4T\tilde{\kappa}_{[12]} \begin{pmatrix} V^{(M}\tilde{\epsilon}^{N)(R}\zeta^{S} + \zeta^{(M}\tilde{\epsilon}^{N)(R}V^{S} & \vdots \\ \vdots & \vdots \end{pmatrix}$	$-2\tilde{\kappa}_{[12]}V^{(M}\tilde{V}_{2}^{N)}$ $+2\tilde{\kappa}_{[12]}\epsilon^{(M}\tilde{V}_{2}^{N)}$		
$2T\tilde{\kappa}_{[13]} \begin{pmatrix} V^{(_{R}\tilde{\epsilon}^{S})_{M}} & \cdot \\ -V^{(_{R}\tilde{\epsilon}^{S})_{M}} & \cdot & \cdot \end{pmatrix}$	$-2\tilde{\kappa}_{[13]}V^{(M}\tilde{V}_{3}^{N)}$	$2\tilde{\kappa}_{[13]}\tilde{V}_1^{\scriptscriptstyle M}$	
$2T\tilde{\kappa}_{[23]} \begin{pmatrix} \zeta^{(M}\tilde{\epsilon}^{N)R} & \vdots \\ -\zeta^{(R}\tilde{\epsilon}^{S)M} & \vdots & \vdots \end{pmatrix}'$	$-2\tilde{\kappa}_{[23]}\zeta^{(\scriptscriptstyle M}\tilde{V}_3^{\scriptscriptstyle N)}$	$2\tilde{\kappa}_{[23]}\tilde{V}_2^{\scriptscriptstyle M}$	
$-T\eta \left(\begin{smallmatrix} P_{\zeta}^{\scriptscriptstyle M\langle R} P_{\zeta}^{\scriptscriptstyle S\rangle \scriptscriptstyle N} & \cdot & \cdot \\ \cdot & \cdot & \cdot \\ \cdot & \cdot & \cdot \end{smallmatrix} \right)$	$-\eta\sigma_\zeta^{_{MN}}$		

Table 5.6: One-derivative Class D constitutive relations for a (4 + 1)-dimensional null superfluid. We have defined $\tilde{\epsilon}^{_{MN}} = \epsilon^{_{MNRST}} V_R u_S \zeta_T$. We have also included $\beta_{55} = \alpha/T$ in K for completeness.

$\overline{\mathfrak{D}}_0$	$T^{\mu u}$	J^{μ}	K
$-2T\beta_{[12]} \begin{pmatrix} P_{\zeta}^{\scriptscriptstyle MN}\zeta^{\scriptscriptstyle (R}V^{\scriptscriptstyle S)} - P_{\zeta}^{\scriptscriptstyle RS}\zeta^{\scriptscriptstyle (M}V^{\scriptscriptstyle N)} & \vdots \\ \vdots & \vdots & \vdots \end{pmatrix}$	$\begin{array}{c} -\beta_{[12]} P_{\zeta}^{_{MN}} S_{2} \\ +2\beta_{[12]} \zeta^{(_{M}} V^{_{N})} S_{1} \end{array}$		
$-T\beta_{[13]}\begin{pmatrix} P^{\scriptscriptstyle M\!N}_{\zeta}\zeta^{\scriptscriptstyle R}\zeta^{\scriptscriptstyle S}-P^{\scriptscriptstyle R\!S}_{\zeta}\zeta^{\scriptscriptstyle M}\zeta^{\scriptscriptstyle N}&\vdots\\\vdots&\vdots&\vdots\end{pmatrix}$	$-eta_{[13]}P^{_MN}_\zeta S_3 \ +eta_{[13]}\zeta^{_M}\zeta^{_N}S_1$		
$-T\beta_{[14]}\begin{pmatrix} & P_{\zeta}^{\scriptscriptstyle NN}\zeta^{\scriptscriptstyle R} & \cdot \\ -P_{\zeta}^{\scriptscriptstyle RS}\zeta^{\scriptscriptstyle M} & \cdot & \cdot \end{pmatrix}$	$-eta_{(14)} P_{\zeta}^{\scriptscriptstyle MN} S_4$	$\beta_{[14]}\zeta^{\scriptscriptstyle M}S_1$	
$-T\beta_{[15]} \begin{pmatrix} \vdots & \vdots & P_{\zeta}^{\scriptscriptstyle MN} \\ -P_{\zeta}^{\scriptscriptstyle RS} & \vdots & \vdots \end{pmatrix}$	$-\beta_{[15]}P_\zeta^{\rm\scriptscriptstyle MN}S_6$		$\beta_{[15]}S_1$
$-2T\beta_{[23]} \begin{pmatrix} V^{(\scriptscriptstyle M}\zeta^{\scriptscriptstyle N})\zeta^{\scriptscriptstyle R}\zeta^{\scriptscriptstyle S} - V^{(\scriptscriptstyle R}\zeta^{\scriptscriptstyle S})\zeta^{\scriptscriptstyle M}\zeta^{\scriptscriptstyle N} \\ \vdots \end{pmatrix}$	$-2\beta_{[23]}V^{({}_{M}\zeta^{{}_{N}})}S_{3} +\beta_{[23]}\zeta^{{}_{M}}\zeta^{{}_{N}}S_{2}$		
$-2T\beta_{[24]}\begin{pmatrix} V^{({}^{\scriptscriptstyle N}\!\zeta^{\scriptscriptstyle N})}\zeta^{\scriptscriptstyle R} \\ -V^{({}^{\scriptscriptstyle R}\!\zeta^{\scriptscriptstyle S})}\zeta^{\scriptscriptstyle M} & \vdots \end{pmatrix}$	$-2\beta_{[24]}V^{(M}\zeta^{N)}S_4$	$\beta_{[24]}\zeta^{\scriptscriptstyle M}S_2$	
$-2T\beta_{[25]}\begin{pmatrix} \cdot & \cdot & V^{(\scriptscriptstyle M}\zeta^{\scriptscriptstyle N})\\ -V^{(\scriptscriptstyle R}\zeta^{\scriptscriptstyle S)} & \cdot & \cdot \end{pmatrix}$	$-2\beta_{[25]}V^{(M}\zeta^{N)}S_{6}$		$\beta_{[25]}S_2$
$-T\beta_{[34]}\begin{pmatrix} \zeta^{R}\zeta^{S}\zeta^{M} & \zeta^{M}\zeta^{N}\zeta^{R} \\ -\zeta^{R}\zeta^{S}\zeta^{M} & \vdots \end{pmatrix}$	$-\beta_{[34]}\zeta^{(M}\zeta^{N)}S_4$	$\beta_{[34]}\zeta^{\scriptscriptstyle M}S_3$	
$-T\beta_{[35]}\begin{pmatrix} \vdots & \vdots & \zeta^{\scriptscriptstyle M}\zeta^{\scriptscriptstyle N} \\ -\zeta^{\scriptscriptstyle R}\zeta^{\scriptscriptstyle S} & \vdots & \vdots \end{pmatrix}$	$-\beta_{[35]}\zeta^{\scriptscriptstyle M}\zeta^{\scriptscriptstyle N}S_6$		$\beta_{[35]}S_3$
$-Teta_{[45]}\left(egin{array}{ccc} \cdot & \cdot & \zeta^{\scriptscriptstyle M} \ \cdot & -\zeta^{\scriptscriptstyle R} & \cdot \end{array} ight)$		$-\beta_{[45]}\zeta^{\scriptscriptstyle M}S_5$	$\beta_{[45]}S_4$
$-4T\kappa_{[12]} \begin{pmatrix} V^{({}^{\scriptscriptstyle M}}P^{{}^{\scriptscriptstyle N})({}^{\scriptscriptstyle R}}\zeta^{{}^{\scriptscriptstyle S})}-\zeta^{({}^{\scriptscriptstyle M}}P^{{}^{\scriptscriptstyle N})({}^{\scriptscriptstyle R}}V^{{}^{\scriptscriptstyle S})} \\ \vdots \\ \vdots \\ \ddots \end{pmatrix}$	$-2\kappa_{[12]}V^{({}_{M}}V_{2}^{{}_{N})} +2\kappa_{[12]}\zeta^{{}_{M}}V_{1}^{{}_{N}}$		
$-2T\kappa_{[13]}\begin{pmatrix} & V^{(M}P_{\zeta}^{N)R} & \\ -V^{(R}P_{\zeta}^{S)M} & & \\ & \ddots & & \end{pmatrix}$	$-2\kappa_{[13]}V^{({}_{M}}V^{{}_{N})}_{3}$	$\kappa_{[13]}V_1^{\scriptscriptstyle M}$	
$-2T\kappa_{[23]}\begin{pmatrix} & \zeta^{(M}P_{\zeta}^{N)R} & \cdot\\ -\zeta^{(R}P_{\zeta}^{S)M} & \cdot & \cdot\\ & \cdot & \cdot \end{pmatrix}$	$-2\kappa_{(23)}\zeta^{({}_MV_3^{N)}}$	$\kappa_{[23]}V_2^{\scriptscriptstyle M}$	
$4T\tilde{\kappa}_{11}\left(\begin{smallmatrix} V^{(\scriptscriptstyle M}\tilde{\epsilon}^{\scriptscriptstyle N)(\scriptscriptstyle R}V^{\scriptscriptstyle S}&\vdots&\vdots\\\vdots&\vdots&\vdots\end{smallmatrix}\right)$	$-2\tilde{\kappa}_{11}V^{({}_M}\tilde{V}_1^{{}_N)}$		
$4T\tilde{\kappa}_{(12)}\begin{pmatrix} V^{(M}\tilde{\epsilon}^{N)(R}\zeta^{S}-\zeta^{(M}\tilde{\epsilon}^{N)(R}V^{S}&\vdots&\vdots\\\vdots&\vdots&\vdots\end{pmatrix}$	$-2\tilde{\kappa}_{(12)}V^{(M}\tilde{V}_{2}^{N)}$		
$2T\tilde{\kappa}_{(13)}\left(\begin{array}{c} V^{(M}\tilde{\epsilon}^{N)R} \\ V^{(R}\tilde{\epsilon}^{S)M} \end{array}\right)$	$-2\kappa_{(12)}\zeta^{(m)}V_{1}^{(m)}$ $-2\tilde{\kappa}_{(13)}V^{(m)}\tilde{V}_{2}^{(n)}$	$-2\tilde{\kappa}_{(13)}\tilde{V}_1^M$	
$4T\tilde{\kappa}_{22}\begin{pmatrix} \zeta^{(M}\tilde{\epsilon}^{N)(R}\zeta^{S} & \vdots & \vdots \end{pmatrix}$	$-2\tilde{\kappa}_{22}\zeta^{(M}\tilde{V}_{2}^{N)}$	(10) 1	
$2T\tilde{\kappa}_{(23)}\left(\begin{smallmatrix} \ddots & \zeta^{(M}\tilde{\epsilon}^{N)R} \\ \zeta^{(R}\tilde{\epsilon}^{S)M} & \vdots \\ \vdots \end{smallmatrix}\right)$	$-2\tilde{\kappa}_{(23)}\zeta^{(\scriptscriptstyle M}\tilde{V}^{\scriptscriptstyle N)}_3$	$-2\tilde{\kappa}_{(23)}\tilde{V}_2^{\scriptscriptstyle M}$	
$T \tilde{\kappa}_{33} \left(\begin{array}{c} & \tilde{\epsilon}^{\scriptscriptstyle MR} & \cdot \\ & \tilde{\epsilon}^{\scriptscriptstyle MR} & \cdot \end{array} \right)$		$-2 ilde\kappa_{33} ilde V_3^{\scriptscriptstyle M}$	
$T ilde\eta \left(egin{smallmatrix} P_{\zeta}^{^{T(M}} ilde{\epsilon}^{^{N)(R}}P_{\zeta}^{^{S)}}{}_{T}& \cdot \ \cdot \ \cdot & \cdot \ \cdot & \cdot \ \cdot & \cdot \ \cdot \end{pmatrix} ight.$	$- ilde\eta ilde\sigma_\zeta^{_{M\!N}}$		

Table 5.7: One-derivative Class \overline{D} constitutive relations for a (4+1)-dimensional null superfluid. We have defined $\tilde{\epsilon}^{_{MN}} = \epsilon^{_{MNRST}} V_R u_S \zeta_T$.

where

$$[a_{ij}] = \begin{pmatrix} 0 & \frac{\tilde{\kappa}_{[12]}}{\kappa_{11}} & \frac{\kappa_{11} \left(\kappa_{22} \tilde{\kappa}_{[13]} - \kappa_{(12)} \tilde{\kappa}_{[23]}\right) - \tilde{\kappa}_{[12]} \left(\kappa_{(12)} \kappa_{(13)} + \zeta^{M} \zeta_{M} \tilde{\kappa}_{[12]} \tilde{\kappa}_{[13]}\right)}{\kappa_{11} \left(\kappa_{11} \kappa_{22} - \kappa_{(12)}^{2} - \zeta^{M} \zeta_{M} \tilde{\kappa}_{[12]}^{2}\right)} \\ 0 & 0 & \frac{\kappa_{11} \tilde{\kappa}_{[23]} - \kappa_{(12)} \tilde{\kappa}_{[13]} + \kappa_{(13)} \tilde{\kappa}_{[12]}}{\kappa_{11} \kappa_{22} - \kappa_{(12)}^{2} - \zeta^{M} \zeta_{M} \tilde{\kappa}_{[12]}^{2}}} \\ 0 & 0 & 0 \end{pmatrix},$$

$$[k_{ij}] = \begin{pmatrix} 0 & 0 & 0 \\ 0 & -\zeta^{M} \zeta_{M} \frac{\left(\tilde{\kappa}_{[12]}\right)^{2}}{\kappa_{11}} & -\zeta^{M} \zeta_{M} \left(\frac{\kappa_{[13]}\right)^{2}}{\kappa_{11}} + \frac{\left(\kappa_{11} \tilde{\kappa}_{[23]} - \kappa_{(12)} \tilde{\kappa}_{[13]} + \kappa_{(13)} \tilde{\kappa}_{[12]}\right)^{2}}{\kappa_{11} \left(\kappa_{11} \kappa_{22} - \kappa_{(12)}^{2} - \zeta^{M} \zeta_{M} \tilde{\kappa}_{[12]}^{2}\right)} \end{pmatrix} \end{pmatrix}. \quad (5.38)$$

The rationale is that in this basis, the quadratic form Δ can be arranged into a form which can be made explicitly positive semi-definite

$$T\Delta = \sum_{i,j=1}^{5} S_i \beta_{(ij)} S_j + \sum_{i,j=1}^{3} V_i^{\prime M} \kappa_{(ij)}^{\prime} V_{j,M}^{\prime} + \eta \sigma^{MN} \sigma_{MN}.$$
(5.39)

Given $T \ge 0$, the condition $\Delta \ge 0$ implies that $\eta \ge 0$ and the matrices $[\beta_{(ij)}]_{5\times 5}$ and $[\kappa'_{(ij)}]_{3\times 3}$ have all non-negative eigenvalues. This gives us 9 inequalities among the 25 dissipative transport coefficients, while 16 remain completely arbitrary.

This finishes our discussion of the first order Galilean superfluid dynamics. Due to their technical form, we have left the explicit constitutive relations in their null fluid form. We can always perform the null reduction prescribed in section 4.2 and express them in the Newton-Cartan language. For reference, we have null reduced all the one-derivative superfluid data in table 5.4. The explicit null reduction of the constitutive relations can be found in our paper [4].

5.3 Galilean fluid surfaces

For the final example of this thesis, we consider the breaking of the spacetime translational symmetry to form surfaces in a Galilean fluid. The discussion presented here is taken from [6], and is a straight forward generalisation of our relativistic discussion in section 3.3. We introduce a Goldstone mode f(x) corresponding to a broken momentum generator, with the symmetry transformation $\delta_{\chi}f = \chi^M \partial_M f$. We require that $V^M \partial_M f = 0$. We also introduce a distribution functional $\theta(f)$, with properties similar to section 3.3: its derivative $\theta'(f)$ is strictly positive and is only supported in a thin band around f = 0; all the *f*-dependence in the constitutive relations comes via $\theta(f)$; the function $f \sim \mathcal{O}(\partial^{-1})$, while the distribution $\theta(f) \sim \mathcal{O}(\partial^0)$. Derivatives of $\theta(f)$ can be characterised in terms of the derivatives of the unit normal vector

$$z_{M} = -\frac{\partial_{M}f}{\sqrt{g^{MN}\partial_{M}f\partial_{N}f}},\tag{5.40}$$

along with the distributions

$$\tilde{\delta}^{(n)}(f) = (-)^{n+1} z^{M_1} \dots z^{M_{n+1}} \mathcal{D}_{M_1} \dots \mathcal{D}_{M_{n+1}} \theta(f).$$
(5.41)

Denoting the equation of motion for f by $Y/\sqrt{g^{MN}\partial_M f \partial_N f} \approx 0$, we can write the Galilean adiabaticity equation as

$$D_{M}N^{M} - N_{H}^{\perp} = \frac{1}{2}T^{MN}\delta_{\mathcal{B}}g_{MN} + J^{M}\delta_{\mathcal{B}}A_{M} + \frac{Y\delta_{\mathcal{B}}f}{\sqrt{g^{MN}\partial_{M}f\partial_{N}f}} + \Delta, \qquad \Delta \ge 0.$$
(5.42)

In an exact correspondence with the relativistic results in section 3.3, up to one-derivative order, this equation allows a set of Class H_S constitutive relations parametrised by

$$\mathcal{N} = \theta(f) P_{\rm in}(T,\mu,\mu_{\rm m}) + \bar{\theta}(f) P_{\rm out}(T,\mu,\mu_{\rm m}) - \tilde{\delta}(f) \gamma(T,\mu,\mu_{\rm m}).$$
(5.43)

Here $\bar{\theta}(f) = 1 - \theta(f)$. In Class D we have a term like $Y \sim -\alpha \theta'(f) \delta_{\rm B} f$ for some non-negative coefficient α . Plus, following section 5.1.2, we have two copies of Class D and $\overline{\rm D}$ constitutive relations η , ζ , κ , σ , κ_Q , and $\overline{\kappa}_Q$, one on either side of the surface, and one copy of Class A and Class H_V constants. Together, they imply a set of constitutive relations

$$T^{MN} = \theta(f)T^{MN}_{in} + \bar{\theta}(f)T^{MN}_{out} + \tilde{\delta}(f) \Big(R_{sur}u^{M}u^{N} + 2(E_{sur} - \gamma)u^{(M}V^{N)} - \gamma(g^{MN} - z^{M}z^{N}) \Big) + \mathcal{O}(\partial^{2}), J^{M} = \theta(f)J^{M}_{in} + \bar{\theta}(f)J^{M}_{out} + \tilde{\delta}(f)Q_{sur}u^{M} + \mathcal{O}(\partial^{2}), Y = \tilde{\delta}(f) \Big(\frac{\alpha}{T}u^{M}z_{M} + (P_{in} - P_{out}) - D_{M}(\gamma z^{M})\Big) + \mathcal{O}(\partial^{2}).$$
(5.44)

The bulk contributions $T_{\text{in/out}}^{MN}$ and $J_{\text{in/out}}^{M}$ to the energy-momentum tensor and charge current are taken directly from section 5.1.2. In addition, there are surface contributions to the constitutive relations, which are basically ideal Galilean fluids living at the boundary, with the negative of surface tension acting as a pressure.

Upon null reduction they imply the Galilean fluid constitutive relations in the Newton-Cartan language

$$\rho^{\mu} = \theta(f)\rho_{\rm in}^{\mu} + \bar{\theta}(f)\rho_{\rm out}^{\mu} + \tilde{\delta}(f)R_{\rm sur}u^{\mu} + \mathcal{O}(\partial^{2}),$$

$$\epsilon^{\mu} = \theta(f)\epsilon_{\rm in}^{\mu} + \bar{\theta}(f)\epsilon_{\rm out}^{\mu} + \tilde{\delta}(f)\left[\left(E_{\rm sur} + \frac{1}{2}R_{\rm sur}u^{a}u_{a}\right)u^{\mu} - \gamma\left(\bar{u}^{\mu} + z^{\mu}v^{\nu}z_{\nu}\right)\right] + \mathcal{O}(\partial^{2}),$$

$$p^{\mu}{}_{a} = \theta(f)p_{\rm in}{}^{\mu}{}_{a} + \bar{\theta}(f)p_{\rm out}{}^{\mu}{}_{a} + R_{\rm sur}u^{\mu} - \gamma\left(e_{a}{}^{\mu} - z^{\mu}z_{a}\right) + \mathcal{O}(\partial^{2}),$$

$$j^{\mu} = \theta(f)j_{\rm in}^{\mu} + \bar{\theta}(f)j_{\rm out}^{\mu} + \tilde{\delta}(f)Q_{\rm sur}u^{\mu} + \mathcal{O}(\partial^{2}).$$
(5.45)

Note that $z^{\mu} = h^{\mu\nu} z_{\nu}$ along with the normalisation condition $z_{\mu} z_{\nu} h^{\mu\nu} = 1$. On the other hand, we get the Young-Laplace equation

$$u^{\mu}z_{\mu} \approx -\frac{T}{\alpha} \left(\Delta P - h^{\mu\nu} \tilde{\mathbf{D}}_{\mu} \left(\gamma z_{\nu} \right) \right) + \mathcal{O}(\partial), \quad \text{where} \quad \Delta P = P_{\text{in}} - P_{\text{out}}, \quad (5.46)$$

which is reminiscent of its relativistic incarnation. We see that the motion of the surface is

governed by the balance of pressures inside and outside, and the surface tension term.

This finishes our discussion of Galilean fluids with surfaces. The generalisation of these results to Galilean superfluids is quite straight forward and has been discussed in [6].

In this thesis, we discussed the principles of relativistic and Galilean hydrodynamics. We built upon the fundamental considerations of symmetries, thermal field theories, and thermodynamics, and presented a universal framework for hydrodynamics that allows for arbitrary gapless modes in its spectrum. This brings together a variety of otherwise distinct hydrodynamic treatments like superfluid dynamics, surface fluid dynamics, magnetohydrodynamics, and obviously ordinary fluid dynamics. We argued that the time-evolution of these gapless modes can also be derived from within the hydrodynamic framework by a suitable generalisation of the second law of thermodynamics to off-shell configurations. This provides a neat hydrodynamic argument to derive the Josephson equation in superfluid dynamics and Young-Laplace equation in surface fluid dynamics. Our treatment of hydrodynamics also includes an arbitrary spin current and background torsion, which has not been discussed in full generality in the literature before.

Using this framework, we presented an all-order analysis of the second law of thermodynamics and classified the most generic hydrodynamic transport compatible with it. We found that the second law imposes some strict constraints in the hydrostatic sector, by forcing the constitutive relations to be characterised by a free energy density (Class H_S) and a finite set of constants (Class A and H_V). In the non-hydrostatic sector, however, we found the second law to be much more lenient; it classifies the hydrodynamic transport into Class D and Class \overline{D} , but only imposes some inequalities on the leading order Class D transport coefficients and none thereafter. Prior to this work, a similar classification analysis had been done in the literature for the restricted case of ordinary fluid dynamics [52]. Although our classification scheme draws heavily from that of [52] in the hydrostatic sector, the two are quite distinct in the non-hydrostatic sector. Most notably, the classification presented in this thesis is non-redundant, i.e. there is no overlap between various classes, and it purifies the true dissipative transport from the mere redundancies (Class S) in the choice of an entropy current that satisfies the second law for a given set of constitutive relations.

Apart from reproducing some known results in hydrodynamics, we applied the ideas discussed in this thesis to study some novel hydrodynamic systems. We presented a theory of non-Abelian superfluid dynamics with a partially broken (semisimple) Lie group of internal symmetries. We introduced a pair of projection operators that allowed us to define a superfluid velocity corresponding to the Goldstone modes valued in the Lie-algebra quotient, irrespective of the details of the explicit symmetry breaking. Expressed in terms of these projectors, we illustrated that the non-Abelian superfluid dynamics can be made structurally similar to its Abelian counterpart. We derived the non-Abelian generalisation of the Goldstone modes in terms of the corresponding Lie-algebra components of the chemical potential. Recently, non-Abelian superfluids have started to gain some attention

in the literature regarding the modelling of p-wave superfluidity observed in liquid ³He [97]. It will be interesting to see if the results found in this thesis have an analogue in these physical systems.

We also studied how the hydrodynamic constitutive relations are modified near a surface or an interface in a fluid. We introduced a new gapless hydrodynamic mode, called a shapefield, to characterise such surfaces and used its dynamics to reproduce the Young-Laplace equation. We worked out the hydrodynamic constitutive relations up to one-derivative order, which amounts to zero-derivative order on the surface, and identified the well-known transport coefficient, surface tension, in the surface thermodynamics. Hydrodynamic configurations with surfaces are quite interesting as they can be used to model fluids undergoing a phase transition, like the fluid-superfluid phase transition in liquid helium or the confinement-deconfinement phase transition in the quark-gluon plasma. Further exploration in this direction, perhaps holographic, will be quite intriguing.

In the introduction of this thesis, we briefly outlined how the theory of magnetohydrodynamics (MHD) can also be incorporated into the same framework by allowing for a dynamical U(1) gauge field as an additional gapless mode in the hydrodynamic description. Although this preliminary analysis checks out quite naturally, we have not considered MHD in rigorous detail in this thesis. As a natural next step, we should verify that the calculation does indeed work out at arbitrarily high orders in the derivative expansion without any surprises. As an interesting aside, note that the dynamical equation of MHD (1.51) forces us to set the charge density to be an order one quantity in the derivative expansion, which algebraically determines the associated chemical potential in terms of the other fields in the theory. The same equation is also present in the conventional formulation of MHD as well; see e.g. [54]. However, the chemical potential is still typically taken to be an independent hydrodynamic variable in MHD, which is puzzling. It will be interesting to see if the off-shell formalism of magnetohydrodynamics can shed some light on this issue.

On another front, in a recent paper [60], an alternative interpretation of MHD has been proposed. Instead of introducing a dynamical gauge field, the authors considered hydrodynamics with a global 1-form symmetry corresponding to the conservation of magnetic field lines and reproduced the spectrum of MHD. It was realised in [8], however, that there are some issues with defining hydrostatic configurations in these theories. In particular, unlike regular hydrodynamics, in the presence of a higher-form symmetry, the second law of thermodynamics does not seem to be sufficient to guarantee the hydrostatic principle. Since the off-shell formalism of hydrodynamics detailed in this thesis renders the connection between the second law and the hydrostatic principle quite transparent, we would like to see if it can be utilised to resolve this discrepancy.

In the second part of this thesis, we offered a new viewpoint of Galilean hydrodynamics. Realising that a generic Galilean theory can be covariantly coupled to a one-higher dimensional null background, we formalised hydrodynamics on such null backgrounds from an axiomatic standpoint. The fluids thus obtained, which we call null fluids, are a one-higher dimensional relativistic embedding of Galilean fluids, which manifest all the Galilean symmetries, especially boosts, in terms of a Poincaré invariant structure. They are essentially relativistic fluids coupled to a spacetime background carrying a null isometry. Due to the anisotropy introduced by this additional background vector field, null/Galilean fluids are characterised by many more transport coefficients compared to their relativistic cousins. One way to understand this distinction is to note that the Galilean fluids carry an additional conserved current, i.e. mass current, in their spectrum, with its own dynamical chemical potential and background gauge field. Correspondingly, we have additional constitutive relations to write down and additional fields to generate the associated tensor structures, which eventually leads to the increase in the number of allowed transport coefficients.

Owing to our handle on the Poincaré symmetry, relativistic fluids are typically much better understood in the literature compared to their Galilean counterparts. Null fluids can potentially bridge this gap in our understanding by providing a mechanism to import all the exotic relativistic machinery at our disposal directly into Galilean hydrodynamics. We witnessed two non-trivial applications of this idea in this thesis. Firstly, we imported the classification scheme of relativistic hydrodynamics into Galilean hydrodynamics and used it to perform an all-order analysis of the second law of thermodynamics. Similar to the relativistic case, we found that the second law imposes some strict constraints in the hydrostatic sector, while only requires some leading-order transport coefficients to be non-negative in the non-hydrostatic sector. We studied the explicit application of this to ordinary Galilean fluids, Galilean superfluids, and surfaces in Galilean fluids. Secondly, we used the preexisting transgression machinery of relativistic hydrodynamics [94] to work out the effects of anomalies on the Galilean hydrodynamic transport. We also used this language to classify the set of constants, called transcendental anomalies, which are left undetermined by the second law of thermodynamics. We found that Galilean hydrodynamics admits many more of these constants compared to the relativistic case. There has been some work in the literature which associates these undetermined constants in relativistic hydrodynamics to anomaly coefficients by demanding the consistency of Euclidean vacuum [87]. We are not aware if a similar argument exists for Galilean hydrodynamics as well.

During this construction, we devised an anomaly inflow mechanism for Galilean field theories using null backgrounds, classifying the possible 't Hooft anomalies. The anomalies we found only affect the rotations and internal symmetries, while leaving the space-time translations, mass conservation, and Galilean boost symmetries non-anomalous. It is interesting to note that the Galilean anomaly polynomial is structurally equivalent to the relativistic anomaly polynomial, and hence the number of anomaly coefficients on both sides match. Owing to this, the structure of Hall currents that enter the conservation laws is also quite similar in both the cases. Hence the results we have obtained promise to be the genuine non-relativistic anomalies and not just a mathematical manifestation of the Galilean invariance. That being said, we still need to explicitly construct a Galilean field theory that exhibits these anomalies, which we leave for the future explorations.

As we discussed in the introduction, a fundamental motivation for the continued research interest in hydrodynamics is to get insights into the physics of out-of-equilibrium thermal field theories, especially the irreversible dissipative phenomena. A part of this program is to write down a Wilsonian effective action for hydrodynamics, which should provide a first principle understanding of the dissipative hydrodynamic transport. The off-shell formalism of hydrodynamics discussed here provides a natural starting point to frame these questions. Much progress has already been made towards writing down a Schwinger-Keldysh effective action for ordinary fluid dynamics [68, 71]. As a prospective direction, it will be interesting to see if these constructs can be extended to include the arbitrary gapless modes discussed in this thesis, thereby providing an extension to more generic hydrodynamic treatments like superfluid dynamics and magnetohydrodynamics. For now, we leave these ambitious ventures open for speculation.

Bibliography

- A. Jain, Theory of non-Abelian superfluid dynamics, Phys. Rev. D95 (2017) 121701, [1610.05797].
- [2] N. Banerjee, S. Dutta and A. Jain, Equilibrium partition function for nonrelativistic fluids, Phys. Rev. D92 (2015) 081701, [1505.05677].
- [3] N. Banerjee, S. Dutta and A. Jain, Null Fluids A New Viewpoint of Galilean Fluids, Phys. Rev. D93 (2016) 105020, [1509.04718].
- [4] N. Banerjee, S. Dutta and A. Jain, First Order Galilean Superfluid Dynamics, Phys. Rev. D96 (2017) 065004, [1612.01550].
- [5] A. Jain, Galilean Anomalies and Their Effect on Hydrodynamics, Phys. Rev. D93 (2016) 065007, [1509.05777].
- [6] J. Armas, J. Bhattacharya, A. Jain and N. Kundu, On the surface of superfluids, JHEP 06 (2017) 090, [1612.08088].
- [7] N. Banerjee, S. Atul Bhatkar and A. Jain, Second order Galilean fluids & Stokes' law, Phys. Rev. D97 (2018) 096018, [1711.09076].
- [8] J. Armas, J. Gath, A. Jain and A. V. Pedersen, Dissipative hydrodynamics with higher-form symmetry, JHEP 05 (2018) 192, [1803.00991].
- [9] P. Burda, R. Gregory and A. Jain, *Holographic Reconstruction of Bubbles*, 1804.05202.
- [10] L. Landau and E. Lifshitz, Fluid Mechanics. Teoreticheskaia fizika. Pergamon Press, 1959.
- [11] VIRGO, LIGO SCIENTIFIC collaboration, B. P. Abbott et al., Observation of Gravitational Waves from a Binary Black Hole Merger, Phys. Rev. Lett. 116 (2016) 061102, [1602.03837].
- [12] G. 't Hooft, Dimensional reduction in quantum gravity, Conf. Proc. C930308 (1993) 284–296, [gr-qc/9310026].
- [13] L. Susskind, The World as a hologram, J. Math. Phys. 36 (1995) 6377-6396, [hep-th/9409089].
- [14] R. Bousso, The Holographic principle, Rev. Mod. Phys. 74 (2002) 825-874, [hep-th/0203101].
- [15] J. M. Maldacena, The Large N limit of superconformal field theories and supergravity, Int. J. Theor. Phys. 38 (1999) 1113-1133, [hep-th/9711200].
- [16] S. S. Gubser, I. R. Klebanov and A. M. Polyakov, Gauge theory correlators from noncritical string theory, Phys. Lett. B428 (1998) 105–114, [hep-th/9802109].
- [17] E. Witten, Anti-de Sitter space and holography, Adv. Theor. Math. Phys. 2 (1998) 253-291, [hep-th/9802150].
- J. M. Maldacena, TASI 2003 lectures on AdS / CFT, in Progress in string theory. Proceedings, Summer School, TASI 2003, Boulder, USA, June 2-27, 2003, pp. 155-203, 2003. hep-th/0309246.
- [19] M. Van Raamsdonk, Building up spacetime with quantum entanglement, Gen. Rel. Grav. 42 (2010) 2323–2329, [1005.3035].
- [20] S. Ryu and T. Takayanagi, Holographic derivation of entanglement entropy from AdS/CFT, Phys. Rev. Lett. 96 (2006) 181602, [hep-th/0603001].
- [21] A. Lewkowycz and J. Maldacena, Generalized gravitational entropy, JHEP 08 (2013) 090, [1304.4926].
- [22] J. Maldacena and L. Susskind, Cool horizons for entangled black holes, Fortsch. Phys. 61 (2013) 781–811, [1306.0533].

- [23] A. R. Brown, D. A. Roberts, L. Susskind, B. Swingle and Y. Zhao, *Holographic Complexity Equals Bulk Action?*, Phys. Rev. Lett. **116** (2016) 191301, [1509.07876].
- [24] L. Thorlacius, Black holes and the holographic principle, Comment. Phys. Math. Soc. Sci. Fenn. 166 (2004) 209–231, [hep-th/0404098].
- [25] D. Simmons-Duffin, The Conformal Bootstrap, in Proceedings, Theoretical Advanced Study Institute in Elementary Particle Physics: New Frontiers in Fields and Strings (TASI 2015): Boulder, CO, USA, June 1-26, 2015, pp. 1–74, 2017. 1602.07982. DOI.
- [26] J. Smit, Introduction to quantum fields on a lattice: A robust mate, Cambridge Lect. Notes Phys. 15 (2002) 1–271.
- [27] S. A. Hartnoll, Lectures on holographic methods for condensed matter physics, Class. Quant. Grav. 26 (2009) 224002, [0903.3246].
- [28] R. H. Price and K. S. Thorne, Membrane Viewpoint on Black Holes: Properties and Evolution of the Stretched Horizon, Phys. Rev. D33 (1986) 915–941.
- [29] W. M. Suen, R. H. Price and I. H. Redmount, Membrane Viewpoint on Black Holes: Gravitational Perturbations of the Horizon, Phys. Rev. D37 (1988) 2761–2789.
- [30] G. Policastro, D. T. Son and A. O. Starinets, From AdS / CFT correspondence to hydrodynamics, JHEP 09 (2002) 043, [hep-th/0205052].
- [31] S. Bhattacharyya, V. E. Hubeny, S. Minwalla and M. Rangamani, Nonlinear Fluid Dynamics from Gravity, JHEP 02 (2008) 045, [0712.2456].
- [32] R. Gregory and R. Laflamme, Black strings and p-branes are unstable, Phys. Rev. Lett. 70 (1993) 2837–2840, [hep-th/9301052].
- [33] M. M. Caldarelli, O. J. C. Dias, R. Emparan and D. Klemm, Black Holes as Lumps of Fluid, JHEP 04 (2009) 024, [0811.2381].
- [34] R. Emparan, T. Harmark, V. Niarchos and N. A. Obers, Essentials of Blackfold Dynamics, JHEP 03 (2010) 063, [0910.1601].
- [35] P. F. Kolb and U. W. Heinz, Hydrodynamic description of ultrarelativistic heavy ion collisions, nucl-th/0305084.
- [36] E. Shuryak, Physics of Strongly coupled Quark-Gluon Plasma, Prog. Part. Nucl. Phys. 62 (2009) 48–101, [0807.3033].
- [37] E. Shuryak, Why does the quark gluon plasma at RHIC behave as a nearly ideal fluid?, Prog. Part. Nucl. Phys. 53 (2004) 273–303, [hep-ph/0312227].
- [38] G. Policastro, D. T. Son and A. O. Starinets, The Shear viscosity of strongly coupled N=4 supersymmetric Yang-Mills plasma, Phys. Rev. Lett. 87 (2001) 081601, [hep-th/0104066].
- [39] P. Kovtun, D. T. Son and A. O. Starinets, Viscosity in strongly interacting quantum field theories from black hole physics, Phys. Rev. Lett. 94 (2005) 111601, [hep-th/0405231].
- [40] P. M. Chesler and L. G. Yaffe, Holography and off-center collisions of localized shock waves, JHEP 10 (2015) 070, [1501.04644].
- [41] W. van der Schee, Gravitational collisions and the quark-gluon plasma. PhD thesis, Utrecht U., 2014. 1407.1849.
- [42] O. J. C. Dias, G. T. Horowitz, N. Iqbal and J. E. Santos, Vortices in holographic superfluids and superconductors as conformal defects, JHEP 04 (2014) 096, [1311.3673].
- [43] A. Adams, P. M. Chesler and H. Liu, Holographic Vortex Liquids and Superfluid Turbulence, Science 341 (2013) 368–372, [1212.0281].
- [44] A. Adams, P. M. Chesler and H. Liu, *Holographic turbulence*, Phys. Rev. Lett. **112** (2014) 151602, [1307.7267].
- [45] S. R. Green, F. Carrasco and L. Lehner, Holographic Path to the Turbulent Side of Gravity, Phys. Rev. X4 (2014) 011001, [1309.7940].
- [46] D. T. Son and P. Surowka, Hydrodynamics with Triangle Anomalies, Phys. Rev. Lett. 103 (2009) 191601, [0906.5044].

- [47] N. Banerjee, S. Dutta, S. Jain, R. Loganayagam and T. Sharma, Constraints on Anomalous Fluid in Arbitrary Dimensions, JHEP 03 (2013) 048, [1206.6499].
- [48] M. Kaminski, C. F. Uhlemann, M. Bleicher and J. Schaffner-Bielich, Anomalous hydrodynamics kicks neutron stars, Phys. Lett. B760 (2016) 170–174, [1410.3833].
- [49] K. Jensen, M. Kaminski, P. Kovtun, R. Meyer, A. Ritz and A. Yarom, Towards hydrodynamics without an entropy current, Phys. Rev. Lett. 109 (2012) 101601, [1203.3556].
- [50] N. Banerjee, J. Bhattacharya, S. Bhattacharyya, S. Jain, S. Minwalla and T. Sharma, Constraints on Fluid Dynamics from Equilibrium Partition Functions, JHEP 09 (2012) 046, [1203.3544].
- [51] S. Bhattacharyya, S. Jain, S. Minwalla and T. Sharma, Constraints on Superfluid Hydrodynamics from Equilibrium Partition Functions, JHEP 01 (2013) 040, [1206.6106].
- [52] F. M. Haehl, R. Loganayagam and M. Rangamani, Adiabatic hydrodynamics: The eightfold way to dissipation, JHEP 05 (2015) 060, [1502.00636].
- [53] J. Bhattacharya, S. Bhattacharyya, S. Minwalla and A. Yarom, A Theory of first order dissipative superfluid dynamics, JHEP 05 (2014) 147, [1105.3733].
- [54] J. Hernandez and P. Kovtun, *Relativistic magnetohydrodynamics*, JHEP 05 (2017) 001, [1703.08757].
- [55] J. de Boer, J. Hartong, N. A. Obers, W. Sybesma and S. Vandoren, *Perfect Fluids*, 1710.04708.
- [56] J. de Boer, J. Hartong, N. A. Obers, W. Sybesma and S. Vandoren, Hydrodynamic Modes of Homogeneous and Isotropic Fluids, 1710.06885.
- [57] L. V. Delacretaz, B. Gouteraux, S. A. Hartnoll and A. Karlsson, Theory of hydrodynamic transport in fluctuating electronic charge density wave states, Phys. Rev. B96 (2017) 195128, [1702.05104].
- [58] M. Blake, Momentum relaxation from the fluid/gravity correspondence, JHEP 09 (2015) 010, [1505.06992].
- [59] J. Armas, J. Bhattacharya and N. Kundu, Surface transport in plasma-balls, JHEP 06 (2016) 015, [1512.08514].
- [60] S. Grozdanov, D. M. Hofman and N. Iqbal, Generalized global symmetries and dissipative magnetohydrodynamics, Phys. Rev. D95 (2017) 096003, [1610.07392].
- [61] S. A. Hartnoll, P. K. Kovtun, M. Muller and S. Sachdev, Theory of the Nernst effect near quantum phase transitions in condensed matter, and in dyonic black holes, Phys. Rev. B76 (2007) 144502, [0706.3215].
- [62] M. Muller and S. Sachdev, Collective cyclotron motion of the relativistic plasma in graphene, Phys. Rev. B78 (2008) 115419, [0801.2970].
- [63] S. Grozdanov and J. Polonyi, Viscosity and dissipative hydrodynamics from effective field theory, Phys. Rev. D91 (2015) 105031, [1305.3670].
- [64] P. Kovtun, G. D. Moore and P. Romatschke, Towards an effective action for relativistic dissipative hydrodynamics, JHEP 07 (2014) 123, [1405.3967].
- [65] M. Harder, P. Kovtun and A. Ritz, On thermal fluctuations and the generating functional in relativistic hydrodynamics, JHEP 07 (2015) 025, [1502.03076].
- [66] M. Crossley, P. Glorioso, H. Liu and Y. Wang, Off-shell hydrodynamics from holography, JHEP 02 (2016) 124, [1504.07611].
- [67] J. de Boer, M. P. Heller and N. Pinzani-Fokeeva, Effective actions for relativistic fluids from holography, JHEP 08 (2015) 086, [1504.07616].
- [68] K. Jensen, R. Marjieh, N. Pinzani-Fokeeva and A. Yarom, A panoply of Schwinger-Keldysh transport, 1804.04654.
- [69] F. M. Haehl, R. Loganayagam and M. Rangamani, Schwinger-Keldysh formalism. Part II: thermal equivariant cohomology, JHEP 06 (2017) 070, [1610.01941].

- [70] F. M. Haehl, R. Loganayagam and M. Rangamani, Schwinger-Keldysh formalism. Part I: BRST symmetries and superspace, JHEP 06 (2017) 069, [1610.01940].
- [71] F. M. Haehl, R. Loganayagam and M. Rangamani, Effective Action for Relativistic Hydrodynamics: Fluctuations, Dissipation, and Entropy Inflow, 1803.11155.
- [72] M. Rangamani, S. F. Ross, D. T. Son and E. G. Thompson, Conformal non-relativistic hydrodynamics from gravity, JHEP 01 (2009) 075, [0811.2049].
- [73] J. Maldacena, D. Martelli and Y. Tachikawa, Comments on string theory backgrounds with non-relativistic conformal symmetry, JHEP 10 (2008) 072, [0807.1100].
- [74] A. Adams, K. Balasubramanian and J. McGreevy, Hot Spacetimes for Cold Atoms, JHEP 11 (2008) 059, [0807.1111].
- [75] C. P. Herzog, M. Rangamani and S. F. Ross, *Heating up Galilean holography*, JHEP 11 (2008) 080, [0807.1099].
- [76] M. Rangamani, Holography for non-relativistic CFTs, Acta Phys. Polon. B40 (2009) 3745–3770.
- [77] E. Cartan, Sur les varietes a connexion affine et la theorie de la relativite generalisee. (premiere partie), Annales Sci. Ecole Norm. Sup. 40 (1923) 325–412.
- [78] E. Cartan, Sur les varietes a connexion affine et la theorie de la relativite generalisee. (premiere partie) (Suite)., Annales Sci. Ecole Norm. Sup. 41 (1924) 1–25.
- [79] K. Jensen, On the coupling of Galilean-invariant field theories to curved spacetime, 1408.6855.
- [80] E. Bergshoeff, A. Chatzistavrakidis, L. Romano and J. Rosseel, Newton-Cartan Gravity and Torsion, JHEP 10 (2017) 194, [1708.05414].
- [81] K. Jensen, Aspects of hot Galilean field theory, JHEP 04 (2015) 123, [1411.7024].
- [82] M. Geracie, K. Prabhu and M. M. Roberts, Fields and fluids on curved non-relativistic spacetimes, JHEP 08 (2015) 042, [1503.02680].
- [83] J. Hartong, N. A. Obers and M. Sanchioni, Lifshitz Hydrodynamics from Lifshitz Black Branes with Linear Momentum, JHEP 10 (2016) 120, [1606.09543].
- [84] M. Taylor, Lifshitz holography, Class. Quant. Grav. 33 (2016) 033001, [1512.03554].
- [85] T. Watanabe and M. J. Hayashi, General relativity with torsion, gr-qc/0409029.
- [86] C. G. Callan, Jr. and J. A. Harvey, Anomalies and Fermion Zero Modes on Strings and Domain Walls, Nucl. Phys. B250 (1985) 427–436.
- [87] K. Jensen, R. Loganayagam and A. Yarom, Thermodynamics, gravitational anomalies and cones, JHEP 02 (2013) 088, [1207.5824].
- [88] M. L. Bellac, Thermal Field Theory. Cambridge University Press, 2011.
- [89] J. I. Kapusta and C. Gale, Finite-temperature field theory: Principles and applications. Cambridge Monographs on Mathematical Physics. Cambridge University Press, 2011, 10.1017/CBO9780511535130.
- [90] W. Florkowski, B. Friman, A. Jaiswal and E. Speranza, *Relativistic fluid dynamics with spin*, 1705.00587.
- [91] S. Bhattacharyya, Entropy current and equilibrium partition function in fluid dynamics, JHEP 08 (2014) 165, [1312.0220].
- [92] K. Jensen, R. Loganayagam and A. Yarom, Anomaly inflow and thermal equilibrium, JHEP 05 (2014) 134, [1310.7024].
- [93] F. M. Haehl, R. Loganayagam and M. Rangamani, *Effective actions for anomalous hydrodynamics*, JHEP 03 (2014) 034, [1312.0610].
- [94] K. Jensen, R. Loganayagam and A. Yarom, Chern-Simons terms from thermal circles and anomalies, JHEP 05 (2014) 110, [1311.2935].
- [95] S. Bhattacharyya, Entropy Current from Partition Function: One Example, JHEP 07 (2014) 139, [1403.7639].

- [96] J. Bhattacharya, S. Bhattacharyya and S. Minwalla, Dissipative Superfluid dynamics from gravity, JHEP 04 (2011) 125, [1101.3332].
- [97] C. Hoyos, B. S. Kim and Y. Oz, Odd Parity Transport In Non-Abelian Superfluids From Symmetry Locking, JHEP 10 (2014) 127, [1404.7507].
- [98] D. D. Osheroff, R. C. Richardson and D. M. Lee, Evidence for a new phase of solid he³, Phys. Rev. Lett. 28 (Apr, 1972) 885–888.
- [99] D. D. Osheroff, W. J. Gully, R. C. Richardson and D. M. Lee, New magnetic phenomena in liquid he³ below 3 mk, Phys. Rev. Lett. 29 (Oct, 1972) 920–923.
- [100] S. Weinberg, The quantum theory of fields. Vol. 2: Modern applications. Cambridge University Press, 2013.
- [101] T. Andrade and B. Withers, A simple holographic model of momentum relaxation, JHEP 05 (2014) 101, [1311.5157].
- [102] G. Manfredi, Non-relativistic limits of Maxwell's equations, European Journal of Physics 34 (July, 2013) 859, [1303.5608].
- [103] A. Einstein, On the electrodynamics of moving bodies, Annalen Phys. 17 (1905) 891–921.
- [104] M. Guica, K. Skenderis, M. Taylor and B. C. van Rees, Holography for Schrodinger backgrounds, JHEP 02 (2011) 056, [1008.1991].
- [105] C. Duval, G. Burdet, H. P. Kunzle and M. Perrin, Bargmann Structures and Newton-cartan Theory, Phys. Rev. D31 (1985) 1841–1853.
- [106] C. Duval, G. W. Gibbons and P. Horvathy, Celestial mechanics, conformal structures and gravitational waves, Phys. Rev. D43 (1991) 3907–3922, [hep-th/0512188].
- [107] B. Julia and H. Nicolai, Null Killing vector dimensional reduction and Galilean geometrodynamics, Nucl. Phys. B439 (1995) 291–326, [hep-th/9412002].
- [108] M. H. Christensen, J. Hartong, N. A. Obers and B. Rollier, Boundary Stress-Energy Tensor and Newton-Cartan Geometry in Lifshitz Holography, JHEP 01 (2014) 057, [1311.6471].
- [109] M. Hassaine and P. A. Horvathy, Field dependent symmetries of a nonrelativistic fluid model, Annals Phys. 282 (2000) 218-246, [math-ph/9904022].
- [110] K. Jensen, Anomalies for Galilean fields, 1412.7750.
- [111] H. D. Dombrowski and K. Horneffer, Die differentialgeometrie des galileischen relativitätsprinzips, Mathematische Zeitschrift 86 (Aug, 1964) 291–311.
- [112] H. P. Kuenzle, Galilei and lorentz structures on space-time comparison of the corresponding geometry and physics, Ann. Inst. H. Poincare Phys. Theor. 17 (1972) 337–362.
- [113] C. Duval and H. P. Kunzle, Dynamics of Continua and Particles from General Covariance of Newtonian Gravitation Theory, Rept. Math. Phys. 13 (1978) 351.
- [114] K. Kuchar, GRAVITATION, GEOMETRY, AND NONRELATIVISTIC QUANTUM THEORY, Phys. Rev. D22 (1980) 1285–1299.
- [115] C. Duval and H. P. Kunzle, Minimal Gravitational Coupling in the Newtonian Theory and the Covariant Schrodinger Equation, Gen. Rel. Grav. 16 (1984) 333.
- [116] D. T. Son and M. Wingate, General coordinate invariance and conformal invariance in nonrelativistic physics: Unitary Fermi gas, Annals Phys. 321 (2006) 197–224, [cond-mat/0509786].
- [117] D. T. Son, Toward an AdS/cold atoms correspondence: A Geometric realization of the Schrodinger symmetry, Phys. Rev. D78 (2008) 046003, [0804.3972].
- [118] C. Duval and P. A. Horvathy, Non-relativistic conformal symmetries and Newton-Cartan structures, J. Phys. A42 (2009) 465206, [0904.0531].
- [119] D. T. Son, Newton-Cartan Geometry and the Quantum Hall Effect, 1306.0638.
- [120] E. A. Bergshoeff, J. Hartong and J. Rosseel, Torsional Newton-Cartan geometry and the Schrodinger algebra, Class. Quant. Grav. 32 (2015) 135017, [1409.5555].
- [121] K. Jensen and A. Karch, Revisiting non-relativistic limits, JHEP 04 (2015) 155, [1412.2738].

- [122] M. Geracie, D. T. Son, C. Wu and S.-F. Wu, Spacetime Symmetries of the Quantum Hall Effect, Phys. Rev. D91 (2015) 045030, [1407.1252].
- [123] R. Banerjee, A. Mitra and P. Mukherjee, A new formulation of non-relativistic diffeomorphism invariance, Phys. Lett. B737 (2014) 369–373, [1404.4491].
- [124] R. Banerjee, A. Mitra and P. Mukherjee, Localization of the Galilean symmetry and dynamical realization of Newton-Cartan geometry, Class. Quant. Grav. 32 (2015) 045010, [1407.3617].
- [125] T. Brauner, S. Endlich, A. Monin and R. Penco, General coordinate invariance in quantum many-body systems, Phys. Rev. D90 (2014) 105016, [1407.7730].
- [126] J. Hartong, E. Kiritsis and N. A. Obers, Schrodinger Invariance from Lifshitz Isometries in Holography and Field Theory, Phys. Rev. D92 (2015) 066003, [1409.1522].
- [127] M. Geracie, Galilean Geometry in Condensed Matter Systems. PhD thesis, Chicago U., 2016. 1611.01198.
- [128] M. H. Christensen, J. Hartong, N. A. Obers and B. Rollier, Torsional Newton-Cartan Geometry and Lifshitz Holography, Phys. Rev. D89 (2014) 061901, [1311.4794].
- [129] F. London, The λ-Phenomenon of Liquid Helium and the Bose-Einstein Degeneracy, Nature 141 (Apr., 1938) 643–644.
- [130] L. Landau, Theory of the superfluidity of helium ii, Phys. Rev. 60 (Aug, 1941) 356–358.
- [131] L. Tisza, The theory of liquid helium, Phys. Rev. 72 (Nov, 1947) 838-854.